Clerk's stamp:

COURT FILE NUMBER

COURT OF QUEEN'S BENCH OF ALBERTA

JUDICIAL CENTRE

PLAINTIFF(S)

DEFENDANT(S)

1703 1276

EDMONTON

SERVUS CREDIT UNION LTD.

CRELOGIX ACCEPTANCE CORPORATION, CRELOGIX CREDIT GROUP INC., CRELOGIX PORTFOLIO SERVICES CORPORATION, KARL SIGERIST, NICHOLAS CARTER, MIKE MCKAY AND MICHAEL MILLS

JUL 0 5 2017

# AFFIDAVIT OF DARCY PEELAR

# VOL. 1 OF 2 WITH TABS A - O

MILLER THOMSON LLP Barristers and Solicitors 2700, Commerce Place 10155-102 Street Edmonton, AB, Canada T5J 4G8 Phone: 780.429.9719 Fax: 780.424.5866

Lawyer's Name: Rick T.G. Reeson, Q.C. Lawyer's Email: rreeson@millerthomson.com File No.: 138667.138

# DOCUMENT

ADDRESS FOR SERVICE AND CONTACT INFORMATION OF PARTY FILING THIS DOCUMENT

Clerk's stamp:

COURT FILE NUMBER

COURT OF QUEEN'S BENCH OF ALBERTA

JUDICIAL CENTRE

PLAINTIFF(S)

DEFENDANT(S)

EDMONTON

SERVUS CREDIT UNION LTD.

CRELOGIX ACCEPTANCE CORPORATION, CRELOGIX CREDIT GROUP INC., CRELOGIX PORTFOLIO SERVICES CORPORATION, KARL SIGERIST, NICHOLAS CARTER, MIKE MCKAY AND MICHAEL MILLS

## DOCUMENT

ADDRESS FOR SERVICE AND CONTACT INFORMATION OF PARTY FILING THIS DOCUMENT

# AFFIDAVIT OF DARCY PEELAR

MILLER THOMSON LLP Barristers and Solicitors 2700, Commerce Place 10155-102 Street Edmonton, AB, Canada T5J 4G8 Phone: 780.429.9719 Fax: 780.424.5866

Lawyer's Name: Rick T.G. Reeson, Q.C. Lawyer's Email: rreeson@millerthomson.com File No.: 138667.138

## Sworn on July 5, 2017

I, Darcy Peelar, of the City of Edmonton, in the Province of Alberta, SWEAR AND SAY THAT:

- I am the Chief Credit Compliance and Operational Support Officer for Servus Credit Union Ltd. ("Servus") and as such, have personal knowledge of the facts and matters herein deposed to except where stated to be based on information and belief, and where so stated I do verily believe the same to be true.
- 2. I have reviewed the business records maintained by Servus in respect of the matters at issue, which I do verily believe were made in the ordinary and usual course of business. Where I do not have direct personal knowledge of matters deposed herein, my knowledge is derived from my review of the business records, relevant copies of which are attached to this my affidavit.
- 3. I am authorized to make this affidavit on behalf of Servus.

## Background

- 4. Crelogix Acceptance Corporation ("CAC") is a corporation incorporated pursuant to the laws of the Province of British Columbia, registered extra provincially in the Province of Alberta, and is a customer and debtor of Servus. Attached hereto and marked as **Exhibit "A**" is a true copy of the Alberta Corporate Registry Search results for CAC.
- 5. CAC, is a non-bank consumer financial services corporation. It provides a range of credit, insurance, and product protection solutions through customer programs it has established with a diverse group of merchants, retailers, wholesalers, manufacturers, franchisors, and industry associations (collectively referred to as the "Merchants"). It acquires from the Merchants consumer debt obligations ("Obligations"), and then resells these Obligations to third party financial institutions ("Funders"). CAC also administers the collection of the indebtedness due on the Obligations on behalf of the Funders in consideration a of monthly administration fee.
- 6. By various Funding Agreements between CAC and the following Funders:
  - Prospera Credit Union June 1, 2014 amended on August 5, 2016 and June 2017
  - Servus Credit Union March 31, 2008 amended on September 30, 2015 and October 1, 2015
  - Connect First Credit Union July 27, 2015
  - Bow Valley Credit Union March 6, 2013 amended January 30, 2014
  - Beaumont Credit Union August 10, 2011 amended January 21, 2014

(the "Funding Agreements"), CAC and the Funders agreed, inter alia as follows:

- the Funders would purchase Contracts (as defined in the Funding Agreements) approved by CAC, provided such Contracts meet the credit underwriting standards and transaction structuring standards agreed to in the Funding Agreements;
- (b) CAC would acquire the Contracts on a bulk basis and assign them to the Funders by way of Bulk Assignment (as defined in the Funding Agreements);
- (c) the Funders would acquire the Contracts on a present value basis, and the assignment value for each Contract would be the present value of the Contract Payments (as defined in the Funding Agreements) discounted at a rate that would give each Funder a yield called the Buy Rate (as defined in the Funding Agreement);
- (d) CAC would, in consideration of a monthly fee, service the Contracts, including contract origination, customers service, billing, collecting and any other function normally associated with servicing a consumer lending portfolio;
- (e) all Contract Payments collected by CAC for remittance to a Funder, are to be received by CAC in trust and as agent for the Funder; and
- (f) CAC has various reporting obligations to each Funder.

7. The shareholders of CAC, and their respective shareholdings are as follows:

<u>Shareholder</u>	Type and Number of Shares	Percentage Holdings
Cerebral Capital Inc. (Karl Sigerist)	2,086,668 Common	27.97
Nicholas Carter	993,333 Common	13.31
Michael Mills	993,333 Common	13.31
Michael McKay	993,333 Common	13.31
Servus	2,395,000 Common	32.10
Total	7,461,667 Common	

- 8. Until June 30, 2017, the following individuals were directors and officers of CAC:
  - Wayne Cook-Director/Chairman of the BoardKarl Sigerist-Director/PresidentMarcel Fizell-Director/Secretary
- 9. Senior management of CAC is comprised as follows:

Karl Sigerist -	President and Chief Executive Officer
Michael McKay -	VP Administration and Controller
Michael Mills -	VP Credit, Compliance & Portfolio Service

- 10. Based on information provided to Servus by Alvarez and Marsal Canada Inc. ("Alvarez"), the consultant appointed by Servus:
  - (a) CAC's head office and main place of operation is located in Burnaby, British Columbia, where it employs approximately 60 employees;
  - (b) CAC also operates a business located in Montreal, Quebec, where it employs between 10 to 12 employees; and
  - (c) CAC purchases consumer Contracts from various parties across Canada including in the Province of Alberta. It also sells those Contracts to various financial institutions located in Alberta including Servus Credit Union, Bow Valley Credit Union, Connect First Credit Union, Prospera Credit Union, and Beaumont Credit Union.
- 11. Crelogix Credit Group Inc. ("CCG"), is a corporation incorporated pursuant to the laws of the Province of British Columbia, and is a customer and debtor of Servus. Attached hereto and marked as **Exhibit "B**" is a true copy of the British Columbia Corporate Registry Search results for CCG. CCG is a wholly owned subsidiary of CAC. CCG guaranteed payment to Servus of a portion of the obligations of CAC to Servus.
- 12. Crelogix Portfolio Service Corporation ("CPS"), is a corporation incorporated pursuant to the laws of the Province of British Columbia, is registered extra provincially in the

Province of Alberta, and is a customer and debtor of Servus. Attached hereto and marked as **Exhibit "C"** is a true copy of the Alberta Corporate Registry Search results for CPS. CPS is a wholly owned subsidiary of CAC. CPS guaranteed payment to Servus of a portion of the obligations of CAC to Servus.

## The Credit Facilities

- 13. On the application of CAC, Servus agreed to extend to CAC the following loans and credit facilities (collectively, the "CAC Facilities"):
  - (a) a Demand Overdraft Facility in the maximum amount of \$2,000,000.00 (the "Warehouse Loan"), the terms of which are set out in the Overdraft Protection Agreement dated March 31, 2008, and the Offers of Financing dated March 7, 2008 and May 31, 2013 (collectively, the "Warehouse Loan Agreements"). Attached hereto and marked as **Exhibit** "D" are true copies of the Warehouse Loan Agreements;
  - (b) a Demand Overdraft Facility in the maximum amount of \$1,250,000.00 (the "Evergreen Loan"), the terms of which are set out in the Supplemental Credit Facility Letter, dated September 8, 2016, and all amendments thereto, including the Amending Facility Letters dated November 15, 2016, December 14, 2016, December 22, 2016, January 9, 2017, January 30, 2017, February 14, 2017, February 28, 2017, March 17, 2017, April 10, 2017, April 19, 2017, April 21, 2017, May 1, 2017, May 15, 2017, and June 27, 2017 (the "Evergreen Loan Agreements"). Attached hereto and marked as **Exhibit** "E" are true copies of the Evergreen Loan Agreements; and
  - (c) a Demand Term Loan in the amount of \$179,652.02 (the "Consultant Loan"), the terms of which are set out in the Supplemental Credit Facility Letter, dated September 8, 2016, and all amendments thereto, including the Amending Facility Letters dated December 14, 2016, December 22, 2016, January 9, 2017, January 30, 2017, February 28, 2017, March 17, 2017, April 19, 2017, May 1, 2017, May 15, 2017, and June 27, 2017 (the "Consultant Loan Agreements"). Attached hereto and marked as **Exhibit "F"** are true copies of the Consultant Loan Agreements.
- 14. Servus agreed to further extend to CCG a one-time non-revolving loan in the principal amount of \$2,000,000.00 (the "CCG Loan"), the terms of which are set out in the Unsecured Convertible Note Agreement, dated September 14, 2016 (the "CCG Note"). Attached hereto and marked as **Exhibit "G**" is a true copy of the body of the CCG Note.
- 15. By a series of Amending Agreements, the CCG Note was extended to and including July 7, 2017. Attached hereto and marked as **Exhibit "H"** is a true copy of the most recent CCG Note Amending Agreement No. 10 dated June 28, 2017.

## The Indebtedness

16. As at July 2, 2017, there is outstanding under the CAC Facilities and the CCG Loan (the CAC Facilities and the CCG Loan hereafter collectively referred to as the "Credit Facilities"), plus such further amounts owed in respect of costs and expenses incurred by Servus, together with accruing interest, as follows:

- (a) Warehouse Loan \$1,474,488.78
- (b) Evergreen Loan \$508,798.92
- (c) **Consultant Loan** \$182,693.82 (per diem - \$32.98)
- (d) **CCG Loan** \$2,108,277.15 (per diem - \$381.54)

**Total**: \$4,274,258.67 (plus such further amounts owed in respect of costs and expenses incurred by Servus, including legal costs and disbursements on a solicitor and own client full indemnity basis, together with accruing interest, hereafter called the "**Indebtedness**").

## **Guarantees of the CAC Facilities**

17. To secure the due payment and discharge of certain of the present and future indebtedness and liability of CAC to Servus, the following guarantees were granted in favour of Servus:

Guarantor	Date of the Guarantee	Amount of the Guarantee
CCG	October 30, 2009	\$2,000,000.00
CPS	October 30, 2009	\$2,000,000.00
Karl Sigerist	December 10, 2010	\$530,400.00
Nicholas Carter	December 10, 2010	\$250,200.00
Mike McKay	December 10, 2010	\$250,200.00
Michael Mills	December 10, 2010	\$250,200.00

and attached hereto collectively as **Exhibit "I"** are true copies of the above mentioned guarantees (the "CAC Guarantees").

## Guarantees of the CCG Loan

18. To secure payment and discharge of the indebtedness and liability of CCG to Servus under the CCG Loan, CAC and CPS each executed unlimited continuing guarantees dated September 14, 2016 (collectively, the "Subsidiary Note Guarantee") in favour of Servus with respect to all amounts owing under the CCG Note, plus all costs, loss, damage or liability suffered by Servus in relation to the enforcing and obtaining payment under the Subsidiary Notes Guarantee. Attached hereto and marked as Exhibit "J" is a true copy of the Subsidiary Note Guarantee.

## Securities

 As further security for the due payment of the Indebtedness, each of CAC, CCG and CPS granted or caused to be granted to Servus certain securities (the "Securities"). Attached hereto and marked as the following exhibits hereto are true copies of certain Securities as follows:

Document	Exhibit No.
<ul> <li>General Security Agreement dated March 21, 2008 (originally in the name of Travelers Acceptance Corporation, which subsequently changed its name to CAC) (the "CAC GSA". The CSA GSA secures all present and future obligations of CAC to Servus.</li> </ul>	" <b>K</b> "
<ul> <li>General Security Agreement dated October 30, 2009 from CCG (the "CCG GSA"). The CCG GSA secures all present and future obligations of CCG to Servus.</li> </ul>	" <b>L</b> "
<ul> <li>General Security Agreement dated October 30, 2009 from CPS (the "CPS GSA"). The CPS GSA secures all present and future obligations of CPS to Servus.</li> </ul>	" <b>M</b> "

- 20. Attached collectively hereto and marked as **Exhibit** "N" are true copies of searches obtained from the Personal Property Registry from the Province of Alberta for each of CAC, CCG, and CPS, confirming registration of the Securities.
- 21. Attached collectively hereto and marked as **Exhibit** "**O**" are true copies of searches obtained from the Personal Property Registry in the Province of British Columbia for each of CAC, CCG, and CPS, confirming registration of the Securities.
- 22. The CAC Facilities are payable on demand.

## **Demand for Payment**

- 23. On or about September 13, 2016, Servus, through its legal counsel, demanded payment of the CAC Facilities, inclusive of solicitor and his own client full indemnity costs, from CAC but it failed or neglected, and continues to fail and neglect, to pay the CAC indebtedness to Servus. Concurrent with the issuance of a demand for payment with respect to the CAC Facilities, Servus did serve on CAC a Notice of Intention to Enforce Security pursuant to Section 244 of the *Bankruptcy and Insolvency Act*.
- 24. On or about September 13, 2016, Servus, through its legal counsel, also demanded payment under the CAC Guarantees, inclusive of costs as between a solicitor and his own client on a full indemnity basis, but each of the guarantors of the CAC Facilities failed or neglected, and continue to fail and neglect, to pay the amounts owing under the CAC Guarantees. Concurrent with the issuance of a demand for payment with respect to the CAC Guarantees, Servus did serve on CCG and CPS Notices of Intention to Enforce Security pursuant to Section 244 of the *Bankruptcy and Insolvency Act*.
- 25. Attached hereto and marked collectively as **Exhibit** "**P**" are true copies of the demands and notices and delivery confirmation of such demands and notices set forth in paragraph 22 and 23.

## The Forbearance Agreement

26. On or about September 14, 2016, the Defendants entered into a forbearance agreement (which together with a forbearance acknowledgement letter dated August 26, 2016, and certain amendments thereto dated September 6, 2016, and September 9, 2016 (collectively, the "Letter Agreements") together with all amendments thereto, is collectively called the "Forbearance Agreement"). Attached hereto and marked as

**Exhibit** "**Q**" are true copies of the Letter Agreements and the body of the Forbearance Agreement.

- 27. In the Forbearance Agreement each of the Defendants acknowledged, agreed and represented to Servus, amongst other things, that:
  - (a) they are in default of their obligations to Servus, which obligations were fully due and owing to Servus;
  - (b) the CAC Facilities are fully due and owing to Servus;
  - (c) there is no defence or setoff to the claim by Servus for repayment;
  - (d) all documentation respecting the loans and the guarantees of the loans giving rise to the Defendants respective indebtedness are valid, subsisting and enforceable;
  - (e) all security granted by the Defendants in favour of Servus is valid, subsisting and enforceable; and
  - (f) Servus has issued certain demands and Section 244 Notices to the Defendants, such demands and notices were received by the Defendants, and no further demands or notices would be required to be given by Servus to the Defendants.
- 28. Each of CAC, CCG, and CPS, as a condition of the Forbearance Agreement, granted to Servus a Consent to Early Enforcement with respect to the Section 244 Notices served upon them. Attached hereto and marked collectively as **Exhibit** "**R**" are true copies of the Consents to Early Enforcement.
- 29. Each of CAC, CCG, and CPS, also provided to Servus, as a condition of the Forbearance Agreement, a Consent Receivership Order (the "Consent Order"). Attached hereto and marked as **Exhibit** "S" is a true copy of the Consent Order.
- 30. Pursuant to the Forbearance Agreement, Servus was at liberty to utilize the Consent Order and to enforce the Securities upon an event of default, which included the failure to pay the Indebtedness to Servus in full no later than 2:00 p.m. (Edmonton time) on December 15, 2016 (the "Forbearance Period") or the occurrence of any adverse change in either the legal position or security position of Servus or the financial condition of CAC.
- 31. By a series of Amending Agreements, the Forbearance Agreement was extended to and including 2:00 p.m. on July 7, 2017. Attached hereto and marked as **Exhibit** "**T**" is a true copy of the most recent Forbearance Amending Agreement No. 13 dated June 28, 2017.
- 32. Pursuant to the Forbearance Agreement, CAC also agreed to Servus appointing Alvarez as a financial consultant to provide the Financial Services set out in the Consultant Agreement dated September 6, 2018, a true copy of which is attached hereto and marked as **Exhibit "U**".

## **Financial Concerns**

33. Attached hereto and marked collectively as **Exhibit** "**V**", are draft copies of the Consolidated Financial Statements for the CCG for October 31, 2015 and October 31,

2016. Although these are still in draft form, at present they reflect significant operating losses for both years and significant deficits in shareholders equity.

- 34. Attached hereto and marked as **Exhibit** "**W**" is a true copy of a report dated July 3, 2017 from Alvarez.
- 35. I am advised by Mr. Tim Reid of Alvarez, and do verily believe, that CAC will require approximately \$2 Million to continue operations over the next 90 days, and in the event that Alvarez is appointed as Receiver of CAC by this Court, Alvarez will require the Court's authorization to borrower these funds within the Receivership proceedings.

## **Resignations of Certain Members of the Board and Officers**

- 36. Attached hereto and marked as **Exhibit "X**" is an email from Ms. Joan Chambers of the law firm Blake, Cassels & Graydon LLP, dated July 1, 2017, 8:33 a.m., advising that each of Karl Sigerist, Mike McKay, Mike Mills and Wayne Cook have tendered their resignations as directors and officers of each of the Crelogix companies effective at approximately 4:30 p.m. on Friday, June 30, 2017.
- 37. Attached hereto and marked as **Exhibit** "**Y**" is a true copy of the Resignation of Director, dated July 4, 2017, executed by Marcel Fizell, tendering Mr. Fizell's resignation from the Board of Directors of CCG, effective immediately.
- 38. The resignation of all of the senior management of CAC, CCG, and CPS leaves only more junior and less experienced employees to run the business. Servus is extremely concerned with this development, which will most certainly have a material adverse effect on the business and the affairs of CAC, and will significantly jeopardize the legal and security position of Servus. It is the position of Servus that this represents a material adverse change under the Forbearance Agreement and the CCG Note.

## Appointment of Receiver

- 39. As a result of the facts deposed to herein, including without limitation, the defaults under the Forbearance Agreement, Servus is now entitled to enforce the Securities and use the Consent Order.
- 40. I swear this affidavit in support of an application for the appointment of Alvarez as receiver pursuant to the Consent Order.

**SWORN** BEFORE ME at the City of Edmonton, in the Province of Alberta, this 5th day of July, 2017.

A Commissioner for Oaths in and for Alberta

Patrick T. Harnett Barrister and Solicitor

DARCY PEELAR

This is **Exhibit "A"** referred to in the Affidavit of Darcy Peelar, sworn before me this 5<sup>th</sup> day of July, 2017

Hant

A Commissioner for Oaths in and for Alberta

Patrick T. Harnett Barrister and Solicitor

# Government Corporation/Non-Profit Search of Alberta Corporate Registration System

Date of Search:	2016/09/08
Time of Search:	07:14 AM
Search provided by:	MILLER THOMSON LLP- EDMONTON

Service Request Number: 25718886 Customer Reference Number: 138667.138

Corporate Access Number: 2116009032Legal Entity Name:CRELOGIX ACCEPTANCE CORPORATION

Legal Entity Status:	Active
Extra-Provincial Type:	Other Prov/Territory Corps
Registration Date:	2011/04/13 YYYY/MM/DD
Date Of Formation in Home Jurisdiction:	2005/10/31 YYYY/MM/DD
Home Jurisdiction:	BRITISH COLUMBIA
Home Jurisdiction CAN:	BC0739156

# Primary Attorney:

		Middle Name	Firm Name	Street	City	Droxingo	Postal Code
BENTLEY	ROSS	1	BLAKE, CASSELS & GRAYDON LLP		CALGARY		T2P 4J8

# Head Office Address:

Street:	SUITE 2600, 595 BURRARD STREET, PO BOX 49314
City:	VANCOUVER
Province:	BRITISH COLUMBIA
Postal Code:	V7X 1L3

# **Other Information:**

# Filing History:

List Date (YYYY/MM/DD)	Type of Filing
2011/04/13	Register Extra-Provincial Profit / Non-Profit Corporation
2011/11/03	Change Address
2011/11/05	Change Attorney

This is to certify that, as of this date, the above information is an accurate reproduction of data contained within the official records of the Corporate Registry.



This is **Exhibit "B"** referred to in the Affidavit of Darcy Peelar, sworn before me this 5<sup>th</sup> day of July, 2017

A Commissioner for Oaths in and for Alberta

Patrick T. Harnett Barrister and Solicitor



Mailing Address: PO Box 9431 Stn Prov Govt Victoria BC V8W 9V3 www.corporateonline.gov.bc.ca Location: 2nd Floor - 940 Blanshard Street Victoria BC 1 877 526-1526

# **BC Company Summary** For CRELOGIX CREDIT GROUP INC.

Date and Time of Search: **Currency Date:** 

September 06, 2016 09:01 AM Pacific Time May 31, 2016

# ACTIVE

BC0864537 **Incorporation Number:** CRELOGIX CREDIT GROUP INC. Name of Company: **Recognition Date and Time:** Incorporated on October 23, 2009 11:13 AM Pacific Time In Liguidation: No Last Annual Report Filed: October 23, 2015 **Receiver:** No

# **REGISTERED OFFICE INFORMATION**

Mailing Address:

SUITE 2600, THREE BENTALL CENTRE P.O. BOX 49314, 595 BURRARD STREET VANCOUVER BC V7X 1L3 CANADA

**Delivery Address:** SUITE 2600, THREE BENTALL CENTRE P.O. BOX 49314, 595 BURRARD STREET VANCOUVER BC V7X 1L3 CANADA

# **RECORDS OFFICE INFORMATION**

Mailing Address: SUITE 2600, THREE BENTALL CENTRE P.O. BOX 49314, 595 BURRARD STREET VANCOUVER BC V7X 1L3 CANADA

**Delivery Address:** SUITE 2600, THREE BENTALL CENTRE P.O. BOX 49314, 595 BURRARD STREET VANCOUVER BC V7X 1L3 CANADA

# DIRECTOR INFORMATION

Last Name, First Name, Middle Name: COOK, WAYNE

Mailing Address: SUITE 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 CANADA

**Delivery Address:** 

SUITE 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 CANADA

Mailing Address: 151 KARL CLARK ROAD NW EDMONTON AB T6N 1H5 CANADA

Last Name, First Name, Middle Name: SIGERIST, KARL

Mailing Address: SUITE 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 CANADA Delivery Address: 151 KARL CLARK ROAD NW EDMONTON AB T6N 1H5 CANADA

Delivery Address: SUITE 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 CANADA

## **OFFICER INFORMATION AS AT October 23, 2015**

Last Name, First Name, Middle Name: Carter, Nicholas Office(s) Held: (Vice President)

Mailing Address: SUITE 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 Delivery Address: SUITE 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 CANADA

#### Last Name, First Name, Middle Name: Cook, Wayne

Office(s) Held: (Chair)

## Mailing Address:

CANADA

SUITE 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 CANADA Delivery Address: SUITE 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 CANADA

# Last Name, First Name, Middle Name: Lynde, David

Office(s) Held: (Vice President)

## Mailing Address:

SUITE 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 CANADA Delivery Address: SUITE 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 CANADA

Last Name, First Name, Middle Name: McKay, Michael Office(s) Held: (Secretary, Vice President)

Mailing Address: SUITE 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 CANADA Delivery Address:

SUITE 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 CANADA Last Name, First Name, Middle Name: Mills, Michael Office(s) Held: (Vice President)

Mailing Address:

SUITE 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 CANADA Delivery Address: SUITE 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 CANADA

Last Name, First Name, Middle Name: Sigerist, Karl

Office(s) Held: (CEO, President)

Mailing Address: SUITE 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 CANADA Delivery Address: SUITE 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 CANADA This is **Exhibit "C"** referred to in the Affidavit of Darcy Peelar, sworn before me this 5<sup>th</sup> day of July, 2017

Hand

A Commissioner for Oaths in and for Alberta

Patrick T. Harnett Barrister and Solicitor

# Government Corporation/Non-Profit Search of Alberta Corporate Registration System

Date of Search:	2016/09/06
Time of Search:	09:37 AM
Search provided by:	MILLER THOMSON LLP- EDMONTON

Service Request Number: 25688685 Customer Reference Number: 138667.138

Corporate Access Number: 2114917210Legal Entity Name:CRELOGIX PORTFOLIO SERVICES CORPORATION

Legal Entity Status:	Active
Extra-Provincial Type:	Other Prov/Territory Corps
Registration Date:	2009/10/26 YYYY/MM/DD
Date Of Formation in Home Jurisdiction:	2009/10/23 YYYY/MM/DD
Home Jurisdiction:	BRITISH COLUMBIA
Home Jurisdiction CAN:	BC0864550

# **Primary Attorney:**

	First Name	Middle Name	Firm Name	Street	City	Umoramoo	Postal Code
BENTLEY	ROSS		BLAKE, CASSELS & GRAYDON LLP		CALGARY	1 1	T2P 4J8

# Head Office Address:

Street:	SUITE 2600, 595 BURRARD STREET, P.O. BOX 49314
City:	VANCOUVER
Province:	BRITISH COLUMBIA
Postal Code:	V7X 1L3

# **Other Information:**

# Filing History:

List Date (YYYY/MM/DD)	Type of Filing
2009/10/26	Register Extra-Provincial Profit / Non-Profit Corporation
2011/10/22	Change Address
2011/10/24	Change Attorney

This is to certify that, as of this date, the above information is an accurate reproduction of data contained within the official records of the Corporate Registry.



This is **Exhibit "D"** referred to in the Affidavit of Darcy Peelar, sworn before me this 5<sup>th</sup> day of July, 2017

Hart

A Commissioner for Oaths in and for Alberta

Patrick T. Harnett Barrister and Solicitor

0	V	ΈR	DF	RAF	Т	PR	OT	EC	TIO	Ν
					-					

AGREEMENT (Electing Rate)

Borrower(s

(i loading itato)	

Burnaby, British Columbia

Travelers Acceptance Corporation

Suite 500, 4180 Lougheed Highway

	n ity n Ltd. (the "Credit Unior and Square Bra	
1010.0	suu oquare are	******
<u>4901 ·</u>	- 48th Street	
Red De	er	Alberta
T4N 6	N4	Brand

#### AGREEMENT

**∇5C 6A7** 

In this agreement, the words, "I", "me" and "my" mean the Borrower or if more than one, each Borrower who signs this agreement

In return for the Credit Union permitting me to overdraw my Cheguing Account being account number \_\_\_\_54638920 ("my Account") I agree to the following:

The maximum amount I may overdraw my Account is  $\frac{2,000,000,00}{2,000,000}$  ("Overdraft Limit"). The Credit Union may debit to my Account the amount of any cheque or withdrawal, or other item drawn on my Account and interest as provided in this agreement. The Credit Union may refuse to pay any cheque, withdrawal or any other item if the outstanding debit balance of my Account exceeds, or would after payment of such cheque, withdrawal or other item exceed, my Overdraft Limit on the date such cheque, withdrawal or other item is presented to the Credit Union for payment. 2

I will pay on demand all amounts which overdraw my Account and I will pay both bafore and after demand interest on all з. amounts which overdraw my Account calculated from the date each such amount is debited to my Account until the date of amount is which overaraw my Account calculated from the date each such amount is dediced to my Account until the date of payment at a rate per annum equal to 1.5 % per annum above the Credit Union Prime Rate, calculated monthly not in advance both before and after demand, default and judgment. Overdue interest shall be compounded monthly. Interest shall be determined daily and be calculated and payable monthly. As of the date of this agreement the Credit Union Prime Rate is 5.25% per annum and the interest rate hereunder is 6.75% per annum.

"Credit Union Prime Rate" means the floating annual rate of interest established from time to time by the Credit Union as the base rate it will use to determine rates of interest on Canadian Dollar loans to its members and designated by the Credit Union as its Credit Union Prime Rate. A change in the Credit Union Prime Rate shall on the day of such change also vary the interest rate on the amounts which overdraw my Account in the same manner.

- In addition to interest payable under clause 3, 1 agree to pay to the Credit Union its costs of administration that relate to transactions that overdraw or increase the amount overdrawn in my Account (the "Transaction Fee") as well as those that relate to maintaining the Overdraft Limit (the "Standby Fee"). These fees may change from time to time. The current Transaction Fee and Standby Fee may be ascertained from the Credit Union. These fees may be debited to my Account and any such fees so charged will be reflected on my periodic statement pertaining to my. Account,
- The Credit Union may charge against any other of my accounts with the Credit Union and credit to my Account any payment that I am obligated to make to the Credit Union under this agreement.
- I hereby pledge to the Credit Union all deposits and paid up shares which I now or may have in the Credit Union, the proceeds of which may, upon default of my obligations contained in this agreement, be applied by the Credit Union to amounts due and owing hereunder. This pledge is in addition to any other rights of or security held by the Credit Union In regard to my obligations hereunder.

·	in order to further secure my obligations hereunder I agree to grant to the Credit Union the following security:		
	(a) See Schedule "A" attached	<u> </u>	
	(a) Dee Defiedure A accached		and;
		·. —	
	(d)		and;

- If I fail to make any payment in accordance with this agreement, or if I fail to do anything I am required to do under this agreement, or if I die, become subject to or take advantage of any law relating to bankruptcy or insolvency or for the relief of 8. debtors or upon attachment, execution or levy against me or my property, then the Credit Union shall have no further obligation to pay any cheques or withdrawals which overdraw my Account or increase an overdraft in my Account and at the option of the Credit Union any outstanding debit balance in my Account shall, without limiting any other of the Credit Union's rights, become immediately due and payable without notice or demand.
- Ether the Credit Union or I may terminate this agreement forthwith by giving written notice to the other, but no termination will relieve me of any obligation I have to the Credit Union under this agreement until I have paid to the Credit Union the outstanding debt balance of my Account in full.
- 10. If more than one person has signed this agreement, the obligations of those persons is joint and several. The Credit Union is authorized to pay any cheques or withdrawals on my Account which overdraw my Account or increase an overdraft in my Account signed by any person who has signed this agreement.
- 11. This agreement shall be governed by the laws of the Province of Alberta whose courts shall have non-exclusive jurisdiction in regard to matters pertaining hereto and to whose jurisdiction I do attorn for such purposes. I agree to pay to the Credit Union all costs it incurs in enforcing its rights under this agreement on a solicitor-client basis.

SIGNED at VOM COUVER, BC	314 day of Morch , 2008
Credit Union Officer Angela Douglas	Borrower
	Borrower

(c)

Distribute to: Credit Union / Member

Borrower

# SCHEDULE "A" TO OVERDRAFT PROTECTION AGREEMENT

- 7 (a) A General Security Agreement covering all presently owned and after acquired property, including proceeds thereof;
  - (b) Assignment of Keyman Life Insurance in the name of Karl Sigerest, in the amount of \$500,000.00;
  - (c) Limited Personal Guarantees and Postponements of Claim in the names of:
    - (i) Karl Sigerist \$227,712.00
    - (ii) Nicholas Carter \$107,411.00
    - (iii) Mike McKay \$107,411.00
    - (iv) Michael Mills \$107,411.00
    - (v) Gary Thompson \$66,596.00
    - (vi) Dennis Holmes \$266,379.00
    - (vii) Jim Case \$266,379.00



Offer of Financing To: Travelers Acceptance Corporation Suite 500 - 4180 Lougheed Highway Burnaby, British Columbia V5C 6A7

March 7, 2008

Provided By: Community Credit Union Ltd. Business Banking Centre 2<sup>nd</sup> Floor, 3001-50<sup>th</sup> Avenue Red Deer, Alberta T4N 5Y6

Contact:

K.W. Cook Vice President and Chief Credit Officer P: (403) 342-9211 F: (403) 347-6686 E: kwcook@communitysavings.ca March 7, 2008

Travelers Acceptance Corporation Suite 500 - 4180 Lougheed Highway Burnaby, British Columbia V5C 6A7

#### Attention: Mr. Sigerist, Vice President

Dear Sir:

#### RE: Offer of Financing Our Reference: 54479340

Community Credit Union Ltd. (the "Credit Union") is pleased to advise that the following new credit facilities have been approved on the terms and conditions below. If you agree with these terms please sign the duplicate copy of this letter in the space provided below under the heading "Acceptance" and return same to the writer's attention.

Borrower:	Travelers Acceptance Corporation		
Financing:	Credit Limit: \$2,000,000		
Conditions:	The following schedules form part of this Offer of Financing:		
	<u>Schedule</u> A B C D	<u>Covering</u> Credit Limit Security and Legal Financial Reporting & Conditions Insurance	

Έ

#### ADDITIONAL TERMS

The implementation and continuation of these credit facilities is subject to periodic review by the Credit Union and is also subject to no adverse change in the financial position of the Borrower. The next review date has been established as April 1, 2009, but may be changed at the discretion of the Credit Union.

Standard Covenants & Events of Default

The terms of this commitment are not to be disclosed, publicly or privately, to any other party except legal counsel or other agents of the Borrower who are specifically involved in this transaction. Without limiting the generality of the foregoing, no such persons shall use or refer to the Credit Union's name in any disclosure made in connection with any of the transactions described herein without the Credit Union's prior written consent.

The Credit Union is entitled to rely on any report or certificate provided to the Credit Union by the Borrower or any Guarantor as applicable, by way of e-mail or fax transmission as though it were an original signed document. The Credit Union is further entitled to assume that any communication from the Borrower received by e-mail or fax transmission is a reliable communication from the Borrower.

A non-refundable application fee of .125% of new authorized funds (\$2.500.00) is payable for the arrangement of these facilities

A monthly administration fee of \$250.00 is to be applied

Business Banking Centre - #201, 3001 – 50<sup>th</sup> Ave, Red Deer, Alberta T4N 5Y6 Tel:(403) 346-2221 Fax:(403) 346-2442

TM "Community" and "Community Savings" are trade marks of Community Credit Union Ltd



Your acceptance of this letter will constitute authority for the Credit Union to instruct its solicitors, or otherwise prepare the necessary documentation. This commitment is not assignable without the prior written consent of the Credit Union.

Your signature below will also give the Credit Union authorization to contact your external auditor directly for provision of annual financial statements for the Borrower and Guarantor companies (if applicable).

This agreement may be executed in any number of counterparts and by different parties in separate counterparts, each of which when so executed shall be deemed to be an original and all of which taken together constitute one and the same instrument.

This Offer of Financing shall expire if not accepted by March 31, 2008.

Yours truly, COMMUNITY CREDIT UNION LTD.

K.W. Cook

Vice President and Chief Credit Officer

Jay 14

D. Mykytyshyn VP Governance & Strategy

## ACCEPTANCE

Acceptance of this Offer of Financing provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facilities may be withdrawn or cancelled at the sole discretion of the Credit Union.

We hereby acknowledge and accept the credit facilities on the terms and conditions outlined by the Offer of Financing dated March 7, 2008 on this \_\_\_\_\_ day of \_\_\_\_\_, 20\_\_

**Travelers Acceptance Corporation** 

Per: -Karl-Sigmist

Per:

Micholas-Carter

Guarantors:

Karl Sigerist	<b>-</b>
Mike McKay	-
Winke MacRay	
Gary Thompson	<b></b>
Jim Case	

Nicholas Carter

**Michael Mills** 

Dennis Holmes



### ACCEPTANCE

Acceptance of this Offer of Financing provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved oredit facilities may be withdrawn or cancelled at the sole discretion of the Credit Union.

We hereby acknowledge and accept the credit facilities on the terms and conditions outlined by the Offer of Financing dated March 7, 2008 on this \_\_\_\_\_ day of \_\_\_\_\_ 20\_\_\_\_

Travelers Acceptance Corporation

Per: Karl Sigerist Per: Nicholas Carter

Guarantors:



Nicholas Carter

Page 3

ł

Michael Mills

#### Dennis Holmes .

. . .

#### SCHEDULE "A" Credit Limit

#### Amount:

\$2,000,000 demand revolving facility

#### Purpose:

To provide funding to dealers prior to securitization of contracts with Community Credit Union.

#### Interest Rate:

Community Credit Union Ltd. prime lending rate ("Prime") plus 1.50%, payable monthly not in advance. Prime means the annual rate of interest announced from time to time by the Credit Union, and as of March 7, 2008 is 5.25%.

#### **Repayment:**

Advances and payments will be charged on a "dollar for dollar" basis with no stipulation regarding multiples in the drawing of his facility. This loan is subject to the Credit Union's right to demand, as provided herein. Interest payments are calculated on the outstanding balance, and are due on the last day of each month. The account is to revolve satisfactorily.

#### **Conditions of Margining:**

Advances will be contained within the lessor of: the authorized limit, or

- a) A margin-free allowance against PNW strength in the amount of \$500,000. OR
- b) AOD is not margined when the loans (CSC's) are securitized weekly by CCU or less and month end balance does not exceed \$500,000 (margin free) and has not revolved in the 30 day prior period margining would be required at 85% of good quality accounts receivable under 60 days excluding any offsets, inter-company accounts, work in progress and holdbacks when loans (CSC's) have not been securitized within the past 30 days. Reporting would be required by the 15<sup>th</sup> of the following month.

#### SCHEDULE "B" Security and Legal

The Borrower agrees to provide to the Credit Union in form and substance satisfactory to it and its solicitors, all security and supporting agreements requested by the Credit Union including the following documentation (the "Security") which will be held by the Credit Union as security for the. It is understood and agreed that the execution and delivery of the security documents securing this transaction shall in no way merge or extinguish this commitment letter or the terms and conditions of it, which shall continue in full force and effect.

#### General Security:

To Be Obtained:

- General Security Agreement covering all company assets (presently owned and after-acquired, including proceeds);
- Borrowing Resolutions
- Executed Securitization Agreement between Travelers Acceptance Corporation and Community Credit Union Ltd.
- Evidence of current Liability Insurance for a minimum amount of \$2,000,000
- Key Man Life Insurance in the name of Karl Sigerist in the amount of \$500,000 assigned to CCU.
- Limited Personal Guarantee and Postponement(s) of Claim granted by each of the following:

Name	Amount of Guarantee
Karl Sigerist	\$227,712
Nicholas Carter	\$107,411
Mike McKay	\$107,411
Michael Mills	\$107,411
Gary Thompson	\$66,596
Dennis Holmes	\$266,379
Jim Case	\$266,379

- The Credit Union will entertain the release of the Personal Guarantees at a future date. This will be subject to the discretion of CCU.
- Solicitor's opinion as to the validity and enforceability of the Credit Union's security, which should
  reference the ranking of the Credit Union's charge and any limitations or restrictions against the
  security which may restrict the Credit Union's claim;
- A General Assignment of Book Debts
- All documentation as, in the Lender's opinion, shall be required to complete this transaction.

#### Credit Limit:

• Overdraft Protection Agreement in the amount of \$2,000,000;



#### SCHEDULE "C"

#### Financial Reporting & Conditions

#### **Financial Reporting:**

The Borrower agrees to provide:

#### Monthly:

In-house monthly year-to-date financial statements and operations are to be provided

#### Annually:

- Within ninety (90) days of its fiscal year end, Audited financial statements of the Borrower, executed by the appropriate Company Officers and signed by a qualified firm of accountants. A late reporting fee of \$25.00 per month may be charged when considered appropriate;
- Annual confirmation of insurance renewal.

#### Other:

- Updated personal net worth statements of all guarantors upon request.
- Signed aged listings of accounts receivable and accounts payable to be provided within 15 days following month end if account balance is over the \$500,000 margin free provision and is not revolving weekly.

#### **Pre-disbursement Conditions:**

The Credit Union's obligation to disburse is conditional upon receipt of the following, all in satisfactory form and substance:

- Completion and, where applicable, registration of all Security;
- Executed personal financial statement of the Guarantor(s);
- Copy of the Securitization Agreement
- All property insurance requirements have been satisfied
- Commitment Letter acknowledged by the borrower and all guarantors is to be held;
- Satisfactory Legal and Letters of Opinion to be obtained
- Satisfactory October 31, 2007 Travelers Acceptance Corporation financial statements
- An executed Shareholders Agreement
- Satisfactory PPR search in British Columbia

#### General Conditions:

- It is understood that the securitization will take place weekly.
- All security documentation will be prepared by the Credit Union solicitor and forwarded to the Borrower's solicitor for execution and provision of an opinion as to the ability of the Borrower to borrow and pledge security. The Credit Union's solicitor will register and provide an opinion as to enforceability;
- All legal costs incurred in this transaction are for the account of the Borrower, including those payable to the solicitor representing the Credit Union;
- All costs (i.e. security registration, appraisals, inspections, etc.) will be for the account of the Borrower;

- Without limiting the Credit Union's right to make demand for payment at any time, all loans are subject to review from time to time, at the Credit Union's discretion. The Credit Union reserves the right to withdraw it's support at any time should there be, in the Credit Union's opinion, a material adverse change in the financial condition of the Borrower and or Guarantor(s), or an unacceptable change in ownership;
- Standard terms and conditions as outlined in CCU policy;
- Electronic copies of loan files are to be available should CCU wish to audit individual loan files as requested;
- Access to Travelers Acceptance Corporations third party secured site (currently CASITRON) to view portfolio as requested (must include payments, history, balances etc.);
- Corporate documents to enable account opening;
- Details on Key Man life insurance policy as it relates Karl Sigerist;

#### SCHEDULE "D" Insurance

The Borrower and Guarantor(s), if applicable, will insure and keep fully insured all assets relied upon the Credit Union as security, and provide an assignment, where applicable, of such coverage showing the Credit Union as first loss payee (with the exception of general liability coverage). The policy(ies) shall include the agreement of the insurer that the policy will not be cancelled or permitted to expire on the expiry date without at least thirty days prior written notice to the Credit Union. All coverages are subject to review and approval by the Credit Union.

#### Liability:

Public liability coverage in an amount not less than \$2,000,000 on a per occurrence basis.

#### Key Man:

Key man life insurance policy over Karl Sigerist for a minimum of \$500,000 is to be in place and assigned to CCU.



#### SCHEDULE "E" Standard Covenants & Events of Default

#### Covenants:

The Borrower agrees that, without the written consent of the Credit Union, the Borrower may not:

- Acquire, amalgamate, or enter into partnership or syndicate with any corporation or person;
- Redeem or purchase any shares or reduce the capital of the Borrower in any manner;
- Make any loan to or invest in or give any Guarantee on behalf of, or provide other financial assistance to any other person, firm, or corporation;
- Sell or dispose of any of their assets, except in the ordinary course of business;
- Change the general nature of the Borrower's business;
- Refuse any reasonable request for financial or other information on the status of operating, as may be from time to time, made by the Credit Union;
- Create any encumbrances ranking or purportedly ranking ahead of the Credit Union.

The Borrower undertakes and agrees:

- To use the loan(s) for the stated purpose and to do all things necessary to accomplish that purpose;
- To do all such things as are necessary to complete the aforementioned security documents,
- To comply at all times with the terms and conditions of the commitment and any other requirements stipulated by the Credit Union;
- To permit Credit Union representatives at any reasonable time to make such inspection of the Borrower's business, assets, and records, as such representative may deem necessary;
- To maintain their corporate existence and do such things as may be required in order to permit them to carry on business, and

#### Events of Default:

If any one of the following Events of Default occurs, the Credit Union may, at its option, declare the unpaid principal amount of the advance and accrued interest thereon, to be immediately due and payable and the Borrower shall forthwith pay all such amounts owing; the security pledged by the Borrower shall become enforceable by the Credit Union, and the obligation for any further credit advances shall cease, when:

- The Borrower fails to make any payment of interest or principal on schedule; or
- There is a breach by the Borrower of any other terms or conditions contained in this Offer of Financing; or
- Any default occurs under any security specified in this Offer of Financing or under other credit or loan agreements to which the Borrower is a party; or
- Any material adverse change occurs in the financial position of the Borrower; or
- Any bankruptcy, reorganization, arrangement, insolvency, or litigation proceedings, or other proceedings for the relief of debts are instituted by or against the Borrower, or if a receiver is appointed over any of the Borrower's property or undertaking; or
- Any of the aforementioned events occurs in respect of any Guarantor or connected borrower.

Waiver: Any waiver by the Lender of any default of the Borrower or any omission of the Lender's part in respect of any default by the Borrower shall not extend to or be taken in any manner whatever to affect any subsequent default by the Borrower or the rights resulting from it. The Lender may waive any condition precedent to funding, but the waiver shall not prejudice any subsequent enforcement of the condition.



#### Offer of Financing to:

CRELOGIX ACCEPTANCE CORPORATION #900 - 4445 LOUGHEED HWY Burnaby, BC V5C 0E4

May 31, 2013

#### Provided by:

Servus Credit Union Ltd. Red Deer Regional Office #303 - 4901 48st Red Deer, AB T4N 6M4

#### Contact:

Marcel Fizell Senior Vice President of Credit T: 780.638.8437 C: 587.986.5516 E: marcel.fizell@servus.ca



May 31, 2013

CRELOGIX ACCEPTANCE CORPORATION #900 - 4445 LOUGHEED HWY Burnaby, BC V5C 6A7

Attention: Karl Sigerist, Michael McKay, Michael Mills, Nicholas Carter

Dear Sirs:

## Re: Offer of Financing, \$2,000,000 authorized overdraft

Servus Credit Union Ltd. (the "Credit Union" or "SCU") is pleased to advise that the following credit facility has been approved on the terms and conditions below.

If you agree with these terms and conditions, please sign the duplicate copy of this letter in the space provided below under the heading "Acceptance" and return one original signed / dated copy of this Offer of Financing to the writer's attention.

Borrower:	CRELOGIX ACCEPTANCE CORPORATION	
Guarantors:	Karl Sigerist, Michael McKay, Michael Mills, Nicholas Carter, Crelogix Portfolio Services Corporation, Crelogix Credit Group Inc	
Facility:	Credit Limit	
Amount:	\$2,000,000	

The following schedules are attached hereto and form part of this Offer of Financing:

<u></u>	
"A″	Credit Limit
"B″	Security & Legal
"C″	Pre-Disbursement & General Conditions of Credit
"D"	Reporting Requirements & Pricing
"E"	Events of Default
"F"	Insurance

Description

## Additional Terms

Schedule

The implementation and continuation of this credit facility will be subject to periodic review by the Credit Union and is also subject to no adverse change in the financial position of the Borrower. The next review date has been established as **January 31**, **2014** but may be changed at the discretion of the Credit Union.

This letter is being delivered to you on the understanding that neither this letter nor its substance shall be disclosed, publicly or privately, except to counsel, accountants, employees, and agents of the Borrower who are specifically involved in the proposed transaction. Without limiting the generality of the foregoing, no such persons shall use or refer to the Credit Union's name in any disclosure made in connection with any of the transactions described herein without the Credit Union's prior written consent.

Initials	
1	



This document may be executed in any number of counterparts and by different parties in separate counterparts, each of which when so executed shall be deemed to be an original and all of which when taken together constitute one and the same instrument.

The Credit Union is entitled to rely on any report or certificate submitted by the borrower and / or guarantors, as applicable, by way of e-mail, fax transmission, or electronic upload as though it were an original signed document. The borrower and any guarantors certifies that all reports or certificates are true and complete in all respects and acknowledges that the Credit Union is relying on these representations and warranties.

This Offer of Financing is not assignable without the prior written consent of the Credit Union.

This Offer of Financing shall expire if not accepted by June 14, 2013.

We thank you for allowing Servus Credit Union the opportunity of being of assistance to you.

Yours truly,

## SERVUS CREDIT UNION LTD.

Marcel Éizell Senior Vice President Credit Servus Credit Union Ltd. 151 Karl Clark Road, Edmonton, AB, T6N 1H5 T: 780.638.8437 C: 587.986.5516

Initials					


# <u>Acceptance</u>

Acceptance of this Offer of Financing provides full and sufficient acknowledgement that, if in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility, in whole or in part, may be withdrawn or cancelled at the discretion of the Credit Union.

We hereby acknowledge and accept the credit facility on the terms and conditions outlined by the Offer of Financing dated **May 31, 2013** on this  $\underline{-4th}$  day of  $\underline{-5une}$ , **2013**.

# CRELOGIX ACCEPTANCE CORPORATION Per: Karl Sigerist Per: With Multiple Acceptance Corporation Personal GUARANTORS Karl Sigerist Michael McKay Michael Mills Nicholas Carter

Init	lals



# CORPORATE GUARANTOR(S)

Crelog	ix Portfolio Services Corporation
Per:	
Per:	MM

Crelogix Credit Group Inc.

Per: Per: فترج

Init	als



CRELOGIX ACCEPTANCE CORPORATION

May 31, 2013

# SCHEDULE "A" Credit Limit

#### Amount:

Loan offer to:

\$2,000,000 demand revolving facility.

#### **Purpose:**

To provide funding of contracts prior to securitization to Funders

#### Interest Rate:

Servus Credit Union Ltd. Prime lending rate ("Prime") plus 1.50%, payable monthly not in advance. Servus Credit Union Ltd. Prime is that rate of interest announced from time to time by Servus Credit Union Ltd, and as of May 31, 2013 is 3.00%.

Advances and payments will be charged on a "dollar for dollar" basis with no stipulation regarding multiples in the drawing of his facility. This loan is subject to the Credit Union's right to demand, as provided herein. Interest payments are calculated on the outstanding balance, and are due on the last day of each month.

#### Prepayment:

Prepayment at any time and in any amount at the borrower's option is acceptable without notice, penalty or bonus.

#### Fees:

The monthly monitoring administration fee has been waived. In addition, the annual credit renewal fee has been waived.

#### **Conditions of Margining:**

Advances will be contained within the lesser of: the authorized limit, or

a) A margin free allowance against PNW strength in the amount of \$500,000

or

b) AOD is not margined when loans (CSC's) are securitized weekly by SCU. In the event that the month end balance exceeds the above \$500,000 & does not revolve fully during the prior 30 day period, margining would be required at 85% of good quality accounts receivables (fully executed contracts) under 60 days excluding any offsets, inter company accounts, work in progress, or holdbacks.

Reporting is required on a weekly basis by Friday of each week, seven days after the end of the previous reporting period (see Schedule "D" - Reporting Requirements & Pricing, Weekly)

Ini	tials

Servus Credit Union Ltd.



May 31, 2013

# SCHEDULE "B" Security & Legal

The borrower agrees to provide to the Credit Union in form and substance satisfactory to it and its solicitors, all security and supporting agreements requested by the Credit Union including the following documentation (the "Security") which will be held by the Credit Union as Security for the loan and all other direct and indirect liabilities of the borrower and the guarantors (or any of them) to the Credit Union from time to time:

# **Existing Security (Presently held):**

- 1. All Corporate Documents and Resolutions as required
- 2. General Borrowing Resolution & Certificate of Non-Restriction
- 3. 1st GSA dated March 31, 2008 in the name of Travelers Acceptance Corporation registered in Alberta and British Columbia through March of 2018
- 4. 1st GSA Dated October 30, 2009 in the name of Crelogix Credit Group Inc. Registered in Alberta and British Columbia through October 2014
- 5. 1st GSA dated October 30, 2009 in the name of Crelogix Portfolio Services Corporation Registered in Alberta and British Columbia through October 2014
- 6. Overdraft Protection Agreement in the amount of \$2,000,000
- 7. Personal Guarantee and Postponement in the name of Karl Sigerist in the amount of \$530,400
- Personal Guarantee and Postponement in the name of Nicholas Carter in the amount of \$250,200
- 9. Personal Guarantee and Postponement in the name of Mike McKay in the amount of \$250,200
- 10. Personal Guarantee and Postponement in the name of Michael Mills in the amount of \$250,200
- 11. Corporate Guarantee from Crelogix Portfolio Services Corporation in the amount of \$2,000,000
- 12. Corporate Guarantee from Crelogix Credit Group Inc in the amount of \$2,000,000

	Init	ials
1		



May 31, 2013

# SCHEDULE "C" Pre-Disbursement & General Terms / Conditions

#### Pre-Disbursement Conditions (SCU Account Manager):

- 1. A copy of the updated Commitment Letter will be provided to the borrower's accountant
- 2. Commitment Letter / Offer of Financing signed by the borrower and guarantors

# General Conditions of Credit:

 Without limiting the Credit Union's right to make demand for payment at any time, all loans are subject to review from time to time at the Credit Union's discretion. The Credit Union reserves the right to withdraw its support at any time should there be, in the Credit Union's opinion, a material adverse change in the financial condition of the borrower(s) and/or guarantor(s) or an unacceptable change in ownership

1	Initials
Ì	



May 31, 2013

# SCHEDULE "D" Reporting Requirements & Pricing

The Borrower agrees to provide:

#### Weekly:

1. Copy of the weekly margining "warehouse" report that shows Accounts Receivables (Contracts) that are not already securitized (cutoff date for data to be Friday) (see Schedule "A" – Credit Limit, Conditions of Margining)

# Monthly:

1. In house year to date financial statements (Balance Sheet, Income Statement, list of Accounts Receivables and Accounts Payables) are to be provided

# Yearly:

- 1. Crelogix Credit Group Inc consolidated fiscal financial statements prepared by a CA, CMAS or CGA on a "Review Engagement" basis signed by the company's officers and received within 90 days of fiscal year end
- 2. Annual confirmation of property insurance renewal

# Other:

- 1. Updated personal net worth statements will be provided for Karl Sigerist, Michael McKay, Michael Mills, Nicholas Carter on a biennial basis
- 2. Any request for financial, reporting or other information on the status of operating, as may be from time to time, made by Servus Credit Union Ltd

#### Fees:

- Monthly monitoring administration fee has been waived
- Annual credit renewal fee has been waived

Init	ials



May 31, 2013

# SCHEDULE "E" Events of Default

If any one of the following Events of Default occurs, the Credit Union may, at its option, declare the unpaid principal amount of the advance and accrued interest thereon, to be immediately due and payable and the borrower shall forthwith pay all such amounts owing; the security pledged by the borrower shall become enforceable by the Credit Union, and the obligation for any further credit advances shall cease, when:

- 1. If any representation or warranty on the part of the borrower and set forth in this letter is incorrect in any material respect
- 2. The borrower fails to make any payment of interest or principal on schedule or
- 3. There is a breach by the borrower of any other terms or conditions contained in this Offer of Financing or
- 4. Any default occurs under any security specified in this Offer of Financing or under other credit or loan agreements to which the borrower is a party or
- 5. Any material adverse change occurs in the financial position of the borrower, as defined by the Credit Union or
- 6. Any bankruptcy, re-organization, arrangement, insolvency, or litigation proceedings, or other proceedings for the relief of debts are instituted by or against the borrower or
- 7. If a receiver is appointed over any of the borrower's property or undertaking or
- 8. Any of the aforesaid events occurs in respect of any Guarantor or Connected Borrower

Waiver: Any waiver by the Lender of any default by the borrower or any omission of the Lender's part in respect of any default by the borrower shall not extend to or be taken in any manner whatever to affect any subsequent default by the borrower or the rights resulting from it. This Lender may waive any condition precedent to funding but the waiver shall not prejudice any subsequent enforcement of the condition.

Initials



#### CRELOGIX ACCEPTANCE CORPORATION

# SCHEDULE "F" Insurance

# Insurance:

Loan offer to:

The borrower and the guarantors will insure and keep fully insured against the following perils:

- a. Public liability insurance to an amount not less than \$3,000,000 on an occurrence basis
- b. Key man life insurance policy over Karl Sigerist for a minimum of \$500,000 is to be in place and assigned to Servus Credit Union Ltd

The borrower will furnish to the Credit Union or its solicitors, at least ten (10) days prior to the advance of any funds, a binder policy, with certified copies of the policies being provided within 45 days thereafter, evidencing the above coverage.

Init	als

This is **Exhibit "E"** referred to in the Affidavit of Darcy Peelar, sworn before me this 5<sup>th</sup> day of July, 2017

Han

A Commissioner for Oaths in and for Alberta

Patrick T. Harnett Barrister and Solicitor

Servus Credit Union Ltd. 151 Karl Clark Road NW Edmonton AB T6N 1H5 servus.ca

September 8, 2016

Crelogix Acceptance Corporation

4445 Lougheed Highway #900

Burnaby, B.C.

V5C 0E4

Private & Confidential

Attention: Karl Sigerist - Director, President and Chief Executive Officer

Dear Mr. Sigerist,

Supplemental Credit Facility Letter to Commitment Letter dated May 7, 2008 and renewal Commitment Letter dated May 31, 2013

We are pleased to confirm that Servus Credit Union Ltd. (the "Credit Union") has authorized the following credit facilities in the name of Crelogix Acceptance Corporation subject to the following terms and conditions as well as those Standard Credit Terms and Conditions in Schedule "A" attached to and forming part of this Credit Facility Letter, and those set out in the Forbearance and Preservation of Rights Agreement dated September  $\underline{H}$ , 2016 (the "Forbearance Agreement"). All other terms and conditions of the Letter(s) stated above including all security and guarantees provided therein remain unchanged except as amended by this Credit Facility Letter.

BORROWER: Crelogix Acceptance Corporation

AMOUNT OF LOANS:

New Credil Facilities:

1) \$1,250,000 Authorized Overdraft Facility # TBD : Evergreen revolving operating facility to pay merchants. Expiry December 15, 2016

#### AVAILABILITY:

- 1) All loans available by way of direct Canadian Dollar advances.
- 2) For all of the credit facilities listed below, if funds are not fully disbursed within 90 days of the date of this credit facility letter or as otherwise agreed to by the borrower and the credit union, the availability of such credit facilities from the Credit Union shall, at the Credit Union's discretion, expire and be cancelled.
  - a) \$1,250,000 Authorized Overdraft Facility # TBD : Evergreen revolving operating facility.



Credit Facility Letter (07/10) | Page 1 of 7

#### INTEREST RATES:

Amounts advanced by the Credit Union to the Borrower will bear interest while outstanding, before and after maturity, default and judgment at the following rates:

#### New Credit Facility:

 Authorized Overdraft Facility #TBD: An annual rate of interest equal to the Credit Union's Prime Lending Rate plus 4.00%, floating, calculated daily and payable monthly in arrears.

The "Prime Lending Rate" referred to above shall mean the floating annual rate of Interest established and recorded as such from time to time by the Credit Union as its reference rate for determining rates of interest it will charge for loans denominated in Canadian Dollars and commonly called the Credit Union's Prime Lending Rate, adjusted automatically upon any change by the Credit Union. The Credit Union's Prime Lending Rate is 2.70% per annum as at the date of this Credit Facility Letter.

Notwithstanding anything to the contrary contained herein, the Credit Union may, in its discretion, make an advance under the Authorized Overdraft, or make a reduction from the advance otherwise requested under the loans, to pay any interest that has become due and payable.

Interest shall be payable willhout demand and, unless otherwise specified herein, shall be calculated daily and payable monthly in arrears on the date specified by the Credit Union and will accrue daily. Overdue Interest shall bear interest at the same rate as that used in calculating the interest overdue, while the same remains unpaid.

#### **REPAYMENT TERMS:**

The Borrower shall repay all loans on demand. Prior to demand by the Credit Union, loans shall be repayable as follows:

1) Authorized Overdraft Facility # TBD : Revolving; Interest payable monthly.

#### GENERAL FEES:

At the time of this Credit Facility Letter, the Credit Union's fees are, in addition to those fees set out in the Forbearance Agreement, as follows:

- An Application Fee is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Credit Facility Letter. The Application Fee Is broken down as follows:
  - a) Authorized Overdraft Facility: \$25,000 (2% of the amount)

Total Application Fee: \$25,000

- An Account Review Fee of \$4,500 will be due and payable annually by the Borrower after completion of our Account Review.
- 3) A late reporting fee for annual reporting requirements of \$250 is due and payable by the Borrower at the end of the month in which the reporting due date occurs and will continue monthly until the reporting requirements are met.
- 4) Search and Credit Investigation Fees are to be paid by the Borrower to the Credit Union.

10978407.3

Credil Facility Letter (07/16) | Page 2 of 7



- 5) The Borrower agrees to forthwith pay to the Credit Union a charge for each cheque presented for deposit which is dishonoured or, in the Credit Union's absolute discretion, a late payment fee whenever a payment is not remitted on its due date at the Credit Union's normal charges in effect from time to time respecting dishonoured cheques and/or late payment fees.
- 6) All fees due and payable, as outlined in this Credit Facility Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's toans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all toans by the Credit Union to the Borrower.

#### COMMERCIAL REPORTING REQUIREMENTS:

- 1) Annually, within 90 days of the Borrower's fiscal year end:
  - a) Crelogix Credit Group Inc. consolidated fiscal financial statements prepared by a CA, CMAS or CGA on a "Review Engagement" basis signed by the company's officers and received within 90 days of fiscal year end
  - b) Signed Personal Net Worth Statements on approved Credit Union form, the following updated at least blennially or sooner if requested by the Credit Union:
    - Karl Sigerist
    - Michael McKay
    - Michael Mills
- 1) Periodically, as requested:
  - Any request for financial, reporting or other information on the status of operating, as may be from time to time, made by Servus Credit Union Ltd.
  - b) Ongoing Monitors reports as applicable.
  - c) Securilization information, merchant payable list, etc.
  - d) 30 day cash flow forecast as required

GENERAL CONDITIONS OF CREDIT:

#### Authorized Overdraft Facility #TBD:

- All security documentation to be completed and registered with a solicitor's letter of opinion confirming that the Credit Union's security is a valid and enforceable first charge and that any prior encumbrances will not affect that first registered position.
- The Credit Union's solicitor is to confirm all proper signing operating account documents and enabling resolutions have been or will be executed.
- 3) Satisfactory PPR Searches in British Columbia
- CAFT to be transferred from \$2,000,000 existing warehouse line to the new \$1,250,000 AOD Evergreen.
- All legal costs incurred in this transaction are for the account of the Borrower, including those payable to the solicitor representing the Credit Union.
- 6) All costs (i.e. security registration, appraisals, inspections, etc.) will be for the account of the borrower.

19978407.3

Credit Facility Letter (07/16) | Page 3 of 7

7) All security documentation will be prepared by the Credit Union Solicitor and forwarded to the Borrower's solicitor for execution and provision of an opinion as to the ability of the Borrower to borrow and pledge security. The Credit Union's solicitor will register and provide and opinion as to enforceability.

#### CONDITIONS OF CREDIT FOR AOD EVERGREEN FACILITY:

· ...

- 8) Credit Union may at any time, in its sole discretion and without notice to the Borrower, suspend or cancel any credit availability under the AOD Evergreen Facility
- Pay Merchants on Tuesdays and Fridays. (able to adjust as volume dictates)
- 10) Maximum advance is \$1,250,000 per CAFT/cheques to Merchants per tranche, in the Credit Union's sole discretion
- 11) Crelogix to submit list of contracts / merchants to be paid to Credit Union
- 12) Crelogix to submit draft signed Funder securitization package to Credit Union
- 13) Credit Union/Monitor to review lists/Funder Securitization package to ensure they are valid contracts, amounts on contracts correspond with amount payable to merchants, merchants getting paid on the contracts are the same as the contracts being sold to the funder, etc.
- 14) Credit Union to temporarily increase AOD/CAFT limit to accommodate request, in Credit Union's sole discretion
- 15) Crelogix to originate CAFT to pay merchants, not more than \$1,250,000.
- 16) Crelogix to provide executed copy of the securitization package and bulk assignment to Credit Union
- 17) Credit Union and/or Monitor to confirm Credit Union is listed as payee on Funder Bulk Assignment (clause #8) and/or "Irrevocable assignment and Direction to Pay" to be acknowledged by the Funder and it can't be amended without Credit Union prior written consent.
- 18) Crelogix to securitize package with the Funder. All contracts to be securitized to the Funder shall be paid for by Crelogix prior to securitization. Crelogix shall provide to Credit Union and the Monitor, prior to securitizing contracts to the Funder, evidence and confirmation that all contracts to be securitized by the Funder have been paid for in full by Crelogix.
- 19) Crelogix shall execute an Irrevocable Assignment and Direction to Pay in respect of each bundle of contracts to be securitized, directing the Funder to pay to Credit Union any and all amounts payable to Crelogix under each respective Funder Bulk Assignment
- 20) Funder to credit Oredit Union account within 3 days via EFT pursuant to the Irrevocable Assignment and Direction to Pay
- 21) Proceeds to be allocated based on 30 day cash flow provided by Crelogix /Monitor and acknowledged by Credit Union
- 22) Account is not to be in an excess position, the amount to be received from the Funder is always to be greater that the Merchant's payable.

Subject to change at Credit Union's sole discretion at any time.

19976407.3

Credit Facility Letter (07/10) | Page 4 of 7

#### **REVIEW:**

All loans are subject to review at any time, and in any event will be reviewed annually. Our next account review would be scheduled for February 1, 2017 based on a full review of financial statements of the Borrower and Guarantors to the year ended.

#### SECURITY APPLICABLE TO THIS CREDIT FACILITY:

The types of security, supporting resolutions and agreements to be provided by the Borrower will be in a form and content determined by the Credit Union and/or its solicitors and include the following:

Security Presently Held:

- 1) All Corporate Documents and Resolutions as required
- 1st GSA dated March 31, 2008 in the name of Travelers Acceptance Corporation registered in Alberta and British Columbia
- 1st GSA Dated October 30, 2009 In the name of Crelogix Credit Group Inc. Registered in Alberta and British Columbia
- 1st GSA dated October 30, 2009 In the name of Crelogix Portfolio Services Corporation Registered in Alberta and British Columbia
- 5) Corporate Guarantee from Crelogix Perifolio Services Corporation in the amount of \$2,000,000
- 6) Corporate Guarantee from Crelogix Credit Group Inc. in the amount of \$2,000,000
- 7) Personal Guarantee from Karl Sigerist In the amount of \$530,400
- 8) Personal Guarantee from Mike McKay in the amount of \$250,200
- 9) Personal Guarantee from Michael Mills in the amount of \$250,200
- 10) Personal Guarantee from Nicholas Carter in the amount of \$250,200

Security To Be Obtained (if appropriate, all security documents to be replatered in the appropriate government or other registry as required or desirable):

1) Execution of Credit Facility Letter

- 2) General Borrowing Resolution & Certificate of Non-Restriction
- 3) Executed Forbearance Agreement

4) Irrevocable Assignment and Direction to Pay in favour of Credit Union for each transaction subject to a Funder Bulk Assignment

#### SECURITY APPLICABLE TO SPECIFIC CREDIT FACILITIES:

The types of security, supporting resolutions and agreements to be provided by the Borrower will be in a form and content determined by the Credit Union and/or its solicitors and registered in the appropriate government or other registry as required or desirable, and includes the following:

19978-107.3

Credit Facility Letter (07/16) | Page 5 of 7

#### ACCEPTANCE:

- 1) To become effective this Credit Facility Letter must be accepted in writing by the Borrower and Guarantors.
- 2) This Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Credit Facility Letter,

 Your acceptance of this Credit Facility Letter will constitute authority for the Credit Union to instruct its solicitors to prepare the necessary documentation.

Please note that this Credit Facility Letter restates and modifies and takes precedence over any prior Commitment Letters or Credit Facility Letters issued to you; however, it is not a novation. All existing security and guarantees held by the Credit Union continue in full force and effect and are security for the loans.

Nothing herein shall be construed to impair any security, iten, or charge held by the Credit Union to secure the loans and Credit Facilities and nothing herein shall affect or impair any powers which the Credit Union may have against the Borrower, any Guarantors or any other person for recover of the loans and Credit Facilities.

If you are in agreement with the above terms and conditions, as well as the Standard Credit Terms and Conditions as detailed in the attached Schedule "A" – Standard Credit Terms and Conditions which form part of this Credit Facility Letter, please sign this letter and return this letter with the attached Standard Credit Terms and Conditions to the Credit Union together with the applicable fee. This Credit Facility Letter will expire if not accepted or extended in writing by September 15, 2016. The foregoing is offered in good faith and is to be held in strict confidence.

Yours truly,

Servus Credit Union Ltd.

per: **Richard Chariton** VP Commercial Credit, South

per:

19978407.3

Credit Facility Letter (07/16) | Page 6 of 7

Accepted this 15th day of September, 2016

# BORROWER(S):

ني

Crelogix Acceptance Corporation

per: <

per:

GUARANTOR(S):

Crelogix Portfolio Service Corporation

per: 💋 P

Crelogix Credit Group Inc.

per

Karl Sigerist

Mike Mill Mike Mc

19978407.3

Credit Facility Letter (07/16) | Page 7 of 7

Accepted this \_\_\_\_\_ day of September, 2016

# BORROWER(S):

1

Crelogix Acceptance Corporation

per:

per:

GUARANTOR(S):

Crelogix Portfolio Service Corporation

per:

Crelogix Credit Group Inc.

per: \_\_\_\_\_

Karl Sigerist

------

Mike Mills

Mike McKay

19978407.3

Credit Facility Letter (07/16) | Page 7 of 7



#### SCHEDULE "A"

#### STANDARD CREDIT TERMS AND CONDITIONS

Borrower: CRELOGIX ACCEPTANCE CORPORATION

Credit Facility Letter dated SEPTEMBER 8, 2016

and signed on the 15th day of September . 2016

All references to "Credit Union" shall mean Servus Credit Union Ltd. In this Schedule "A" unless otherwise specified. All references herein to "the Credit Facility Letter" shall mean the Credit Facility Letter to which this Schedule "A" is attached and in the event of a conflict between the terms of the Credit Facility Letter and this Schedule "A", the terms of the Credit Facility Letter shall prevail but for clarity the mention of a provision in the Credit Facility Letter and not in Schedule "A" or *vice versa* shall not constitute a conflict but shall be deemed to be supplemental and In addition to any of the terms and conditions under either of the Credit Facility Letter or Schedule "A" as the case may be.

#### ACCEPTANCE, ADVANCES, PAYMENTS, EXPENSES, FEES AND CONSENTS:

- Acceptance: The Borrower and all Guarantors must accept in writing the terms and conditions of the Credit Facility Letter prior to any advances being made.
- 2) Evidence of Advances: The Borrower and all of the Guarantors agree that the Credit Union's records evidencing an advance shall be complete and final proof, absent manifest error, that funds have been advanced under any one or more of the loans set forth in the Credit Facility Letter and may, from time to time dependent upon the type of loan facility made available, be evidenced by other documentation such as, for example and without limitation, promissory notes, direct deposits, drafts or cheques made payable to other parties including solicitors and agents and any other means by which the Credit Union provides value to the Borrower under any one or more of the loan facilities.
- Debit from Borrower Account: Unless otherwise stated in writing, payments for all loans and credits will be automatically transferred or debited from the Borrower's operating account with the Credit Union.
- 4) <u>Payment of Costs:</u> The Borrower agrees to pay all expenses, fees and charges incurred by the Credit Union in relation to all loans and credits, the preparation and registration of all security, enforcement or preservation of any or all of the Credit Union's rights and remedies including those incurred during an annual or any other periodic review of the Borrower's relationship with the Credit Union, whether or not any such documentation is completed or any funds are advanced, including but not limited to legal expenses (on a solicitor-and-his-own-client full indemnity basis), costs of accountants, engineers, architects, consultants, appraisers and the costs of any and all searches and registrations the Credit Union or its solicitor deems either necessary or desirable.
- 5) <u>Change in Fees:</u> The Borrower and all Guarantors acknowledge that the Credit Union may change the fees payable pursuant to the Credit Facility Letter from time to time upon notice to the Borrower in person, by telephone, by letter that is sent either by mail or facsimile transmission or by electronic mall (e-mail) to the address, telephone number and/or electronic mail (e-mail) address on file at the Credit Union.
- 6) <u>Not Assumable</u>: All loans and credits are not assumable without the Credit Union's prior, written consent and if the Borrower does attempt to have some other entity assume any loan or security, any and all loans and credits shall, in the Credit Union's discretion, become immediately due and payable and the Credit Union may commence enforcement.

#### CONDITIONS PRECEDENT TO DRAWDOWN:

Schedule "A" Standard Credit Terms and Conditions 012016

- Security Completion: Prior to funding any loans or credits or making any further advances, all security as contemplated by the Credit Facility Letter must be completed and registered wherever required or desirable by and in form and content acceptable to the Credit Union and its solicitors, and all certificates, searches, solicitor's opinions and other documents as required by the Credit Union shall be delivered to the Credit Union in form and content acceptable to the Credit Union.
- 2) <u>Title Insurance:</u> With respect to real estate transactions including, without limitation, mortgages, the Credit Union may fund upon receipt of an acceptable title insurance policy issued by any major title insurer including at this time Stewart Title Insurance, Chicago Title Insurance and First Canadian Title.

#### **REPRESENTATIONS AND WARRANTIES:**

The Borrower, all Corporate Guarantors and any other Guarantors represent and warrant to the Credit Union that:

- <u>Corporate Status</u>: If a corporation, it is duly incorporated, validly existing and duly registered and qualified to carry on business in the Province of Alberta and in all other jurisdictions where it carries on business and shall maintain such corporate existence and registration at all times during which any money may be owing to the Credit Union or it shall be liable to the Credit Union;
- 2) <u>Authorizations:</u> The execution, delivery and performance by it of this Credit Facility Letter and any and all terms and conditions thereunder including provision of security have been duly authorized by all necessary actions and do not violate its constating documents or any applicable Laws or agreements to which it is subject or by which it is bound;
- 3) Environmental Claims: There are no claims, actions, prosecutions or other proceedings of any kind pending or threatened against it or any of its assets or properties before any court or administrative tribunal or agency which relate to any noncompliance with any environmental law or any release from its lands of a contaminant into the natural environment or which, if adversely determined, might have a material adverse effect upon its financial condition or operations or its ability to perform its obligations under the Credit Facility Letter or under any of the Credit Union's security and that there are no circumstances of which they are aware which might give rise to any such proceeding which has not already been fully disclosed to the Credit Union;
- 4) <u>Claims</u>: There are no claims, actions, prosecutions or other proceedings of any kind pending or threatened against it or against any of their assets or properties before any court or other administrative agency which relate to any noncompliance with any other applicable law or which, if adversely determined, might have a material adverse effect upon their financial condition or operation or their ability to perform their obligations under the Credit Facility Letter or in any of the Credit Union's security, and that there are no circumstances of which they are aware that might give rise to any such proceeding that have not already been fully disclosed to the Credit Union;
- 5) <u>Crime Proceeds and Terrorism</u>: That they are not in violation of any applicable law relating to terrorism or money laundering, including the *Proceeds of Crime (Honey Laundering) and Terrorist Financing Act* (Canada).
- 6) <u>Accuracy of Information and Accounting</u>: That all Information provided to the Credit Union is complete and accurate and does not omit any material fact and all financial statements delivered by it to the Credit Union fairly represent its financial condition as of the date of such financial statement all in accordance with Accounting Standards for Private Enterprises (ASPE) or International Financial Reporting Standards (IFRS), as the case may be;
- 7) <u>Good Title/Interest:</u> In respect of all property and assets subject to Credit Union security, it has good and marketable title or a valid interest in such property and assets free and clear of all claims and encumbrances except those claims and encumbrances to which the Credit Union has provided its prior consent in writing;
- No Default: There is no default or event or default that has occurred or is occurring as of acceptance of the Credit Facility Letter;

- <u>Material Adverse Change:</u> To the best of their knowledge no event or circumstance has occurred or is continuing which has or would reasonably be expected to have a material adverse impact on the operations or financial condition of the Borrower or any Guarantor;
- 10) <u>Government Remittances:</u> That all material remittances required to be made to government authorities have been made, are currently up to date and that there are no outstanding arrears, other than those that are being validly contested and that have been disclosed to the Credit Union;
- 11) <u>Tax Payments</u>: That the Borrower and all Guarantors have duly filed on a timely basis all tax returns or other information required to be filed and have paid all material taxes which are due and payable including all assessments and reassessments other than those that are being validly contested and that have been disclosed to the Credit Union and that they have made adequate provision for, and all required instalment payments have been made in respect of taxes payable for the current period for which returns have not as yet been required to be filed and that there are no agreements, waivers of other arrangements providing for any extension of time with respect to the filing of any tax returns or the payment of any taxes and that no actions or proceedings are being taken or threatened by any taxation authority in any jurisdiction where a Borrower or Guarantor carries on business other than those that are being validly contested and that have been disclosed in writing to the Credit Union; and,
- 12) <u>Intellectual Property:</u> That they have the legal right to use all intellectual property necessary for the operation and conduct of their businesses, affairs, operations and processes and that they will continue to maintain such legal right so long as any monies are owing to the Credit Union.

All representations and warranties set out herein shall survive the acceptance of the Credit Facility Letter and shall be deemed to be repeated at the time of each advance hereunder and the Credit Union may rely upon them upon each advance pursuant to the Credit Facility Letter except for those representations and warranties that the Credit Union has been notified of in writing that can no longer be repeated for any such advance. Failure to provide such notification shall, in the Credit Union's discretion, be considered an evant of default and all credits may become immediately due and payable and the Credit Union may proceed to enforce the same in its discretion.

All other representations, warranties, certifications and statements of the Borrower or any Guarantor contained in any other document delivered pursuant to the Credit Facility Letter or thereafter to the Credit Union shall be deemed to constitute additional representations and warranties made by the Borrower or any Guarantor to the Credit Union.

#### GENERAL COVENANTS:

Each of the Borrower and each Guarantor covenants and agrees with the Credit Union, that for so long as any monies are due and payable to the Credit Union as follows:

- 1) <u>Timely Payment:</u> To pay all sums of money when due to the Credit Union.
- Performance of Obligations: To perform all of the obligations and covenants under the Credit Facility Letter or under any security document.
- Maintenance of Security: To maintain in full force and effect any security contemplated by the Credit Facility Letter and any security that may be granted thereafter.
- 4) <u>Notice of Default:</u> To provide the Credit Union with prompt written notice of any event that constitutes, or which, with notice, lapse of time, or both, would constitute an Event of Default under the Credit Facility Letter or any security.
- 5) <u>Change of Ownership:</u> To give the Credit Union at least 30 days' prior written notice of any intended change in the ownership of its shares and not to consent to or facilitate a change in the ownership of its shares without prior written consent of the Credit Union.
- 6) <u>Insurance</u>: The Borrower, and the Guarantors shall insure and keep fully insured all property and assets in accordance with insurance requirement in the Credit Facility Letter so long as any monies are owing to the Credit Union.

Schedule "A" Standard Credit Terms and Conditions 012016

- 7) <u>Guarantees:</u> Not, without the prior written consent of the Credit Union, guarantee or otherwise provide for, on a direct, indirect or contingent basis, the payment of any monies or performance of any obligations by any other entity, except as may be provided for in the Credit Facility Letter.
- 8) <u>Sale of Property:</u> Not, without the prior written consent of the Credit Union, to sell, transfer, convey, lease or otherwise dispose of any of its property or assets subject to security under the Credit Facility Letter except where such is done in the ordinary course of the Borrower's business operations.
- Merger: Not, without the prior written consent of the Credit Union, to merge, amalgamate, or otherwise enter into any other form of business combination with any other entity.
- 10) <u>Environmental Non-Compliance:</u> To provide the Credit Union with prompt written notice of any noncompliance by the Borrower with any environmental laws or any release from the land of the Borrower by the Borrower of a contaminant into the natural environment and to indemnify and save harmless the Credit Union from all liability of loss as a result of such environmental activity or any noncompliance with any environmental law.
- <u>Illegal Activity</u>: To not engage or allow any person in any of its business premises to engage in any activity that is contrary to any applicable laws and in particular any laws prohibiting criminal or illegal activities.
- 12) <u>Personal Information</u>: The Borrower and each Guarantor acknowledge that the Credit Union is required to verify and record information regarding the Borrower and each Guarantor, their directors, authorized signing officers, shareholders and any persons in control of the Borrower and each Guarantor and that they shall each promptly provide all such information including such other supporting documentation and other evidence as may be required by the Credit Union or any assignee or other entity participating in any credits with the Credit Union in order to comply with not only the Credit Union's internal identification policies but all applicable laws regarding anti-money laundering and "anti-terrorist financing".
- 13) <u>Inspection</u>: To permit the Credit Union or its representatives, from time to time, to visit and inspect the Borrower's premises, properties and assets and examine and obtain copies of the Borrower's records or other Information and to discuss or otherwise communicate about the Borrower's affairs with the auditors, accountants, counsel and other professional advisers of the Borrower from time to time as the Credit Union may deem necessary.
- 14) <u>Monthly Books and Records:</u> That it shall keep proper books of record and account in which complete and correct entries will be made of all of its business transactions in accordance with ASPE or IFRS.
- 15) <u>Prudent Operation</u>: That it will keep all of its properties, assets and operations maintained and operated in accordance with diligent and prudent industry practice and in accordance with the law in compliance with any applicable insurance policy or policies covering such assets or activities.
- 16) Use of Loan Proceeds: That it shall use the proceeds of all loans and credits being made available to it pursuant to the Credit Facility Letter or otherwise solely for the purposes set forth thereunder and for no other purpose whatsoever without the prior, written consent of the Credit Union.
- 17) <u>Related Party Dealings</u>: The Borrower shall not, without the prior written consent of the Credit Union, enter into any contract, agreement or transaction whatsoever including for the sale, purchase, lease or other dealing in any property or provision of any service with any non-arms' length entity or any related party as defined in the *Business Corporations Act* of Alberta except upon fair and reasonable terms, which terms are not less favourable than it would obtain on a arms' length transaction and for a consideration which equals the fair market value of such property or other than a fair market rental as regards lease property.
- 18) <u>Other Encumbrances:</u> Not, without the prior written consent of the Credit Union, grant, create, assume or suffer to exist any mortgage, charge, lien, pledge, security interest or other encumbrance affecting any of its properties, assets or any other rights.

Schedule "A" Standard Credit Terms and Conditions 012016

- 19) <u>Exclusive Account Operation:</u> That for so long as there are any monies due and owing or any Credit Facility outstanding with the Credit Union, maintain all of its operating accounts with the Credit Union.
- 20) <u>Payment of Management or Shareholder Fees:</u> The Borrower will not pay or agree to pay any management or shareholder fees or executive management compensation except as agreed and approved by the Credit Union.

Nothing contained in the foregoing Covenants shall limit any right of the Credit Union under the Credit Facility Letter or any other agreement with the Borrower to terminate or demand payment of, or cancel or restrict availability of any unutilized portion of, any demand or other discretionary loan or credit facility made available by the Credit Union to the Borrower.

#### MISCELLANEOUS:

1. F

- <u>Cumulative Powers of Credit Union</u>: The rights and powers of the Credit Union pursuant to the Credit Facility Letter and the securities taken pursuant hereto are cumulative and not alternative, and not in substitution for any rights, remedies, or powers of the Credit Union.
- Failure: Any failure or delay by the Credit Union to exercise fully its rights and remedies pursuant to this Credit Facility Letter and the securities taken to pursuant hereto shall not be construed as a waiver of such rights and remedies.
- 3) <u>Time:</u> Time is of the essence.
- 4) Non-Assignability: This Credit Facility Letter is not assignable by the Borrower in any manner.
- Laws of Alberta: This Credit Facility Letter and the security documentation to be provided by the Borrower pursuant hereto shall be construed in accordance with and governed by the laws of the Province of Alberta.
- 6) <u>Cross-Defaultr</u> Any default hereunder or under any security document or other agreement between the Borrower and the Credit Union shall be a default under each and every other obligation of the Borrower to the Credit Union, whether or not collateral or supplemental hereto.
- 7) <u>Conflict:</u> The terms and conditions of this Credit Facility Letter shall not be merged by and shall survive the execution, delivery and registration of any and all security documents. In the event of a conflict between the terms of this Credit Facility Letter and the terms of any security document, the terms of this Credit Facility Letter shall prevail. For clarity, the mention of a provision in either the Credit Facility Letter and not in the security or vice versa shall not constitute a conflict but shall be deemed to be supplemental and in addition to any of the terms and conditions available under either the Credit Facility or the security as the case may be.
- 8) <u>Periodic Review of Accounts</u>: The Credit Union may conduct periodic reviews of the affairs of the Borrower, as and when determined by the Credit Union, for the purpose of evaluating the financial condition of the Borrower. The Borrower shall make available to the Credit Union such financial statements and other information and documentation as the Credit Union may reasonably require and shall do all things reasonably necessary to facilitate such review by the Credit Union.
- 9) Limitation Period: The Borrower and all Guarantors by their signature and acceptance of the Credit Facility Letter hereby expressly provide that any and all limitation periods or action on any and all loans and credit facilities made available from time to time pursuant to the Credit Facility Letter or otherwise by the Credit Union shall be extended for a period of six years from the date of any event of default on a non-demand loan and for a period of six years on any demand loan from the time at which a demand for payment is made.
- 10) <u>Application of Account Balances</u>: The Credit Union is authorized, but not obligated, at any time, to apply any credit balance, whether or not then due, to which the Borrower is entitled and any account in any currency at any branch or office of the Credit Union in or towards satisfaction of the loans and obligations of the Borrower to the Credit Union at any time. The Credit Union is authorized to use any

such credit balance to convert such credit balance to any currency required as may be necessary to effect such application.

- 11) <u>Non-Waiver</u>: No amendment or waiver of any provision of this agreement or any agreement amending, supplemental or relating hereto, will be effective unless it is in writing signed by the Borrower and the Credit Union. No failure or delay, on the part of the Credit Union in exercising any right or power hereunder or under any security document taken pursuant to the Credit Facility Letter or otherwise shall operate as a waiver thereof. The Guarantors agree that the amendment or waiver of any provision of this agreement (other than agreements, covenants or representations expressly made by the Guarantors herein, if any) may be made without and does not require the consent or agreement of, or notice to , the Guarantors.
- 12) <u>Enforceability</u>: If any provision of the Credit Facility Letter or any other agreement is or becomes prohibited or unenforceable in any jurisdiction, such prohibition or unenforceability shall not invalidate or render unenforceable the provision concerned in any other jurisdiction nor invalidate, affect or impair any of the remaining provisions of the Credit Facility Letter or any other agreement and such remaining provisions or parts thereof that are not invalid, illegal or unenforceable or severeable from such provision.
- 13) <u>Joint and Several Liability:</u> Where more than one person is liable as a Borrower or Guarantor for any loan or credit facility or obligation under the Credit Facility Letter or otherwise then the liability of each such person is joint and several with each other such person.
- 14) <u>Mandatory Membership</u>: Membership with Servus Credit Union requires that every Borrower invest a minimum of \$1.00 in Common Shares of the Credit Union Ltd. and such ownership and membership must be maintained so long as there are any monies and obligations outstanding by the Borrower to the Credit Union Ltd.
- 15) <u>Accuracy of Information:</u> The Borrower and Guaranter represent and warrant to the Credit Union that all information set out and certified in any electronic transfer and on any accompanying report is true and complete in all respects and acknowledges that the Credit Union Ltd. is relying upon all such representations and warranties from time to time.
- 16) Entire Agreement: This Credit Facility Letter, the security and any other written agreement delivered pursuant to or referred to in the Credit Facility Letter constitute the whole and entire agreement between the Credit Union and the Borrower and Guarantors in respect of the credit facilities hereunder. There are no verbal agreements, undertakings or representations by the Credit Union in connection with any loans or other credit made available to the Borrower at any time.
- 17) <u>Non-Merger</u>: All remedies provided herein shall be deemed to be in addition to and not restrictive of any other remedies of the Credit Union at law or in equity, may be enforced in priority to, or concurrently with, or subsequent to any other remedy or remedies, the Credit Union may rely upon the various securities and parts thereof in such order as it may deem fit without prejudice to any other realization and the powers of sale contained therein. The security provided for in this Credit Facility Letter is in addition to and not in substitution for any other security now or hereafter held by the Credit Union.
- 18) <u>Waiver Under PPSA:</u> The Borrower expressly waives the right to receive any copies of any Financing Statements or Financing Change Statements (or any other jurisdiction equivalent) that might be registered by the Credit Union in connection with any security or any Verification Statement issued with respect therato including all amendments, extensions or renewals of such registration and in any jurisdiction where not otherwise prohibited by law.
- 19) <u>Accounting Standards</u>: In the event that the Borrower or any Guarantor as a result of the changes to Canadian Accounting Standards on January 1, 2011 to the International Financial Reporting Standards ("IFRS"), Accounting Standards for Private Enterprises and/or Accounting Standards for Not-For-Profit Organizations and such adoption has an effect on any provision of the Credit Facility Letter relying on

, 1<sub>1</sub> +

financial statement calculations or other financial reporting requirements, the Credit Union may amend such provision to reflect the original intent of such provision at any time.

#### SALE OR ASSIGNMENT OF CREDIT FACILITIES:

, 1. t

- Assignability: The Credit Union shall have the unrestricted right to sell or assign the Credit Facilities or any loan thereunder, and/or the security documents (including this Credit Facility Letter), in whole or in part, in connection with any syndication, securitization or otherwise, to any other party or parties (each a "Holder"), and the Holder(s) shall thereafter have all the rights herein of the Credit Union, including the right to so sell or assign in turn.
- 2) <u>Syndication:</u> The Credit Union may from time to time, in connection with any syndication or securitization of the Credit Facilities or loans thereunder or otherwise, appoint or designate a custodian or agent for the same, which custodian or agent may be the registered security document holder. The Borrower and each Guarantor, if any, acknowledges that such custodian or agent will have no liability whatsoever to the Borrower or Guarantor, if any, in connection with the Credit Facility or loans thereunder, being merely custodian or agent for the Credit Union and/or Holders.
- 3) Loan Administration: The Credit Union shall have the unrestricted right from time to time to appoint a third party to service or administer the Credit Facilities or loans, and to deal with the Borrower and Guarantor, if any, in place of the Credit Union, provided that until the Credit Union gives notice of such appointment to the Borrower, the Borrower and Guarantor, if any, shall continue to deal with the Credit Union in matters pertaining to the servicing or administration of the Credit Facilities and loans.

# COLLECTION, USE, DISCLOSURE AND RELEASE OF FINANCIAL AND OTHER INFORMATION AND MATERIALS:

For the purposes of making, administering, reporting, selling or assigning in whole or in part, in connection with securitization or otherwise, and collecting the Credit Facilities and loans, the following parties (collectively, the "Authorized Parties") will be reviewing and examining financial and other information and materials provided to or obtained by the Credit Union concerning the Credit Facilities and loans, the Borrower and the Guarantor, if any:

- The Credit Union and/or any Holder or servicer of the Credit Facilities and loans or of an interest therein from time to time and/or their respective affiliates and/or agents;
- 2) Rating agencies, purchasers or investors and prospective purchasers or investors;
- 3) Respective third party advisors of the parties listed in 1) and 2) above, such as lawyers, accountants, real estate brokers, investment dealers and underwriters, consultants, and appraisers; and,
- 4) Credit verification sources.

The Borrower and each Guarantor, if any, acknowledges and irrevocably consents to the foregoing and irrevocably agrees that, in such manner as the Authorized Parties may determine to be necessary or desirable for these purposes, the Authorized Parties may disclose, release, exchange and share such information and materials:

- To and with any Individual(s), corporation(s) or other entities designated from time to time to hold title to the Credit Facilities or loans and/or security documents as custodian(s) or agent(s);
- 2) To and with each other;
- 3) The Borrower and each Guarantor, if any, hereby consents to the Authorized Parties conducting such credit Inquiries, as they may from time to time consider advisable for these purposes; and,
- 4) The provisions of this paragraph shall apply until all loans have been fully and completely repaid and the security documents have been discharged.

#### EVENTS OF DEFAULT:

The occurrence of any one of the following shall constitute an Event of Default, the happening of which shall entitle the Credit Union, in its sole discretion, to demand immediate payment of all loans and credits in full, Schedule "A" Standard Credit Terms and Cenditions 012015 Page 7 of 8

together with outstanding accrued interest and any other costs outstanding, and to realize and enforce on any and all of the security granted in its favour under the Credit Facility Letter or otherwise:

- <u>Payment:</u> Failure of the Borrower to pay any principal, interest or other amount due and owing at any time.
- Breach of Term: Failure or refusal of the Borrower to observe or perform any term, covenant, condition
  or provision contained in this Credit Facility Letter or any documentation or security relating thereto.
- 3) <u>Cross-Default:</u> If the Borrower is in default under any other agreement with the Credit Union or any third party for the granting of the loan or other financial assistance and such default remains unremedied after any cure period provided in any other such agreement.
- 4) <u>Breach of Representation or Warranty:</u> If any representation or warranty made by the Borrower or any Guarantor in any document (including the Credit Facility Letter) is breached, faise or misleading in any material respect or becomes at any time false or misleading in any material respect.
- 5) <u>Accuracy of Documentation:</u> If any schedule, certificate, financial statement report or other writing furnished by the Borrower or any Guarantor to the Credit Union in connection with the Credit Facility Letter or any other agreement is false or misleading in any material respect on the date on which it is certified or stated.
- 6) Insolvency: The Borrower or any Guarantor becomes insolvent or generally fails to pay its just debts as they become due or they apply for, consent to or acquiesce in the appointment of a trustee, receiver or other custodian for the Borrower or any Guarantor or any property thereof or makes a general assignment for the benefit of creditors, or for a trustee, receiver or other custodian is appointed for the Borrower or Guarantor or for a substantial part of the property of such Borrower or Guarantor, or any bankruptcy, reorganization, debt arrangement or other case or proceeding under any bankruptcy or insolvency law, or any dissolution or liquidation proceeding, as commenced in respect of the Borrower or Corporate Guarantor or if they take any action to authorize or further any of the foregoing.
- <u>Cessation of Business</u>: The Borrower or Guarantors ceases or threatens to cease to carry on all or substantially all of their business.
- 8) <u>Maroin Report Delivery</u>: If margining reports are not received within 10 days of a reporting deadline as set forth in the Credit Facility Letter and the Credit Union determines it would issue a Notice of Default with respect to such reporting.
- 9) <u>Margin Report Demand:</u> If such margining reports have still not been received within 14 days from the issuance of a Notice of Default, the Credit Union Ltd. has the right to reduce all lines of credit and outstanding authorizations to zero.
- 10) <u>Material Adverse Change</u>: There occurs, in the sole opinion of the Credit Union, acting reasonably:
  - a) A material adverse change in the financial or environmental condition of the Borrower or any Guarantor,
  - b) An unacceptable, or unapproved change in ownership or control of the Borrower or any Guarantors takes place, or
  - c) Without authorization, the Borrower disposes of all or substantially all of its key business assets or substantially all of its assets.

#### **OTHER SECURITY:**

1 10 2

In addition to the security referred to herein, the Borrower hereby pledges all deposits and paid up shares which it now or may have in the Credit Union, the proceeds of which may, upon default, be applied by the Credit Union to amounts then due and owing under any loan.



Servus Credit Union Ltd. 151 Karl Clark Road NW Edmonton AB T6N 1H5 servus.ca

November 15, 2016

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC VSC DE4

Private & Confidential

Attention: Karl Sigerist

Dear Mr. SIgerist,

RE: Amending Credit Facility Letter to Letter dated September B ,2016

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letter stated above including all security and guarantees provided therein remain unchanged.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

#### FEES:

- An Amendment Fee of \$1,000 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

#### **REVISED TERMS AND CONDITIONS:**

Evergreen Revolving Facility #1 for \$1,475,000 (temporary increase of \$225,000, not margined)

A temporary increase of \$225,000 to the existing \$1,250,000 facility to pay the EPO liability amount due to the Funders on approximately November 10, 2016. The excess amount (\$225,000) is to be covered by securitization proceeds by November 23, 2015. Approximately \$128,000 is to be covered by November 18, 2016 from Bulk Assignment #333. Once the \$128,000 is deposited the account on November 18, 2016, the excess is reduced to \$97,000, which amount shall be deposited in the account on or before November 23, 2016, from a further Bulk Assignment contract.

Your acceptance of this letter will constitute authority for the Credit Union to Instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by November 16, 2016.



Servus Credit Union Ltd.

Page 1 of 3

Yours truly,

Servus Credit Union Ltd.

per:

Richard Charlton Regional Manager Credit, Commercial

#### ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this \_\_\_\_\_ day of November, 2016

BORROWER(S):

Crelogix Acceptance Corporation per:

Karl Sigerist

GUARANTOR(S):

Crelogix Portfolio Services Corporation per: Karl Sigerist

Credit Facility Letter Amendment 032015 21168652.2 Servus Credit Union Ltd.

Page 2 of 3

Crelogix Credit Group Inc. per: Karl Sigerist \_ per Michael Mills 22 7 ¢ 2 per KarlySiperist per: Mike McKey 1

Credit Facility Letter Amendment 032015 21168652.2

Servus Credit Union Ltd.

Page 3 of 3



Servus Credit Union Ltd. 151 Karl Clark Road NW Edmonton AB T6N 1H5 servus.ca

December 14, 2016

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC V5C 0E4

Private & Confidential

Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8 ,2016 and Amending letter dated November 15, 2016

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letter stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Chariton.

FEES:

- An Amendment Fee of \$1,000 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

**REVISED TERMS AND CONDITIONS:** 

Evergreen Revolving Facility #1 for \$1,250,000

Delete: Expiry date December 15, 2016

Insert: Explry date December 22, 2016

Your acceptance of this letter will constitute authority for the Credit Union to instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by December 15, 2016.

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 1 of 3

Yours truly,

Servus Credit Union Ltd.

per: Richard Charlton

Regional Manager Credit, Commercial

#### ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this  $\int \sum_{i=1}^{H} day$  of December, 2016

BORROWER(S):



Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 2 of 3

GUARANTOR(S):

ç

	Crelogix Portfolio Service Corporation
per:	At
	Crelogix Credit Group Inc.

per:

Karl Sigerist

.

----22 Mike Mills Mike McKay

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 3 of 3



Servus Credit Union Ltd. 151 Karl Clark Road NW Edmonton AB T6N 1H5 servus.ca

December 22, 2016

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC VSC 0E4

#### Private & Confidential

Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated November 15, 2016 and Amending letter dated December 14, 2016

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letter stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Chariton.

FEES:

- 1) An Amendment Fee of \$1,000 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

#### **REVISED TERMS AND CONDITIONS:**

Evergreen Revolving Facility #1 for \$1,250,000

Delete: Expiry date December 22, 2016

Insert: Expiry date January 10, 2017

Your acceptance of this letter will constitute authority for the Credit Union to Instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by December 22, 2016.

(JUE) n Letter Amendment 032015

Servus Credit Union Ltd.

Page 1 of 3

Yours truly,

Servus Credit Union Ltd.

per: **Richard Charlton** 

Regional Manager Credit, Commercial

#### ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this \_ 12 day of December, 2016

#### BORROWER(S):

Crelogix Acceptance Corporation

per:	Bigned with Agreement Express 22-Dec-16 22-Dec-16	
	11 1 1	*
рег:	MVV	an a

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 2 of 3

# GUARANTOR(5):

	Crelogix Portfolio Service Corporation
per:	Cipred with Agreemant Espiremant Espiremant 22-bes-10
	Crelogix Credit Group Inc.
per:	Signed with Acrement Express 22 Over-16 23 Over-16
-	
	Brandt Allo Diversion Resources Estatione 22 Oct-10
	Karl Sigerist
	Contena de la co
	Mike Makay

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 3 of 3

í

2 4 8

1



Servus Credit Union Ltd. 151 Karl Clark Road NW Edmonton AB T6N 1H5 servus.ca

January 9, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC VSC 0E4

Private & Confidential

Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated November 15, 2016, Amending letter dated December 14, 2016 and Amending letter dated December 22, 2016

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit (acilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

FEES:

- An Amendment Fee of \$1,000 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

#### **REVISED TERMS AND CONDITIONS:**

Evergreen Revolving Facility #1 for \$1,250,000

Delete: Expiry date January 10, 2017

Insert: Expline date January 30, 2017

Your acceptance of this letter will constitute authority for the Credit Union to Instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This an and the Credit Facility Letter shall expire if not accepted by January 10, 2017. Credit Facility Letter Americanent 032015 Sarvus Credit Union Ud,

Page 1 of 3

Yours truly,

Servus Credit Union Ltd.

part **Richard Charlton** Regional Manager Credit, Commercial

#### ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit fadiities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this 10th day of January, 2017

BORROWER(5):



Credit Facility Latter Amendment 031015

Sarvus Credit Union Ltd.

Page 2 of 3

ż

t
# GUARANTOR(S):

----

Crelogix Portfolio Service Corporation تر متدر per; ZE t descritet for t Nexaer - Serie Nexaer - Serie Crelogix Credit Group Inc. فيترجع per: ៍ា <u>,</u> Karl Sigorist a Mike Mills Mike McKdy

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 3 of 3

í

Calcular Andrew S

1

1

ŝ



January 30, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC V5C 0E4

Private & Confidential

Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated November 15, 2016, Amending letter dated December 14, 2016, Amending letter dated December 22, 2016 and Amending letter dated January 9, 2017

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Chariton.

# FEES:

- 1) An Amendment Fee of \$1,000 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

# **REVISED TERMS AND CONDITIONS:**

Evergreen Revolving Facility #1 for \$1,250,000

Delete: Expiry date January 30, 2017

Insert: Expiry date February 28, 2017

Your acceptance of this letter will constitute authority for the Credit Union to instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

Credit Pacility Petter Amendment 032015

Servus Credit Union Ltd.

Page 1 of 3

This amendment to the Credit Facility Letter shall expire if not accepted by January 30, 2017.

Yours truly,

Servus Credit Union Ltd.

per:

Richard Charlton Regional Manager Credit, Commercial

# ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this 30th day of January, 2017

BORROWER(S):



Credit Facility Letter Amendment 032015

# and a second second

GUARANTOR(S):

Crelogix Portfolio Service Corporation per: \_\_\_\_

Crelogix Credit Group Inc.

pera Karl Sigerist Mike Mills ۲ Mike

Credit Facility Letter Amendment 032015



February 14, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC V5C 0E4

Private & Confidential

Page 1 of 3

# Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated November 15, 2016, Amending letter dated December 14, 2016, Amending letter dated December 22, 2016, Amending letter dated January 9, 2017 and Amending letter dated January 30, 2017

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

# FEES:

- 1) An Amendment Fee of \$1,000 is considered earned by the Credit Union and Is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

# **REVISED TERMS AND CONDITIONS:**

# Evergreen Revolving Facility #1 for \$1,250,000

A temporary "unmargined" advance of \$250,000 on the existing \$1,250,000 facility is available to pay the EPO liability amount due to the Funders (~Feb 14<sup>th</sup>). The amount (\$250,000) is to be covered by securitization proceeds by February 24, 2017. Approximately \$53,000 is to be covered from Bulk Assignment #348 on February 15, 2017, \$55,000 is to be covered by February 16, 2017 from Bulk Assignment #103. Once these funds are deposited to the account on or before February 16, 2017, the excess is reduced to \$142,000. All of the future tranches are to cover the residual "unmargined" amount, future proceeds are not to be used for anything other than reducing the "unmargined" amount,



Your acceptance of this letter will constitute authority for the Credit Union to Instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by February 16, 2017.

Yours truly,

Servus Credit Union Ltd.

per: **Richard Charlton** 

Regional Manager Commercial Credit

# ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this \_\_\_\_\_ day of February, 2017

BORROWER(S):

Crelogix Acceptance Corporation per: per: Credit Facility Letter Amendment 032015 Servus Credit Union Ltd.

Page 2 of 3

GUARANTOR(S):

Crelogix Portfoljo Service Corporation per: Crelogix Credit Group Inc. per: Karl Sigerist Mike Mills Mike McKay

Credit Facility Letter Amendment 032015



February 28, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC VSC 0E4

Private & Confidential

Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated November 15, 2016, Amending letter dated December 14, 2016, Amending letter dated December 22, 2016, Amending letter dated January 9, 2017, Amending letter dated January 30, 2017 and Amending letter dated February 14, 2017

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

# FEES:

- 1) An Amendment Fee of \$1,000 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

# **REVISED TERMS AND CONDITIONS:**

Evergreen Revolving Facility #1 for \$1,250,000

Delete: Expiry date February 28, 2017

Insert: Explry date March 31, 2017





Your acceptance of this letter will constitute authority for the Credit Union to instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by February 28, 2017,

Yours truly,

Servus Credit Union Ltd.

nete per: John McLellan

Senior Credit Manager

## ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and If, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this 28 day of February, 2017

### BORROWER(S):

Crelogix Acceptance Corporation Der ner: Credit Facility Letter Amendment 032015 Servus Credit Union Ltd.

Page 2 of 3

GUARANTOR(5): Crelogix Portfolio Service Corporation 9 per: per: Crelogix Credit Group Inc. ι Karl Sigerist Mike Mills Mike McKay . . Credit Facility Letter Amendment 032015 Servus Credit Union Ltd. Page 3 of 3

1



March 17, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC V5C 0E4

Private & Confidential

### Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated November 15, 2016, Amending letter dated December 14, 2016, Amending letter dated December 22, 2016, Amending letter dated January 9, 2017, Amending letter dated January 30, 2017, Amending letter dated February 14, 2017 and Amending letter dated February 28, 2017

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Chariton.

# FEES:

- 1) An Amendment Fee of \$1,000 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

# **REVISED TERMS AND CONDITIONS:**

# Evergreen Revolving Facility #1 for \$1,250,000

A temporary increase of \$865,000 (the "Excess Amount") to the existing \$1,250,000 facility to pay i) the EPO llability amounts due to the Funders for the February EPO and li) to pay critical payables as outlined in March 15, 2017 critical supplier payments listing provided by Crelogix Acceptance Corporation ("Crelogix"). Up to and including March 31, 2017, the Excess Amount shall be repaid by 70% of all amounts received by Crelogix in respect of future securitization net proceeds/debt external and any cash received by Crelogix, which amounts received shall be applied to this.

Credit Facilitation Amendment 032015 2335026122 Servus Credit Union Ltd.

Page 1 of ]

facility as a permanent reduction. After March 31, 2017, and continuing up to and including the earlier of the closing of the sale of all issued and outstanding shares of Crelogix Credit Group Inc. to Enliven Financail Inc. (the "Closing Date") and April 15, 2017, the Excess Amount shall be repaid by 100% of all amounts received by Crelogix in respect of future securitization net proceeds, which amounts received shall be applied to this facility as permanent reduction. Any balance outstanding on this facility shall be due in full and repaid on the Closing Date.

Pdf coples of supplier cheques to be provided for ~\$101,000, the cheques must align with the listing provided by Crelogix to Servus Credit Union dated March 15, 2017.

The 30% residual of all payments received by Crelogix to up to March 31, 2017 is to be used, first, for month end payroll, and second, any amounts remaining after paying month end payroll shall be paid in respect of supplier payables.

Your acceptance of this letter will constitute authority for the Credit Union to instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by March 20, 2017.

Yours truly,

Servus Credit Union Ltd.

per:

Richard Charlton Regional Manager Credit, Commercial

### ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and If, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

Page 2 of 3

Credit Facility Letter Amendment 032015 Servus Credit Union Ltd. 23350281.2

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this  $17^{\text{th}}$  day of March, 2017

# BORROWER(S):

Crelogix Acceptance Corporation per: per:

# GUARANTOR(S):

Crelogix Portfolio Service Corporation per:

Crelogix Credit Group Inc.

per:

Karl Sigerist

Mike Mills Mike McKa



April 10, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC V5C DE4

Private & Confidential

#### Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated November 15, 2016, Amending letter dated December 14, 2016, Amending letter dated December 22, 2016, Amending letter dated January 9, 2017, Amending letter dated January 30, 2017, Amending letter dated February 14, 2017, Amending letter dated February 28, 2017 and Amending letter dated March 17, 2017

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

## FEES:

- 1) An Amendment Fee of \$1,000 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

### **REVISED TERMS AND CONDITIONS:**

#### Evergreen Revolving Facility #1 for \$1,250,000

A temporary increase of \$258,000 (the "Excess Amount") to the existing \$1,250,000 facility () to pay payroll (~\$173,000) on April 11, 2017 and ii) to pay the highlighted critical payables (~\$65,000) as outlined in April 10, 2017 critical supplier payments listing provided by Crelogix Acceptance Corporation ("Crelogix"). The Excess Amount shall be repaid by 100% of all amounts received by Crelogix in respect of future securitization net proceeds/debt sales/refunds and any cash received by ेंद्रिविश्विंग, which amounts received shall be applied to this facility as a permanent reduction. Any Weilance outstanding on this facility shall be due in full on or before April 15, 2017. Servus Credit Union Ltd.

Credit Fad antither Amendment 032015

Page 1 of 3

Pdf copies of supplier cheques to be provided for  $\sim$ \$246,000 (including with respect to, but not limited to, the amounts paid by Crelogix from the  $\sim$ \$88,000 currently held in the Credit Union account and from the pending  $\sim$ \$73,000 transfer from Propsera to the Credit Union account), the cheques must align with the listing (highlighted items) provided by Crelogix to the Credit Union dated April 10, 2017.

Delete: Explry date the earlier of the closing of the sale of all Issued and outstanding shares of Crelogix Credit Group Inc. to Enliven Financial Inc. and April 15, 2017

Insert: Expiry date of April 15, 2017

Your acceptance of this letter will constitute authority for the Credit Union to instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by April 12, 2017.

Yours truly,

Servus Credit Union Ltd.

per: **Richard Charlton** 

Director Commercial Credit

# ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material advarse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 2 of 3

Accepted this \_\_\_\_\_ day of April, 2017

# BORROWER(S):



# GUARANTOR(S):

Crelogix Portfolio Service Corporation per:

Crelogix Credit Group Inc. pet -----Karl Sigerist Mike Mills Mike M ٢a

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

ę.



April 21, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, 8C VSC 0E4

Private & Confidential

Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Lotter to Letter dated September 8, 2016, Amending letter dated November 15, 2016, Amending letter dated December 14, 2016, Amending letter dated December 22, 2016, Amending letter dated January 9, 2017, Amending letter dated January 30, 2017, Amending letter dated February 14, 2017, Amending letter dated February 28, 2017, Amending letter dated March 17, 2017 and Amending letter dated April 10, 2017

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Latters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

# FEES:

- An Amendment Fee of \$1,000 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

#### **REVISED TERMS AND CONDITIONS:**

Evergreen Revolving Facility #1 for \$1,250,000

A temporary increase of \$657,000 (the "Excess Amount") to the existing \$1,250,000 facility.i) to pay the March EPO due ~April 10, 2017 of \$499,000 and II) to pay 'payroli of ~\$228,000 due April 26/ 2017 (this is an estimate, only the required amount will be advanced for payroli). The current excess of ~\$80,000 is to be covered by the funds already set to be transferred to the Credit Union from Prospera Credit Union (~\$149,000 by April 24, 2017). The excess amount of \$657,000 takes into account the \$69,000 that is set to be transferred to the Credit Union from Prospera Gredit Union on \$24, 2017. The Excess Amount shall be repaid by 100% of all amounts received by Credos. Account the \$69,000 that is set to be transferred to the Credit Union from Prospera Gredit Union on \$24, 2017. The Excess Amount shall be repaid by 100% of all amounts received by Credos. Account the \$69,000 that is set to be transferred to the Credit Union from Prospera Gredit Union on the \$24, 2017. The Excess Amount shall be repaid by 100% of all amounts received by Credos. Account the \$61,000 that is set to be transferred to the Credit Union of the Strate and Acceptance Corporation ("Crelogix") In respect of future securitization net proceeds/debt sales/refunds and any cash received by Crelogix, which amounts received shall be applied to this facility as a permanent reduction.

Delete: Expiry date of April 15, 2017

Insert: Expiry date is the earlier of i) the closing the transaction set forth in the Confidential Non-Binding Term Sheet dated April 18, 2017 (the "Dealnet Term Sheet"), entered into as between the Credit Union, Crelogix Credit Group Inc. ("CCGI") and Dealnet Capital Corp. ("Dealnet") ii) either of the Credit Union, CCGI or Dealnet electing not to proceed with the transaction set forth in the Dealnet Term Sheet and iii) May 1, 2017.

Your acceptance of this letter will constitute authority for the Credit Union to Instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by April 24, 2017.

Yours truly,

Servus Credit Union Ltd.

Def: **Richard Chariton** 

**Director Commercial Credit** 

#### ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amondment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if 50 signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this 219 day of April, 2017

Credit Facility Letter Amondmont 032015

Servus Credit Union Ltd.

Page 2 of 3

ł

1

And the set of the set

BORROWER(S):

:

per: per:

Crelogix Acceptance Corporation

# GUARANTOR(S):

	Crelggix Portfollo Service Corporation
per: .	
	Crelogia Credit Group Inc.
per:	IVI
	Karl Sigerist
	III IA.
موي	THE MUS Mike Mikey
	Mike Heker Mills

Credit, Facility Latter Amendment 032015

Servus Creck Union Ltd.

Page 3 of 3

B	0	R	R	0	W	E	R	(S)	:	
---	---	---	---	---	---	---	---	-----	---	--

Crelogix Acceptance Corporation

per: \_\_\_\_\_

per:

GUARANTOR(S):

Crelogix Portfolio Service Corporation

per: \_\_\_\_\_

Crelogix Credit Group Inc.

per:  ${}^{\circ}$ Karl/Sigerist

Mike Mills

Mike McKay

Credit Facility Letter Amendment 032015 Servus Credit Union Ltd.

New York State State State

Page 3 of 3



May 1, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC VSC 0E4

Private & Confidential

age Lof. 3

#### Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated November 15, 2016, Amending letter dated December 14, 2016, Amending letter dated December 22, 2016, Amending letter dated January 9, 2017, Amending letter dated January 30, 2017, Amending letter dated February 14, 2017, Amending letter dated February 28, 2017, Amending letter dated March 17, 2017, Amending letter dated April 10, 2017, Amending letter dated April 19, 2017 and Amending letter dated April 21, 2017

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

# FEES:

- 1) An Amendment fee of \$1,000 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

#### **REVISED TERMS AND CONDITIONS:**

Letter Amendment 032015

Evergreen Revolving Facility 9771221 sub #1

#### **Expiry Date:**

Delete: Expiry date is the earlier of i) the closing the transaction set forth in the Confidential Non-Binding Term Sheet dated April 18, 2017 (the "Dealnet Term Sheet"), entered into as between the Credit Union, Crelogix Credit Group Inc. ("CCGI") and Dealnet Capital Corp. ("Dealnet") ii) either of the Credit Union, CCGI or Dealnet electing not to proceed with the transaction set forth in the Dealnet Term Sheet and iii) May 1, 2017.

Insert: Expiry date is the earlier of i) the closing the transaction set forth in the Confidential Non-Binding Term Sheet dated April 18, 2017 (the "Dealnet Term Sheet"), entered into as between the Credit Union, Crelogix Credit Group Inc. ("CCGI") and Dealnet Capital Corp. ("Dealnet") ii) either of the Credit Union, CCGI or Dealnet electing not to proceed with the transaction set forth in the Dealnet Term Sheet and iii) May 15, 2017.

Your acceptance of this letter will constitute authority for the Credit Union to instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by May 1, 2017.

Yours truly,

Servus Credit Union Ltd.

per: **Richard Charlton** Director Commercial Credit

#### ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and If, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this \_\_\_\_\_ day of May, 2017

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 2 of 3

BORROWER(5):
Crelogix Acceptance Corporation
per:

GUARANTOR(S):

Crelogix Portfolio Service Corporation

5 per:

Crelogix Credit Group Inc. per: Karl Sigerist 2 Mike Mills 1 Mike McKe

.

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

•

.

Page 3 of 3



May 15, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC **V5C 0E4** 

Private & Confidential

Page

#### Attention: Karl Sigerist

#### Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated November 15, 2016, Amending letter dated December 14, 2016, Amending letter dated December 22, 2016, Amending letter dated January 9, 2017, Amending letter dated January 30, 2017, Amending letter dated February 14, 2017, Amending letter dated February 28, 2017, Amending letter dated March 17, 2017, Amending letter dated April 10, 2017, Amending letter dated April 19, 2017, Amending letter dated April 21, 2017 and Amending letter dated May 1, 2017

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

#### FEES:

Cred

- 1) An Amendment fee of \$1,000 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

#### **REVISED TERMS AND CONDITIONS:**

#### Evergreen Revolving Facility 9771221 sub #1

A temporary increase of \$762,000 (the "Excess Amount") to the existing \$1,250,000 facility i) to pay the April EPO due ~May 15, 2017 of ~\$412,000, ii) pay critical payables of \$134,000 based on the May 15, 2017 list provided by Crelogix Acceptance Corporation ("Crelogix") and iii) continuation of current excess of ~\$216,000. The excess amount of \$762,000 takes into account the ~\$113,000 that is set to be transferred to the Credit Union from Prospera Credit Union on May 16, 2017, The., Excess Amount shall be repaid by 100% of all amounts received by Crelogix in respect of future Securitization net proceeds/debt sales/refunds and any cash received by Crelogix, which amounts Meceived shall be applied to this facility as a permanent reduction. adity Letter Amendment 032015 Servus Gredit Union Ltd.

ł.

#### Expiry Date:

Delete: Expiry date is the earlier of i) the closing the transaction set forth in the Confidential Non-Binding Term Sheet dated April 18, 2017 (the "Dealnet Term Sheet"), entered into as between the Credit Union, Crelogix Credit Group Inc. ("CCGI") and Dealnet Capital Corp. ("Dealnet") ii) either of the Credit Union, CCGI or Dealnet electing not to proceed with the transaction set forth in the Dealnet Term Sheet and iii) May 15, 2017.

Insert: Expiry date is the earlier of i) the closing the transaction set forth in the Confidential Non-Binding Term Sheet dated April 18, 2017 (the "Dealnet Term Sheet"), entered into as between the Credit Union, Crelogix Credit Group Inc. ("CCGI") and Dealnet Capital Corp. ("Dealnet") ii) either of the Credit Union, CCGI or Dealnet electing not to proceed with the transaction set forth in the Dealnet Term Sheet and iii) June 2, 2017.

Your acceptance of this letter will constitute authority for the Credit Union to instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by May 15, 2017.

Yours truly,

Servus Credit Union Ltd.

Der:

Richard Charlton Director Commercial Credit

## ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this 16 day of May, 2017

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 2 of 3

BORROWER(S):

	Crelogix Acceptance Corporation
per:	ZT Sz
per:	M M
P-41.4	

# GUARANTOR(S):

Crelogix Portfolio Service Corporation per:

Crelogix Credit Group Inc.

per: Karl Sigerist 2 Mike Mills

Credit Facility Letter Amendment 032015

Mike McKay

Servus Credit Union Ltd.

Page 3 of 3

------



June 27, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC V5C 0E4

Private & Confidential

#### Attention: Karl Sigerist

## Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated November 15, 2016, Amending letter dated December 14, 2016, Amending letter dated December 22, 2016, Amending letter dated January 9, 2017, Amending letter dated January 30, 2017, Amending letter dated February 14, 2017, Amending letter dated February 28, 2017, Amending letter dated March 17, 2017, Amending letter dated April 10, 2017, Amending letter dated April 19, 2017, Amending letter dated April 21, 2017, Amending letter dated May 1, 2017 and Amending letter dated May 15, 2017

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

### FEES:

- 1) An Amendment fee of \$5,000 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

#### REVISED TERMS AND CONDITIONS:

#### Evergreen Revolving Facility 9771221 sub #1

A temporary increase of \$1,749,000 (the "Excess Amount") to the existing \$1,250,000 facility I) to pay payroll in the amount of ~\$209,000, II) to pay for merchant contracts up to the amount of \$2,500,000 (the "Merchant Payables") (a list of the merchant contracts to be purchased with the Excess Amount shall be provided to the Credit Union by Crelogix Acceptance Corporation ("Crelogix), on June 27, 2017 (the "Merchant List"), and III) continuation of current excess of ~\$290,000, All Interchant contracts purchased by Crelogix by way of the Merchant Payables shall be the first interchant contracts to be securitized by Crelogix in July 2017. All amounts utilized by Crelogix of the Excess Credit Accept Exter Amendment 032015 Servus Credit Union Lid. 25224991 Amount for Merchant Payables shall be repaid by a securitization tranche to Prospera Credit Union on or before July 7, 2017. It is further expressly acknowledged by Crelogix that the Excess Amount shall be repaid by 100% of all amounts received by Crelogix in respect of future securitization net proceeds/debt sales/refunds and any cash received by Crelogix, which amounts received shall be applied to this facility as a permanent reduction. All amounts utilized by Crelogix of the Excess Amount shall be used solely for the purpose of i) paying payroll in an amount not exceeding \$209,000, and ii) purchase of the merchant contracts set out in the Merchant List.

÷

#### Condition of Credit:

Servus shall advance to Crelogix from the Excess Amount an amount not exceeding \$209,000 immediately upon being provided with a fully executed version of i) this agreement, and ii) the Forbearance Amending Agreement No. 13 (the "Payroll Advance"). Crelogix acknowledges and agrees that any amounts advanced to Crelogix by way of the Payroll Advance shall be used solely for the purposes of fulfilling Crelogix's current payroll obligations. Servus shall not advance any amounts to Crelogix from the Excess Amount that exceed the Payroll Advance until such time as Prospera Credit Union has agreed in writing to purchase, on or before July 7, 2017, all merchant contracts purchased by Crelogix by way of the Merchant Payables.

#### Expiry Date:

Delete: Expiry date is the earlier of i) the closing the transaction set forth in the Confidential Non-Binding Term Sheet dated April 1B, 2017 (the "Dealnet Term Sheet"), entered into as between the Credit Union, Crelogix Credit Group Inc. ("CCGI") and Dealnet Capital Corp. ("Dealnet") ii) either of the Credit Union, CCGI or Dealnet electing not to proceed with the transaction set forth in the Dealnet Term Sheet and iii) June 2, 2017.

Insert: Expiry date is July 7, 2017.

Your acceptance of this letter will constitute authority for the Credit Union to instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by June 28, 2017.

Yours truly,

Servus Credit Union Ltd.

Richard Charlton Director Commercial Credit

### ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

Credit Facility Letter Amendment 032015 Servus Credit Union Ltd. 25224999.3

Page 2 of 4

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this 28th day of June, 2017

# BORROWER(S):

**Crelogix Acceptance Corporation** per: per:

GUARANTOR(5):

**Crelogix Portfolio Service Corporation** per:

Crelogix Credit Group Inc.

per: Karl Sigerist -----Mike Mills Mike McK

Credit Facility Letter Amendment 032015 25224999.3 Servus Credit Union Ltd.

Page 3 of 4

This is **Exhibit "F"** referred to in the Affidavit of Darcy Peelar, sworn before me this 5<sup>th</sup> day of July, 2017

A Commissioner for Oaths in and for Alberta

Patrick T. Harnett Barrister and Solicitor

September 8, 2016

**Crelogix Acceptance Corporation** 

4445 Lougheed Highway #900

Burnaby, B.C.

V5C 0E4

Ŧ

# Private & Confidential

Attention: Karl Sigerist - Director, President and Chief Executive Officer

Dear Mr. Sigerist,

RE: Supplemental Credit Facility Letter to Commitment Letter dated May 7, 2006 and renewal Commitment Letter dated May 31, 2013

We are pleased to confirm that Servus Cradit Union Ltd. (the "Credit Union") has authorized the following credit facilities in the name of Crelogix Acceptance Corporation subject to the following terms and conditions as well as those Standard Credit Terms and Conditions in Schedule "A" attached to and forming part of this Credit Facility Letter, and those set out in the Forbearance and Preservation of Rights Agreement dated September 14, 2016 (the "Forbearance Agreement"). All other terms and conditions of the Letter(s) stated above including all security and guarantees provided therein remain unchanged except as amended by this Credit Facility Letter.

BORROWER: Crelogix Acceptance Corporation

AMOUNT OF LOANS:

New Credit Facilities:

1) \$100,000 Demand Term Loan #TBD - To pay Monilors fees - Expiry December 15, 2016

AVAILABILITY:

- 1) All loans available by way of direct Canadian Dollar advances.
- 2) For all of the credit facilities listed below, if funds are not fully disbursed within 90 days of the date of this credit facility letter or as otherwise agreed to by the borrower and the credit union, the availability of such credit facilities from the Credit Union shall, at the Credit Union's discretion, expire and be cancelled.
  - a) \$100,000 Demand Term Loan #TBD



Credit Facility Letter (07/16) | Page 1 of 6

# INTEREST RATES:

Amounts advanced by the Credit Union to the Borrower will bear Interest while outstanding, before and after maturity, default and judgment at the following rates:

#### New Gredit Facility:

1) <u>Demand Term Loan #TBD:</u> An annual rate of interest equal to the Credit Union's Prime Lending Rate plus 4.00%, floating, calculated daily and payable monthly in arrears.

The "Prime Lending Rate" referred to above shall mean the floating annual rate of interest established and recorded as such from time to time by the Credit Union as its reference rate for determining rates of interest it will charge for loans denominated in Canadian Dollars and commonly called the Credit Union's Prime Lending Rate, adjusted automatically upon any change by the Credit Union. The Credit Union's Prime Lending Rate is 2.70% per annum as at the date of this Credit Facility Letter.

Interest shall be payable without demand and, unless otherwise specified herein, shall be calculated daily and payable monthly in arrears on the date specified by the Credit Union and will accrue daily. Overdue interest shall be an interest at the same rate as that used in calculating the interest overdue, while the same remains unpaid.

#### **REPAYMENT TERMS:**

The Borrower shall repay all loans on demand. Prior to demand by the Credit Union, loans shall be repayable as follows:

The full principal balance owed, inclusive of accrued interest, fees and expenses, including all of Credit Union's solicitor and client costs on a full indemnity basis, shall be repaid on or before December 15, 2016.

The Borrower, at any time and from time to time, when not in default, shall have the privilege of paying the whole amount or any part thereof without notice or bonus.

#### **GENERAL FEES:**

At the time of this Credit Facility Letter, in addition to those feas set out in the Forbearance Agreement, the Credit Union's fees are as follows:

- An Application Fee is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Credit Facility Letter. The Application Fee is broken down as follows:
  - a) Demand Loan Facility: \$1,000

Total Application Fee: \$1,000

- An Account Review Fee of \$4,500 will be due and payable annually by the Borrower after completion of our Account Review.
- 3) A late reporting fee for annual reporting requirements of \$250 is due and payable by the Borrower at the end of the month in which the reporting due date occurs and will continue monthly until the reporting requirements are met.
- 4) Search and Credit Investigation Fees are to be paid by the Borrower to the Credit Union.
- 5) The Borrower agrees to forthwith pay to the Credit Union a charge for each cheque presented for deposit which is dishonoured or, in the Credit Union's absolute discretion, a late payment fee whenever a payment is not remitted on its due date at the Credit Union's normal charges in effect from time to time respecting dishonoured cheques and/or late payment fees.

19979029.3

Credit Facility Letter (07/16) | Page 2 of 6

6) All fees due and payable, as outlined in this Credit Facility Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

# COMMERCIAL REPORTING REQUIREMENTS:

- 1) Annually, within 90 days of the Borrower's fiscal year end:
  - a) Crelogix Credil Group Inc. consolidated fiscal financial statements prepared by a CA, CMAS or CGA on a "Review Engagement" basis signed by the company's officers and received within 90 days of fiscal year end
  - b) Signed Personal Net Worth Statements on approved Credit Union form, the following updated at least blennially or sooner if requested by the Credit Union:
    - Karl Sigerisi
    - Michael McKay
    - Michael Mills
- 1) Periodically, as requested:
  - Any request for financial, reporting or other information on the status of operating, as may be from time to time, made by Servus Credit Union Ltd.

## **GENERAL CONDITIONS OF CREDIT:**

# Demand Term Loan Facility #TBD:

- All security documentation to be completed and registered with a solicitor's letter of opinion confirming that the Credit Union's security is a valid and enforceable first charge and that any prior encumbrances will not affect that first registered position.
- The Credit Union's solicitor is to confirm all proper signing operating account documents and enabling resolutions have been or will be executed.
- 3) Satisfactory PPR Search In British Columbia
- 4) All legal costs incurred in this transaction are for the account of the Borrower, including those payable to the solicitor representing the Credit Union.
- All costs (i.e. security registration, appraisals, inspections, etc.) will be for the account of the borrower.
- 6) All security documentation will be prepared by the Credit Union Solicitor and forwarded to the Borrower's solicitor for execution and provision of an opinion as to the ability of the Borrower to borrow and pledge security. The Credit Union's solicitor will register and provide and opinion as to enforceability.

#### **REVIEW:**

All loans are subject to review at any time, and in any event will be reviewed annually. Our next account review would be scheduled for February 1, 2017 based on a full review of financial statements of the Borrower and Guarantors to the year ended.

19979029.3

Credit Facility Letter (07/16) | Page 3 of 6

# SECURITY APPLICABLE TO THIS CREDIT FACILITY:

The types of security, supporting resolutions and agreements to be provided by the Borrower will be in a form and content determined by the Credit Union and/or its solicitors and include the following:

# Security Presently Held:

٤

- 1) All Corporate Documents and Resolutions as required
- 1st GSA dated March 31, 2008 in the name of Travelers Acceptance Corporation registered in Alberta and British Columbia
- 1st GSA Dated October 30, 2009 in the name of Crelogix Credit Group Inc. Registered in Alberta and British Columbia
- 1st GSA dated October 30, 2009 in the name of Crelogix Portfolio Services Corporation Registered in Alberta and British Columbia
- 5) Corporate Guarantee from Crelogix Portfolio Services Corporation in the amount of \$2,000,000
- 6) Corporate Guarantee from Crelogix Credit Group Inc. in the amount of \$2,000,000

Security To Be Obtained (if appropriate, all security documents to be registered in the appropriate government or other registry as required or desirable):

- 1) Execution of Credit Facility Letter
- 2) General Borrowing Resolution & Certificate of Non-Restriction
- 3) Executed Forbearance Agreement

## SECURITY APPLICABLE TO SPECIFIC CREDIT FACILITIES:

The types of security, supporting resolutions and agreements to be provided by the Borrower will be in a form and content determined by the Credit Union and/or its solicitors and registered in the appropriate government or other registry as required or desirable, and includes the following:

## ACCEPTANCE:

- 1) To become effective this Credit Facility Letter must be accepted in writing by the Borrower and Guarantors.
- 2) This Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Credit Facility Letter.

 Your acceptance of this Credit Facility Letter will constitute authority for the Credit Union to instruct its solicitors to prepare the necessary documentation.

19976029.3

Credit Facility Letter (07/16) | Page 4 of 6

Please note that this Gredit Facility Letter restates and modifies and takes precedence over any prior Commitment Letters or Credit Facility Letters issued to you; however, it is not a novation. All existing security and guarantees held by the Credit Union continue in full force and effect and are security for the loans.

Nothing herein shall be construed to impair any security, iten, or charge held by the Credit Union to secure the loans and Credit Facilities and nothing herein shall affect or impair any powers which the Credit Union may have against the Borrower, any Guarantors or any other person for recover of the loans and Credit Facilities.

If you are in agreement with the above terms and conditions, as well as the Standard Credit Terms and Conditions as detailed in the attached Schedule "A" – Standard Credit Terms and Conditions which form part of this Credit Facility Letter, please sign this letter and raturn this letter with the attached Standard Credit Terms and Conditions to the Credit Union together with the applicable fee. This Credit Facility Letter will expire if not accepted or extended in writing by September 16, 2018. The foregoing is offered in good faith and is to be held in strict confidence.

Yours truly,

Servus Credit Union Ltd.

per: **Richard Charlton** 

VP Commercial Credit, South

per;

Accepted this \_\_\_\_\_ day of September, 2016

BORROWER(S):

Crelogix Acceptance Corporation

per: \_\_\_\_\_

per:

19979029,3

Credit Facility Letter (07/16) | Page 5 of 6

Please note that this Credit Facility Letter restates and modifies and takes precedence over any prior Commitment Letters or Credit Facility Letters issued to you; however, it is not a novation. All existing security and guarantees held by the Credit Union continue in full force and effect and are security for the loans.

Nothing herein shall be construed to impair any security, lien, or charge held by the Credit Union to secure the loans and Credit Facilities and nothing herein shall affect or impair any powers which the Credit Union may have against the Borrower, any Guarantors or any other person for recover of the loans and Credit Facilities.

If you are in agreement with the above terms and conditions, as well as the Standard Credit Terms and Conditions as detailed in the attached Schedule "A" – Standard Credit Terms and Conditions which form part of this Credit Facility Letter, please sign this letter and return this letter with the attached Standard Credit Terms and Conditions to the Credit Union together with the applicable fee. This Credit Facility Letter will expire if not accepted or extended in writing by September 15, 2016. The foregoing is offered in good faith and is to be held in strict confidence.

Yours truly,

ť

Servus Gredit Union Ltd.

per:

Richard Charllon VP Commercial Credit, South

per:

Accepted this 15th day of September, 2016

### BORROWER(S):

**Crelogix Acceptance Corporation** per:

5 - **1** - 61-

19979029.3

Credit Facility Letter (07/16) | Page 5 of 6
GUARANTOR(S):

,

ţ

Crelogix Portfolio Service Corporation -0 per; 70

Crelogix Credit Group Inc.

per:

19979029.3



Credit Facility Letter (07/16) | Page 6 of 6



## SCHEDULE "A"

## STANDARD CREDIT TERMS AND CONDITIONS

Borrower: CRELOGIX ACCEPTANCE CORPORATION

Credit Facility Letter dated SEPTEMBER 8, 2016

and signed on the 15th day of September , 2016

All references to "Credit Union" shall mean Servus Credit Union Ltd. in this Schedule "A" unless otherwise specified. All references herein to "the Credit Facility Letter" shall mean the Credit Facility Letter to which this Schedule "A" is attached and in the event of a conflict between the terms of the Credit Facility Letter and this Schedule "A", the terms of the Credit Facility Letter shall prevail but for clarity the mention of a provision in the Credit Facility Letter and not in Schedule "A" or *vice versa* shall not constitute a conflict but shall be deemed to be supplemental and in addition to any of the terms and conditions under either of the Credit Facility Letter or Schedule "A" as the case may be.

## ACCEPTANCE, ADVANCES, PAYMENTS, EXPENSES, FEES AND CONSENTS:

- Acceptance: The Borrower and all Guarantors must accept in writing the terms and conditions of the Credit Facility Letter prior to any advances being made.
- 2) Evidence of Advances: The Borrower and all of the Guarantors agree that the Credit Union's records evidencing an advance shall be complete and final proof, absent manifest error, that funds have been advanced under any one or more of the loans set forth in the Credit Facility Letter and may, from time to time dependent upon the type of loan facility made available, be evidenced by other documentation such as, for example and without limitation, promissory notes, direct deposits, drafts or cheques made payable to other parties including solicitors and agents and any other means by which the Credit Union provides value to the Borrower under any one or more of the loan facilities.
- Debit from Borrower Account: Unless otherwise stated in writing, payments for all loans and credits will be automatically transferred or debited from the Borrower's operating account with the Credit Union.
- 4) <u>Payment of Costs:</u> The Borrower agrees to pay all expenses, fees and charges incurred by the Credit Union in relation to all loans and credits, the preparation and registration of all security, enforcement or preservation of any or all of the Credit Union's rights and remedies including those incurred during an annual or any other periodic review of the Borrower's relationship with the Credit Union, whether or not any such documentation is completed or any funds are advanced, including but not limited to legal expenses (on a solicitor-and-his-own-client full indemnity basis), costs of accountants, engineers, architects, consultants, appraisers and the costs of any and all searches and registrations the Credit Union or its solicitor deems either necessary or desirable.
- 5) <u>Change in Fges:</u> The Borrower and all Guarantors acknowledge that the Credit Union may change the fees payable pursuant to the Credit Facility Letter from time to time upon notice to the Borrower in person, by telephone, by letter that is sent either by mail or facsimile transmission or by electronic mail (e-mail) to the address, telephone number and/or electronic mail (e-mail) address on file at the Credit Union.
- 6) <u>Not Assumable</u>: All loans and credits are not assumable without the Credit Union's prior, written consent and if the Borrower does attempt to have some other entity assume any loan or security, any and all loans and credits shall, in the Credit Union's discretion, become immediately due and payable and the Credit Union may commence enforcement.

#### CONDITIONS PRECEDENT TO DRAWDOWN:

Schedule "A" Stendard Credit Terms and Conditions 012016

Page 1 of B

- Security Completion: Prior to funding any loans or credits or making any further advances, all security as contemplated by the Credit Facility Letter must be completed and registered wherever required or desirable by and in form and content acceptable to the Credit Union and its solicitors, and all certificates, searches, solicitor's opinions and other documents as required by the Credit Union shall be delivered to the Credit Union in form and content acceptable to the Credit Union.
- 2) <u>Title Insurance:</u> With respect to real estate transactions including, without limitation, mortgages, the Credit Union may fund upon receipt of an acceptable title Insurance policy issued by any major title insurer including at this time Stewart Title Insurance, Chicago Title Insurance and First Canadian Title.

#### **REPRESENTATIONS AND WARRANTIES:**

ş

The Borrower, all Corporate Guarantors and any other Guarantors represent and warrant to the Credit Union that:

- <u>Corporate Status</u>: If a corporation, it is duly incorporated, validly existing and duly registered and qualified to carry on business in the Province of Alberta and in all other jurisdictions where it carries on business and shall maintain such corporate existence and registration at all times during which any money may be owing to the Credit Union or it shall be liable to the Credit Union;
- 2) <u>Authorizations:</u> The execution, delivery and performance by it of this Credit Facility Letter and any and all terms and conditions thereunder including provision of security have been duly authorized by all necessary actions and do not violate its constating documents or any applicable Laws or agreements to which it is subject or by which it is bound;
- 3) <u>Environmental Claims</u>: There are no claims, actions, prosecutions or other proceedings of any kind pending or threatened against it or any of its assets or properties before any court or administrative tribunal or agency which relate to any noncompliance with any environmental law or any release from its lands of a contaminant into the natural environment or which, if adversely determined, might have a material adverse effect upon its financial condition or operations or its ability to perform its obligations under the Credit Facility Letter or under any of the Credit Union's security and that there are no circumstances of which they are aware which might give rise to any such proceeding which has not already been fully disclosed to the Credit Union;
- 4) <u>Claims</u>: There are no claims, actions, prosecutions or other proceedings of any kind pending or threatened against it or against any of their assets or properties before any court or other administrative agency which relate to any noncompliance with any other applicable law or which, if adversely determined, might have a material adverse effect upon their financial condition or operation or their ability to perform their obligations under the Credit Facility Letter or in any of the Credit Union's security, and that there are no circumstances of which they are aware that might give rise to any such proceeding that have not already been fully disclosed to the Credit Union;
- 5) <u>Crime Proceeds and Terrorism</u>: That they are not in violation of any applicable law relating to terrorism or money laundering, including the *Proceeds of Crime (Noney Laundering) and Terrorist Financing Act* (Canada).
- 6) <u>Accuracy of Information and Accounting:</u> That all information provided to the Credit Union is complete and accurate and does not omit any material fact and all financial statements delivered by it to the Credit Union fairly represent its financial condition as of the date of such financial statement all in accordance with Accounting Standards for Private Enterprises (ASPE) or International Financial Reporting Standards (IFRS), as the case may be;
- 7) <u>Good Title/Interest:</u> In respect of all property and assets subject to Credit Union security, it has good and marketable title or a valid interest in such property and assets free and clear of all claims and encumbrances except those claims and encumbrances to which the Credit Union has provided its prior consent in writing;
- No Default: There is no default or event or default that has occurred or is occurring as of acceptance of the Credit Facility Letter;

Schedula "A" Standard Credit Terms and Conditions 012016

Page 2 of 8

- 9) <u>Material Adverse Change:</u> To the best of their knowledge no event or circumstance has occurred or is continuing which has or would reasonably be expected to have a material adverse impact on the operations or financial condition of the Borrower or any Guarantor;
- 10) <u>Government Remittances</u>: That all material remittances required to be made to government authorities have been made, are currently up to date and that there are no outstanding arrears, other than those that are being validly contested and that have been disclosed to the Credit Union;
- 11) <u>Tax Payments</u>: That the Borrower and all Guarantors have duly filed on a timely basis all tax returns or other information required to be filed and have paid all material taxes which are due and payable including all assessments and reassessments other than those that are being validly contested and that have been disclosed to the Credit Union and that they have made adequate provision for, and all required instalment payments have been made in respect of taxes payable for the current period for which returns have not as yet been required to be filed and that there are no agreements, waivers of other arrangements providing for any extension of time with respect to the filing of any tax returns or the payment of any taxes and that no actions or proceedings are being taken or threatened by any taxation authority in any jurisdiction where a Borrower or Guarantor carries on business other than those that are being validly contested and that have been disclosed in writing to the Credit Union; and,
- 12) <u>Intellectual Property:</u> That they have the legal right to use all intellectual property necessary for the operation and conduct of their businesses, alfairs, operations and processes and that they will continue to maintain such legal right so long as any monies are owing to the Credit Union.

All representations and warranties set out herein shall survive the acceptance of the Credit Facility Letter and shall be deemed to be repeated at the time of each advance hereunder and the Credit Union may rely upon them upon each advance pursuant to the Credit Facility Letter except for those representations and warranties that the Credit Union has been notified of in writing that can no longer be repeated for any such advance. Failure to provide such notification shall, in the Credit Union's discretion, be considered an event of default and all credits may become immediately due and payable and the Credit Union may proceed to enforce the same in its discretion.

All other representations, warranties, certifications and statements of the Borrower or any Guarantor contained in any other document delivered pursuant to the Credit Facility Letter or thereafter to the Credit Union shall be deemed to constitute additional representations and warranties made by the Borrower or any Guarantor to the Credit Union.

#### GENERAL COVENANTS:

Each of the Borrower and each Guarantor covenants and agrees with the Credit Union, that for so long as any monies are due and payable to the Credit Union as follows:

- 1) <u>Timely Payment:</u> To pay all sums of money when due to the Credit Union.
- Performance of Obligations: To perform all of the obligations and covenants under the Credit Facility Letter or under any security document.
- <u>MaIntenance of Security</u>: To maintain in full force and effect any security contemplated by the Credit Facility Letter and any security that may be granted thereafter.
- 4) <u>Notice of Default</u>: To provide the Credit Union with prompt written notice of any event that constitutes, or which, with notice, lapse of time, or both, would constitute an Event of Default under the Credit Facility Letter or any security.
- 5) <u>Change of Ownership</u>: To give the Credit Union at least 30 days' prior written notice of any intended change in the ownership of its shares and not to consent to or facilitate a change in the ownership of its shares without prior written consent of the Credit Union.
- 6) <u>Insurance:</u> The Borrower, and the Guarantors shall insure and keep fully insured all property and assets in accordance with insurance requirement in the Credit Facility Letter so long as any monies are owing to the Credit Union.

Schedule "A" Standard Credit Terms and Conditions 012016

- 7) <u>Guarantees:</u> Not, without the prior written consent of the Credit Union, guarantee or otherwise provide for, on a direct, indirect or contingent basis, the payment of any monies or performance of any obligations by any other entity, except as may be provided for in the Credit Facility Letter.
- 8) <u>Sale of Property:</u> Not, without the prior written consent of the Credit Union, to sell, transfer, convey, lease or otherwise dispose of any of its property or assets subject to security under the Credit Facility Letter except where such is done in the ordinary course of the Borrower's business operations.
- Merger: Not, without the prior written consent of the Credit Union, to merge, amalgamate, or otherwise enter into any other form of business combination with any other entity.
- 10) <u>Environmental Non-Compliance:</u> To provide the Credit Union with prompt written notice of any noncompliance by the Borrower with any environmental laws or any release from the land of the Borrower by the Borrower of a contaminant into the natural environment and to Indemnify and save harmless the Credit Union from all liability of loss as a result of such environmental activity or any noncompliance with any environmental law.
- Illegal Activity: To not engage or allow any person in any of its business premises to engage in any activity that is contrary to any applicable laws and in particular any laws prohibiting criminal or illegal activities.
- 12) Personal Information: The Borrower and each Guarantor acknowledge that the Credit Union is required to verify and record information regarding the Borrower and each Guarantor, their directors, authorized signing officers, shareholders and any persons in control of the Borrower and each Guarantor and that they shall each promptly provide all such information including such other supporting documentation and other evidence as may be required by the Credit Union or any assignee or other entity participating in any credits with the Credit Union in order to comply with not only the Credit Union's internal identification policies but all applicable laws regarding anti-money laundering and "anti-terrorist financing".
- 13) <u>Inspection</u>: To permit the Credit Union or its representatives, from time to time, to visit and inspect the Borrower's premises, properties and assets and examine and obtain copies of the Borrower's records or other information and to discuss or otherwise communicate about the Borrower's affairs with the auditors, accountants, counsel and other professional advisers of the Borrower from time to time as the Credit Union may deem necessary.
- 14) <u>Monthly Books and Records</u>: That it shall keep proper books of record and account in which complete and correct entries will be made of all of its business transactions in accordance with ASPE or IFRS.
- 15) <u>Prudent Operation</u>: That it will keep all of its properties, assets and operations maintained and operated in accordance with diligent and prudent industry practice and in accordance with the law in compliance with any applicable insurance policy or policies covering such assets or activities.
- 16) Use of Loan Proceeds: That it shall use the proceeds of all loans and credits being made available to it pursuant to the Credit Facility Letter or otherwise solely for the purposes set forth thereunder and for no other purpose whatsoever without the prior, written consent of the Credit Union.
- 17) <u>Related Party Dealings</u>: The Borrower shall not, without the prior written consent of the Credit Union, enter into any contract, agreement or transaction whatsoever including for the sale, purchase, lease or other dealing in any property or provision of any service with any non-arms' length entity or any related party as defined in the *Business Corporations Act* of Alberta except upon fair and reasonable terms, which terms are not less favourable than it would obtain on a arms' length transaction and for a consideration which equals the fair market value of such property or other than a fair market rental as regards lease property.
- 18) <u>Other Encumbrances:</u> Not, without the prior written consent of the Credit Union, grant, create, assume or suffer to exist any mortgage, charge, lien, pledge, security interest or other encumbrance affecting any of its properties, assets or any other rights.

Schedule "A" Standard Credit Terms and Conditions 012016

- 19) <u>Exclusive Account Operation:</u> That for so long as there are any monies due and owing or any Credit Facility outstanding with the Credit Union, maintain all of its operating accounts with the Credit Union.
- 20) <u>Payment of Management or Shareholder Fees:</u> The Borrower will not pay or agree to pay any management or shareholder fees or executive management compensation except as agreed and approved by the Credit Union.

Nothing contained in the foregoing Covenants shall limit any right of the Credit Union under the Credit Facility Letter or any other agreement with the Borrower to terminate or demand payment of, or cancel or restrict availability of any unutilized portion of, any demand or other discretionary loan or credit facility made available by the Credit Union to the Borrower.

#### MISCELLANEOUS:

- <u>Cumulative Powers of Credit Union</u>: The rights and powers of the Credit Union pursuant to the Credit Facility Letter and the securities taken pursuant hereto are cumulative and not alternative, and not in substitution for any rights, remedies, or powers of the Credit Union.
- Failure: Any failure or delay by the Credit Union to exercise fully its rights and remedies pursuant to this Credit Facility Letter and the securities taken to pursuant hereto shall not be construed as a waiver of such rights and remedies.
- 3) <u>Time:</u> Time is of the essence.
- 4) Non-Assignability: This Credit Facility Letter Is not assignable by the Borrower In any manner.
- 5) <u>Laws of Alberta:</u> This Credit Facility Letter and the security documentation to be provided by the Borrower pursuant hereto shall be construed in accordance with and governed by the laws of the Province of Alberta.
- 6) <u>Cross-Default</u>: Any default hereunder or under any security document or other agreement between the Borrower and the Credit Union shall be a default under each and every other obligation of the Borrower to the Credit Union, whether or not collateral or supplemental hereto.
- 7) <u>Conflict:</u> The terms and conditions of this Credit Facility Letter shall not be merged by and shall survive the execution, delivery and registration of any and all security documents. In the event of a conflict between the terms of this Credit Facility Letter and the terms of any security document, the terms of this Credit Facility Letter shall prevail. For clarity, the mention of a provision in either the Credit Facility Letter and not in the security or vice versa shall not constitute a conflict but shall be deemed to be supplemental and in addition to any of the terms and conditions available under either the Credit Facility or the security as the case may be.
- 8) <u>Periodic Review of Accounts</u>: The Credit Union may conduct periodic reviews of the alfairs of the Borrower, as and when determined by the Credit Union, for the purpose of evaluating the financial condition of the Borrower. The Borrower shall make available to the Credit Union such financial statements and other information and documentation as the Credit Union may reasonably require and shall do all things reasonably necessary to facilitate such review by the Credit Union.
- 9) <u>Limitation Period</u>: The Borrower and all Guarantors by their signature and acceptance of the Credit Facility Letter hereby expressly provide that any and all limitation periods or action on any and all loans and credit facilities made available from time to time pursuant to the Credit Facility Letter or otherwise by the Credit Union shall be extended for a period of six years from the date of any event of default on a non-demand loan and for a period of six years on any demand loan from the time at which a demand for payment is made.
- 10) <u>Application of Account Balances</u>: The Credit Union is authorized, but not obligated, at any time, to apply any credit balance, whether or not then due, to which the Borrower is entitled and any account in any currency at any branch or office of the Credit Union in or towards satisfaction of the loans and obligations of the Borrower to the Credit Union at any time. The Credit Union is authorized to use any

Schedule "A" Standard Credit Terms and Conditions 012016

such credit balance to convert such credit balance to any currency required as may be necessary to effect such application.

- 11) <u>Non-Waiver</u>: No amendment or waiver of any provision of this agreement or any agreement amending, supplemental or relating hereto, will be effective unless it is in writing signed by the Borrower and the Credit Union. No failure or delay, on the part of the Credit Union in exercising any right or power hereunder or under any security document taken pursuant to the Credit Facility Letter or otherwise shall operate as a waiver thereof. The Guarantors agree that the amendment or waiver of any provision of this agreement (other than agreements, covenants or representations expressly made by the Guarantors herein, if any) may be made without and does not require the consent or agreement of, or notice to , the Guarantors.
- 12) <u>Enforceability:</u> If any provision of the Credit Facility Letter or any other agreement is or becomes prohibited or unenforceable in any jurisdiction, such prohibition or unenforceability shall not invalidate or render unenforceable the provision concerned in any other jurisdiction nor invalidate, affect or impair any of the remaining provisions of the Credit Facility Letter or any other agreement and such remaining provisions or parts thereof that are not invalid, illegal or unenforceable or severeable from such provision.
- 13) <u>Joint and Several Liability</u>: Where more than one person is liable as a Borrower or Guarantor for any loan or credit facility or obligation under the Credit Facility Letter or otherwise then the liability of each such person is joint and several with each other such person.
- 14) <u>Mandatory Membership</u>: Membership with Servus Credit Union requires that every Borrower invest a minimum of \$1.00 in Common Shares of the Credit Union Ltd. and such ownership and membership must be maintained so long as there are any monies and obligations outstanding by the Borrower to the Credit Union Ltd.
- 15) Accuracy of Information: The Borrower and Guarantor represent and warrant to the Credit Union that all information set out and certified in any electronic transfer and on any accompanying report is true and complete in all respects and acknowledges that the Credit Union Ltd. is relying upon all such representations and warranties from time to time.
- 16) Entire Agreement: This Credit Facility Letter, the security and any other written agreement delivered pursuant to or referred to in the Credit Facility Letter constitute the whole and entire agreement between the Credit Union and the Borrower and Guarantors in respect of the credit facilities hereunder. There are no verbal agreements, undertakings or representations by the Credit Union in connection with any loans or other credit made available to the Borrower at any time.
- 17) <u>Non-Merger:</u> All remedies provided herein shall be deemed to be in addition to and not restrictive of any other remedies of the Credit Union at law or in equity, may be enforced in priority to, or concurrently with, or subsequent to any other remedy or remedies, the Credit Union may rely upon the various securities and parts thereof in such order as it may deem fit without prejudice to any other realization and the powers of sale contained therein. The security provided for in this Credit Facility Letter is in addition to and not in substitution for any other security now or hereafter held by the Credit Union.
- 18) <u>Walver Under PPSA:</u> The Borrower expressly waives the right to receive any copies of any Financing Statements or Financing Change Statements (or any other jurisdiction equivalent) that might be registered by the Credit Union in connection with any security or any Verification Statement issued with respect thereto including all amendments, extensions or renewals of such registration and in any jurisdiction where not otherwise prohibited by law.
- 19) <u>Accounting Standards</u>: In the event that the Borrower or any Guarantor as a result of the changes to Canadian Accounting Standards on January 1, 2011 to the International Financial Reporting Standards ("IFRS"), Accounting Standards for Private Enterprises and/or Accounting Standards for Not-for-Profit Organizations and such adoption has an effect on any provision of the Credit Facility Letter relying on

Schedule "A" Standard Credit Terms and Conditions 012016

ž

financial statement calculations or other financial reporting requirements, the Credit Union may amend such provision to reflect the original intent of such provision at any time.

## SALE OR ASSIGNMENT OF CREDIT FACILITIES:

ŧ

- Assignability: The Credit Union shall have the unrestricted right to sell or assign the Credit Facilities or any loan thereunder, and/or the security documents (including this Credit Facility Letter), in whole or in part, in connection with any syndication, securitization or otherwise, to any other party or parties (each a "Holder"), and the Holder(s) shall thereafter have all the rights herein of the Credit Union, including the right to so sell or assign in turn.
- 2) <u>Syndication:</u> The Credit Union may from time to time, in connection with any syndication or securitization of the Credit Facilities or loans thereunder or otherwise, appoint or designate a custodian or agent for the same, which custodian or agent may be the registered security document holder. The Borrower and each Guarantor, if any, acknowledges that such custodian or agent will have no liability whatsoever to the Borrower or Guarantor, if any, in connection with the Credit Facility or loans thereunder, being merely custodian or agent for the Credit Union and/or Holders.
- 3) Loan Administration: The Credit Union shall have the unrestricted right from time to time to appoint a third party to service or administer the Credit Facilities or loans, and to deal with the Borrower and Guarantor, if any, in place of the Credit Union, provided that until the Credit Union gives notice of such appointment to the Borrower, the Borrower and Guarantor, if any, shall continue to deal with the Credit Union in matters pertaining to the servicing or administration of the Credit Facilities and loans.

# COLLECTION, USE, DISCLOSURE AND RELEASE OF FINANCIAL AND OTHER INFORMATION AND MATERIALS:

For the purposes of making, administering, reporting, selling or assigning in whole or in part, in connection with securitization or otherwise, and collecting the Credit Facilities and loans, the following parties (collectively, the "Authorized Parties") will be reviewing and examining financial and other information and materials provided to or obtained by the Credit Union concerning the Credit Facilities and loans, the Borrower and the Guarantor, if any:

- The Credit Union and/or any Holder or servicer of the Credit Facilities and loans or of an interest therein from time to time and/or their respective affiliates and/or agents;
- 2) Rating agencies, purchasers or investors and prospective purchasers or investors;
- Respective third party advisors of the parties listed in 1) and 2) above, such as lawyers, accountants, real estate brokers, investment dealers and underwriters, consultants, and appraisers; and,
- Credit verification sources.

The Borrower and each Guarantor, if any, acknowledges and irrevocably consents to the foregoing and irrevocably agrees that, in such manner as the Authorized Parties may determine to be necessary or desirable for these purposes, the Authorized Parties may disclose, release, exchange and share such information and materials:

- To and with any individual(s), corporation(s) or other entities designated from time to time to hold title to the Credit Facilities or loans and/or security documents as custodian(s) or agent(s);
- 2) To and with each other;
- 3) The Borrower and each Guarantor, if any, hereby consents to the Authorized Parties conducting such credit inquiries, as they may from time to time consider advisable for these purposes; and,
- 4) The provisions of this paragraph shall apply until all loans have been fully and completely repaid and the security documents have been discharged.

## EVENTS OF DEFAULT:

The occurrence of any one of the following shall constitute an Event of Default, the happening of which shall entitle the Credit Union, In its sole discretion, to demand immediate payment of all loans and credits in full, Schedule "A" Standard Credit Terms and Conditions 012016 Page 7 of B

together with outstanding accrued interest and any other costs outstanding, and to realize and enforce on any and all of the security granted in its favour under the Credit Facility Letter or otherwise:

- <u>Payment:</u> Failure of the Borrower to pay any principal, interest or other amount due and owing at any time.
- Breach of Term: Failure or refusal of the Borrower to observe or perform any term, covenant, condition
  or provision contained in this Credit Facility Letter or any documentation or security relating thereto.
- 3) <u>Cross-Default:</u> If the Borrower is in default under any other agreement with the Credit Union or any third party for the granting of the loan or other financial assistance and such default remains unremedied after any cure period provided in any other such agreement.
- 4) <u>Breach of Representation or Warranty:</u> If any representation or warranty made by the Borrower or any Guarantor in any document (including the Credit Facility Letter) is breached, false or misleading in any material respect or becomes at any time false or misleading in any material respect.
- 5) Accuracy of Documentation: If any schedule, certificate, financial statement report or other writing furnished by the Borrower or any Guarantor to the Credit Union in connection with the Credit Facility Letter or any other agreement is false or misleading in any material respect on the date on which it is certified or stated.
- 6) <u>Insolvency</u>: The Borrower or any Guarantor becomes insolvent or generally fails to pay its just debts as they become due or they apply for, consent to or acquiesce in the appointment of a trustee, receiver or other custodian for the Borrower or any Guarantor or any property thereof or makes a general assignment for the benefit of creditors, or for a trustee, receiver or other custodian is appointed for the Borrower or Guarantor or for a substantial part of the property of such Borrower or Guarantor, or any bankruptcy, reorganization, debt arrangement or other case or proceeding under any bankruptcy or Insolvency law, or any dissolution or liquidation proceeding, as commenced in respect of the Borrower or Corporate Guarantor or if they take any action to authorize or further any of the foregoing.
- Cessation of Business: The Borrower or Guarantors ceases or threatens to cease to carry on all or substantially all of their business.
- B) Margin Report Delivery: If margining reports are not received within 10 days of a reporting deadline as set forth in the Credit Facility Letter and the Credit Union determines it would issue a Notice of Default with respect to such reporting.
- 9) <u>Margin Report Demand:</u> If such margining reports have still not been received within 14 days from the issuance of a Notice of Default, the Credit Union Ltd. has the right to reduce all lines of credit and outstanding authorizations to zero.
- 10) <u>Material Adverse Change:</u> There occurs, in the sole opinion of the Credit Union, acting reasonably:
  - a) A material adverse change in the financial or environmental condition of the Borrower or any Guarantor,
  - b) An unacceptable, or unapproved change in ownership or control of the Borrower or any Guarantors takes place, or
  - c) Without authorization, the Borrower disposes of all or substantially all of its key business assets or substantially all of its assets.

## **OTHER SECURITY:**

In addition to the security referred to herein, the Borrower hereby pledges all deposits and paid up shares which it now or may have in the Credit Union, the proceeds of which may, upon default, be applied by the Credit Union to amounts then due and owing under any loan.

Schedule "A" Standard Credit Terms and Conditions D12016

ŧ



December 14, 2016

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC V5C 0E4

Private & Confidential

Attention: Karl Sigerist

Dear Mr. Sigerist,

#### RE: Amending Credit Facility Letter to Letter dated September 8, 2016

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit fadilities has been approved on the terms and conditions below. All other terms and conditions of the Letter stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

FEES:

- An Amendment Fee of \$250 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

#### **REVISED TERMS AND CONDITIONS:**

Demand Loan to pay Monitor's fees for \$100,000

Delete: Expiry date December 15, 2016

Insert: Expiry date December 22, 2016

Your acceptance of this letter will constitute authority for the Credit Union to Instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by December 15, 2016.

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 1 of 3

Yours truly,

Servus Credit Union Ltd.

per: <u>KG</u> Richard Charlton

Regional Manager Credit, Commercial

#### ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this  $\frac{15^{n}}{15}$  day of December, 2016

#### BORROWER(S):



Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 2 of 3

200

GUARANTOR(S): • Crelogix Portfolio Service Corporation per: Crelogix Credit Group Inc. per; de

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 3 of 3

1



December 22, 2016

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC VSC 0E4

#### Private & Confidential

#### Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016 and Amending letter dated December 14, 2016

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letter stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Chariton.

FEES:

- 1) An Amendment Fee of \$1,000 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

## REVISED TERMS AND CONDITIONS:

Demand Loan to pay Monitor's fees

Delete: Expiry date December 22, 2016

Insert: Explry date January 10, 2017

Delete: Existing Loan amount \$100,000

Insert: Revised loan amount \$150,000



Servus Credit Union Ltd.

Page 1 of 3

Your acceptance of this letter will constitute authority for the Credit Union to Instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by December 22, 2016.

Yours truly,

Servus Credit Union Ltd.

per: **Richard Charlton** 

Regional Manager Credit, Commercial

## ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this 27 day of December, 2016

#### BORROWER(S):

Crelogix Acceptance Corporation



Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 2 of 3

	• A 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	
	Crelogix Portfollo Service Corporation	
	<b>*</b>	
	Signed with Agreement karl Signerist Zaperist 22-Dec-16	
per:		
	Crelogix Credit Group Inc.	
per:	tigned with Agenetics Register Register 22 Case 16	
	·	
	5 1.00	
P		
	•	
	· .	
		r.
	• • •	
8		
	 • • <del></del>	
	·	
	·	
		* *.
	••	
	~	
n anna ann an a' ann an an an	ار دینی بیش این از این	n nya ang ang kanang manang
Credit Fadilt	y Laiter Amendment 032015 Servus Credit Union Ltd.	Page 3 of 3

٠

14 · 104



January 9, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC V5C 064

Private & Confidential

#### Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated December 14, 2016 and Amending letter dated December 22, 2016

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Chariton.

## FEES:

- An Amendment Fee of \$250 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

#### **REVISED TERMS AND CONDITIONS:**

Demand Loan to pay Monitor's fees + +

Delete: Expiry date January 10, 2017

Insert: Explry date January 30, 2017

Your acceptance of this letter will constitute authority for the Credit Union to instruct its solicitors to prepara any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by January 10, 2017.

isk: Creat Facility for Amendment 032015

Servius Credit Union Ltd.

Page 1 of 3

2

Yours truly,

Servus Credit Union Ltd.

per: **Richard Chariton** Regional Manager Credit, Commercial

#### ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amondment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this 10th day of January, 2017

## BORROWER(S):



Paga 2 of 3

GUARANTOR(S):

	Crelogix Portfolio Service Corporation
per:	

Crelogix Credit Group Inc. per .....

Credit Fadility Letter Amendment 032015

Servus Credit Union Ltd.,

Fage 3 of 3

Walance -

in an

2



January 30, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC V5C 0E4

Private & Confidential

Page 1 of 3

#### Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated December 14, 2016, Amending letter dated December 22, 2016 and Amending letter dated January 9, 2017

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

#### FEES:

- An Amendment Fee of \$250 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

#### **REVISED TERMS AND CONDITIONS:**

Demand Loan to pay Monitor's fees

Delete: Expiry date January 30, 2017

Insert: Expiry date February 28, 2017

Your acceptance of this letter will constitute authority for the Credit Union to Instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This friendment to the Credit Facility Letter shall expire if not accepted by January 30, 2017. Credit Facility Letter Shall expire if not accepted by January 30, 2017. Servus Credit Union Ltd. Yours truly,

Servus Credit Union Ltd.

per: Charlton

Regional Manager Credit, Commercial

## ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this  $\frac{30^{7.1}}{10^{10}}$  day of January, 2017

BORROWER(S):



Credit Facility Letter Amendment 032015

GUARANTOR(S):

Crelogix Portfolio Service Corporation Z per: Crelogix Credit Group Inc. pe

Credit Facility Letter Amendment 032015



February 28, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC VSC 0E4

. . .

#### Private & Confidential

Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated December 14, 2016, Amending letter dated December 22, 2016, Amending letter dated January 9, 2017 and Amending letter dated January 30, 2017

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

FEES:

- 1) An Amendment Fee of \$250 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Lettor, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

#### **REVISED TERMS AND CONDITIONS:**

Demand Loan to pay Monitor's fees

Delete: Explry date February 28, 2017

Insert: Expiry date March 31, 2017

Your acceptance of the letter will constitute authority for the Credit Union to Instruct its solicitors to prepare any necessary accurity or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This Frequencies of the Credit Facility Letter shall expire if not accepted by February 28, 2017. Credit Facility Letter shall expire if not accepted by February 28, 2017. Yours truly,

Servus Credit Union Ltd.

ne felles per: John McCellan Senior Credit Manager

## ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this 28 day of February, 2017

BORROWER(S):

Crelogix Acceptance Corporation + per: per

Credit Facility Letter Amendment 032015

GUARANTOR(5):

Crelogix Portfolio Service Corporation

per:

Crelogix Credit Group Inc. per

Page 3 of 3

\*

Servus Gredit Union Ltd.

Credit Facility Letter Amondment 032015

\*. \*

herrachtebaction -



March 17, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC V5C 0E4

Private & Confidential

Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated December 14, 2016, Amending letter dated December 22, 2016, Amending letter dated January 9, 2017, Amending letter dated January 30, 2017 and Amending letter dated February 28, 2017

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

FEES:

- An Amendment Fee of \$250 is considered earned by the Credit Union and Is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

#### REVISED TERMS AND CONDITIONS:

Demand Loan to pay Monitor's fees

Delete: Expiry date March 31, 2017

Insert: Expiry date the earlier of the closing of the sale of all issued and outstanding shares of ... Crelogix Credit Group Inc. to Enliven Financial Inc. (the "Closing Date") and April 15, 2017

Your acceptance of this letter will constitute authority for the Credit Union to instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

Credit Facility Letter Amendment 032015

This amendment to the Credit Facility Letter shall expire if not accepted by March 20, 2017.

Yours truly,

Servus Credit Union Ltd.

per:

Richard Charlton Regional Manager Credit, Commercial

## ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this 17 day of March, 2017

BORROWER(S):



Credit Facility Letter Amendment 032015

GUARANTOR(S):

r. ..

£

Crelogix Portfolio Service Corporation per:

Crelogix Credit Group Inc. per:

Credit Facility Letter Amendment 032015



April 19, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC V5C 0E4

Private & Confidential

#### Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated December 14, 2016, Amending letter dated December 22, 2016, Amending letter dated January 9, 2017, Amending letter dated January 30, 2017, Amending letter dated February 28, 2017 and Amending letter dated March 17, 2017

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

FEES:

- An Amendment Fee of \$250 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

**REVISED TERMS AND CONDITIONS:** 

Demand Loan to pay Monitor's fees

Delete: Expiry date the earlier of the closing of the sale of all issued and outstanding shares of Crelogix Credit Group Inc. to Enliven Financial Inc. (the "Closing Date") and April 15:2017/

Insert: Expiry data is the earlier of I) the closing the transaction set forth in the Confidential Ngn-Binding Term Sheet dated April 18, 2017 (the "Dealnet Term Sheet"), entered into as between the Credit Union, Crelogix Credit Group Inc. ("CCGI") and Dealnet Capital Corp. ("Dealnet") ii) either of the Credit Union, CCGI or Dealnet electing not to proceed with the transaction set forth in the Dealnet-Term Sheet and III) May 1, 2017.



Your acceptance of this letter will constitute authority for the Credit Union to instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by April 21, 2017.

Yours truly,

Servus Credit Union Ltd.

per: harlton **Director Commercial Credit** 

#### ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material advarse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

134 Accepted this day of April, 2017 BORROWER(S): Creloa eptance Corporation Der Der:

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 2 of 3

## GUARANTOR(5):

Crelogix fortfolio Service Corporation per: \_ Crelogia Credit Group Inc. . per:

Credit Facility Letter Amendment 032015

Servits Credit Union Ltd.

.

٠

Page 3 of 3

-----

.

.



May 1, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC V5C 0E4

Private & Confidential

Page 1 of

Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated December 14, 2016, Amending letter dated December 22, 2016, Amending letter dated January 9, 2017, Amending letter dated January 30, 2017, Amending letter dated February 28, 2017, Amending letter dated March 17, 2017 and Amending letter dated April 19, 2017

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

## FEES:

- An Amendment Fee of \$250 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

#### **REVISED TERMS AND CONDITIONS:**

Demand Loan to pay Monitor's fees

Delete: Expiry date is the earlier of i) the closing the transaction set forth in the Confidential Non-Binding Term Sheet dated April 18, 2017 (the "Dealnet Term Sheet"), entered into as between the Credit Union, Crelogix Credit Group Inc. ("CCGI") and Dealnet Capital Corp. ("Dealnet") lij, either of the Credit Union, CCGI or Dealnet electing not to proceed with the transaction set forth in the Dealnet Term Sheet and iii) May 1, 2017.

Insert: Expiry date is the earlier of i) the closing the transaction set forth in the Confidential Non-Binding Term Sheet dated April 18, 2017 (the "Dealnet Term Sheet"), entered into as between the Credit Union, Crelogix Credit Group Inc. ("CCGI") and Dealnet Capital Corn. ("Dealnet") ii) either of the Gredit Union, CCGI or Dealnet electing not to proceed with the transaction/set forth in the Dealnet Ferm Sheet and iii) May 15, 2017.

Servus Credit Union

redit Facility Letter Amendment 032015

Your acceptance of this letter will constitute authority for the Credit Union to instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by May 1, 2017.

Yours truly,

Servus Credit Union Ltd.

per:

Richard Charlton Director Commercial Credit

## ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this \_\_\_\_ day of May, 2017

BORROWER(S):

Crelogix Acceptance Corporation per: per

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 2 of 3

GUARANTOR(S):

Crelogix Portfolio Service Corporation per:

Crelogix Credit Group Inc.

per: G

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

.

Page 3 of 3



May 15, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC V5C 0E4

Private & Confidential

Page 1 of 3

Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated December 14, 2016, Amending letter dated December 22, 2016, Amending letter dated January 9, 2017, Amending letter dated January 30, 2017, Amending letter dated February 28, 2017, Amending letter dated March 17, 2017, Amending letter dated April 19, 2017 and Amending letter dated May 1, 2017

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

#### FEES:

- 1) An Amendment Fee of \$250 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter.
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

#### **REVISED TERMS AND CONDITIONS:**

Demand Loan to pay Monitor's fees

Delete: Expiry date is the earlier of i) the closing the transaction set forth in the Confidential Non-Binding Term Sheet dated April 18, 2017 (the "Dealnet Term Sheet"), entered into as between the Credit Union, Crelogix Credit Group Inc. ("CCGI") and Dealnet Capital Corp. ("Dealnet") ii) either, of the Credit Union, CCGI or Dealnet electing not to proceed with the transaction set forth in the Dealnet Term Sheet and iii) May 15, 2017.

Insert: Expiry date is the earlier of i) the closing the transaction set forth-in the Confidential Non-Binding Term Sheet dated April 18, 2017 (the "Dealnet Term Sheet"), entered into as between the Credit Union, Crelogix Credit Group Inc. ("CCGI") and Dealnet Capital Corp, ("Dealnet") ii) either



of the Credit Union, CCGI or Dealnet electing not to proceed with the transaction set forth in the Dealnet Term Sheet and iii) June 2, 2017.

Your acceptance of this letter will constitute authority for the Credit Union to instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by May 15, 2017.

Yours truly,

Servus Credit Union Ltd.

per: **Richard Charlton Director Commercial Credit** 

## ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter.

Accepted this 16 day of May, 2017

#### BORROWER(S):

Crelogix Acceptance Corporation ner: per:

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 2 of 3

GUARANTOR(S):

Crelogix Portfolio Service Corporation per: Crelogix Credit Group Inc. Sales State per:

the end of reference and the enderson of the other statements

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 3 of 3


Servus Credit Union Ltd. 151 Karl Clark Road NW Edmonton AB T6N 1H5 servus.ca

June 27, 2017

Crelogix Acceptance Corporation 4445 Lougheed Highway #900 Burnaby, BC VSC 0E4

Private & Confidential

'age 1 bi

#### Attention: Karl Sigerist

Dear Mr. Sigerist,

RE: Amending Credit Facility Letter to Letter dated September 8, 2016, Amending letter dated December 14, 2016, Amending letter dated December 22, 2016, Amending letter dated January 9, 2017, Amending letter dated January 30, 2017, Amending letter dated February 28, 2017, Amending letter dated March 17, 2017, Amending letter dated April 19, 2017, Amending letter dated May 1, 2017 and Amending letter dated May 15, 2017

Servus Credit Union Ltd. (the "Credit Union") advises that the following amendment(s) to your credit facilities has been approved on the terms and conditions below. All other terms and conditions of the Letters stated above including all security and guarantees provided therein remain unchanged except as amended by this Letter.

In consideration of the Credit Union providing or continuing to provide credit facilities and if you agree with these terms please sign this letter in the space provided below under the heading "Acceptance" and return it to the Credit Union, Attention: Richard Charlton.

#### FEES:

- 1) An Amendment Fee of \$250 is considered earned by the Credit Union and is payable by the Borrower upon acceptance of this Amendment Letter,
- 2) All fees due and payable, as outlined in this Amendment Letter, if not paid forthwith by the Borrower may be charged to the Borrower's account(s) or may be added to the Borrower's loans, at the discretion of the Credit Union, and shall be secured by all of the security taken in support of all loans by the Credit Union to the Borrower.

#### **REVISED TERMS AND CONDITIONS:**

Demand Loan to pay Monitor's fees

Delete: Loan amount \$150,000.00

Insert: Loan amount \$179,652.02 (plus accrued interest of \$2,909.89 as at June 27, 2017). Loan debited \$35,070.00 to pay the Alvarez & Marsai Canada ULC invoice dated May 5/2017.

Delete: Expiry date is the earlier of i) the closing the transaction set forth in the Confidential Non-) Binding Term Sheet dated April 18 2017 (the "Dealnet Term Sheet"), intered into as between the Credit Union, Crelogix Credit Group Inc. ("CCGI") and Dealnet Capital Corp. ("Dealnet") i) either.

redit Uhior



of the Credit Union, CCGI or Dealnet electing not to proceed with the transaction set forth in the Dealnet Term Sheet and iii) June 2, 2017.

#### Insert: July 7, 2017

Your acceptance of this letter will constitute authority for the Credit Union to instruct its solicitors to prepare any necessary security or other documentation required. This amendment commitment is not assignable without the prior written consent of the Credit Union.

This amendment to the Credit Facility Letter shall expire if not accepted by June 28, 2017.

Yours truly,

Servus Credit Union Ltd.

per: **Richard Charlton** 

Director Commercial Credit

#### ACCEPTANCE

Acceptance of this Amendment to the Credit Facility Letter provides full and sufficient acknowledgement that the Credit Union has no obligation to advance any funds under this agreement and if, in the opinion of the Credit Union, any material adverse change in risk occurs, the approved credit facility may be withdrawn or cancelled at the sole discretion of the Credit Union.

This Amendment to the Credit Facility Letter may be signed by the Credit Union and the Borrower and Guarantors in any number of counterparts, each of which shall be deemed to be an original, but all of which together shall constitute one and the same document.

In the discretion of the Credit Union, this Amendment to the Credit Facility Letter may also be transmitted by facsimile or by other electronic means and if so signed and transmitted, this Agreement shall be for all purposes as if the Credit Union and the Borrower and Guarantors had delivered an executed original Amendment to the Credit Facility Letter.

We hereby acknowledge and accept the credit facilities based on the terms and conditions outlined in the Letter(s) stated above and this Amendment Letter. 201%

Accepted this 🖂 day of June, 2017

BORROWER(S):

**Crelogix Acceptance Corporation** 

Credit Facility Letter Amendment 032015

Servus Credit Union Ltd.

Page 2 of 3

1 1 1 1 ... -20 per: per: GUARANTOR(S): Crelogix Portfolio Service Corporation per: Crelogix Credit Group Inc. per: a the same in the state of the second Credit Facility Letter Amendment 032015 Servus Credit Union Ltd. Page 3 of 3 .

This is **Exhibit "G"** referred to in the Affidavit of Darcy Peelar, sworn before me this 5<sup>th</sup> day of July, 2017

A Commissioner for Oaths in and for Alberta

Patrick T. Harnett Barrister and Solicitor

## CRELOGIX CREDIT GROUP INC.

## UNSECURED CONVERTIBLE NOTE AGREEMENT

#### Date: September 14, 2016

#### Principal Amount: CAD. \$2,000,000

FOR VALUE RECEIVED, Crelogix Credit Group Inc., a company incorporated under the laws of the Province of British Columbia (the "Company"), hereby promises to pay to the order of Servus Credit Union Ltd., ("Servus" or "Holder"), the amount set out above as Principal Amount (or so much thereof as may be outstanding from time to time, the "Principal Amount") on the Maturity Date (as hereafter defined), or earlier upon acceleration, or prepayment (in each case in accordance with the terms hereof) and to pay interest on any Principal Amount (and interest on any outstanding interest) (collectively, "Interest") at a rate per annum equal to the Prime Rate plus 4 % per annum compounded monthly in advance (the "Interest Rate"), from the date hereof until due and payable on the Maturity Date, or upon acceleration, Conversion, or prepayment (in each case in accordance with the terms hereof). This unsecured convertible note agreement (as the same may be amended, restated, or replaced from time to time) shall be referred to as the "Note". All capitalized terms used but not otherwise defined herein shall have the respective meanings attributed thereto in Schedule "A" attached hereto. All references to Dollars or \$ shall be to Canadian Dollars.

## ARTICLE 1 TERMS OF THIS NOTE AND CONDITIONS OF THE LOAN

#### Section 1.1 Loan

The Holder has agreed to extend a non-revolving loan (the "Loan") by way of a one-time advance to and in favour of the Company in the Principal Amount of \$2,000,000, subject to the terms hereof. The Loan is non-revolving. Any amount that is repaid may not be re-borrowed.

#### Section 1.2 Purpose of Loan

The Loan shall only be used by the Company as follows (A) on the date of the advance of the Loan the Company shall apply the Loan proceeds to pay in full the accounts payable of merchants of the Company identified in Schedule B attached hereto and the early payout liability identified in Schedule B attached hereto, and (B) thereafter the Company shall apply the Loan proceeds to finance the general operating purposes of the Company (including reasonable legal fees of the Company).

#### Section 1.3 Maturity; Payment

- (a) Unless otherwise converted or accelerated as provided herein, this Note will automatically mature and shall immediately be due and payable on December 15, 2016, as such date may be extended by the Holder in accordance with paragraph (b) below) (the "Maturity Date").
- (b) At any time on or before December 15, 2016 the Holder shall be entitled, but not obligated, to extend the then current Maturity Date for a period of up to a further 90 days by written notice of extension delivered by the Holder to the Company on or before Friday, December 9, 2016, which notice shall set forth the new extended Maturity Date, and thereafter all references to the Maturity Date in any Transaction Document shall be to the Maturity Date as so extended.

## Section 1.4 Interest on this Note

- (a) Interest shall accrue on the unpaid Principal Amount from the date hereof until the Principal Amount outstanding is paid in full at the Interest Rate, compounded monthly and computed on the basis of a 365-day year and actual days elapsed. All accrued Interest shall be payable on the earlier of (i) acceleration of the Obligations in accordance with the terms hereof and (ii) the Maturity Date. To the extent that any Principal Amount of this Note is Converted prior to the Maturity Date, accrued and unpaid Interest with respect to such converted Principal Amount shall be paid up to the Conversion Date on the earlier of (i) acceleration of the Obligations in accordance with the terms hereof and (ii) the Conversion Date.
- (b) Notwithstanding anything to the contrary contained in any Transaction Document, the interest paid or agreed to be paid under the Transaction Documents shall not exceed the maximum rate of non-usurious interest permitted by applicable law (the "Maximum Rate"). If the Holder shall receive interest in an amount that exceeds the Maximum Rate, the excess interest shall be applied to the outstanding Principal Amount of the Note or, if it exceeds such unpaid Principal Amount, refunded to the Company. For purposes of disclosure pursuant to the *Interest Act* (Canada), the annual rates of interest to which the rates of interest provided in this Note and the other Transaction Documents (and stated herein or therein, as applicable, to be computed on the basis of 365 days or any other period of time less than a calendar year) are equivalent to the rates so determined multiplied by the actual number of days in the applicable calendar year and divided by 365 or such other period of time, respectively.

# Section 1.5 Prepayments

- (a) Voluntary Prepayments.
  - (i) On or before November 15, 2016, provided that a binding commitment for a Qualified Sale has not been accepted by the Company or its shareholders, as applicable, all but not less than all of the outstanding Principal Amount may be prepaid by the Company (together with all accrued and unpaid Interest thereon), at the option of the Company subject to delivery by the Company to the Holder of a Prepayment Notice not less than 10 days prior to the proposed date of prepayment (or such shorter period as may be agreed to by the Holder in its absolute discretion) (such 10 day period referred to as the "Prepayment Notice Period") subject to the Prepayment Conversion rights of the Holder set forth in Section 3.1 hereof.
  - (ii) On and after November 15, 2016, the Company shall not have the right to prepay the outstanding Principal Amount prior to the Maturity Date without the prior written consent of the Holder, which consent shall be in the Holder's absolute and unfettered discretion.
- (b) Mandatory Prepayments. Without limiting the Holder's Conversion rights pursuant to Article 3 hereof, 100% of the net proceeds from any Qualified Sale shall, subject to any required prior repayment of the Senior Secured Debt in accordance with the Senior Secured Loan Documents (and the payment of amounts owed to Raymond James on account of placement fees, reasonable

legal fees owing to Blake, Cassels & Graydon LLP in advising the Company in respect of the sale process, and costs of closing the Qualified Sale transaction), be applied in repayment of the Obligations hereunder, up to a maximum amount equal to the Obligations then outstanding.

(c) Any prepayment of all or portion of the Principal Amount as permitted hereunder shall be accompanied by a payment of all accrued and unpaid Interest thereon, and other Obligations relating thereto and, shall be applied by the Holder first to unpaid expenses and indemnities, next to accrued and unpaid Interest and thereafter to Principal Amount.

## Section 1.6 Time and Place of Payments

Whenever any payment of the Obligations is to be made by the Company to the Holder pursuant to this Note, such payment shall be made in lawful money of the Canada by wire transfer of immediately available funds to such account or accounts as shall be specified by the Holder from time to time. Whenever any amount expressed to be due by the terms of this Note is due on any day which is not a Business Day, the same shall instead be due on the next succeeding day which is a Business Day.

## Section 1.7 Obligations Absolute

The Obligations under this Note and other Transaction Documents shall be absolute, unconditional and irrevocable and shall be paid or performed strictly in accordance with the terms of this Note and the other Transaction Documents under all circumstances including (i) the existence of any claim, set-off, or other right which the Company may have at any time against Holder (or any Persons for whom Holder may be acting), or any other Person, whether in connection with this Note or the other Transaction Documents, or the transactions contemplated thereby or hereby or any unrelated transaction or matter or (ii) any other circumstances or happening whatsoever.

#### Section 1.8 Obligations Unsecured

Notwithstanding anything to the contrary in any Senior Secured Debt Documents granted by the Company to Servus in its capacity as lender of the Senior Secured Debt, the Obligations hereunder shall be unsecured.

## ARTICLE 2 CONDITIONS PRECEDENT TO THE LOAN.

#### Section 2.1 Conditions Precedent

The following shall have been met to the satisfaction of the Holder as a condition precedent to the obligation of the Holder to make the Loan by no later than 1:00 p.m. (Edmonton time) on September 15, 2016. Where delivery of documents is referred to, the documents must be in form and substance satisfactory to the Holder, duly executed by all parties and in full force and effect, and all matters disclosed by the documents must be satisfactory to the Holder:

- (a) the Holder shall have received the following items, each of which shall be, unless otherwise noted, dated as of the date hereof:
  - (i) this Note, a joint and several guarantee of the Obligations in favour of the Holder by each of the Company's Subsidiaries (the **"Subsidiary**

**Guarantees**"), the Senior Management Shareholder Waiver and Agreement and the other Transaction Documents;

- (ii) without limiting (i) immediately above, any necessary waivers required to enable Servus on the exercise of the Conversion rights to hold the percentages of Shares contemplated herein without dilution (including without limitation waivers of pre-emptive rights or adjustment rights) including any waivers required from any employees, directors, officers or of the Company entitled to Options including under the Stock Option Plan;
- a currently dated certificate of good standing or equivalent of the appropriate Governmental Authority of the jurisdiction of incorporation of the Company and each of the Subsidiaries, stating that such Person is in good standing in such jurisdiction;
- (iv) a certificate from a senior officer of each of the Company and the Subsidiaries addressed to the Holder and its counsel attaching a certified and true copy of (i) a resolution of the board of directors and Shareholders of the Company or Subsidiary as applicable approving the due authorization, execution, delivery and performance of the Transaction Documents to which it is a party,(ii) the share register of the Company and the Subsidiaries (iii) all Option Agreements (iv) the Stock Option Plan (v) the Shareholders Agreement, (vi) an incumbency certificate, and certifying such related matters as the Holder may reasonably require;
- (b) all necessary authorizations, consents and approvals necessary in connection with the consummation of the transactions contemplated by this Note and the other Transaction Documents shall have been given unconditionally and without containing any onerous terms and the Company shall have satisfied all applicable waiting periods necessary in connection with the consummation of the transactions contemplated by this Note and the other Transaction Documents;
- (c) the Holder shall have received an opinion from counsel for the Company and the Subsidiaries regarding the due incorporation and existence of the Company, the corporate power and capacity of the Company and the Subsidiaries, the issued and authorized share capital of the Company and the Subsidiaries, the due authorization, execution, delivery of the Transaction Documents to which the Company and the Company' Subsidiaries are a party, a no breach opinion in respect of the Company's and Subsidiaries' Organizing Documents and all applicable laws, requirement of regulatory approvals and such other matters as the Holder and its counsel may reasonably require, in form and substance satisfactory to the Holder;
- (d) all of the conditions required to be satisfied under the other Transaction Documents, if any shall have been satisfied;
- (e) all representations and warranties of the Company contained in this Note and of the Company and the other parties in the other Transaction Documents shall be true and correct in all material respects, and there shall exist no Event of Default under this Note or any default under any Transaction Document or any event which, with the giving of notice or lapse of time, or both, could become an Event of Default or an event of default or default under any Transaction Document as of

the date hereof, after giving effect to the transaction contemplated hereby or thereby; and

(f) the Holder shall have received such additional evidence, documents, undertakings or opinions as it may reasonably require to complete the transactions contemplated hereby in accordance with the terms and conditions contained herein.

#### Section 2.2 Waiver

All such conditions precedent, are imposed hereby solely for the benefit of Holder, and no other party may require satisfaction of any such condition precedent or be entitled to assume that Holder will refuse to make the Loan in the absence of strict compliance with such conditions precedent. Any requirement of this Note may be waived, in whole or in part, in a specific written waiver intended for that purpose and signed by Holder. No waiver by Holder of any condition precedent or obligation shall preclude Holder from requiring such condition or obligation to be met from thereafter declaring the failure to satisfy such condition or obligation to be an Event of Default.

#### ARTICLE 3 CONVERSION

#### Section 3.1 Conversion Rights

The Holder shall have the right at its sole option to convert this Note as follows (each a "Conversion"):

- (a) Conversion Rights on Prepayment. At any time, during the Prepayment Notice Period, the Holder shall have the right at its sole option, to convert ("Prepayment Conversion") into fully paid, validly issued Shares, \$1,000,000 ("Prepayment Conversion Amount") of the outstanding Principal Amount into that number of Shares which would result in the Holder holding at the time of the Prepayment Conversion, in the aggregate, 49.9% of the Issued Common Shares of all classes, calculated as of the effective date of such Prepayment Conversion.
- (b) Conversion Rights Upon Qualified Sale or Prior to Maturity. In addition to the Prepayment Conversion Rights, the Holder shall have the right at any time during the Voluntary Conversion Periods (but no later than the closing of the consummation of subject Qualified Sale or the Maturity Date, as applicable) at its sole option, to convert ("Voluntary Conversion") all or any part of the outstanding Principal Amount (the "Voluntary Conversion Amount") into fully paid and non-assessable Issued Common Shares in accordance with the following conversion rate: (i) if the Voluntary Conversion Amount is equal to the Principal Amount of \$2,000,000, then the number of Shares issuable to the Holder on conversion shall, subject to Section 3.2 hereof, be that number of Shares that results in the Holder holding at the time of Conversion, in the aggregate, 80% of the Issued Common Shares of all classes (the "Conversion Threshold"), calculated as of the effective date of such Voluntary Conversion; (ii) if the Voluntary Conversion Amount is less than the full Principal Amount of \$2,000,000, then the Conversion Threshold (and the corresponding number of Shares issuable to the Holder on Conversion) shall, subject to Section 3.2 hereof, be proportionately adjusted downward from 80% to a percentage which is relative to the reduced Voluntary Conversion Amount. The determination of such

adjusted Conversion Threshold shall be mutually determined by the Holder and the Company prior to the Voluntary Conversion.

(c) In addition to the Prepayment Conversion and Voluntary Conversion rights, the Holder shall have the right at its option to convert the outstanding Principal Amount upon the occurrence of an Event of Default in accordance with Section 7.2 hereof.

#### Section 3.2 Sale of Shares pursuant to Qualified Sale transaction

In the event the Company and the Shareholders have received and accepted a binding Offer for a Qualified Sale by way of a sale of at least all of the issued and outstanding Shares in the capital of the Company other than the Subsidiary Preferred Shares, with a purchase price of not less than \$11,000,000 (net of amounts owed to Raymond James on account of placement fees, reasonable legal fees owing to Blake, Cassels & Graydon LLP in advising the Company in respect of the sale process, and costs of closing the Qualified Sale transaction), and the Holder has issued a Conversion Notice in accordance with Section 3.1(a) with a Voluntary Conversion Amount equal to all but not less than all of the then outstanding Principal Amount, then the Conversion Threshold shall be downwardly adjusted from 80% and the number of Shares to which the Holder shall be entitled on Conversion shall be correspondingly adjusted as follows:

- (a) Where the net proceeds from a Qualified Sale transaction are in excess of \$11,000,000 but less \$20,000,000, then the Conversion Threshold shall be reduced by 3.50% for every \$1,000,000 in such net proceeds in excess of \$11,000,000 (pro-rated for such excess amounts that are increments less than \$1,000,000); and
- (b) Where the net proceeds from a Qualified Sale transaction are equal to or greater than \$20,000,000, then in addition to the adjustments referred to in paragraph (a) above, the Conversion Threshold shall be reduced by 1.25% for every \$1,000,000 in such net proceeds equal to or in excess of \$20,000,000 (pro-rated for such excess amounts that are increments less than \$1,000,000); provided that the minimum Conversion Threshold shall not be adjusted to below 42% in any circumstances.

#### Section 3.3 Mechanics of Conversion

- (a) To convert the Prepayment Conversion Amount or any Voluntary Conversion Amount as applicable into Shares on any date as permitted hereunder, the Holder shall deliver to the Company, for receipt on or prior to 3:30 p.m., Edmonton time, on such date, a copy of an executed notice of conversion in the form attached hereto as Schedule C (the "Conversion Notice").
- (b) The Conversion shall be deemed to have occurred (i) in the case of a Prepayment Conversion, concurrently with the applicable prepayment triggering the Conversion right, (ii) in the case of a Voluntary Conversion in connection with a Qualified Sale pursuant to Section 3.1(b), concurrently with the consummation of the Qualified Sale transaction, and (iii) in the case of a Voluntary Conversion on the Maturity Date pursuant to Section 3.1(b) on the close of business on the day immediately preceding the Maturity Date, (each a "Conversion Date"). On the Conversion Date, the Company shall issue and deliver to Holder at the address as specified in the Conversion Notice, a certificate, registered in the name of the Holder and dated as of the Conversion Date, for the number of

Shares to which the Holder shall be entitled and update the register of Shareholders of the Company to reflect the Holder as the holder of such Shares to which the Holder shall be entitled, and the Holder shall be treated for all purposes as the record holder of such conversion Shares as of the Conversion Date. All Shares issued upon the Conversion of this Note, or pursuant to Section 3.4, shall be validly issued, fully-paid and non-assessable.

- (c) The Company will not by amendment of Organizational Documents or through any reorganization, recapitalization, transfer of assets, consolidation, merger, dissolution, issue or sale of Capital Stock or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Note or any other Transaction Document to be observed or performed hereunder by the Company but will at all times in good faith assist in the carrying out of all the provisions of this Note and the other Transaction Documents and in the taking of all such action as may be necessary or appropriate in order to protect the Conversion rights of the Holder of this Note against impairment.
- (d) Any failure by the Company within three (3) Business Days of the Conversion Date, to issue and deliver a certificate to the Holder representing the number of Shares to which the Holder is entitled upon Conversion of any Conversion Amount and to update the register of Shareholders of the Company to reflect the Holder of such Shares to which the Holder is entitled shall constitute a "Conversion Failure".
- (e) Notwithstanding anything to the contrary set forth herein, upon Conversion of any portion of this Note in accordance with the terms hereof, the Holder shall not be required to physically surrender this Note to the Company unless (A) the full Conversion Amount represented by this Note is being converted or (B) the Holder has provided the Company with prior notice (which notice may be included in a Conversion Notice) requesting reissuance of this Note upon physical surrender of this Note. The Holder and the Company shall maintain records showing the Conversion Amount converted and the dates of such Conversion or shall use such other method, reasonably satisfactory to the Holder and the Company, so as not to require physical surrender of this Note upon Conversion (other than as provided in the immediately preceding sentence).
- (f) In the event of a dispute as to the number of Shares issuable to the Holder in connection with a conversion of this Note, the Company shall issue to the Holder the number of Shares not in dispute and resolve such dispute in accordance with Section 8.5.
- (g) Notwithstanding anything to the contrary in this Note, or in any other Transaction Document or related document, including without limitation, the Shareholders Agreement and any Option Agreements, nothing in any such documents, or any part thereof, or in any other related document, security, guarantee, agreement, or the like shall require any adjustments, conversions, trigger any pre-emptive rights or anti-dilution mechanisms of any kind, including without limitation, exercise price adjustments, or similar rights due to the carrying out any Conversion or any part thereof, of the Company or any Affiliate thereof.
- (h) The Company shall not issue any fraction of a Share upon any Conversion. If the issuance would result in the issuance of a fraction of a Share, the Company shall round such fraction of a Share up to the nearest whole Share. The

Company shall pay any and all Taxes and Other Taxes that may be payable with respect to the issuance and delivery of Shares upon conversion of any Conversion Amount.

#### Section 3.4 Dilution Protection

If on or after a Conversion Date and for so long as Servus is a Shareholder, the holder of an Option should exercise its rights to acquire Shares of the Company (an "**Option Exercise**"), then at the time of the Option Exercise the Company shall issue such number and class of Shares to the Holder, for an aggregate issue price of \$1.00, as may be necessary for the Holder to maintain the same Percentage Interest after Shares are issued under the Option Exercise as the Percentage Interest of the Holder immediately prior to the Option Exercise, without any dilutive effect. The provisions of this Section 3.4 shall survive any termination of this Note and satisfaction of the Obligations.

# ARTICLE 4

# QUALIFIED SALE AND SENIOR MANAGEMENT SUBSCRIPTION RIGHTS

#### Section 4.1 Offer

In the event that the Company or its Shareholders as applicable receives an offer ("Offer") prior to the repayment in full of the Obligations hereunder from another Person for the purchase of the Company (or of all or substantially all of the assets and undertaking of the Company), the Company shall immediately provide a copy of such Offer to the Holder together with such additional information as the Holder may request in respect of the Offer (including without limitation, the anticipated aggregate proceeds and the date the sale is expected to be completed. Not later than 10 days following the receipt of the Offer, the Holder shall advise the Company in writing whether or not the terms of the purchase and sale of the Offer constitute a Qualified Sale. In the event the Holder so advises the Company that Offer constitutes a Qualified Sale, the Company (or its Shareholders as applicable) shall accept the Offer without variation and consummate the Qualified Sale in accordance with the Offer without variation or waiver. This provision shall supersede and replace the requirements under section 4.6 of the Sale Process Agreement for the Approving Shareholders (as defined in the Sale Process Agreement) to approve a Proposed Transaction (as defined in the Sale Process Agreement), and a Qualified Sale hereunder shall be, and be deemed to be, an "Approved Transaction" under the Sale Process Agreement.

#### ARTICLE 5 COMPANY REPRESENTATIONS AND WARRANTIES

#### Section 5.1 Representations and Warranties

The Company makes the following representations and warranties to the Holder as to itself and, as applicable, its Subsidiaries, as of the date of this Note:

(a) The Company and each of its Subsidiaries (i) is a duly organized or incorporated, as the case may be, and validly existing entity in good standing under the laws of the Province of British Columbia and has the corporate power and authority to own its property and assets and to conduct its business as currently conducted in any jurisdiction where it is required to be so qualified and where the failure to be so qualified could not reasonably be expected to have a Material Adverse Effect.

- (b) The authorized and issued equity interests (including options, warrants, convertible securities and rights to acquire the same) and ownership of the Company and each of its Subsidiaries is set forth in Schedule D hereof. Except as set forth on Schedule D hereof, no Options, warrants, conversion rights, preemptive rights or other statutory or contractual rights to purchase or otherwise acquire Capital Stock of the Company or the Subsidiaries now exist, nor has the Company or the Subsidiaries authorized any such right, nor is the Company or any Subsidiary obligated in any other manner to issue Capital Stock/share capital or other equity securities. The Company has no Subsidiaries other than as listed in Schedule D attached hereto.
- (C) Each of the Company and the Subsidiaries has the corporate power and authority to execute, deliver and perform the Transaction Documents to which it is a party and has taken all necessary corporate action to authorize the execution, delivery and performance by it of each of such Transaction Documents. This Note and the other Transaction Documents constitutes valid and legally binding obligations of the Company and the Subsidiaries to the extent a party thereto enforceable in accordance with their terms, except to the extent that enforceability thereof may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws generally affecting creditors' rights and by equitable principles regardless of whether enforcement is sought in equity or at law. The execution, delivery and performance of the Transaction Documents by the Company and the Subsidiaries and the other parties thereto and the consummation by the Company and the Subsidiaries and the other parties thereto and hereto of the transactions contemplated hereby and thereby (including, without limitation, the issuance of the Shares of the Company) will not: (i) result in a violation of any of the Organizational Documents of the Company or the Subsidiaries or the terms of the Shares of the Company; (ii) conflict with, or constitute a default (or an event which with notice or lapse of time or both would become a default) under, or give to others any rights of termination, amendment, acceleration or cancellation of, any agreement, indenture or instrument to which the Company or a Subsidiary is a party; or (iii) result in a violation of any law, rule, regulation, order, judgment or decree applicable to the Company or any of its Subsidiaries or by which any property or asset of the Company or any of its Subsidiaries is bound or affected.
- (d) Except to the extent of the defaults disclosed in this Note and in the Forbearance Agreement and other Transaction Documents:
  - all financial and other information provided by or in respect of the Company and any of its Subsidiaries to the Holder was true, correct and complete in all material respects when provided; and
  - (ii) no information, exhibit, or report furnished by the Company or any of its Subsidiaries to the Holder contains any material misstatement of fact or omits to state a material fact or any fact necessary to make the statement contained therein not materially misleading in the circumstances in which it was made.
- (e) No consent, approval, order or authorization of, or registration, qualification, designation, declaration or filing with, any federal, provincial or local Governmental Authority is required that has not been obtained, and no consent of any third party is required that has not been obtained, in order to avoid (i) any

agreement to which the Company or any of its Subsidiaries is party being in default, or (ii) such third party having a right of termination after giving effect to the issuance of the Note and the consummation of the transactions contemplated by this Note and the other Transaction Documents, other than defaults and rights of termination as of the date of this Note and as disclosed in Schedule 'H' hereto, that may be exercised by merchants, suppliers and funders of the Business who were not paid amounts owing to them when due by the Company and its Subsidiaries including, without limitation, those who remain unpaid as of the date of this Note. None of the Company or any of its Subsidiaries is required to obtain any consent, authorization or order of, or make any filing or registration with, any court, Governmental Authority or any regulatory or self-regulatory agency or any other Person in order for it to execute, deliver or perform any of its obligations under or contemplated by the Transaction Documents, in each case, in accordance with the terms thereof. All consents, authorizations, orders, filings and registrations which the Company or any of its Subsidiaries is required to obtain pursuant to the preceding sentence have been obtained or effected on or prior to the date hereof, and the Company is unaware of any facts or circumstances which might prevent the Company or any of its Subsidiaries from obtaining or effecting any of the registration, application or filings pursuant to the preceding sentence.

- (f) On the date hereof, and after giving effect to the transactions contemplated by the Transaction Documents, (i) each Material Contract of the Company and each of its Subsidiaries is in full force and effect, and no party thereunder is in breach of or default under any provision thereof and (ii) neither the Company or any of its Subsidiaries has received any notice of repudiation, termination, cancellation or non-renewal of any of the Material Contracts to which it is a party or is aware of any fact or circumstance which is likely to give rise to a breach of, default under, or termination of, any such contract.
- (g) The issuance of the Shares of the Company in connection with any issuance or Conversion in accordance with this Note, has been duly authorized and such securities, upon issuance, shall be free from all Liens and charges created by the Company with respect to the issue thereof. Upon issuance or conversion in accordance with this Note, the Shares issued in connection therewith, will be validly issued, fully paid and non-assessable and free from all Taxes, Liens and charges with respect to the issue thereof and all pre-emptive or similar rights shall have been waived, with the Holder being entitled to all rights accorded to a holder of Shares except as those rights may be affected by the provisions contained in the Organizational Documents of the Company and the Shareholders Agreement (other than those provisions waived pursuant to the terms of the Senior Management Shareholder Waiver and Agreement).
- (h) The Company acknowledges that its obligation to issue Shares upon Conversion in accordance with this Note is absolute and unconditional regardless of the dilutive effect that such issuance may have on the ownership interests of other Shareholders of the Company.
- (i) The Company is not a reporting issuer (as such term is defined in the *Securities Act* (British Columbia)) and there is no published market for any of the shares in the capital of the Company.

- (j) The issuance of this Note and the issuance of Shares upon Conversion as contemplated herein, are exempt from the registration requirements of the prospectus requirements of Canadian securities laws and other applicable securities laws. Neither Company nor any authorized agent acting on its behalf will take any action hereafter that would cause the loss of such exemption.
- (k) Other than as disclosed in writing to the Holder:
  - the Company has not entered into any written or oral agreement or understanding providing for severance or termination payments to any director, officer or employee in connection with the termination of their position or their employment;
  - (ii) the Company is not subject to any claim for severance, wrongful dismissal, constructive dismissal or any other tort claim, actual or, to the knowledge of the Company, threatened, or any litigation actual, or to the knowledge of the Company, threatened, relating to employment or termination of employment of employees or independent contractors; and
  - (iii) there has been no amendment to, announcement by the Company relating to, or change in compensation arrangement with any employee or contractor that will (i) entitle any employees or contractors of the Company to severance pay or any increase in severance pay upon any termination of employment after the date hereof, (ii) accelerate the time of payment or vesting or result in any payment or funding (through a grantor trust or otherwise) of compensation or benefits, or increase the amount payable or result in any other material obligation pursuant to, any compensation arrangement.
- (I) With the exception of the Obligations to the Holder (including those obligations under the Senior Secured Debt Documents), employment compensation and benefits payable to employees in the Ordinary Course, the Option Agreements, and transactions between the Company and its Subsidiaries (as disclosed in Schedule 'I' hereto):
  - no amounts are payable by the Company or its Subsidiaries to the officers, directors, employees of the Company or any of its Subsidiaries or any Shareholders; and
  - (ii) the Company and its Subsidiaries are not presently a party to any transaction with any Person who does not deal at arm's length with the Company and its Subsidiaries.
- (m) Except for:
  - the Obligations to the Holder (including obligations under the Senior Secured Debt Documents);
  - the unpaid merchants, suppliers and funders of the Business and any intercorporate loans made among the Company and its Subsidiaries as more particularly set forth in Schedule 'F'; and
  - (iii) trade payables owing in the Ordinary Course,

neither the Company nor any of its Subsidiaries has any outstanding Indebtedness or is in violation of any term or in default under any contract, agreement or instrument relating to any Indebtedness.

- (n) There is no action, litigation, investigation or proceeding pending or, to the knowledge of the Company, threatened with respect to the Company or any of its Subsidiaries by or before any Governmental Authority that could reasonably be expected to have a Material Adverse Effect.
- (o) The Company and each of its Subsidiaries has (i) made or filed all foreign, federal and provincial income and all other tax returns, reports and declarations required by any jurisdiction to which it is subject, (ii) paid all taxes and other governmental assessments and charges that are material in amount, shown or determined to be due on such returns, reports and declarations, except those being contested in good faith and (iii) set aside on its books provisions reasonably adequate for the payment of all taxes for periods subsequent to the periods to which such returns, reports or declarations apply. There are no unpaid taxes in any material amount claimed to be due by the taxing authority of any jurisdiction, and the Company and each of its Subsidiaries knows of no basis for any such claim.
- (p) Except as expressly disclosed elsewhere in this Note, the Forbearance Agreement and other Transaction Documents, there are no facts known to the Company which could reasonably be expected to have a Material Adverse Effect on the Company or the Subsidiaries including their ability to observe and perform their respective obligations under the Transaction Documents.
- (q) The Company understands and confirms that the Holder has relied on the foregoing representations in effecting transactions in securities of the Company. All disclosure provided to the Holder regarding the Company and/or any of its Subsidiaries, their business and the transactions contemplated hereby and in the Transaction Documents, including the Schedules to this Note, furnished by or on behalf of the Company or any Subsidiary or Shareholder of the Company is true and correct and does not contain any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading.

#### Section 5.2 Survival of Representations and Warranties

The Company acknowledges that the Holder is relying upon the foregoing representations and warranties in connection with the establishment and advancement of the Loan. Notwithstanding any investigations which may be made by the Holder, the said representations and warranties shall survive the execution and delivery of this Note until full and final payment and satisfaction of the Obligations hereunder.

#### ARTICLE 6 COVENANTS

#### Section 6.1 Positive Covenants.

The Company covenants and agrees that, until all of its Obligations have been paid in full and this Note ceases to be outstanding that it will, and will cause each of its Subsidiaries as applicable to, perform the following covenants:

- (a) Payment of Obligations. The Company and each of its Subsidiaries will duly and punctually pay all Obligations owing under this Note and the other Transaction Documents to which it is a party when due. The Company and each of its Subsidiaries will observe and comply with all other requirements applicable to it pursuant to this Note and the other Transaction Documents to which it is a party and any other agreement or instrument entered by it into in connection with this Note.
- (b) Use of Proceeds. The Company will apply the proceeds of this Note exclusively in accordance with Section 1.2 hereof.
- (c) Maintenance of Existence; Compliance with Contracts, Laws, etc. The Company will, and will cause each of its Subsidiaries to, (i) preserve and maintain its legal existence (ii) take all steps reasonably necessary to obtain, preserve, renew, extend and keep in full force and effect the rights, licenses, permits, franchises, authorizations, patents, copyrights, trademarks and trade names material to the conduct of its business, (iii) comply with all applicable requirements of law, whether now in effect or hereafter enacted, except where the failure to comply, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect, and (iv) perform in all material respects its obligations under Material Contracts to which it is a party.
- (d) *Reporting.* The Company shall deliver to the Holder:
  - prompt notice (and in any event within five Business Days) after any (i) officer of the Company or any of its Subsidiaries obtains knowledge (i) of the existence of any Event of Default, or becomes aware that the holder (other than Servus) of any Indebtedness of the Company or any of its Subsidiaries has given any notice or taken any other action with respect to a claimed default thereunder, (ii) of any change in the Company's or any of its Subsidiaries' certified accountant or any resignation, or decision not to stand for re-election, by any member of the Company's board of directors, (iii) any material change in the Company's or its Subsidiaries' accounting or financial reporting practice (it being acknowledged by the Holder that the Company and its Subsidiaries are in the process of changing their financial reporting from Accounting Standards for Private Enterprise (ASPE) to International Financial Reporting Standards (IFRS)), (iv) that any Person has given notice to the Company or any of its Subsidiaries or taken any other action with respect to a claimed default under any Material Contract (but excluding therefrom notice relating to those unpaid Material Contracts which will be paid from the proceeds of the Loan, as contemplated under this Note), and (v) of the institution of any litigation or any adverse determination in any litigation, in any case specifying the nature thereof and what actions the Company or any of its

Subsidiaries has taken, is taking or proposes to take with respect thereto (but excluding therefrom actions involving the Company or its Subsidiaries in the Ordinary Course, provided the aggregate amount of these actions does not exceed \$50,000);

- (ii) the Company shall within 5 days following each calendar month end deliver to the Holder a report (with such detail as may be requested by the Holder) of all Reimbursable Expenses incurred in the immediately prior calendar month.
- (e) Taxes and Other Charges. The Company will, and will cause each of its Subsidiaries to, duly pay and discharge, or cause to be paid and discharged, before the same becomes in arrears, all Taxes and other governmental charges imposed upon such Person and its properties, sales or activities, or upon the income or profits therefrom, and duly pay and discharge all claims for labor, materials or supplies which if unpaid might by law become a Lien upon any of its property, prior to its becoming such a Lien; provided, however, that any such Tax, charge or claim need not be paid if the validity or amount thereof shall at the time be contested in good faith by appropriate proceedings and if such Person shall, in accordance with GAAP, have set aside on its books adequate reserves with respect thereto; and provided, further, that the Company will, and will cause each of its Subsidiaries to, pay or bond, or cause to be paid or bonded, all such Taxes, charges or other governmental claims immediately upon the commencement of proceedings to foreclose any Lien which may have attached as security therefor (except to the extent such proceedings have been dismissed or stayed).
- (f) The Company will keep, and will cause each of its Subsidiaries to keep, proper books of record and account in accordance with GAAP in which full, true and correct entries shall be made of all dealings and transactions in relation to its business and activities.

# Section 6.2 Negative Covenants

The Company covenants and agrees that, until all of the Obligations have been paid in full:

- (a) Indebtedness. The Company shall not (and shall not permit any of its Subsidiaries to) create, incur, assume, or otherwise become or remain directly or indirectly liable with respect to any Indebtedness, except:
  - (i) Indebtedness incurred pursuant to this Note and the other Transaction Documents;
  - (ii) the Indebtedness incurred pursuant to the Senior Secured Debt Documents;
  - (iii) Indebtedness in connection with operating and capital leases in the Ordinary Course;
  - (iv) Indebtedness incurred in connection with the funding/securitization arrangements of the Business as described in Schedule 'G' hereto; and
  - (v) additional Indebtedness incurred by the Company and its Subsidiaries consented to in writing by Servus in its sole and absolute discretion.

- (b) *Liens and Related Matters.* The Company shall not and shall not permit its Subsidiaries or Shareholders, to the extent of the Shares held by such Shareholders from time-to-time, to directly or indirectly:
  - create, incur, assume or permit to exist any Lien on or with respect to any property or asset of the Company or any of its Subsidiaries or any Capital Stock of the Company, whether now owned or hereafter acquired, or any income or profits therefrom (except Permitted Liens).
  - (ii) enter into or assume any agreement (x) prohibiting the creation or assumption of any Lien upon its properties, revenues or assets, whether now owned or hereafter acquired or (y) restricting the ability of the Company to amend or otherwise modify any Transaction Document or other agreement or instrument related to this transaction.
- (c) Restrictions on Issuance of Shares, New Subsidiaries and Related Matters. Except as expressly permitted by the Transaction Documents or the Option Agreements, subject to Section 3.4 hereof, or otherwise in connection with the closing of a Qualified Sale, the Company shall not issue any additional Capital Stock or organize any new Subsidiaries subsequent to the date hereof, or issue any purchase rights, options, convertible notes, securities, phantom shares or stock, share or stock appreciation rights, or any similar form of interest (in the form of equity, rights to equity or similar vehicles designed to give value directly or indirectly to a Person).
- (d) Sales of Assets. Dispose of the Company's or any of its Subsidiary's assets to any Person in one transaction or series of related transactions (except in the Ordinary Course and except pursuant to a Qualified Sale transaction strictly in accordance with the Offer thereto without amendment or variation).
- (e) Restricted Payments. The Company shall not, and shall not permit any of its Subsidiaries to directly or indirectly, declare, order, pay, make or set apart any sum for any Restricted Payment other than:
  - (i) repayment of the Obligations;
  - (ii) repayment of the Senior Secured Debt;
  - (iii) Restricted Payments made to officers and employees of the Company and its Subsidiaries pursuant to employee stock option plans or employee stock incentive arrangements approved by the Holder in writing and to the extent that such repurchase is in connection with the cessation of the applicable recipient's employment by the Company;
  - (iv) Restricted Payments made to officers and employees of the Company and its Subsidiaries with respect to expenses incurred by such officers and employees which expenses are (A) ordinary and necessary business expenses incurred in the Ordinary Course and (B) reimbursable by the Company and its Subsidiaries ("Reimbursable Expenses"), as applicable, and in no event exceed the sum of \$25,000 per calendar month in the aggregate, outstanding at any one time; and

- (v) Restricted Payments consisting of salary and benefits payable to employees in the Ordinary Course.
- (f) *Investments, Notes and Advances to Employees.* The Company shall not, and shall not permit any of its Subsidiaries to directly or indirectly, make or own any Investment in any Person, except:
  - (i) Investments contemplated by the Transaction Documents;
  - (ii) Investments existing on the Closing Date and identified on Schedule E;
  - (iii) cash and cash equivalents;
  - (iv) Investments constituting (A) accounts receivable arising, (B) trade debt granted, or (C) deposits made in connection with the purchase price of goods or services, in each case in the Ordinary Course, (D) the purchase of secured or unsecured loans from merchants in the Ordinary Course.
- (g) Limitation on Modifications of Indebtedness; Modifications of Organizational Documents. Other than as permitted by the Transaction Documents, the Company shall not, and shall not permit any of its Subsidiaries to, consent to any amendment, supplement, waiver or other modification of, or enter into any forbearance from exercising any rights with respect to the terms or provisions contained in:
  - (i) any Indebtedness; or
  - (ii) the Organizational Documents of the Company or any of its Subsidiaries;
  - (iii) any Material Contract of the Company or any of its Subsidiaries,

without the prior written consent of the Holder which consent shall be in the Holder's absolute and unfettered discretion.

- (h) *Fiscal Year End.* The Company shall not and shall not permit any of its Subsidiaries to change its respective fiscal year.
- (i) *Business*. The Company will not, and will not permit any of its Subsidiaries to, engage in any business activity except the Business and reasonable extensions thereof, and activities reasonably incidental thereto.
- (j) Prepayments of Other Indebtedness. Except as permitted under the Transaction Documents, the Company shall not, and shall not permit any of its Subsidiaries to:
  - make any payment or prepayment of principal of, or premium or interest on any Indebtedness (other than the Senior Secured Debt), other than the stated, scheduled date for payment of Principal Amount and interest set forth in the applicable documents;
  - (ii) redeem, retire, purchase, defease or otherwise acquire any Indebtedness; or

- (iii) make any deposit (including the payment of amounts into a sinking fund or other similar fund) for any of the foregoing purposes.
- (k) Consolidation, Merger; etc. The Company shall not, and shall not permit any of its Subsidiaries to wind up, liquidate or dissolve, or consolidate with, or merge into or with, any other Person, or purchase or otherwise acquire all or substantially all of the assets of any Person (or any division thereof), other than in respect of a Qualified Sale.
- (I) Non-Circumvention. The Company shall not, and shall cause its Subsidiaries to not, by amendment of its Organizational Documents or through any reorganization, transfer of assets, consolidation, merger, scheme of arrangement, dissolution, issue or sale of securities, or any other voluntary action, avoid or seek to avoid the observance or performance of any of the terms of this Note, the Transaction Documents (or any other agreement and instrument entered into by it in connection with this transaction) and will at all times in good faith carry out all of the provisions of this Note, and the Transaction Documents (and any other agreement or instrument entered into in by it connection with this transaction).
- (m) Senior Management Shareholders. The Company shall not hire or terminate the employment of any Senior Management Shareholder or related Principal without the consent of the Holder.

#### ARTICLE 7 EVENT OF DEFAULT

#### Section 7.1 Event of Default

This Note shall be in default upon the occurrence of the following (each of which being an "Event of Default"):

- (a) the Company's (A) failure to cure a Conversion Failure by delivery of the required number of Shares within three (3) Business Days after the Conversion Date, or (B) notice, written or oral, to the Holder, including by way of public announcement or through any of its agents, at any time, of its intention not to comply with a request for Conversion of this Note into Shares that is tendered in accordance with the provisions of this Note; or
- (b) the Company's failure to pay to the Holder any of the Principal Amount, Interest, or other amounts when and as due under this Note, or any failure by the Company, or any Subsidiary to pay any amounts due under any other Transaction Document; or
- (c) if the Company uses any portion of the Loan for any reason other than as specifically contemplated by Section 1.2 hereof; or
- (d) if any representation or warranty made in this Note or any other Transaction Documents is false or erroneous in any material respect when made or in the event of any material default or material breach of any representation or warranty or covenant made in this Note at any time hereafter; or
- (e) the Company or any of its Subsidiaries shall (A) default in the due performance or observance by it of any term, covenant or agreement contained herein, or any

other Transaction Document (other than those specifically set forth elsewhere in this Article 7) and such default shall continue unremedied for a period of 10 days after the date on which written notice thereof is given to the Company by the Holder; or

- (f) any of the Transaction Documents shall cease to be in full force or effect, or the Company or any of its Subsidiaries shall deny its obligations thereunder; or
- (g) (A) the Company or any of its Subsidiaries shall (x) default in any payment of any Indebtedness (other than the Obligations) including the Senior Secured Debt, beyond the period of grace, if any, provided in an instrument or agreement under which such Indebtedness was created or (y) default in the observance or performance of any agreement or condition relating to any Indebtedness (other than the Obligations), including the Senior Secured Debt Documents or contained in any instrument or agreement evidencing, securing or relating thereto, or any other event shall occur or condition exist, the effect of which default or other event or condition is to cause any such Indebtedness to become due prior to its stated maturity, or (B) any Indebtedness (other than the Obligations) of the Company or any of its Subsidiaries shall be declared to be (or shall become) due and payable, or required to be prepaid other than by a regularly scheduled required prepayment, prior to the stated maturity thereof; or
- (h) if the Company or any Subsidiary or Shareholder denies, to any material extent, its obligations under this Note or other Transaction Documents to which it or he is a party or claims this Note or any other Transaction Document to be invalid or withdrawn in whole or in part; or
- (i) if the Company or any of its Subsidiaries taken as a whole is unable to pay debts as such debts become due, or is, or is adjudged or declared to be, or admits to being, bankrupt or insolvent; or
- (j) if the Company or any of its Subsidiaries ceases or threatens to cease to carry on its business, makes or agrees to make as assignment or bulk sale of its assets without prior written consent of the Holder, or makes a proposal, compromise or arrangement with or to its creditors; or
- (k) if the Company or any of its Subsidiaries makes a general assignment for the benefit of creditors; or any proceeding or filing is instituted or made by the Company or any of its Subsidiaries seeking relief on its behalf as debtor, or to adjudicate it a bankrupt or insolvent, or seeking liquidation, winding-up, reorganization, arrangement, adjustment or composition of it or its debts under any law relating to bankruptcy, insolvency, reorganization or relief of debtors, or seeking appointment of a receiver, trustee, custodian or other similar official for it or for any substantial part of its properties or assets; or the Company or any of its Subsidiaries takes any corporate action to authorize any of the actions set forth in this subparagraph (k); or
- (I) if any notice of intention is filed or any proceeding or filing is instituted or made against the Company or any of its Subsidiaries in any jurisdiction seeking to have an order for relief entered against it as debtor or to adjudicate it a bankrupt or insolvent, or seeking liquidation, winding-up, reorganization, arrangement, adjustment or composition of it or its debts under any law relating to bankruptcy, insolvency, reorganization or relief of debtors, or seeking appointment of a

receiver, trustee, custodian or other similar official for it or for all or any substantial part of its properties or assets or seeking possession, foreclosure or retention, or sale or other disposition of, or other proceedings to enforce security over, all or a substantial part of the assets of the Company or any of its Subsidiaries unless the same is being contested actively and diligently in good faith by appropriate and timely proceedings and is dismissed, vacated or stayed within 30 days of institution thereof; or

- (m) if a receiver, liquidator, trustee, sequestrator or other officer with like powers is appointed with respect to, or an encumbrancer pursuant to a security interest or otherwise takes possession of, or forecloses or retains, or sells or otherwise disposes of, or otherwise proceeds to enforce security over any of the properties or assets of the Company or any of its Subsidiaries or gives notice of its intention to do so; or
- (n) if any writ, attachment, execution, sequestration, distress or any other similar process becomes enforceable against the Company or if a distress or any analogous process is levied against any of the properties or assets of the Company, except where the same is being contested actively and diligently in good faith by appropriate and timely proceedings and the enforcement or levy has been stayed; or
- (o) if the Company sells or otherwise disposes of, or threatens to sell or otherwise dispose of, all or a substantial part of its undertaking and property and assets whether in one transaction or a series of related transactions (other than as permitted hereunder); or
- (p) if any provision of this Note or the other Transaction Documents will for any reason cease to be valid, binding and enforceable in accordance with its terms; or
- (q) if the Company assigns its rights or obligations under this Note or the other Transaction Documents or any interest herein to a third party without the prior written consent of the Holder; or
- (r) if one or more final judgments for the payment of money (whether or not covered by insurance) will be rendered against the Company and the Company fails to discharge or appeal the same within thirty (30) days from the date of notice of entry thereof; or
- (s) if at any time there occurs an event or circumstance which, in the opinion of the Holder, acting reasonably and in good faith, which has a Material Adverse Effect.

#### Section 7.2 Rights Upon Event of Default

(c) Upon the occurrence of an Event of Default, the Holder may, in the exercise of its sole discretion and at its sole option, (i) declare the Principal Amount and any accrued Interest in respect of the Note and all Obligations owing to be immediately due and payable to the Holder without further notice, demand or other formality, whereupon the same shall become, forthwith due and payable, (ii) exercise its rights to convert the Note pursuant to Section 7.2(b) hereof and/or (iii) pursue such other rights and remedies Holder may have pursuant to the Transaction Documents or at law.

(d) Upon the occurrence of an Event of Default, the Holder shall have the right at its sole option to convert the unpaid Principal Amount into that number of fully paid, validly issued Shares which would result in the Holder holding in the aggregate 80% of the Issued Common Shares of the Company.

# Section 7.3 Remedies, Characterizations, Other Obligations, Breaches and Injunctive Relief

The remedies provided in this Note shall be cumulative and in addition to all other remedies available to the Holder under this Note, any Transaction Documents or other agreements and instruments entered into in connection with this transaction, at law or in equity (including a decree of specific performance and/or other injunctive relief) and nothing herein shall limit the Holder's right to pursue actual and consequential damages for any failure by the Company to comply with the terms of this Note. Amounts set forth or provided for herein with respect to payments, conversion and the like (and the computation thereof) shall be the amounts to be received by the Holder and shall not, except as expressly provided herein, be subject to any other obligation of the Company (or the performance thereof). The Company acknowledges that a breach by it of its obligations hereunder will cause irreparable harm to the Holder and that the remedy at law for any such breach may be inadequate. The Company therefore agrees that, in the event of any such breach or threatened breach, the Holder shall be entitled, in addition to all other available remedies, to an injunction restraining any breach, without the necessity of showing economic loss and without any bond or other security being required.

## ARTICLE 8 MISCELLANEOUS

#### Section 8.1 Expenses

The Company shall reimburse the Holder for any and all reasonable fees, costs, charges and expenses incurred by it in connection with the negotiation, preparation, execution, delivery and performance of this Note, the other Transaction Documents, any other agreement, instrument or document executed pursuant thereto and the transactions hereby contemplated.

#### Section 8.2 Construction; Headings

This Note shall be deemed to be jointly drafted by the Company and the Holder and shall not be construed against any Person as the drafter hereof. The headings of this Note are for convenience of reference and shall not form part of, or affect the interpretation of, this Note.

# Section 8.3 Waiver

The failure or delay by the Holder in exercising any right or privilege with respect to the noncompliance with any provisions of this Note or any other Transaction Document by the Company, the Shareholders or any Subsidiary of the Company and any course of action on the part of the Holder, shall not operate as a waiver of any rights of the Holder unless made in writing by the Holder. Any such waiver shall be effective only in the specific instance and for the purpose for which it is given and shall not constitute a waiver of any other rights and remedies of the Holder with respect to any other or future non-compliance.

# Section 8.4 Assignment

This Note and any Shares issued upon conversion of this Note may be offered, sold, assigned or transferred by the Holder without the consent of the Company, subject only to the provisions of the Organizational Documents of the Company and the Shareholders Agreement (other than

those provisions waived pursuant to the Senior Management Shareholder Waiver and Agreement). This Note may not be assigned by the Company without the prior written consent of the Holder which consent shall be in the Holders sole and absolute discretion.

#### Section 8.5 Dispute Resolution

In the case of a dispute as to the determination of the arithmetic calculation of the Conversion Rate, or Shares to be held by the Company on Conversion, the Company shall submit the disputed determinations or arithmetic calculations within one Business Day of receipt, or deemed receipt, of the Conversion Notice or other event giving rise to such dispute, as the case may be, to the Holder. If the Holder and the Company are unable to agree upon such determination or calculation within one Business Day of such disputed determination or arithmetic calculation being submitted to the Holder, then the Company shall, within one Business Day submit the disputed arithmetic calculation of the Conversion Rate to an independent, reputable accounting firm selected by the Holder and approved by the Company. The Company, at their sole expense, shall cause the accounting firm to perform the determinations or calculations and notify the Company and the Holder of the results no later than two Business Days from the time it receives the disputed determinations or calculations. Such accountant's determination or calculation, as the case may be, shall be binding upon all parties absent demonstrable error.

#### Section 8.6 Notices

Without prejudice to any other method of giving notice, all communications provided for or permitted hereunder shall be in writing and delivered to the addressee by prepaid private courier or sent by pdf to the applicable address and to the attention of the officer of the addressee as follows:

Servus Credit Union Ltd.

151 Karl Clark Road, Edmonton, Alberta T6N 1H5

(a) if to the Holder, at its address at:

Servus Credit Union Ltd. 151 Karl Clark Road Edmonton, AB T6N 1H5 Attention: Jay Hamblin

Email: jay.hamblin@servus.ca

With a copy to:

Miller Thomson LLP 2700 Commerce Place 10155 - 102 Street Edmonton, AB T5J 4G8 Attention: Rick Reeson

Email: rreeson@millerthomson.com

22

(b) if to the Company, at its address at:

900 - 4445 Lougheed Highway Burnaby, BC V5C 0E4 Attention: Karl Sigerist

Fax: (604) 357-1095 Email: ksigerist@crelogix.com

With a copy to:

Blake, Cassels & Graydon LLP 595 Burrard Street Vancouver, BC V7X 1L3 Attention: Joan C. Chambers

Fax: 604-631-3309 Email: joan.chambers@blakes.com

Any communication transmitted by prepaid private courier shall be deemed to have been validly and effectively given or delivered at the time of delivery to the recipient. Any communication transmitted by pdf shall be deemed to have been validly and effectively given or delivered on the day on which it is transmitted, if transmitted on a Business Day on or before 5:00 p.m. (local time of the intended recipient), and otherwise on the next following Business Day, with evidence of a successful transmission. Any party may change its address for service by notice given in the foregoing manner.

## Section 8.7 Further Assurances

The Company shall, at its expense, promptly execute and deliver or cause to be executed and delivered to the Holder upon request, acting reasonably, from time to time all such other and further documents, agreements, opinions, certificates and instruments in compliance with this Note and the other Transaction Documents, or if necessary to more fully record or evidence the obligations intended to be entered into herein, or to make any recording, file any notice or obtain any consent.

#### Section 8.8 Cancellation

After all Principal Amount, accrued Interest, and other amounts at any time owed on this Note have been paid in full in cash or otherwise satisfied in full through any other means of which Holder may have availed itself, this Note shall automatically be deemed canceled, shall be surrendered to the Company for cancellation and shall not be reissued.

#### Section 8.9 Survival of Certain Obligations Despite Cancellation

The termination of this Note shall not relieve the Company from its obligations to the Holder arising prior to such cancellation. For greater certainty, the obligations of the Company set forth in Section 3.4 shall survive any termination or cancellation of this Note and fulfilment of the Obligations.

# Section 8.10 Waiver of Notice

To the extent permitted by law, the Company hereby waives demand, notice, protest and all other demands and notices in connection with the delivery, acceptance, performance, default or enforcement of this Note.

# Section 8.11 Taxes

Any and all payments hereunder or with respect to this Note shall be made free and clear of and without deduction for any and all present or future Taxes, excluding taxes imposed or based on the Holder's overall net income, franchise or capital taxes imposed on it in lieu of net income taxes, any estate, inheritance, gift or personal property tax, any branch profits taxes, and any taxes similar to the foregoing, and any taxes other than Other Taxes, but only to the extent any of such foregoing Taxes are imposed as a result of a connection between the Holder and the relevant jurisdiction or political subdivision other than a connection resulting from entering into, receiving payment under, or exercising rights or performing obligations under this Note. If the Company shall be required by law to deduct any Taxes from or in respect of any sum payable under this Note to the Holder. (i) the sum payable shall be increased as may be necessary so that after making all required deductions the Holder receives an amount equal to the sum it would have received had no such deductions been made, (ii) the Company shall make such deductions and (iii) the Company shall remit the full amount deducted to the relevant Governmental Authority in accordance with applicable law. Within thirty (30) days after the date of any payment of Taxes or, if later, as soon as practicable following the Company' receipt of the same, the Company shall furnish to the Holder the original or certified copy of a receipt evidencing payment thereof. In addition, the Company shall pay any present or future stamp or documentary taxes or any other excise or property taxes, charges or similar levies which arise from any payment made hereunder or from the execution, delivery or registration of, performance under, or otherwise with respect to, this Note ("Other Taxes").

# Section 8.12 Confidentiality

The Company agrees not to file a copy of this Note or any other Transaction Document in any public manner, or otherwise publicly disclose any information contained therein, except (i) on a confidential basis to its officers, directors, employees, accountants, lawyers and other professional advisors; (ii) to any bona fide existing or prospective investor or purchaser of the Shares of the Company or all or substantially all of the assets of any of the Company, provided that such Person agrees in writing with the Holder to maintain the confidentiality of such information in accordance with the provisions of this section; and (iii) as may be required pursuant to applicable law. If any such disclosure is required pursuant to applicable law, the Company will provide at least seven (7) days' prior written notice to the Holder before making such disclosure and during such period the Holder acting reasonably may advise the Company as to which portions of such Transaction Documents shall be redacted in order to protect the rights of the Holder to maintain the confidentiality of information which the Holder believes is confidential and proprietary to it. The Company agrees to comply with any such request unless such compliance would contravene applicable law. The terms of this paragraph shall survive the cancellation hereof.

# Section 8.13 Execution by Fax and Counterparts

This Note may be executed in several counterparts, each of which, when so executed, shall be deemed to be an original and which counterparts together shall constitute one and the same agreement. This Note may be executed by facsimile or pdf, and any signature contained hereon by facsimile or pdf shall be deemed to be equivalent to an original signature for all purposes.

#### Section 8.14 Governing Law;

All questions concerning the construction, validity, enforcement and interpretation of this Note shall be governed by the internal laws of the Province of Alberta and the federal laws applicable therein.

# Section 8.15 Time of the Essence

Time shall be of the essence of this Note.

#### Section 8.16 Entire Agreement; Waivers and Amendments to be in Writing

This Note and the other Transaction Documents supersede all discussion papers, term sheets and other writings which may have been issued by the Holder prior to the date hereof relating to the Loan, which shall have no force or effect; and this Agreement and any other documents or instruments contemplated herein or therein shall constitute the entire agreement and understanding among the Holder and the Company relating to the subject-matter hereof. No provision of this Note, or any other Transaction Document in existence among the parties may be modified, waived or terminated except by an instrument in writing executed by the party against whom such modification waiver or termination is sought to be enforced.

#### Section 8.17 Rules of Construction

The words "herein", "hereof" and "hereunder" and other words of similar import refer to this Note as a whole and not to any particular Section or subsection. Reference herein to any Section or subsection refers to such Section or subsection (as the case may be) hereof. Words in the singular include the plural, and words in the plural include the singular, and pronouns stated in the masculine, feminine or neuter gender shall include the masculine, feminine and neuter genders. The words "including", "includes" and "include" shall be deemed to be followed by the words "without limitation", whether or not so followed. Each covenant or agreement contained herein shall be construed (absent express provision to the contrary) as being independent of each other covenant or agreement contained herein, so that compliance with any one covenant or agreement shall not (absent such an express contrary provision) be deemed to excuse compliance with any other covenant or agreement. Where any provision herein refers to action to be taken by any Person, or which such Person is prohibited from taking, such provision shall be applicable whether such action is taken directly or indirectly by such Person. Unless stated otherwise, all references to any instruments or agreements, including references to any of the Transaction Documents, shall include any and all modifications or amendments thereto and any and all extensions or renewals thereof, in each case, made in accordance with the terms of the Transaction Documents including this Note. All references to Persons include their respective successors and assigns (to the extent permitted under the applicable Transaction Documents). All references to Statutes and related regulations shall include any amendments of the same and any successor Statutes and regulations. Whenever any provision in any Transaction Document refers to the knowledge (or an analogous phrase) of the Company or any of its Subsidiaries, such words are intended to signify that the Company or any of its Subsidiaries has actual knowledge or awareness of a particular fact or circumstance or that the Company or any of its Subsidiaries, if it had exercised reasonable diligence, would have known or been aware of such fact or circumstance.

#### Section 8.18 Confidentiality.

By acceptance of this Note, the Holder hereby acknowledges that all records, material and information pertaining to the Company and any copies thereof obtained by the Holder are and

shall remain the exclusive property of the Company. For so long as the Company carries on business, the Holder shall keep in the strictest confidence, not disclose and not use, without the consent of the Company, all non-public information pertaining to or concerning the Company. Notwithstanding the foregoing, a Holder may disclose such information to its officers, employees, managers, agents and representatives (collectively "**Representatives**") who have a need to know such information in connection with the oversight of the Holder's investment in the Company and only for that purpose, <u>provided</u> that the Holder will advise its Representatives of the proprietary nature of the information and of the obligations set forth herein, shall require such Representatives to be bound by written confidentiality restrictions no less stringent than those contained herein, and shall assume full liability for acts or omissions by its Representatives that are inconsistent with the obligations set forth herein. The foregoing restrictions shall not apply to information which:

- (a) was already in the public domain or comes into the public domain without any breach of this provision;
- (b) was known to the Holder prior to the date of this Note and not as a result of a breach of the Note;
- (c) is required to be disclosed by applicable law; or

is made to a professional or other adviser of the Holder, who has agreed to or is obligated hold it in confidence

#### Section 8.19 Severability

If any portion of this Note or the application thereof becomes or is declared to be illegal, void or unenforceable by a court of competent jurisdiction, the remainder of the terms of this Note will continue to be in full force and effect.

#### Section 8.20 Indemnification

In consideration of the Holder's execution and delivery of the Transaction Documents and entering into the transaction contemplated by this Note and in addition to all of the Company' other obligations under the Transaction Documents, the Company shall hold the Holder and its Affiliates harmless from any and all fees and expenses incurred by the Holder in connection with all of the transactions contemplated by the Transaction Documents, (including, without limitation, all reasonable expenses, and attorney's fees incurred in the preparation of the Transaction Documents, the negotiation of the same, and any filings in connection therewith, regardless of whether any disbursements are made under this Note) and shall defend, protect, indemnify and hold harmless the Holder, its Affiliates and all of its stockholders, partners, members, officers, directors and employees and direct or indirect investors and any of the foregoing Persons' agents or other representatives (including without limitation, those retained as counsel or other capacity in connection with the transaction contemplated by this Note and the Transaction Documents (collectively, the "Indemnitees") from and against any and all actions, causes of action, suits, claims, losses, costs, penalties, fees, liabilities and damages, and expenses in connection therewith (irrespective of whether any such Indemnitee is a party to the action for which indemnification hereunder is sought), and including reasonable attorneys' fees and disbursements (the "Indemnified Liabilities"), incurred by any Indemnitee as a result of, or arising out of, or relating to (a) any misrepresentation or breach of any representation or warranty made by the Company in the Transaction Documents or any other certificate, instrument or document contemplated thereby, (b) any breach of any covenant, agreement or obligation of the Company contained in the Transaction Documents or any other certificate,

instrument or document contemplated thereby or (c) any cause of action, suit or claim brought or made against such Indemnitee by a third party (including for these purposes a derivative action brought on behalf of the Company) and arising out of or resulting from (i) the execution, delivery, performance or enforcement of the Transaction Documents or any other certificate, instrument or document contemplated thereby, and (ii) any transaction financed or to be financed in whole or in part, directly or indirectly, with the proceeds of this Note. To the extent that the foregoing undertaking by the Company may be unenforceable for any reason, the Company shall make the maximum contribution to the payment and satisfaction of each of the Indemnified Liabilities which is permissible under applicable law. The agreements contained in this Section 8.20 shall survive the termination of all of the transactions contemplated by this Note and the cancellation of the same.

#### Section 8.21 Binding Effect

This Note shall be binding upon and shall enure to the benefit of the parties and their respective successors and permitted assigns; "successors" includes any corporation resulting from the amalgamation of any party with any other corporation.

[SIGNATURES ON NEXT PAGE]

IN WITNESS WHEREOF, the undersigned have caused this Note to be duly executed as of the date first above written.

CRELOGIX CREDIT GROUP INC.

Per:

Name: KARL SIGERIST

Title: DIRECTOR

Per:

Name:

Title:

I/We have the authority to bind the corporation

# SERVUS CREDIT UNION LTD.

Per:

Name: Title:

Per:

Name:

Title:

I/We have the authority to bind the corporation

[Signature page of unsecured convertible note agreement between Servus and Crelogix]

The undersigned Shareholders hereby acknowledge and agree to the provisions of Section 4.1, and confirm that the provisions of Section 4.1 supersede and replace the requirements under section 4.6 of the Sale Process Agreement for the Approving Shareholders (as defined in the Sale Process Agreement) to approve a Proposed Transaction (as defined in the Sale Process Agreement), and that a Qualified Sale under this Note shall be, and be deemed to be, an "Approved Transaction" under the Sale Process Agreement.

CEREBRAL CAPITAL INC. Per: Name: Karl Sigerist Title: Director I have the authority to bind the corporation NICHOLAS CARTER MICHAEL MILLS

MICHAEL MCKAY

# [SIGNATURES CONTINUED ON NEXT PAGE]

[Signature page of Shareholders of unsecured convertible note agreement between Servus and Crelogix]

The undersigned Shareholders hereby acknowledge and agree to the provisions of Section 4.1, and confirm that the provisions of Section 4.1 supersede and replace the requirements under section 4.6 of the Sale Process Agreement for the Approving Shareholders (as defined in the Sale Process Agreement) to approve a Proposed Transaction (as defined in the Sale Process Agreement), and that a Qualified Sale under this Note shall be, and be deemed to be, an "Approved Transaction" under the Sale Process Agreement.

#### CEREBRAL CAPITAL INC.

Per:

Name: Karl Sigerist Title: Director I have the authority to bind the corporation

NICHOLAS CARTER

Sepred with Alfreement Alfreement Same 15-Sep-16

MICHAEL MILLS

MICHAEL MCKAY

[SIGNATURES CONTINUED ON NEXT PAGE]

[Signature page of Shareholders of unsecured convertible note agreement between Servus and Crelogix]

The undersigned Subsidiary Guarantors hereby acknowledge and agree to the terms of the Note as of the date first written above:

Address:

**CRELOGIX ACCEPTANCE** CORPORATION AUS Per:

Name: KARL SIGERIST Title: DIRECTOR

Per:

Name:

Title:

I/We have the authority to bind the corporation

-

Address:

# **CRELOGIX PORTFOLIO SERVICES** CORPORATION

Per:	
	Name:
	Title:
Per:	the
	Name: KARL SIGERIST
	Title: DIRECTOR

I/We have the authority to bind the corporation

[Signature page of Guarantor acknowledgement of unsecured convertible note agreement between Servus and Crelogix]

# SCHEDULE A DEFINITIONS

"Affiliate" with respect to any Person, means (i) any person who, directly or indirectly, through one or more intermediaries, controls or is controlled by or is under common control with the specified Person, (ii) any person who is an officer of, partner in or trustee of, or serves in a similar capacity with respect to, the specified Person or of which the specified Person is an officer, partner or trustee, or with respect to which the specified Person serves in a similar capacity, (iii) any person who, directly or indirectly, is the beneficial owner of more than ten percent (10%) of any class of equity securities of, or otherwise has a substantial beneficial interest in, the specified Person or of which the specified Person is directly or indirectly the owner of more than ten percent (10%) of any class of equity securities or in which the specified Person is directly or indirectly the owner of more than ten percent (10%) of any class of equity securities or in which the specified Person is directly or indirectly the owner of more than ten percent (10%) of any class of equity securities or in which the specified Person has a substantial beneficial interest; or (iv) any spouse, ancestor, sibling, or lineal or collateral descendant of any of the foregoing, or any relationships described in (i) – (iii) above with respect to any of the foregoing.

"Business" means the consumer finance business of the Company and its Subsidiaries which is currently carried on by way of acceptance and factoring of dealer receivables and direct loans to consumers.

"Business Day" means any day other than Saturday, Sunday or statutory holiday in the Provinces of Alberta and British Columbia.

"Capital Stock" means and includes, with respect to any Person (a) any and all shares, securities, interests, participations or other equivalents of or interests in or over (however designated, including options and warrants) corporate stock or share capital, including preferred or preference stock or shares of such Person, (b) all partnership interests (whether general or limited) in such Person which is a partnership, (c) all membership interests or limited liability company interests in such Person which is a limited liability company, (d) any interest or participation that confers on a Person the right to receive a share of the profits and/or losses of, or distributions of assets of such Person, and (e) all equity or ownership interests in such Person of any other type, and any and all warrants, rights or options to purchase any of the foregoing.

"Company" has the meaning set forth in the first paragraph of this Note.

"Conversion Date" is defined in Section 3.3(b).

"Conversion Failure" is defined in Section 3.3(d).

"Conversion Notice" is defined in Section 3.3(a).

"Conversion Threshold" is defined in Section 3.1(b).

"Convertible Securities" means any shares or securities (other than Options, but including for the avoidance of doubt, warrants) directly or indirectly convertible into or exercisable or exchangeable for Shares.

"Credit Facility Letters" has the meaning ascribed thereto in the Forbearance Agreement.

"disposition" (or similar words such as "dispose") means any sale, transfer, lease, contribution or other conveyance (including by way of merger) of, or the granting of options, warrants or

other rights to, the Company's or its Subsidiaries' assets (including, without limitation, accounts receivable and Capital Stock of Subsidiaries and any Capital Stock or other securities of, or equity interests in, another Person) to any other Person in a single transaction or series of transactions, other than those occurring in the Ordinary Course of carrying-on the Business.

"Event of Default" has the meaning set forth in Section 7.1.

"Forbearance Agreement" means the forbearance agreement between Crelogix Acceptance Corporation, Crelogix Credit Group Inc., Crelogix Portfolio Services Corporation, Karl Sigerist, Mike McKay and Michael Mills dated September 14, 2016, as the same may be amended, restated, or replaced from time to time.

"GAAP" means generally accepted accounting principles in Canada as in effect from time to time.

"Governmental Authority" means any nation or government, any state, province or other political subdivision thereof, any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative functions of or pertaining to government, any securities exchange and any self-regulatory organization.

"Holder" has the meaning set forth in the first paragraph of this Note.

"Indebtedness" of any Person means, without duplication (i) all indebtedness of such Person for borrowed money, (ii) all reimbursement or payment obligations of such Person with respect to letters of credit, surety bonds and other similar instruments, (iii) all obligations of such Person evidenced by notes, bonds, debentures or similar instruments, (iv) all contingent obligations of such Person in respect of indebtedness or obligations of others of the kinds referred to in clauses (i) through (iii) above. Notwithstanding the foregoing, Indebtedness shall not include accrued expenses and deferred tax and other credits incurred by any Person in accordance with customary practices and in the Ordinary Course of business of such Person.

"Indemnified Liabilities" is defined in Section 8.20.

"Indemnities" is defined in Section 8.20.

"Interest" has the meaning set forth in the first paragraph of this Note.

"Interest Rate" has the meaning set forth in the first paragraph of this Note.

"Investment" means, relative to the Company:

- note, advance or extension of credit made by the Company or its Subsidiaries to any other Person, including the purchase by the Company or its Subsidiaries of any bonds, notes, debentures or other debt securities of any other Person (but does not include the funding/securitization arrangements of the Business); and
- (ii) any Capital Stock held by such Person in any other Person.

"Issued Common Shares" means any class of common shares of the Company that are issued and outstanding.

"Lien" means, with respect to any property, any mortgage, deed of trust, charge, lien, pledge, encumbrance, claim, fixed or floating charge, collateral assignment, assignment by way of
security, hypothec or hypothecation, security interest, restriction, prior claim, regulatory right, trust, option, lease or other encumbrance of any kind whatsoever, any other type of preferential arrangement in respect of such property, in each of the foregoing cases whether voluntary or imposed by law, and any agreement to give any of the foregoing.

"Loan" is defined in Section 1.1.

"Material Adverse Effect" shall mean at any time a material adverse effect on the business, operations, assets, condition (financial or otherwise) or prospects of the Company and its Subsidiaries considered as a whole, or a material adverse effect on the ability of the Company, the Senior Management Shareholders or any Subsidiary of the Company to perform any of its obligations under any of the Transaction Documents to which it is a party or on the ability of Servus to enforce such obligations.

"Material Contracts" means: (a) the Senior Secured Debt Documents; (b) the Option Agreements (c) the Shareholder's Agreement (d) the Stock Option Plan (d) any contract to which are attached obligations on the part of the Company or any Subsidiary in excess of \$100,000 per annum; (e) any contract under which the Company or any Subsidiary would receive in the Ordinary Course an amount in excess of \$100,000 per annum or that otherwise has an economic value to the Company or any Subsidiary in excess of \$100,000; or (f) any Contract to which the Company or any Subsidiary is a party that, the breach, non-performance, cancellation or non-renewal of which would impair the ability of the Company or such Subsidiary as applicable to carry on business in the Ordinary Course or would have a Material Adverse Effect.

"Maturity Date" is defined in Section 1.3(a).

"Maximum Rate" is defined in Section 1.4(B).

"Note" has the meaning set forth in the first paragraph of this Note.

"Obligations" means, collectively, (a) the obligations of the Company to pay any and all of the unpaid Principal Amount and Interest on (including interest accruing after the filing of any petition in bankruptcy, or the commencement of any insolvency, reorganization or like proceeding, relating to the Company, whether or not a claim for post-filing or post-petition interest is allowed in such proceeding) this Note, including by prepayment, or otherwise, (b) the obligations of the Company and any of its Subsidiaries to pay any and all fees, expenses, costs, indemnities and other amounts, whether direct or indirect, absolute or contingent, due or to become due, or now existing or hereafter incurred, which may arise under, out of, or in connection with, this Note or the other Transaction Documents, and (c) the obligations of the Company and any of its Subsidiaries to pay, perform, discharge, observe and comply with any and all covenants, agreements and other obligations required to be performed, discharged, observed or complied with by it pursuant to this Note or the other Transaction Documents.

"Offer" is defined in Section 4.1.

"Option Agreements" means all agreements granting Options in the Capital Stock of the Company described in Schedule D attached hereto.

"Option Exercise" is defined in Section 3.4.

"**Options**" means any rights or options to subscribe for or purchase Capital Stock or Convertible Securities of the Company or any Capital Stock of any Subsidiary of the Company.

"Ordinary Course" means with respect to an action taken by the Company or any of its Subsidiaries that the action is consistent with the past practices of the Company or its Subsidiaries and is taken in the usual course of the normal operations of the Company or its Subsidiaries, as applicable.

"Organizational Documents" means, as to any Person, its certificate of incorporation or articles of association and memorandum of association, articles of association, by-laws, its partnership agreement, its certificate of formation and limited liability company agreement or operating agreement and/or other organizational or governing documents of such Person.

"Other Taxes" has the meaning set forth in Section 8.11.

"Permitted Liens" means Liens securing the Senior Secured Debt and any other Liens securing the Obligations and permitted by Servus in writing.

"Percentage Interest" means the percentage of each class of Shares of the Company held by the Holder at the time such percentage is calculated.

"Person" means an individual, a limited liability company, a partnership, a joint venture, a Company, a trust, an unincorporated organization, any other entity and a government or any department or agency thereof.

"Prepayment Conversion" is defined in Section 3.1(a).

"Prepayment Conversion Amount" is defined in Section 3.1(a).

"**Prepayment Notice**" means the written notice that the Company must give to the Holder stating the Company's intention to make a prepayment of Principal Amount on this Note on a date prior to the Maturity Date. The Prepayment Notice must state the amount of the Principal Amount to be prepaid and the exact date such prepayment will be made.

"Prepayment Notice Period" is defined in Section 1.5(a)(i).

"Prime Rate" means the floating annual rate of interest established and recorded by Servus as such from time to time as its reference rate for determining rates of interest it will charge for loans denominated in Canadian dollars and commonly called Servus' Prime Lending Rate, adjusted automatically upon any change by Servus. The Prime Rate as of the date hereof is 2.7% per annum

"**Principal**" means the individual employed by the Company as a senior manager of the Company who is either (i) a Senior Management Shareholder or (ii) an Affiliate of a Senior Management Shareholder.

"Principal Amount" has the meaning set forth in the first paragraph of this Note.

"Qualified Sale" means a sale of the Company (or of all or substantially all of the assets and undertaking of the Company) to another Person at arm's length on terms and substance satisfactory to the Company in its sole and unfettered discretion.

"Reimbursable Expenses" is defined in Section 6.2(e)(iv).

"Representative" is defined in Section 8.18.

"Restricted Payment" means (a) the declaration or payment of any dividend on, or the making of any payment or distribution on account of, or setting apart assets for a sinking or other analogous fund for the purchase, redemption, defeasance, retirement or other acquisition of, any class of Shares of the Company or any of its Subsidiaries or any warrants, options or other right or obligation to purchase or acquire any such Shares, whether now or hereafter outstanding; or (b) the making of any other distribution in respect of such Shares, in each case either directly or indirectly, whether in cash, property or obligations of the Company or any of its Subsidiaries or otherwise; (c) repayment of Shareholder and Affiliate debt (other than to the Holder); and (d) payments of management fees, advances, salaries, bonuses, incentive compensation or bonuses or other payment to any director, officer, Shareholder, Principal or employee;

"Sale Process Agreement" means the Sale Process Agreement dated February 12, 2016 among Cerebral Capital Inc., Mike McKay, Nicholas Carter, Mike Mills, the Holder and the Company.

"Senior Management Shareholder Waiver and Agreement" means the Waiver and Consent Agreement dated September 14, 2016 among the Senior Management Shareholders, the Holder and the Company.

"Senior Management Shareholders" means Cerebral Capital Inc., Michael McKay, Nicholas Carter, Michael Mills and Karl Sigerist.

"Senior Secured Debt" means any secured loans or indebtedness from Servus as lender to the Company or its Subsidiaries, including without limitation pursuant to the Credit Facilities Letters.

"Senior Secured Debt Documents" means all loan agreements evidencing the Senior Secured Debt, including the Credit Facilities Letters and the Forbearance Agreement, and all documents, instruments and assurances contemplated therein to be provided by the Company or any other obligor thereunder, including any security in respect thereof.

"Servus" has the meaning set forth in the first paragraph of this Note.

"Shares" means the Capital Stock of the Company, as constituted form time to time, issued to the Shareholders of the Company.

"Shareholders" means any Person that is a holder of Shares, but excludes the purchaser under a Qualified Sale.

"Shareholders Agreement means the amended and restated shareholder agreement among Cerebral Capital Inc., the Senior Management Shareholders, Servus and the Company dated as of May 14, 2012.

"Stock Option Plan" means the employee stock option plan established by the Company on March 31, 2008 as amended and restated on July 8, 2013 providing for the grant to employees, officers and directors of the Company of Options to purchase Shares in the capital of the Company representing up to 800,000 Common Shares in the capital of the Company.

"Subsidiary" means, with respect to any Person, any company, Company or other entity of which more than 50% of the outstanding Shares with voting rights is at the time directly or indirectly owned or controlled by such Person or by one or more of any entities directly or indirectly owned or controlled by such Person. For the purposes of this definition, "control" of a Person means the possession, directly or indirectly, of the power to direct or cause the direction

of the management and policies of such Person (whether by ownership of Capital Stock, by contract or otherwise), and as of the date of this Note the "Subsidiaries" of the Company are Crelogix Acceptance Corporation and Crelogix Portfolio Services Corporation.

"Subsidiary Guarantees" is defined in Section 2.1(a)(i).

"Subsidiary Preferred Shares" means the Preferred shares of the Company registered in the name of Crelogix Acceptance Corporation.

"Taxes" means all income, stamp or other taxes, duties, levies, imposts, charges, assessments, fees, deductions or withholdings, now or hereafter imposed, levied, collected, withheld or assessed by any Governmental Authority, and all interest, penalties or similar liabilities with respect thereto.

"**Transaction Documents**" means this Note, the Subsidiary Guarantees and the Senior Management Shareholder Waiver and Agreement.

"Voluntary Conversion" is defined in Section 3.1(b).

"Voluntary Conversion Amount" is defined in Section 3.1(b).

"Voluntary Conversion Periods" means (i) any time following receipt and acceptance of a binding offer in respect of a Qualified Sale, until the consummation of the Qualified Sale and (ii) any time during the period commencing on the 15<sup>th</sup> day preceding the Maturity Date until the Loan has been indefeasibly paid in full.

# This is **Exhibit "H"** referred to in the Affidavit of Darcy Peelar, sworn before me this 5<sup>th</sup> day of July, 2017

A Commissioner for Oaths in and for Alberta

Patrick T. Harnett Barrister and Solicitor

## CONVERTIBLE NOTE AGREEMENT AMENDING AGREEMENT NO. 10

THIS AGREEMENT made effective the  $28^{\text{tr}}$  day of June, 2017.

**BETWEEN:** 

#### SERVUS CREDIT UNION LTD.

OF THE FIRST PART

- and -

#### CRELOGIX CREDIT GROUP INC.

OF THE SECOND PART

#### WHEREAS:

- A. The Company and the Holder are parties to an Unsecured Convertible Note Agreement dated September 14, 2016, as amended by Convertible Note Agreement Amending Agreement No.1 made effective the 15<sup>th</sup> day of December, 2016, as amended by Convertible Note Agreement Amending Agreement No.2 made effective the 22<sup>nd</sup> day of December, 2016, as amended by Convertible Note Agreement Amending Agreement No. 3 made effective the 10<sup>th</sup> day of January, 2017, as amended by Convertible Note Agreement Amending Agreement No. 4 made effective the 30<sup>th</sup> day of January, 2017, as amended by Convertible Note Agreement No. 5 made effective the 28<sup>th</sup> day of February, 2017, as amended by Convertible Note Amending Agreement No. 6 made effective the 17<sup>th</sup> day of March, 2017, as amended by Convertible Note Amending Agreement No. 7 made effective the 21<sup>st</sup> day of April, 2017, as amended by Convertible Note Amending Agreement No. 8 made effective the 16<sup>th</sup> day of May, 2017, together, the "Convertible Note Agreement"), whereby the Holder agreed, among other things, to extend the Loan to the Company;
- B. The Borrower and the Guarantors and the Bank are desirous of further amending the Convertible Note Agreement and have agreed to enter into this Amending Agreement (the "Amending Agreement");

NOW THEREFORE, in consideration of the mutual promises, covenants and agreements contained in the Convertible Note Agreement and herein contained, the receipt and sufficiency of which are hereby acknowledged by each of the parties, the parties hereto agree each as follows:

- 1. Except where specifically otherwise defined in this Amending Agreement, each of the capitalized terms used in this Amending Agreement, shall have the same meaning as set out in the Convertible Note Agreement.
- 2. Paragraph 1.3(a) of the Convertible Note Agreement shall be amended to change the date from "June 2, 2017" to "July 7, 2017".
- 3. Paragraph 1.3(b) of the Convertible Note Agreement is deleted in its entirety and replaced with the following:
  - "(b) At any time on or before July 7, 2017, the Holder shall be entitled, but not obligated, to extend the then current Maturity Date for a period of up to a further 90 days by written notice of extension delivered by the Holder to the Company on or before July 7, 2017, which notice shall set forth the new extended Maturity Date, and thereafter all references to the Maturity Date in any Transaction Document shall be to the Maturity Date as so extended."
- 4. In all other respects, the terms and conditions of the Convertible Note Agreement shall remain unamended and in full force and effect.
- 5. This Amending Agreement may be executed in counterparts and such counterparts together shall be deemed to be an original and shall constitute a single instrument. Notwithstanding the date of execution, such counterparts shall be deemed to bear a date as of the date of this Amending Agreement. Delivery of an executed counterpart of this Amending Agreement by electronic means, including, without limitation, by facsimile transmission or by electronic delivery in portable document format (".pdf") or tagged image file format (".tif"), shall be equally effective as delivery of a manually executed counterpart hereof. Any party delivering an executed counterpart of this Amending Agreement by electronic means shall also deliver a manually executed counterpart hereof by mail or courier upon demand.

[Signature Pages Follow]

24758230.2

- 2 -

- 3 -IN WITNESS WHEREOF the undersigned have caused this Amending Agreement to be executed through their authorized representatives, all as of the date and year first above written.

## **CRELOGIX CREDIT GROUP INC.**

Per: Name:

Karl 2 colden Title:

Per:

Mekay Name: Mille Title: VP I/We have the authority to bind the corporation

#### SERVUS CREDIT UNION LTD.

Per:

Name: IAN GLASSFORD Title: CFO

Per:

Name: Title: Darcy Peelar I/We have this authority formaliance and Operational Support Officer

### [SIGNATURES CONTINUED ON NEXT PAGE]

- 4 -**CRELOGIX ACCEPTANCE** CORPORATION Per: Name: Karl Sig en Title: Presigler Per: mike MCKay Name: VP Title: I/We have the authority to bind the corporation

## CRELOGIX PORTFOLIO SERVICES CORPORATION

Per:

Name: Title: MCKay

Per:

Narfie: Mille VP Title:

i/We have the authority to bind the corporation

à.

## [SIGNATURES CONTINUED ON NEXT PAGE]

- 5 -

CEREBRAL CAPITAL INC.

Per: ورلا

Name: Karl Sigerest Title: Director I have the authority to blnd the corporation

. . I

Witness

Witness

Pro

Witness

NICHOLAS CARTER

MIKE MOKAY

é

MICHAEL MILLS

24758230;2

## This is **Exhibit "I"** referred to in the Affidavit of Darcy Peelar, sworn before me this 5<sup>th</sup> day of July, 2017

A Commissioner for Oaths in and for Alberta

Patrick T. Harnett Barrister and Solicitor

#### GUARANTEE AND POSTPONEMENT

SERVUS Credit Union Ltd. (the "Credit Union")

#### TO: THE CREDIT UNION

## In consideration of the Credit Union dealing with \_TRAVELERS ACCEPTANCE CORPORATION

(hereinafter referred to as the "Customer"), the undersigned and each of them, if more than one, hereby jointly and severally guarantee payment to the Credit Union of all present and future debts and liabilities (direct or indirect, absolute or contingent, matured or otherwise), now or at any time and from time to time hereafter due or owing to the Credit Union by the Customer, whether incurred by the Customer alone or jointly with any corporation, person or persons, or otherwise howsoever, including all costs (and in respect of legal costs on a solicitor and his own client basis) and disbursements incurred by the Credit Union in recovering or attempting to recover the said debts and liabilities from the Customer including the enforcement of any security held by the Credit Union therefor. Provided, however, that the liability of the undersigned, and of each of the undersigned herein is limited to <u>-TWO MILLION DOLLARS--</u> Dollars (2,000,000.00) plus (i) interest thereon (such interest to be calculated and compounded monthly). or on so much thereof as

Dollars (2,000,000.00 plus (i) interest thereon (such interest to be calculated and compounded monthly), or on so much thereof as may be outstanding from time to time, from the date of demand of payment until paid, at the rate(a) agreed upon, between the Credit Union and the Customer, and (ii) legal costs incurred by the Credit Union in enforcing and obtaining payment under this guarantee including the enforcement of any security held by the Credit Union therefore on a solicitor and his own olient basis (the "Legal Costs").

2. In this guarantee, the word "Guarantor" shall mean the undersigned and if there is more than one, it shall mean each of them.

3. This guarantee shall not be affected by the death or loss or diminution of capacity of the Customer or of the Guarantor or by any change in the name of the Customer or in the membership of the firm of the Customer through the death or retirement of one or more partners or the introduction of one or more other partners or otherwise, or by the acquisition of the business of the Customer by a corporation, firm or person, or by any change whatsoever in the objects, capital structure or constitution of the Customer, or by the Customer or the business of the Customer or comparison, firm or person, is if such event has not happened. The Guarantor agrees to monitor changes in the financial position of the Customer and hereby releases the Credit Union from any liability resulting therefrom.

4. All monies, advances, renewals and credits borrowed or obtained by the Customer from the Credit Union shall be deemed to form part of the debts and liabilities, notwithstanding any lack or limitation of status or of power, incapacity or disability of the Customer or of the directors, partners or agents thereof, or that the Customer may not be a legal or suable entity, or any irregularity, defect or informality in the borrowing or obtaining or such monies, advances, renewals or credits, whether known to the Credit Union or not; and any sum which may not be recoverable from the Guarantor on the basis of a guarantee shall be recoverable from the Guarantor as sole and principal debtor in respect thereof and shall be paid to the Credit Union on demand with interest and accessories as herein provided.

5. This guarantee shall continue and be enforceable notwithstanding any amalgamation of the Credit Union with any other Credit Union(s), financial institution(s) or other corporation(s), and any further amalgamation, in which event this guarantee shall also extent to all debts and liabilities then or thereafter owed by the Customer to the amalgamated Credit Union. Furthermore, all security, real or personal, moveable or immoveable, which has been or will be given by the Guarantor of the said debts and liabilities shall be valid in the hands of the Credit Union, as well as its successors and assigns.

6. It is further agreed that this shall be a continuing guarantee, and shall cover and secure any ultimate balance owing to the Credit Union.

7. This guarantee shall bind the Guarantor and Guarantor's, executors, administrators, legal representatives, successors and assigns until termination thereof by notice in writing to the manager of the branch of the Credit Union at which the account of the Customer is kept, but such termination by any of the Guarantors or their respective heirs, executors, administrators, legal representatives, successors or assigns shall not prevent the continuance of the liability hereunder of any other Guarantor. Such termination shall apply only to those debts or liabilities of the Customer incurred or arising after ninety days after receipt of the notice by the Credit Union, but not in respect of any prior debts or liabilities, matured or not. The notice of termination shall have no effect on those debts or liabilities incurred more than ninety days after receipt of said notice, which result or will result from express or implied commitments made prior to the ninetieth day after receipt of the notice.

8. This guarantee will not be diminished or modified on account of any act on the part of the Credit Union which would prevent subrogation from operating in favour of the Guarantor. It is further agreed that the Credit Union, without exonerating in whole or in part the Guarantor, may grant time, renewals, extensions, indulgences, release and discharges to, may take security from, and give up or release any or part of the security held, may abstain from taking, perfecting, registering or renewing security or from realizing on security, may accept compositions and otherwise deal with the Customer and with any other person or persons, including any of the Guarantors, and dispose of any security held by the Credit Union as it may see fit, and that all dividends and monies received by the Credit Union from the Customer or from any other person, capable of being applied by the Credit Union in reduction of the debts and liabilities hereby guaranteed, shall be considered for all purposes as payment in gross which the Credit Union shall have the right to apply as it may see fit, not being bound by the law of imputation, and the Credit Union shall be entitled to prove against the estate of the Customer upon any insolvency or winding up, in respect of the whole said debts and liabilities. The Guarantor shall have no right to be subrogated to the Credit Union until the Credit Union shall have received payment in full of its claims against the Customer with interest and costs.

9. If any circumstances arise necessitating the Credit Union to file its claim against the estate of the Customer and to value its security, it will be entitled to place such valuation as the Credit Union may in its discretion see fit, and the filing of such claim and the valuation of its security shall in no way prejudice or restrict its rights against the Guarantor.

10. The Credit Union shall not be obligated to exhaust its recourse against the Customer or other persons or security it may hold before being entitled to payment from the Guarantor of each and every of the debts and liabilities hereby guaranteed and it shall not be obliged to offer or deliver its security before its whole claim has been paid. The Guarantor waives all benefits of discussion and division.

11. All indebtedness and liability, present and future, of the Customer to the Guarantor are hereby assigned to the Credit Union and postponed to the present and future debts and liabilities of the Customer to the Credit Union. All monies received from the Customer or on the Customer's behalf by the Guarantor shall be held as agent, mandatary and trustee for the Credit Union and shall be paid over to the Credit Union forthwith. This provision will remain in full force and effect, notwithstanding the termination of this guarantee pursuant to the provisions of paragraph 7 in which event it will terminate when the debts and liabilities of the Customer to the Credit Union covered by this guarantee pursuant to paragraph 7 hereof have been paid in full.

12. This guarantee is in addition to and not in substitution for any other guarantee, by whomsoever given, at any time held by the Credit Union, and without prejudice to any other security by whomsoever given held at any time by the Credit Union and the Credit Union shall be under no obligation to marshall in favour of the Guarantor any such security or any of the funds or assets the Credit Union may be entitled to receive or have a claim upon.

13. The Guarantor shall be bound by any account settled between the Credit Union and the Customer and, if no such account has been so settled, any account stated by the Credit Union shall be accepted by the Guarantor as conclusive evidence of the amount which at the date of the account so stated is due by the Customer to the Credit Union.

14. The Guarantor shall make payment to the Credit Union of the Guarantor's liability hereunder forthwith after demand therefor is made in writing. Such demand shall be deemed to have been effectually made when an envelope containing it addressed to the Guarantor's last address known to the Credit Union is deposited postage prepaid in the Post Office. The liability of the Guarantor shall bear interest from the date of such demand at the rate or rates then applicable to the debts and liabilities of the Customer to the Credit Union.

15. For better certainty, and in addition to any limitation of the liability of the Guarantor hereunder, the Guarantor upon the demand of the Credit Union shall pay to the Credit Union all Legal Costs.

16. This guarantee and agreement shall be operative and binding upon every signatory hereto and notwithstanding the non-execution thereof by any other proposed signatory or signatories, and possession of the instrument by the Credit Union shall be conclusive evidence against the Guarantor that this instrument was not delivered in escrow or pursuant to any agreement that it should not be effective until any condition has been complied with. None of the parties shall be bound by any representation or promise made by any person relative thereto which is not embodied herein. The liability of the Guarantor hereunder begins on the date of the Guarantor's signature on this Guarantee and Postponement.

17. This guarantee shall be binding upon the undersigned and each of them, and if more than one, then jointly and severally between them and also upon the heirs, executors, administrators, successors and assigns of the Guarantor and will extend to and enure to the benefit of the successors and assigns of the Credit Union. Each and every provision hereof is severable and should any provision hereof be illegal or not enforceable for any reason whatsoever, such illegality or invalidity shall not affect the other provisions hereof which shall remain in force and be binding on the parties hereto.

18. The Guarantor acknowledges having read and taken cognizance of this Guarantee and Postponement before signing it and declares that the Guarantor understands perfectly the terms, conditions and undertakings contained therein.

19. This Guarantee and Postponement shall be construed in accordance with the laws of the Province of Alberta and the Guarantor agrees that any legal suit, action or proceeding arising out of or relating to this Guarantee and Postponement may be instituted in the courts of such province, and the Guarantor hereby accepts and irrevocably submits to the jurisdiction of the said courts, and acknowledges their competence and agrees to be bound by any judgement thereof, provided that nothing herein shall limit the Credit Union's right to bring proceedings against the Guarantor elsewhere.

20. The Guarantor acknowledges and agrees that this guarantee is in addition to and not in substitution for any other guarantee(s) that any of the undersigned (alone or jointly with any other person) may have granted to the Credit Union in respect of the Customer's obligations and liability of the Customer to the Credit Union or to the Credit Union and any other person(s). This guarantee shall remain in full force and effect until all debts and obligations hereby secured have been irrevocably and indefeasibly paid and released.

KANKAKK this	30TH	day of OCTOBER	2009	
AVAILUTATE? ITTO	00111			

SIGNED, SEALED AND DELIVERED

WITNESS	SIGNATURE OF GUARANTOR(S)
	CRELOGIX CREDIT GROUP INC. PER: [Seal(s)]
	KARL SIGERIST

Note: Certificate under Guarantees Acknowledgement Act (Alberta) is required in respect of any individual Guarantor.

DM\_CGY/255194-00018/33275.3

#### GUARANTEE AND POSTPONEMENT

SERVUS Credit Union Ltd. (the "Credit Union")

#### TO: THE CREDIT UNION

## In consideration of the Credit Union dealing with TRAVELERS ACCEPTANCE CORPORATION

(hereinafter referred to as the "Customer"), the undersigned and each of them, if more than one, hereby jointly and severally guarantee payment to the Credit Union of all present and future debts and liabilities (direct or indirect, absolute or contingent, matured or otherwise), now or at any time and from time to time hereafter due or owing to the Credit Union by the Customer, whether incurred by the Customer alone or jointly with any corporation, person or persons, or otherwise howsoever, including all costs (and in respect of legal costs on a solicitor and his own client basis) and disbursements incurred by the Credit Union in recovering or attempting to recover the said debts and liabilities from the Customer including the enforcement of any security held by the Credit Union therefor. Provided, however, that the liability of the undersigned, and of each of the undersigned herein is limited to \_\_\_\_\_WOM MILLION DOLLARS--

Dollars (2,000,000.00) plus (i) interest thereon (such interest to be calculated and compounded monthly), or on so much thereof as may be outstanding from time to time, from the date of demand of payment until paid, at the rate(s) agreed upon, between the Credit Union and the Customer, and (ii) legal costs incurred by the Credit Union in enforcing and obtaining payment under this guarantee including the enforcement of any security held by the Credit Union therefore on a solicitor and his own client basis (the "Legal Costs").

2. In this guarantee, the word "Guarantor" shall mean the undersigned and if there is more than one, it shall mean each of them.

3. This guarantee shall not be affected by the death or loss or diminution of capacity of the Customer or of the Guarantor or by any change in the name of the Customer or in the membership of the firm of the Customer through the death or retirement of one or more partners or the introduction of one or more other partners or otherwise, or by the acquisition of the business of the Customer by a corporation, firm or person, or by any change whatsoever in the objects, capital structure or constitution of the Customer, or by the Customer or the business of the Customer by a corporation, firm or person, or by any change whatsoever in the objects, capital structure or constitution of the Customer, or by the Customer or the business of the Customer or the business of the Customer and apply to the full extent as if such event has not happened. The Guarantor agrees to monitor changes in the financial position of the Customer and hereby releases the Credit Union from any liability resulting therefrom.

4. All monies, advances, renewals and credits borrowed or obtained by the Customer from the Credit Union shall be deemed to form part of the debts and liabilities, notwithstanding any lack or limitation of status or of power, incapacity or disability of the Customer or of the directors, partners or agents thereof, or that the Customer may not be a legal or suable entity, or any irregularity, defect or informality in the borrowing or obtaining or such monies, advances, renewals or credits, whether known to the Credit Union or not; and any sum which may not be recoverable from the Guarantor on the basis of a guarantee shall be recoverable from the Guarantor as sole and principal debtor in respect thereof and shall be paid to the Credit Union on demand with interest and accessories as herein provided.

5. This guarantee shall continue and be enforceable notwithstanding any amalgamation of the Credit Union with any other Credit Union(s), financial institution(s) or other corporation(s), and any further amalgamation, in which event this guarantee shall also extent to all debts and liabilities then or thereafter owed by the Customer to the amalgamated Credit Union. Furthermore, all security, real or personal, moveable or immoveable, which has been or will be given by the Guarantor of the said debts and liabilities shall be valid in the hands of the Credit Union, as well as its successors and assigns.

6. It is further agreed that this shall be a continuing guarantee, and shall cover and secure any ultimate balance owing to the Credit Union.

7. This guarantee shall bind the Guarantor and Guarantor's, executors, administrators, legal representatives, successors and assigns until termination thereof by notice in writing to the manager of the branch of the Credit Union at which the account of the Customer is kept, but such termination by any of the Guarantors or their respective heirs, executors, administrators, legal representatives, successors or assigns shall not prevent the continuance of the liability hereunder of any other Guarantor. Such termination shall apply only to those debts or liabilities of the Customer incurred or arising after ninety days after receipt of the notice by the Credit Union, but not in respect of any prior debts or liabilities, matured or not. The notice of termination shall have no effect on those debts or liabilities incurred more than ninety days after receipt of said notice, which result or will result from express or implied commitments made prior to the ninetieth day after receipt of the notice.

8. This guarantee will not be diminished or modified on account of any act on the part of the Credit Union which would prevent subrogation from operating in favour of the Guarantor. It is further agreed that the Credit Union, without exonerating in whole or in part the Guarantor, may grant time, renewals, extensions, indulgences, release and discharges to, may take security from, and give up or release any or part of the security held, may abstain from taking, perfecting, registering or renewing security or from realizing on security, may accept compositions and otherwise deal with the Customer and with any other person or persons, including any of the Guarantors, and dispose of any security held by the Credit Union as it may see fit, and that all dividends and monies received by the Credit Union from the Customer or from any other person, capable of being applied by the Credit Union in reduction of the debts and liabilities hereby guaranteed, shall be considered for all purposes as payment in gross which the Credit Union shall have the right to apply as it may see fit, not being bound by the law of imputation, and the Credit Union shall have no right to be subrogated to the Credit Union until the Credit Union shall have received payment in full of its claims against the Customer with interest and costs.

9. If any circumstances arise necessitating the Credit Union to file its claim against the estate of the Customer and to value its security, it will be entitled to place such valuation as the Credit Union may in its discretion see fit, and the filing of such claim and the valuation of its security shall in no way prejudice or restrict its rights against the Guarantor.

10. The Credit Union shall not be obligated to exhaust its recourse against the Customer or other persons or security it may hold before being entitled to payment from the Guarantor of each and every of the debts and liabilities hereby guaranteed and it shall not be obliged to offer or deliver its security before its whole claim has been paid. The Guarantor waives all benefits of discussion and division.

the r

11. All indebtedness and liability, present and future, of the Customer to the Guarantor are hereby assigned to the Credit Union and postponed to the present and future debts and liabilities of the Customer to the Credit Union. All monies received from the Customer or on the Customer's behalf by the Guarantor shall be held as agent, mandatary and trustee for the Credit Union and shall be paid over to the Credit Union forthwith. This provision will remain in full force and effect, notwithstanding the termination of this guarantee pursuant to the provisions of paragraph 7 in which event it will terminate when the debts and liabilities of the Customer to the Credit Union covered by this guarantee pursuant to paragraph 7 hereof have been paid in full.

12. This guarantee is in addition to and not in substitution for any other guarantee, by whomsoever given, at any time held by the Credit Union, and without prejudice to any other security by whomsoever given held at any time by the Credit Union and the Credit Union shall be under no obligation to marshall in favour of the Guarantor any such security or any of the funds or assets the Credit Union may be entitled to receive or have a claim upon.

13. The Guarantor shall be bound by any account settled between the Credit Union and the Customer and, if no such account has been so settled, any account stated by the Credit Union shall be accepted by the Guarantor as conclusive evidence of the amount which at the date of the account so stated is due by the Customer to the Credit Union.

14. The Guarantor shall make payment to the Credit Union of the Guarantor's liability hereunder forthwith after demand therefor is made in writing. Such demand shall be deemed to have been effectually made when an envelope containing it addressed to the Guarantor's last address known to the Credit Union is deposited postage prepaid in the Post Office. The liability of the Guarantor shall bear interest from the date of such demand at the rate or rates then applicable to the debts and liabilities of the Customer to the Credit Union.

15. For better certainty, and in addition to any limitation of the liability of the Guarantor hereunder, the Guarantor upon the demand of the Credit Union shall pay to the Credit Union all Legal Costs.

16. This guarantee and agreement shall be operative and binding upon every signatory hereto and notwithstanding the non-execution thereof by any other proposed signatory or signatories, and possession of the instrument by the Credit Union shall be conclusive evidence against the Guarantor that this instrument was not delivered in escrow or pursuant to any agreement that it should not be effective until any condition has been complied with. None of the parties shall be bound by any representation or promise made by any person relative thereto which is not embodied herein. The liability of the Guarantor hereunder begins on the date of the Guarantor's signature on this Guarantee and Postponement.

17. This guarantee shall be binding upon the undersigned and each of them, and if more than one, then jointly and severally between them and also upon the heirs, executors, administrators, successors and assigns of the Guarantor and will extend to and enure to the benefit of the successors and assigns of the Credit Union. Each and every provision hereof is severable and should any provision hereof be illegal or not enforceable for any reason whatsoever, such illegality or invalidity shall not affect the other provisions hereof which shall remain in force and be binding on the parties hereto.

18. The Guarantor acknowledges having read and taken cognizance of this Guarantee and Postponement before signing it and declares that the Guarantor understands perfectly the terms, conditions and undertakings contained therein.

19. This Guarantee and Postponement shall be construed in accordance with the laws of the Province of Alberta and the Guarantor agrees that any legal suit, action or proceeding arising out of or relating to this Guarantee and Postponement may be instituted in the courts of such province, and the Guarantor hereby accepts and irrevocably submits to the jurisdiction of the said courts, and acknowledges their competence and agrees to be bound by any judgement thereof, provided that nothing herein shall limit the Credit Union's right to bring proceedings against the Guarantor elsewhere.

20. The Guarantor acknowledges and agrees that this guarantee is in addition to and not in substitution for any other guarantee(s) that any of the undersigned (alone or jointly with any other person) may have granted to the Credit Union in respect of the Customer's obligations and liability of the Customer to the Credit Union or to the Credit Union and any other person(s). This guarantee shall remain in full force and effect until all debts and obligations hereby secured have been irrevocably and indefeasibly paid and released.

AS WITNESS the hands and seals of the Guarantor(s), at VANCOUVER, BRITISH COLUMBIA

Allerta this 3	OTH day of	OCTOBER .	2009

SIGNED, SEALED AND DELIVERED

WITNESS	SIGNATURE OF GUARANTOR(S)
	CRELOGIX PORTFOLIO SERVICES CORPORATION
	PER: [Seal(s)]
	An
	KARL SIGERIST

Note: Certificate under Guarantees Acknowledgement Act (Alberta) is required in respect of any individual Guarantor.

DM\_CGY/255194-00018/33275.3

#### GUARANTEE AND POSTPONEMENT

Servus Credit Union 1.td. (the "Credit Union")

#### TO: THE CREDIT UNION

į

1. In consideration of the Credit Union dealing with <u>Crelogix Acceptance Corporation</u> (hereinafter referred to as the "Customer"), the undersigned and each of them, if more than one, hereby jointly and severally guarantee payment to the Credit Union of all present act future debt and liabilities differed in the set action of the customer alone or jointly with any corporation, person or persons, or otherwise howsover, including all costs (and in respect of legal costs on a solicitor and his own client basis) and disbursements incurred by the Credit Union in recovering or attempting to recover the said debt and liabilities of the undersigned herein is limited to <u>Five Hundred & Thirty Thouseand Four Hundred Dollars</u> Dollars (<u>\$530, 400, 00</u>) plus (i) interest thereon (such interest to be calculated and compounded monthly), or on so much thereof as may be outstanding from time to line, from the date of demand of payment unit paid, in the rate(s) agreed upon, between the Credit Union and the Customer, and (ii) legal costs incurred by the Credit Union in enforcing and obtaining payment under this guarantee including the Credit Union and the Customer, and (ii) legal costs incurred by the Credit Union in enforcing and obtaining payment under this guarantee including the enforcement of any security held by the Credit Union in enforcing and obtaining payment under this guarantee including the enforcement of any security held by the Credit Union therefore on a solicitor and his own client basis (the "Legal Costs").

#### 2. In this guarantee, the word "Guaranter" shall mean the undersigned and if there is more than one, it shall mean each of them.

3. This guarantee shall not be affected by the death or loss or diminution of capacity of the Customer or of the Guaranter or by any change in the Austomer or in the membership of the firm of the Customer through the death or retirement of one or more partners or the latroduction of one or more other partners or otherwise, or by the acquisition of the business of the Customer to the outperform or by any change whatsoever in the objects, capital structure or constitution of the Customer, or by any change whatsoever in the objects, capital structure or constitution of the Customer, or by the Customer or the bulness of the Customer being amalgamated with a firm or corporation but shall, norwithstanding the happening of any such event, continue to exist and apply to the full extent as if such event has not happened. The Guarantor agrees to monitor changes in the financial position of the Customer and hereby releases the Credit Ualon from any liability resulting thereform.

4. All monies, advances, renewals and credits borrowed or obtained by the Customer from the Credit Union shall be deemed to form part of the debts and liabilities, notwithstanding any lack or limitation of status or of power, incapacity or disability of the Customer or of the directors, partners or agents thereof, or that the Customer may not be a legal or subjle online, or any irregularity, dofect or informality in the borrowing or obtaining or such monies, advances, renewals or credits, whether known to the Credit Union or not; and any sum which may not be recoverable from the Guaranter on the basis of a guarantee shall be recoverable from the Guaranter as sole and principal debtor in respect thereof and shall be paid to the Credit Union on demand with interest and accessories as herein provided.

5. This guarantee shall continue and be enforceable notwithstanding any amalgamation of the Credit Union with any other Credit Union(s), financial institution(s) or other corporation(s), and any further amalgamation, in which event this guarantee shall also extent to all debts and liabilities then or thereafter owed by the Customer to the amalgamated Credit Union. Furthermore, all security, real or personal, moveable or immoveable, which has been or will be given by the Guaranter of the said debts and liabilities shall be valid in the hands of the Credit Union, as well as its accessors and assigns.

6. It is further agreed that this shall be a continuing guarantee, and shall cover and scoure any ultimate balance owing to the Credit Union.

7. This guarantee shall bind the Guarantor and Guarantor's, executors, administrators, legal representatives, successors and assigns until fermination thereof by notice in writing to the manager of the branch of the Credit Union at which the neceunt of the Customer is kept, but such termination by may of the Guarantors or their respective heirs, executors, administrators, legal representatives, successors or assigns shall not prevent the continuance of the liability hereunder of any other Guarantor. Such termination shall upply only to those debts or liabilities of the Customer incurred or not. The notice of termination shall have no effect on those debts or liabilities incurred more than ninety days after receipt of said notice, which result or will result from express or implied commitments made prior to the ninetieth day after receipt of the notice.

5. This guarantee will not be diminished or modified on account of any act on the part of the Credit Union which would prevent subrogation from operating in favour of the Guanantor. It is further agreed dust the Credit Union, without exonerating in whole or in part the Guanantor, may grant line, renewals, extensions, indulgences, release and discolarges to, may take security from, and give up or release any or part of the security hold, may absiain from taking, perfecting, registering or recewing security or from realizing on sourity, may accept compositions and otherwise deal with the Customer and with any other person or persons, including any of the Guarantors, and dispose of any security beld by the Credit Union from the Customer or form any other person, capable of being applied by the Credit Union shall have the right to apply as it may see fit, and being sound by its law of imputation, and its all dividends and monies received by the Credit Union from the Oustomer or form any other person, capable of being applied by the Credit Union shall have the right to apply as it may see fit, not being bound by its law of imputation, and discolar dividences are part or the cost or work or working up, in respect of the whole said debts and liabilities. The Guarantor shall have no right to be subrogated to the Credit Union shall have received payment in foll of its claims against the Customer or with interest and costs.

9. If any circumstances arise necessitating the Credit Union to file its claim against the estate of the Customer and to value its security, it will be cutited to place such valuation as the Credit Union may in its discretion see fit, and the filing of such claim and the valuation of its security shall in no way prejudice or testrict its rights against the Guarantor.

10. The Credit Union shall not be obligated to exhaust its recourse against the Customer or other persons or security it may hold before being entitled to payment from the Guarantor of each and every of the debts and habilities hereby guaranteed and it shall not be obliged to offer or deliver its security before its whole claim has been paid. The Guarantor waives all benefits of discussion and division.

.. ...

Ł,

11. All indebtedness and liability, present and future, of the Customer to the Guaranter are hereby assigned to the Credit Union and postponed to the present and future debts and liabilities of the Customer to the Credit Union. All monies received from the Customer are on the Customer's behalf by the Guaranter shall be hald as agent, mandatary and trustee for the Credit Union and shall be paid over to the Credit Union forthwith. This provision will remain in full force and effect, notwithstanding the termination of this guarantee pursuant to the provisions of paragraph 7 in which event it will terminate when the debts and liabilities of the Customer to the Credit Union covered by this guarantee pursuant to paragraph 7 hereof have been paid in full.

12. This guarantee is in addition to and not in substitution for any other guarantee, by whomspever given, at any time held by the Credit Union, and without projudice to any other security by whomspever given held at any time by the Credit Union and the Credit Union shall be under no obligation to marshall in favour of the Guarantor any such security or any of the funds or assets the Credit Union may be entitled to receive or have a claim upon.

13. The Guarantor shall be bound by any account settled between the Credit Union and the Customer and, if no such account has been so settled, any account stated by the Credit Union shall be accepted by the Guarantor as conclusive evidence of the amount which at the date of the account so stated is due by the Customer to the Credit Union.

14. The Guarantor shall make payment to the Credit Unico of the Guarantor's liability hereunder forthwith after denmad therefor is made in writing. Such demand shall be deemed to have been effectually made when an envelope containing it addressed to the Guarantor's last address known to the Credit Union is deposited postage prepaid in the Post Office. The liability of the Guarantor shall bear interest from the date of such demand at the rate or rates then applicable to the debts and liabilities of the Customer to the Credit Union.

15. For better certainty, and in addition to any limitation of the liability of the Guarantor hereunder, the Guarantor upon the demand of the Credit Union shall pay to the Credit Union all Legal Costs.

16. This guarantee and agreement shall be operative and binding upon every signatory hereto and notwithstanding the non-execution thereof by any other proposed signatory or signatories, and possession of the instrument by the Credit Union shall be conclusive evidence against the Guarantor that this instrument was not delivered in escrow or pursuant to any agreement that it should not be effective until any condition has been complicat with. None of the parties shall be bound by any representation or promise made by any person relative thereto which is not embodied herein. The liability of the Guarantor thereauder begins on the date of the Guarantor's signature on this Guarantee and Postponentent.

17. This guarantee shall be binding upon the undersigned and each of them, and if more than one, then jointly and severally between them and also upon the heirs, executors, administrators, successors and assigns of the Guarantor and will extend to and enure to the benefit of the successors and assigns of the Credit Union. Each and every provision hereof is severable and should any provision hereof be illegal or not enforceable for any reason whatsoever, such filegality or invalidity shall not affect the other provisions hereof which shall remain in force and be binding on the parties hereto.

18. The Guarantor acknowledges having read and taken cognizance of this Guarantee and Postponement before signing it and deciares that the Guarantor understands perfectly the terms, conditions and undertakings contained therein.

19. This Guarantee and Postponentent shall be construed in accordance with the laws of the Province of Alberta and the Guarantee regress that any legal suit, action or proceeding arising out of or relating to this Guarantee and Postponentent may be instituted in the courts of such province, and the Guaranter hereby accepts and irrevocably submits to the jurisdiction of the said courts, and acknowledges their competence and agrees to be bound by any judgement thereof, provided that nothing herein shall limit the Credit Union's right to bring proceedings against the Guarantee elsewhere.

20. The Guarantor acknowledges and agrees that this guarantee is in addition to and not in substitution for any other guarantee(s) that any of the undersigned (alone or jointly with any ofter person) may have granted to the Credit Union in respect of the Customer's obligations and liability of the Customer to the Credit Union or to the Credit Union and any other person(s). This guarantee shall remain in full force and effect until all debts and obligations hereby secured have been irrevocably and indefeasibly paid and released.

British	Column Sia	rember	
	SIGNED, SEALED AND DELIVERED	SIGNATURE OF GUARANSORTS	
	Seter Alchan	Rarl Sigerist	[SERT([8)]]

Note: Certificate under Guarantees Acknowledgement Act (Alberta) is required in respect of any individual Guarantor,

DM\_CGY/155194-00018/33275.3

#### GUARANTEES ACKNOWLEDGEMENT ACT

#### CERTIFICATE OF NOTARY PUBLIC

#### I HEREBY CERTIFY THAT:

- 1. Karl Sigerist of \_\_\_\_\_\_, in the Province of <u>British Columbia</u>, the Guarantor (or one of the Guarantors) in the guarantee dated <u>10</u> day of <u>Declandee</u>, 2010, made between Karl Sigerist and the Credit Union named in the Guarantee and Postponement to which this certificate is attached or noted upon, appeared in person before me and acknowledged that he/she had executed the guarantee.
- 2. I satisfied myself by examination of him/her that he/she is aware of the contents of the guarantee and understands it.

.

GIVEN at	Butish Columbithis 10 day of December, 2010 under
FRANCIS CHANG Barrister & Solicitor BLAKE, CASSELS & GRAYDON LLP Suite 2600, Three Bentail Centre 595 Burrard St., P.O. Box 49314 Vancouver, B.C. V7X 1L3 (804) 691*9682	A Notary Public in and for the Province of XXXXXXXX British Columbia
s	TATEMENT OF GUARANTOR
I am the person named in this certif	ficate. Signature of Guaranter
	Karl Sigerest

NOTE: A separate Certificate must be completed for each individual Guarantor.

Microsoft Word > File > New > L-Commercial & Ag > Certiflaste of Notary Public 10585nd - (06/2000 JB)

#### GUARANTEE AND POSTPONEMENT

Servus Credit Union Ltd. (the "Credit Union")

#### TO: THE CREDIT UNION

#### Crelogix Acceptance Corporation

1

1. In consideration of the Credit Union dealing with <u>Crelcgix</u> ACCEPTARCE COTOTERCION (hereinafter referred to as the "Customer"), the undersigned and each of them, if more than one, hereby jointly and severally guarantee payment to the Credit Union of all present and future debts and liabilities (direct or indirect, absolute or contingent, matured or otherwise), now or at any time and from time to reme hereafter due or owing to the Credit Union by the Customer, whether incurred by the Customer alone or jointly with any corporation, parson or persons, or otherwise howsover, including all costs (and in respect of legal costs on a solicitor and his own olient basis) and disbursements incurred by the Credit Union in recovering or attempting to recover the said debts and liabilities from the Customer including the enforcement of any security held by the Credit Union therefor. Provided, howsever, that the liability of the undersigned, and of each of the undersigned herein is limited to <u>Two Hundred & Fifty Thousand Two Hundred</u> Dollars (<u>\$250,200,00</u>) plus (i) interest thereon (such interest to be calculated and compounded monthily), or en so much thereof as may be outstanding from time to time, from the date of demand of payment until paid, at the rate(s) agreed upon, between the Credit Union and the Customer, and (I) legal costs incurred by the Credit Union therefore on a solicitor and his own client has (a the "Legal Costs"). In consideration of the Credit Union dealing with \_

2. In this guarantee, the word "Guaranter" shall mean the undersigned and if there is more than one, it shall mean each of them.

3. This guarantee shall not be affected by the death or loss or diminution of capacity of the Customer or of the Guarantor or by any change in the name of the Customer or in the membership of the firm of the Customer through the death or retirement of one or more partners or the introduction of one or more other partners or otherwise, or by the acquisition of the business of the Customer by a corporation, firm or person, or by any change whatsoever in the objects, capital structure or constitution of the Customer, or by the Customer or the business of the Customer being amalgamated with a firm or corporation but shall, notwithstanding the happening of any such event, continue to exist and apply to the full extent as it such event has not happened. The Guarantor agrees to monitor changes in the financial position of the Customer and hereby releases the Credit Union from any liability resulting therefrom.

All monies, advances, renewals and credits borrowed or obtained by the Customer from the Credit Union shall be deemed to form part 4. All mones, advances, renewals and credits borrowed or obtained by the Customer from the Credit Union shall be deemed to form part of the directors, partners or agents thereof, or that the Customer ones not be a legal or suble entity, or any irregularity, defect or informality in the borrowing or obtaining or such monies, advances, renewals or credits, whether known to the Cetil Union or not and any sum which may not be a legal or suble entity, or any irregularity, defect or informality in the borrowing or obtaining or such monies, advances, renewals or credits, whether known to the Cetil Union or not and any sum which may not be recoverable from the Guarantor on the basis of a guarantee shall be recoverable from the Guarantor on the basis of a guarantee shall be recoverable from the Guarantor on the basis of a guarantee shall be recoverable from the Guarantor on the basis of a guarantee shall be recoverable from the Guarantor on the basis of a guarantee shall be recoverable from the Guarantor on the basis of a guarantee shall be recoverable from the Guarantee shall be paid to the Credit Union or not rand and shall be paid to the Credit Union on demand with interest and accessories as herein provided.

5. This guarantee shall continue and be enforceable notwithstanding any smalgamation of the Credit Union with any other Credit Union(s), financial institution(s) or other corporation(s), and any further analgamation, in which event this guarantee shall also extent to all debts and liabilities than or thereafter owed by the Customer to the smalgamated Credit Union. Furthermore, all security, real or personal, movesible or immoveable, which has been or will be given by the Guaranter of the said debts and liabilities shall be valid in the bands of the Credit Union, as well as its successors and assigns.

It is further agreed that this shall be a continuing guarantee, and shall cover and secure any ultimate balance owing to the Credit Union,

This guarantee shall bind the Guarantee and Guarantee's executors, administrators, legal representatives, successors and assigns until 7. In guarantee shall bed the Cutanator and Guarantor's, executors, administrators, legal representatives, successors and assigns until termination thereof by notice in writing to the manager of the branch of the Cutanators, legal representatives, successors or assigns until termination by any of the Quarantors or their respective heirs, executors, administrators, legal representatives, successors or assigns shall not prevent the continuance of the liability horeunder of any other Guarantor. Such termination shall apply only to those debts or liabilities of the Customer incurred or arising after ninety days after receipt of the notice by the Credit Union, but not in respect of any prior debts or liabilities, matured or not. The notice of formination shall have no effect on those debts or liabilities incurred more than ninety days after receipt of said notice, which result from express or implied commitments made prior to the ninetieth day after receipt of the notice.

8. This guntantee will not be diminished or modified on account of any set on the part of the Credit Union which would prevent subrogation from operating in favour of the Guarantor. It is further agreed that the Credit Union, without exonerating in whole or in part the Guarantor, may grant time, renewals, extensions, indulgences, release and discharges to, may take accurity from, and give up or release any or part of the security held, may absult from taking, perfecting, registering or renewing security or from realizing on security, may uccept compositions and otherwise deal with the Customer and with any other perior or percens, including any of the Guaranteed, and the Guaranteed end with any other perior or percens, including any of the Guaranteed, shall be considered for any security held by the Credit Union as it may see fit, and that all dividends and monies received by the Credit Union the Customer or from any other person, capable of being applied by the Credit Union in reduction of the debts and liabilities hereby guaranteed, shall be considered for all the Credit Union shall have the right to apply as it may see fit, not being bound by the law of imputation, and the Cadit Union shall be entitled to prove against the estate of the Customer upon any insolvency or winding up, in respect of the whole said debts and liabilities. The Guaranter shall have no right to be subrogated to the Credit Union shall have received pryment in full of its claims against the Customer with interest and costs. in full of its claims against the Customer with interest and costs.

9. If any circumstances arise necessitating the Credit Union to file its claim against the estate of the Customer and to value its security, it will be entitled to place such valuation as the Credit Union may in its discretion see fit, and the filing of such claim and the valuation of its security shall in no way prejudice or restrict its rights against the Guarantor.

The Credit Union shall not be obligated to exhaust its recourse against the Customer or other persons or security it may hold before being entitled to payment from the Guarantor of each and every of the debts and liabilities hereby guaranteed and it shall not be obliged to offer or deliver its security before its whole claim has been paid. The Guarantor waives all benefits of discussion and division.

11. All indebtedness and liability, present and future, of the Customer to the Guarantor are hereby assigned to the Credit Union and postponed to the present and future debts and liabilities of the Customer to the Credit Union. All monies received from the Customer or on the Customer's behalf by the Guarantor shall be held as agent, mandatary and trustee for the Credit Union and shall be paid over to the Credit Union forthwith. This provision will remain in full force and effect, notwithstanding the termination of this guarantee pursuant to the provisions of paragraph 7 in which event it will terminate when the debts and liabilities of the Customer to the Credit Union covered by this guarantee pursuant to paragraph 7 hereof have been paid in full.

12. This guarantee is in addition to and not in substitution for any other guarantee, by whomsoever given, at any time held by the Credit Union, and without prejudice to any other security by whomsoever given held at any time by the Credit Union and the Credit Union shall be under no obligation to marshall in favour of the Guarantor any such security or any of the funds or assets the Credit Union may be entitled to receive or have a claim upon.

13. The Guarantor shall be bound by any account settled between the Credit Union and the Customer and, if no such account has been so settled, any account stated by the Credit Union shall be accepted by the Guarantor as conclusive evidence of the amount which at the date of the account so stated is due by the Customer to the Credit Union.

14. The Guarantor shall make payment to the Credit Union of the Guarantor's liability hereunder forthwith after domand therefor is made in writing. Such demand shall be doemed to have been effectually made when an envelope containing it addressed to the Guarantor's last address known to the Credit Union is deposited postage prepaid in the Post Office. The liability of the Guarantor shall bear interest from the date of such demand at the rate or rates then applicable to the debts and liabilities of the Customer to the Credit Union.

15. For better certainty, and in addition to any limitation of the liability of the Guarantor hereunder, the Guarantor upon the demand of the Credit Union shall pay to the Credit Union all Legal Costs,

16. This guarantee and agreement shall be operative and binding upon every signatory hereto and norwithstanding the non-execution thereof by any other proposed signatory or signatories, and possession of the instrument by the Credit Union shall be conclusive evidence against the Guarantor that this instrument was not delivered in escrow or pursuant to any agreement that it should not be effective until any condition hese bound by any representation or promise made by any person relative thereto which is not embodied therein. The liability of the Guarantor hereundor begins on the date of the Guarantor's signature on this Guarantee and Postponement.

17. This guarantee shall be binding upon the undersigned and each of them, and if more than one, then jointly and severally between them and also upon the heirs, executors, administrators, successors and assigns of the Guarantee and will extend to and enure to the benefit of the successors and assigns of the Credit Union. Each and every provision hereof is severable and should any provision hereof be illegal or not enforceable for any reason whatsoever, such Illegality or invalidity shall not affect the other provisions hereof which shall remain in force and be binding on the paties hereto.

18. The Guarantor acknowledges having read and taken cognizance of this Guarantee and Postponement before signing it and declares that the Guarantor understands perfectly the terms, conditions and undertakings contained therein.

19. This Guarantee and Postponement shall be construed in accordance with the laws of the Province of Alberta and the Ouaranter agrees that any legal suit, action or proceeding arising out of or rolating to this Guarantee and Postponement may be instituted in the courts of such province, and the Guaranter hereby accepts and irrevacably submits to the jurisdiction of like sold courts, and acknowledges their competence and agrees to be bound by any judgement thereof, provided that nothing herein shall limit the Credit Union's right to bring proceedings against the Guarantee lisewhere.

20. The Guarantor acknowledges and agrees that this guarantee is in addition to and not in substitution for any other guarantee(3) their any of the undersigned (alone or jointly with any other person) may have granted to the Credit Union in respect of the Customer's obligations and liability of the Customer to the Credit Union or to the Credit Union and any other person(s). This guarantee shall remain in full force and effect until all debts and obligations hereby secured have been irrevocably and indefensibly paid and released.

Bit ish Colevation a 2010 Mey \_\_\_\_\_ this \_\_\_\_ ĺØ December day of

SIGNED, SEALED AND DELIVERED

WITNESS Alha

[Seal(s)] Michael Mills

SIGNATURE OF GUARANTOR(S

Note: Certificate under Guarantees Acknowledgement Act (Alberta) is required in respect of any individual Guarantor.

DM\_CCY/255194-90818/53276.3

## GUARANTEES ACKNOWLEDGEMENT ACT

-----

#### CERTIFICATE OF NOTARY PUBLIC

#### I HEREBY CERTIFY THAT:

- 1. Michael Mills of \_\_\_\_\_\_, in the Province of British Columbia \_\_\_\_\_, the Guarantor (or one of the Guarantors) in the guarantee dated 10 day of \_\_\_\_\_\_\_\_, 2010 made between Michael Mills and the Credit Union named in the Guarantee and Postponement to which this certificate is attached or noted upon, appeared in person before me and acknowledged that he/she had executed the guarantee.
- 2. I satisfied myself by examination of him/her that he/she is aware of the contents of the guarantee and understands it.

GIVEN at Von Cover M my hand and seal of office. Brith Colubin, this 10 day of December, 2010 under FRANCIS CHANG Barrister & Solicitor BLAKE, CASSELS & GRAYDON LLP Sulte 2600, Three Bentail Centre 595 Burrard St., P.O. Box 49314 Vancouver, B.C. V7X 1L3 (804) 631-8332 A Notary Public in and for the Province of Alberta British Columbia

#### STATEMENT OF GUARANTOR

I am the person named in this certificate.

Signature of Guarantor Michael Mills

NOTE: A separate Certificate must be completed for each individual Guarantor,

Microsoft Word > File > New > L-Commercial & Ag > Certificate of Notary Public 10585nd - (06/2000 JB)

. . . . . . .

. .

#### GUARANTEE AND POSTPONEMENT

Servus

#### Credit Union Ltd. (the "Credit Union")

#### TO: THE CREDIT UNION

1. In consideration of the Credit Union dealing with <u>Crelogix</u> Acceptance Corporation (hereinafter referred to as the "Customer"), the undersigned and each of them, if more than one, hereby jointly and severally guarantee payment to the Credit Union of all present and future debts and liabilities (direct or indirec, absolute or contingent, matured or otherwise), now or at any time and from time to time, hereafter due or owing to the Credit Union by the Customer, whether incurred by the Customer and so no one of an operation, any corporation, person or persons, or otherwise howsoever, including all costs (and in respect of legal costs on a solicitor and his own olient basis) and disbursements incurred by the Credit Union in recovering or attempting to recover the said debts and liabilities from the Customer including the enforcement of any security held by the Credit Union therefor. Provided, however, that the liability of the undersigned, and of each of the undersigned herein is limited to <u>Two Hundred & Fifty Thousand Two Hundred</u> Dellers (<u>\$250, 200,00</u>) plus (f) interest thereo if (union in enforcing and obtaining payment under the Credit Union and the Customer, and (ii) legal costs incurred by the Credit Union in enforcing and obtaining payment under this guarantee including the enforcement of any security held by the Credit Union in enforcing and obtaining payment under this guarantee including the enforcement of any security held by the Credit Union therefor on a solicitor and his own client basis (the "Legal Costs").

2. In this guarantee, the word "Guarantor" shall mean the undersigned and if there is more than one, it shall mean each of them.

3. This guarantee shall not be affected by the death or loss or diminution of capacity of the Customer or of the Guarantor or by any change in the name of the Customer or in the membership of the firm of the Customer through the death or retirement of one or more partners or the introduction of one or more other partners or otherwise, or by the acquisition of the business of the Customer by a corporation, firm or person, or by any change whateoever in the objects, capital stucture or constitution of the Customer, or by the Customer or the business of the Customer business of the Customer business of the Customer to the business of the Customer to even the source to even the source to the business of the Customer or the business of the Customer busing analgumated with a firm or corporation but shall, notwithstanding the happening of any such event, continue to exist and apply to the full extent as if such event has not heppened. The Guarantor agrees to monitor changes in the financial position of the Customer and hereby releases the Customer in the libition group customer that the object. the Credit Union from any liability resulting therefrom.

All monies, advances, renewals and credits borrowed or obtained by the Customer from the Credit Union shall be deemed to form part 4. All montes, auvances, renewals and creats corrowed of optimized by the Customer from the Creat Union shall be deented to form part of the debts and likelifics, notwithstanding any lack or limitation of status or of power, incomparity or disability of the Customer or of the directors, partners or agents thereof, or that the Customer may not be a legal of suble entity, or any irregularity, defect or informality in the borrowing or obtaining or such monies, advances, renewals or ordelis, whether known to the Creatil Union or not; and any sum which may not be recoverable from the Guarantor on the basis of a guarantee shall be recoverable from the Guarantor on the basis of a guarantee shall be recoverable from the Guarantor on the basis of a guarantee shall be recoverable from the Guarantor on the basis of a guarantee shall be recoverable from the Guarantor on the basis of a guarantee shall be recoverable from the Guarantor on the basis of a guarantee shall be recoverable from the Guarantor as sole and principal debtor in respect thereof and shall be paid to the Credit Union on demand with interest and accessories as herein provided.

5. This guarantee shall continue and be enforceable netwithstanding any amalgamation of the Credit Union with any other Credit Union(s), financial institution(s) or other corporation(s), and any further amalgamation, in which event this guarantee shall also extent to all debts and liabilities then or thereafter owed by the Customer to the annalgamated Credit Union. Furthermore, all security, real or personal, moveable or immoveable, which has been or will be given by the Guaranter of the said debts and liabilities shall be valid in the hands of the Credit Union, as well as its successors and assigns.

It is further agreed that this shall be a continuing guarantee, and shall cover and secure any ultimate balance owing to the Credit Union

7. This guarantee shall bind the Guarantor and Guarantor's, executors, administrators, legal representatives, successors and assigns until termination thereof by notice in writing to the manager of the branch of the Credit Union at which the account of the Customer is kept, but such termination by any of the Guarantors or their respective heirs, executors, administrators, legal representatives, successors or assigns shall not prevent the continuance of the liability hereunder of any other Guarantor. Such termination shall apply only to those debts or liabilities, the Customer incurred or arising after ninety days after receipt of the notice by the Credit Union, but not in respect of any prior debts or liabilities, matured or not. The notice of termination shall have no effect on those debts or liabilities incurred more than anoty days after receipt of said to the such apply only the prior debt or liabilities. notice, which result or will result from express or implied commitments made prior to the ninetieth day after receipt of the notice.

8. This guarantee will not be diminished or modified on account of any act on the part of the Credit Uulon which would prevent subrogation from operating in favour of the Guarantor. It is further agreed that the Credit Uulon, without exonersting in whole or in part the Guarantor, may grant time, renewals, extensions, indulgences, release and discharges to, may take socurity from, and give up or release any or part of the security held, may abstain from taking, perfecting, registering or renewing security for from realizing on security, may sees any or part of the security held, may abstain from taking, perfecting, registering or renewing security or from realizing on security, may sees to empositions and otherwise deal with the Customer and with any other person or persons, including any of the Customer of from any other person, expable of being applied by the Credit Union in reduction of the dekis and liabilities bereby guaranteed, shall be ensidered for all purposes as payment in gross which the Credit Union shell have the right to apply as it may see fit, net being bound by the law of imputation, and debts and liabilities. The Guarantor shell have no right to be subrogated to the Credit Union shell have received payment in full of its claims against the Customer with interest and costs.

9. If any circumstances arise necessitating the Credit Union to file its claim against the estate of the Customer and to value its security, it will be entitled to place such valuation as the Credit Union may in its disoretion see fit, and the filing of such claim and the valuation of its security shall in no way prejudice or restrict its rights against the Guaranter.

10. The Credit Union shall not be obligated to exhaust its recourse against the Customer or other persons or security it may hold before being entitled to payment from the Guarantor of each and every of the debts and liabilities hereby guatanteed and it shall not be obliged to offer or deliver its security before its whole claim has been paid. The Guarantor waives all benefits of discussion and division.

11. All indebtedness and liability, present and future, of the Customer to the Guaranter are horeby assigned to the Credit Union and postponed to the present and future debts and liabilities of the Customer to the Credit Union. All monier received from the Customer or on the Customer's behalf by the Guaranter shall be held as agent, mandatary and trustee for the Credit Union and shall be paid over to the Credit Union forthwith. This provision will remain in full force and effect, notwithstanding the termination of this guarantee pursuant to the provisions of paragraph 7 in which event it will terminate when the dobts and liabilities of the Customer to the Credit Union covered by this guarantee pursuant to paragraph 7 hereof have been paid in full.

12. This guarantee is in addition to and not in substitution for any other guarantee, by whomsoaver given, at any time held by the Credit Union, and without prejudice to any other security by whomsoaver given held at any time by the Credit Union and the Credit Union shall be under no obligation to marshall in favour of the Guarantor any such accurity or any of the funds or assets the Credit Union may be cutilied to receive or have a claim upon.

13. The Guarantor shall be bound by any account settled between the Credit Union and the Customer and, if no such account has been so sottled, any account stated by the Credit Union shall be accepted by the Guarantor as conclusive avidence of the amount which at the date of the account so stated is due by the Customer to the Credit Union.

14. The Quarantor shall make payment to the Credit Union of the Quatantor's liability hereunder forthwith after demand therefor is made In writing. Such demand shall be deemed to have been effectually made when are envelope containing it addressed to the Quarantor's last address known to the Credit Union is deposited postage prepaid in the Post Office. The liability of the Quarantor shall bear interest from the date of such demand at the rate or rates then applicable to the debts and liabilities of the Customer to the Credit Union.

15. For botter certainty, and in addition to any limitation of the liability of the Ousrantor hereunder, the Guarantor upon the demand of the Credit Union shall pay to the Credit Union all Legal Costs.

16. This guarantee and agreement shall be operative and binding upon every signatory hereto and notwithstanding the non-execution thereof by any other proposed signatory or signatories, and possession of the instrument by the Credit Union shall be conclusive evidence against the Guarantor that this instrument was not delivered in escrow or pursuant to any agreement that it should not be effective until any condition has been compiled with. None of the parties shall be bound by any representation or promise made by any person relative thereto which is not embodied herein. The liability of the Guarantor hereunder begins on the date of the Guarantor's signature on this Guarantee and Postponement.

17. This guarantee shall be binding upon the undersigned and each of them, and if more than one, then joinily and severally between them and also upon the heirs, executors, administrators, successors and assigns of the Guarantor and will extend to and enure to the benefit of the successors and assigns of the Credit Union. Each and every provision hereof is severable and should any provision hereof be illegal or not enforceable for any reason whatsoever, such illegality or invalidity shall not affect the other provisions hereof which shall remain in force and be binding on the parties hereto.

18. The Guarantor acknowledges having read and taken cognizance of this Guarantee and Postponement before signing it and declares that the Guarantor understands perfectly the terms, conditions and undertakings contained therein.

19. This Guarantee and Postponement shall be construed in accordance with the laws of the Province of Alberta and the Guarantor agrees that any legal suit, action or proceeding arising out of or relating to this Guarantee and Postponement may be instituted in the courts of such province, and the Guarantor hereby accepts and irrevocably submits to the jurisdiction of the suid courts, and acknowledges their competence and agrees to be bound by any judgement thereof, provided that nothing herein shall limit the Credit Union's right to bring proceedings against the Guarantor elsewhere.

20. The Guarantor acknowledges and agrees that this guarantee is in addition to and not in substitution for any other guarantee(s) that any of the undersigned (alone or jointly with any other person) may have grauted to the Credit Union in respect of the Customer's obligations and liability of the Customer to the Credit Union or to the Credit Union and any other person(s). This guarantee shall remain in full force and effect until all debts and obligations hereby secured have been irrevocably and indefeasibly paid and released.

vitish	Columbia	SS the hands and seals of the Guara	untor(s), et		t	
$\sim$	_Alberia, this	day of	Ram	ber	2010	
m	SIGNED, SEALED AN	ID DELIVERED		ì.		
	WITNESS			SIGNATURE OF GUA	RANTOR(S)	
	Detal	All	~	MikaMeka	$\sim$	[Scal(s)]
				v ·	v	<b>L</b> e (e /)
			1			

Note: Certificate under Guarantees Acknowledgement Act (Alberta) is required in respect of any individual Guaranter.

DM\_C0Y/253194-00018/33275.3

b

#### GUARANTEES ACKNOWLEDGEMENT ACT

#### CERTIFICATE OF NOTARY PUBLIC

#### I HEREBY CERTIFY THAT:

- 1. Mike McKay of \_\_\_\_\_\_, in the Province of British Columbia , the Guarantor (or one of the Guarantors) in the guarantee dated 10 day of \_\_\_\_\_\_\_\_\_, 2010made between Mike McKay and the Credit Union named in the Guarantee and Postponement to which this certificate is attached or noted upon, appeared in person before me and acknowledged that he/she had executed the guarantee.
- 2. I satisfied myself by examination of him/her that he/she is aware of the contents of the guarantee and understands it.

GIVEN at Vancower, Brits Clambia, this 10 day of <u>December</u>, 2010 under my hand and seal of office.

FRANCIS CHANG Barrister & Solicitor BLAKE, CASSELS & GRAYDON LLP Suite 2600, Tinree Bentall Centre 595 Burrard St., P.O. Box 49914 Vancouver, B.C. V7X 1L9 (604) 691-6932 A Notary Public in and for the Province of Aberia British Columbia

#### STATEMENT OF GUARANTOR

I am the person named in this certificate.

Signature of Guarantor Mike McKay

NOTE: A separate Certificate must be completed for each individual Guarantor.

Microsoft Word > File > New > L-Commercial & Ag > Certificate of Notary Public 10585nd – (06/2000 JB)

#### GUARANTEE AND POSTFONEMENT

#### Servus Credit Union Ltd. (the "Credit Union")

#### TO: THE CREDIT UNION

- 1

1. In consideration of the Credit Union dealing with <u>Crelogix Acceptance Corporation</u> (hereinafter referred to as the "Customer"), the undersigned and each of them, if more than one, hereby jointly and severally guarantee payment to the Credit Union of all present and finure debts and liabilities (direct or indirect, absolute or contingent, matured or otherwise), now or at any time and from time to time hereafter due or owing to the Credit Union by the Customer, whicher incurred by the Customer alone or jointhy with any corporation, person or persons, or otherwise howsover, including all costs (and in respect of legal costs on a soliciter and his own client basis) and disbursements incurred by the Credit Union in recovering or attempting in respect of legal costs on a soliciter and his own client basis) and disbursements incurred by the Credit Union in recovering or attempting in respect of legal costs on a soliciter and his own client basis) and disbursements of any security held by the Credit Union in recovering or attempting in respect of legal costs on a soliciter and his own client including the senforement of any security held by the Credit Union in person. Provided, however, that the liability of the undersigned, and of each of the undersigned herein is limited to <u>Two Hundred & Firty Thousant Two Hundred</u> Dollars (<u>\$250,2000</u>) plus (i) interest thereon (such interest to be calculated and compounded monthly), or on so much thereof as may be outstanding from time to time, from the date of demand of payment until pid, at the rate(s) agreed upco, between the Credit Union and the Customer, and (ii) legal costs incurred by the Credit Union in enforcing and obtaining payment under this guarantee including the enforcement of any security held by the Credit Union therefore on a solicitor and his own elicit basis (the "Legal Costs").

2. In this guarantee, the word "Guarantor" shall mean the undersigned and if there is more than one, it shall mean each of them.

This guarantee shall not be affected by the dauth or loss or diminution of capacity of the Customer or of the Guarantor or by any 3. This guarantee shall not be affected by the death or loss or dimutation of capacity of the Customer or of the Guarantor or by any change in the name of the Castomer or in the membership of the firm of the Customer through the death or retirement of one or more partners or the htroduction of one or more other partners or otherwise, or by the acquisition of the business of the Customer by a corporation, firm or person, or by any change whatsoever in the objects, capital structure or constitution of one or the vestices of the Customer or the business of the Customer or cuptor the business of the customer or customer the business of the Customer or the business of the Customer and hereby releases the Credit Union from any liability resulting therefrom.

4. All monies, advances, renewals and credits borrowed or obtained by the Customer from the Credit Union shall be deemed to form part of the debts and Habilities, notwithstanding ony lack or binitation of status or of power, incapacity or disability of the Customer or of the directors, partners or agents thereof, or that the Customer may not be a legal or subble entity, or any irregularity, defect or informality in the borrowing or obtaining or such munies, advances, renewals or credits, whether known to the Credit Union or oot; and any sum which may not be, recoverable from the Guarantor on the basis of a guarantee shall be recoverable from the Guarantor as sole and principal debtor in respect thereof and shall be paid to the Credit Union on demand with interest and accessories as horin provided,

5. This guarantee shall continue and be enforceable norwithstanding any analgamation of the Credit Union with any other Credit Union(s), financial institution(s) or other corporation(s), and any further analgamation, in which event this guarantee shall also extent to all debts and lisbilities fluen or thereafter owed by the Customer to the analgamated Credit Union. Furthermore, all security, real or personal, moveable or immoveable, which has been or will be given by the Ouaranter of the said debts and hiabilities shell be valid in the hands of the Credit Union, as well as its successors and assigns.

It is further agreed that this shall be a continuing guarantee, and shall cover and secure any ultimate balance owing to the Credit 6. Un lon,

7. This guarantee shall bind the Guarantor and Guarantor's, executors, administrators, legal representatives, successors and assigns until termination thereof by notice in writing to the manager of the branch of the Credit Union at which the account of the Customer is kept, but such termination by any of the Guarantors or their respective beirs, executors, administrators, legal representatives, successors or assigns shall not prevent the continuance of the liability hereunder of any other Guarantor. Such termination shall apply only to these debts or liabilities of the Customer incurred or arising after anicely days after receipt of the notice by the Credit Union, but not in respect of any prior debts or liabilities manured or not. The notice of termination shall have no effect on those debts or liabilities incurred more than ninety days after receipt of said notice, which result or will result from express or implied commitments made prior to the ninetieth day after receipt of the colice.

8. This guarantee will not be diminished or modified on account of any act on the part of the Credit Union which would prevent subrogation from operating in favour of the Guarantor. It is further agreed that the Credit Union, without exonerating in whole or in part the Guarantor, may grant time, renewals, extensions, indulgences, release and discharges to, may take security form, and give up or release any or part of the security beld, may abstuin from taking, perfecting, registering or renewing security or from relizing on security, may accept compositions and otherwise deal with the Customer and with any other parson or persons, including any of the Guarantors, and dispose of any security held by the Credit Union as the Credit Union in reduction of the debt and linbilities hereby guaranteed, shall be considered for all purposes as payment in gross which the Credit Union shall have the right to apply as it may see fit, not being both by the Security but no fit will be united by the security or four set to prove against the estate of the Customer or gon may insolvency or windoing up, in respect of the whole said debts and liabilities. The Guarantor shall have no right to be subrogated to the Credit Union shall have received payment in full of its claims against the Customer with interest and costs.

9. If any circumstances arise necessitating the Credit Union to file its claim against the estate of the Customer and to value its security, it will be entitled to place such valuation as the Credit Union may in its discretion see fit, and the filing of such claim and the valuation of its security shall in any any prejudice or restrict its rights against the Guarantor.

10. The Credit Union shall not be obligated to exhaust its recourse against the Customer or other persons or security it may hold before being entitled to payment from the Quarautor of each and every of the debts and liabilities hereby guaranteed and it shall not be obliged to offer or deliver its security before its whole claim has been paid. The Guaranter weives all benefits of discussion and division.

.. .

• • • • • • • • • • • • • • •

. . . . . . .

11. All indebtodness and liability, present and future, of the Customer to the Guarantor are hereby assigned to the Credit Union and postponed to the present and future debts and liabilities of the Customer to the Credit Union. All monies received from the Customer or on the Customer's behalf by the Guarantor shall be held as agent, mandatary and trustee for the Credit Union and shall be paid over to the Credit Union forthwith. This provision will remain in full force and effect, notwithstanding the termination of this guarantee pursuant to the provisions of paragraph 7 in which event it will terminate when the debts and liabilities of the Customer to the Credit Union covered by this guarantee pursuant to paragraph 7 knowled here been paid in full.

12. This guarantee is in addition to and not in substitution for any other guarantee, by whomsoever given, at any time hold by the Credit Union, and without prejudice to any other security by whomsoever given held at any time by the Credit Union and the Credit Union shall be under to obligation to marshall In favour of the Guarantee any such security or any of the funds or assets the Credit Union may be entitled to receive or have a claim upon.

13. The Guarantor shall be bound by any account settled between the Credit Union and the Customer and, if no such account has been so sottled, any account stated by the Credit Union shall be accepted by the Guarantor as conclusive evidence of the amount which at the date of the account so stated is due by the Customer to the Credit Union.

14. The Guarantor shall make payment to the Credit Union of the Guarantor's liability hereunder forthwith after demand therefor is made in writing. Such demand shall be deemed to have been effectually made when an envelope containing it addressed to the Guarantor's last address known to the Credit Union is deposited postage proposid in the Post Office. The liability of the Guarantor shall bear interest from the date of such demand at the rate or rates then applicable to the debts and liabilities of the Customer to the Credit Union.

15. For better certainty, and in addition to any limitation of the liability of the Guarantor hereunder, the Guarantor upon the demand of the Credit Union shall pay to the Credit Union all Legal Costs.

16. This guarantee and agreement shall be operative and binding upon every signatory hereto and notwithstanding the non-execution thereof by any other proposed signatory or signatory or against the Guarantor that this instrument was not delivered in escrow or pursuant to any agreement that it should not be effective until any condition has been complied with. None of the parties shall be bound by any representation or promise made by any person relative thereto which is not embodied herein. The liability of the Guarantor hereunder begins on the date of the Guarantor's signature on this Guarantee and Postponement.

17. This guarantee shall be binding upon the undersigned and each of them, and if more than one, then jointly and severally between them and also upon the hoirs, executors, administrators, successors and assigns of the Guarantor and will extend to and enture to the benefit of the successors and assigns of the Credit Union. Each and every provision hereof is severable and should any provision hereof be illegal or not enforceable for any reason whatsoever, such illegality or invalidity shall not affect the other provisions hereof which shall remain in force and be binding on the parties hereto.

18. The Guarantor acknowledges having read and taken cognizance of this Guarantee and Postporement before signing it and declares that the Guarantor understands porfectly the terms, conditions and undertakings contained therein.

19. This Guarantee and Postponement shall be construed in accordance with the laws of the Province of Alberta and the Guarantor agrees that any legal suit, action or proceeding arising out of or relating to this Guarantee and Postponement may be instituted in the courts of such province, and the Guarantor hereby accepts and irrevocably submits to the jurisdiction of the said courts, and acknowledges their competence and agrees to be bound by any judgement thereof, provided that nothing herein shall limit the Credit Union's right to bring proceedings against the Guarantor elsewhere.

20. The Guarantor acknowledges and agrees that this guarantee is in addition to and not in substitution for any other guarantee(s) that any of the undersigned (alone or jointly with any other person) may have granted to the Credit Union in respect of the Customer's obligations and liability of the Customer to the Credit Union or to the Credit Union and any other person(s). This guarantee shall remain in full force and effect until all debts and obligations hereby scenred have been irrevocably and indefeasibly paid and released.

$\mathbb{C}$	AS WITNESS th	AS WITNESS the hands and seals of the Guarantor(s), at					
W.	British Colembia	10 day of	Dece	mber	2010	- d	
)	SIGNED, SEALED AND D				/	11	
1	WITINESS			SIGNATURE OF	GUARANTOP S)	6	
					AB	KA .	
				Nicholas	Gatter	6	[Seal(s)]
	Latat	prehal	_				

Note: Cortificate under Guarantaes Acknowledgement Act (Alberta) is required in respect of any individual Quarantor.

DM\_CDY/255194-00918/33275.3

## GUARANTEES ACKNOWLEDGEMENT ACT

#### CERTIFICATE OF NOTARY PUBLIC

#### I HEREBY CERTIFY THAT:

• •

.....

e

1.	Nicholas Carter of, in the Province of British Columbia , the Guarantor (or one of the Guarantors) in the guarantee dated <u>10</u> day of <u>December</u> , <u>2010</u> made between Nicholas Carter and the Credit Union named in the Guarantee and Postponement to which this certificate is attached or noted upon, appeared in person before me and acknowledged that he/she had executed the guarantee.
2.	I satisfied myself by examination of him/her that he/she is aware of the contents of the guarantee and understands it.
	GIVEN at Vancoule, Brhsh Columbia, this /D day of December, 2010 under my hand and seal of office.
	A Notary Public in and for the Province of Allar Biritish Columbia
	J am the person named in this certificate, Nicholas Carter

NOTE: A separate Certificate must be completed for each individual Guarantor.

Microsoft Word > File > New > L-Commercial & Ag > Certificete of Notery Public 10585nd - (06/2000 JB)

. . . .

.

· • • • •

,

## This is **Exhibit "J"** referred to in the Affidavit of Darcy Peelar, sworn before me this 5<sup>th</sup> day of July, 2017

ant

A Commissioner for Oaths in and for Alberta

Patrick T. Harnett Barrister and Solicitor

## GUARANTEE

## TO: SERVUS CREDIT UNION LTD.

DATE: As of September 14, 2016

## RECITALS:

- A. The undersigned obligor (the "**Obligor**") is required to deliver this guarantee and indemnity (the "**Agreement**") under the terms of the Note.
- B. The Obligor will derive substantial direct and indirect benefits and advantages from the financial accommodations to the Company under the Transaction Documents, and it will be to the Obligor's direct interest and economic benefit to deliver this Agreement in order to allow the Company to obtain those financial accommodations.
- C. The Obligor acknowledges the value of that benefit.

**NOW THEREFORE**, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Obligor agrees as follows:

## 1. INTERPRETATION

## 1.1 Capitalized Terms

In this Agreement, except where the context otherwise requires, capitalized terms that are used and not otherwise defined have the meanings defined in the Servus (as defined below), and:

- (a) "Company" means Crelogix Credit Group Inc.
- (b) "Event of Default" means the occurrence of (i) an Event of Default as defined in the Note, (ii) a "default," "event of default" or similar circumstance identified in any Transaction Document other than the Note that entitles the Company's counterparty to enforce its rights under that Transaction Document, (iii) the failure of the Company to pay any of the Obligations when due.
- (c) "Note" means the unsecured convertible note agreement dated as of the date hereof in the principal amount of Two Million Canadian Dollars (CAD \$2,000,000) made by the Company as borrower in favour of Servus as lender and note holder, as amended, supplemented, restated and replaced from time to time.
- (d) "Obligations" means all debts, liabilities and obligations of the Company to Servus under or in connection with the Note and other Transaction Documents, whether present or future, direct or indirect, absolute or contingent, matured or not, at any time owing or remaining unpaid by the Company to Servus in any currency under or in connection with the Transaction Documents, whether arising from dealings between Servus and the Company or from other dealings or proceedings by which Servus may be or become in any manner whatever creditors of the Company under or in connection with the Transaction Documents, and wherever incurred, and whether incurred by the Company alone or with another or others and whether as principal or surety (including obligations

under or in connection with any guarantee or indemnity given by the Company under or in connection with the Transaction Documents), and all interest, fees, commissions and legal and other costs, charges and expenses owing or remaining unpaid by the Company to Servus in any currency under or in connection with the Transaction Documents.

(e) "Transaction Documents" has the meaning ascribed thereto in the Note.

## 1.1 No Contra Proferentum

. a

This Agreement has been negotiated by the Obligor and Servus with the benefit of legal representation, and any rule of construction to the effect that ambiguities are to be resolved against the drafting party shall not apply to the construction or interpretation of this Agreement.

#### 1.2 Conflict With Note

If there is any conflict or inconsistency between the terms of the Note and the terms of this Agreement, the provisions of the Note shall govern to the extent necessary to remove the conflict or inconsistency.

## 1.3 Other Interpretation Rules

In this Agreement:

- (a) The division into Sections and the insertion of headings are for convenience of reference only and do not affect the construction or interpretation of this Agreement.
- (b) Unless otherwise specified or the context otherwise requires, (i) "including" or "includes" means "including (or includes) but is not limited to" and shall not be construed to limit any general statement preceding it to the specific or similar items or matters immediately following it, (ii) a reference to any legislation, statutory instrument or regulation or a section thereof is a reference to the legislation, statutory instrument, regulation or section as amended, restated and re-enacted from time to time, and (iii) words in the singular include the plural and vice-versa and words in one gender include all genders.
- (c) Unless otherwise specified or the context otherwise requires, any reference in this Agreement to payment of the Obligations includes performance of the Obligations.

#### 2. GUARANTEE AND INDEMNITY

#### 2.1 Guarantee

The Obligor unconditionally guarantees payment to Servus of the Obligations.

## 2.2 Indemnity

The Obligor also unconditionally agrees that, if the Company does not unconditionally and irrevocably pay any Obligations when due and those Obligations are not recoverable from the Obligor for any reason under Section 2.1, the Obligor shall indemnify Servus immediately on demand against any cost, loss, damage, expense or liability suffered by Servus as a result of the Company's failure to do so.

## 2.3 Separate Liabilities

The liabilities of the Obligor under Sections 2.1 and 2.2 are separate and distinct from each other, but the provisions of this Agreement shall apply to the liabilities under both of those Sections unless the context otherwise requires.

#### 2.4 Limit on Liability

The liability of the Obligor under this Agreement is unlimited.

#### 2.5 Irrevocable

This Agreement is irrevocable by the Obligor, and the Obligor expressly and unconditionally waives any right to terminate this Agreement.

#### 3. CONTINUING AGREEMENT AND REINSTATEMENT

#### 3.1 Continuing Agreement

This Agreement is a continuing guarantee and indemnity for a current or running account and will extend to the ultimate balance of the Obligations, regardless of any intermediate payment or discharge of the Obligations in whole or in part. Without limiting the foregoing, the Obligations may include advances and readvances under revolving credit facilities, which permit borrowing, repayment of all or part of the amount borrowed and re-borrowing of amounts previously paid.

#### 3.2 Payments in Gross

Until this Agreement has been terminated in accordance with Section 3.4, all amounts of any kind received by Servus from any source in respect of the Obligations shall be regarded for all purposes as payments in gross without any right on the part of the Obligor to claim the benefit of those amounts in reduction of its liabilities under this Agreement.

### 3.3 Reinstatement

If at any time any payment of the Obligations is or must be rescinded or returned by Servus as a result of insolvency or reorganization of the Company or any other person, or for any other reason whatsoever, the Obligations will be deemed to have continued in existence and this Agreement shall continue to be effective, or be reinstated, as if the payment had not occurred. Servus may concede or compromise any claim that any payment ought to be rescinded or returned without diminishing the liability of the Obligor under this Section.

#### 3.4 Termination

If the Obligations have been indefeasibly paid in full (whether by Conversion as defined in in and in accordance with the Note or in cash) and if all obligations of Servus to extend credit under the Transaction Documents have been cancelled, then Servus shall, at the request and expense of the Obligor, execute and deliver whatever documents are reasonably required to acknowledge the termination of this Agreement.

#### 4. WAIVER OF DEFENCES AND OTHER MATTERS

#### 4.1 In Addition to Other Rights; No Marshalling

This Agreement is in addition to and is not in any way prejudiced by or merged with any other guarantee, indemnity or security now or subsequently held by Servus in respect of any Obligations. Servus shall be under no obligation to marshal in favour of the Obligor any other guarantees or other securities or any money or other property that Servus may be entitled to receive or may have a claim upon.

#### 4.2 Liabilities Unconditional

The liabilities of the Obligor under this Agreement are absolute and unconditional, and will not be affected by any act, omission, matter or thing that, but for this Section, would reduce, release or prejudice any of its liabilities under this Agreement, or that might constitute a legal or equitable defence to or a discharge, limitation or reduction of the Obligor's liabilities under this Agreement, including the following, whether or not known to it or Servus or consented to by it or Servus:

- (a) any discontinuance, reduction, increase, extension or other variance in the credit granted by Servus to the Company or any time, waiver or consent granted to, or any release of or compromise with, the Company or any other person;
- (b) any amendment, supplement or restatement (however fundamental) or replacement of any Transaction Document;
- any unenforceability, illegality or invalidity of any obligation of any person under or in connection any Transaction Document, including any bar to recovery under any statute of limitations;
- (d) the loss of capacity of the Company, any change in the name of the Company, or in the ownership, objects, capital structure or constitution of the Company, the sale of all or any part of the Company's business or the Company being amalgamated or merged with one or more other entities, but shall, notwithstanding any such event, continue to apply to all Obligations whether previously or subsequently incurred; and in the case of the Company being amalgamated or merged with one or more other entities, this Agreement shall also apply to the liabilities of the resulting or continuing entity, and the term "Company" shall include each resulting or continuing entity;
- (e) any credit being granted or continued by Servus purportedly to or for the Company after the loss of capacity, bankruptcy or insolvency of the Company;
- (f) any lack or limitation of power, incapacity or disability of the Company or of the directors or agents of the Company, or the Company not being a legal or suable entity, or any irregularity, defect or lack of formality in the obtaining of credit by the Company;
- (g) any bankruptcy, insolvency or similar proceedings, including any stay of or moratorium on proceedings;
- (h) any impossibility, impracticability, frustration of purpose, *force majeure*, illegality or act of governmental authority affecting any Transaction Document;

- (i) any taking or failure to take security, any loss of or loss of value of security for the Obligations, any invalidity, lack of perfection or unenforceability of any security, or any enforcement of, failure to enforce or irregularity or deficiency in the enforcement of any security; or
- (j) the existence of any claim, set-off or other right that the Obligor may have against the Company, Servus or any other person, whether in connection with the Transaction Documents or otherwise.

Each of the defences mentioned above is waived by the Obligor to the fullest extent permitted under applicable law.

#### 4.3 Information Concerning Company

The Obligor acknowledges that it is currently familiar with the Transaction Documents and the terms thereof, the financial condition of the Company and any other circumstances affecting the risk incurred by the Obligor in connection with this Agreement. The Obligor shall be solely responsible for keeping itself informed concerning those matters in the future. The Obligor acknowledges that Servus has no obligation to provide any information concerning those matters now or in the future and that, if it does so at any time, it shall have no obligation to update the information or provide other information subsequently.

#### 4.4 No Obligation to Enforce Other Rights

The Obligor waives any right it may have of requiring Servus (or any trustee or agent on its behalf) to proceed against or enforce any other rights or security or claim payment from any person before claiming from the Obligor under this Agreement and the Obligor waives all benefits of discussion and division. These waivers apply irrespective of any law or any provision of any Transaction Document to the contrary.

## 4.5 Saskatchewan

*The Limitation of Civil Rights Act* (Saskatchewan) shall not have any application to this Agreement, or to any agreement or instrument renewing, extending or collateral to this Agreement, or to the rights, powers or remedies of Servus under this Agreement.

## 5. USE OF AMOUNTS RECEIVED

## 5.1 Use of Amounts Received

Until this Agreement has been terminated in accordance with Section 32.4, Servus (or any trustee or agent on its behalf) may:

- (a) refrain from applying any money received or enforcing any other security or rights held by or on behalf of Servus in respect of the Obligations, or apply any money and enforce any other security or rights in any manner and order as they see fit;
- (b) change any application of money received in whole or in part from time to time; and
- (c) hold in a suspense account any money received from the Obligor or on account of the Obligor's liabilities under this Agreement.

#### 6. POSTPONEMENT OF OBLIGOR'S RIGHTS

#### 6.1 Postponement of Subrogation

Until this Agreement has been terminated in accordance with Section 3.4, the Obligor shall not exercise any rights that it may have by reason of performance by it of its liabilities under this Agreement:

- (a) to be indemnified by the Company;
- (b) to claim contribution from any other guarantor of the Obligations; or
- (c) to take the benefit (in whole or in part and whether by way of subrogation or otherwise) of any rights of Servus under any Transaction Document.

#### 6.2 Postponement of Set-Off

Until this Agreement has been terminated in accordance with Section 3.4, the Obligor shall not claim any set-off or counterclaim against the Company as a result of any liability of the Company to the Obligor, or claim or prove in the bankruptcy or insolvency of the Company in competition with Servus.

### 6.3 **Postponement and Assignment**

The Obligor postpones payment of all present and future debts, liabilities and obligations of the Company to the Obligor until this Agreement has been terminated in accordance with Section 3.4. The Obligor assigns to Servus all present and future debts, liabilities and obligations of the Company to the Obligor as security for payment of the Obligor's liabilities under this Agreement, and agrees that all money received by the Obligor in respect of those debts, liabilities and obligations shall be received in trust for Servus and forthwith upon receipt shall be paid over to Servus, all without in any way lessening or limiting the liabilities of the Obligor under this Agreement. The provisions of this Section 6.3 are independent of the other provisions of this Agreement and shall remain in full force and effect until this Agreement has been terminated in accordance with Section 3.4, notwithstanding that the other liabilities of the Obligor under this Agreement may have been discharged or terminated.

#### 7. OBLIGATION TO MAKE PAYMENT

#### 7.1 Payment Immediately After Demand

The Obligor's liability to make a payment under this Agreement shall arise immediately after demand for payment has been made in writing on the Obligor. In connection with any demand, Servus may treat all Obligations as due and payable and may demand immediate payment from the Obligor of the total amount of its liabilities under this Agreement, whether or not all Obligations are otherwise due and payable at the time of demand.

#### 7.2 Right to Enforce

Demand under this Agreement may be made from time to time upon the occurrence of an Event of Default and during its continuance, and the liabilities of the Obligor under this Agreement may be enforced, irrespective of:

- (a) whether any demands, steps or proceedings are being or have been made or taken against the Company and/or any third party; or
- (b) whether or in what order any security to which Servus may be entitled in connection with any Transaction Document is enforced.

#### 7.3 Certificate as to Amount

A certificate of Servus specifying the outstanding amount of the Obligations shall be conclusive evidence of that amount against the Obligor in the absence of any manifest error.

#### 7.4 Interest

The Obligor's liabilities under this Agreement shall bear interest from the date of demand at the highest rate of interest per annum that is applicable to any part of the Obligations.

### 7.5 Rights Cumulative

No failure on the part of Servus to exercise, nor any delay in exercising, any right or remedy under any Transaction Document or this Agreement shall operate as a waiver, nor shall any single or partial exercise of any right or remedy prevent any further or other exercise or the exercise of any other right or remedy. Neither the taking of any judicial or extra judicial proceeding nor the exercise of rights under any security held from the Obligor shall extinguish the liability of the Obligor to pay and perform its liabilities under this Agreement, nor shall the acceptance of any payment or security create any novation. No covenant, representation or warranty of the Obligor in this Agreement shall merge in any judgment. The rights and remedies provided in this Agreement are cumulative and do not exclude any rights and remedies provided by law or otherwise.

#### 7.6 Limitation Periods

To the extent that any limitation period applies to any claim for payment of the Obligations or remedy for enforcement of the Obligations, the Obligor agrees that:

- (a) any limitation period is expressly excluded and waived entirely if permitted by applicable law;
- (b) if a complete exclusion and waiver of any limitation period is not permitted by applicable law, any limitation period is extended to the maximum length permitted by applicable law;
- (c) any applicable limitation period shall not begin before an express demand for payment of the Obligations is made in writing by the Servus to the Obligor; and
- (d) any applicable limitation period shall begin afresh upon any payment or other acknowledgment of the Obligations by the Obligor.

### 8. PAYMENTS

#### 8.1 Withholdings

Any payment made by the Obligor under this Agreement shall be made without any deduction or withholding for or on account of tax and without any setoff or counterclaim

of any kind. However, if the Obligor is required by law to deduct, withhold or pay any tax in respect of any payment under this Agreement, then (a) the Obligor shall pay additional sums under this Agreement as necessary so that, after making or allowing for all required deductions, withholdings and payments (including deductions, withholdings and payments applicable to additional sums payable under this Section), each Secured Party receives an amount equal to the sum it would have received had no deductions, withholdings or payments been required, (b) the Obligor shall make any deductions, withholdings or payments required by law to be made by it, and (c) the Obligor shall timely pay the full amount required to be deducted, withheld or paid to the relevant governmental authority in accordance with applicable law.

#### 8.2 Currency and Place of Payment

Payment shall be made in the currency or currencies specified in the demand for payment to Servus at Servus's address specified in the Note or another address or account that Servus may specify by written notice to the Obligor from time to time.

## 8.3 Currency Indemnity

If a judgment or order is rendered by any court or tribunal for the payment of any amount owing to Servus under or in connection with this Agreement and the judgment or order is expressed in a currency (the "Judgment Currency") other than the currency payable under or in connection with this Agreement (the "Agreed Currency"), the Obligor shall indemnify and hold each Secured Party harmless against any deficiency in terms of the Agreed Currency in the amount received by that Secured Party arising or resulting from any variation as between (a) the rate at which the Agreed Currency is converted into the Judgment Currency for the purposes of the judgment or order, and (b) the rate at which the Secured Party is able to purchase the Agreed Currency in accordance with normal banking practice with the amount of the Judgment Currency actually received by the Secured Party on the date of receipt. The indemnity in this Section shall constitute a separate and independent liability from the other liabilities of the Obligor under this Agreement, shall apply irrespective of any indulgence granted by Servus, and shall be secured by-any security held by Servus from the Obligor.

## 8.4 Set-Off

Servus is authorized at any time and from time to time to set off and apply any and all deposits (general or special, time or demand, provisional or final, in whatever currency) at any time held and other obligations (in whatever currency) at any time owing by Servus or affiliates to or for the credit or the account of the Obligor against any and all of the liabilities of the Obligor now or in the future existing under this Agreement, irrespective of whether or not Servus has made any demand under this Agreement and although those liabilities of the Obligor may be contingent or unmatured. The rights of Servus under this Section 8.4 are in addition to other rights and remedies (including other rights of set-off, consolidation of accounts and bankers' lien) that Servus may have.

## 9. NOTICES

#### 9.1 Notices in Writing

Any communication to be made under this Agreement shall be made in accordance with the procedures set out in the Note.
# 9.2 Address for Notice

The Obligor's address for notice in accordance with the Note is:

900 - 4445 Lougheed Highway Burnaby, BC V5C 0E4 Attention: Karl Sigerist

Fax: (604) 357-1095 Email: ksigerist@crelogix.com

# 10. ENTIRE AGREEMENT; SEVERABILITY

# 10.1 Entire Agreement

This Agreement embodies all the agreements between the Obligor and Servus relating to the guarantee, indemnity, assignment and postponement contemplated in this Agreement. No party shall be bound by any representation or promise made by any person relating to this Agreement that is not embodied in it. It is specifically agreed that Servus shall not be bound by any representation or promise made by the Company to the Obligor. Any waiver of, or consent to departure from, the requirements of any provision of this Agreement shall be effective only if it is in writing and signed by Servus, and only in the specific instance and for the specific purpose for which it has been given.

# 10.2 Severability

If, in any jurisdiction, any provision of this Agreement or its application to any circumstance is restricted, prohibited or unenforceable, that provision shall, as to that jurisdiction, be ineffective only to the extent of that restriction, prohibition or unenforceability without invalidating the remaining provisions of this Agreement, without affecting the validity or enforceability of that provision in any other jurisdiction and, if applicable, without affecting its application to other circumstances.

# 11. DELIVERY OF AGREEMENT

# 11.1 Delivery and Counterparts

To evidence the fact that it has executed this Agreement, the Obligor may send a signed copy of this Agreement or its signature to this Agreement by facsimile transmission or email in portable document format (PDF) and the signature sent in that way shall be deemed to be its original signature for all purposes. This Agreement may be executed in any number of counterparts and all counterparts taken together shall be deemed to constitute a single agreement.

# 11.2 No Conditions

Possession of this Agreement by Servus shall be conclusive evidence against the Obligor that the Agreement was not delivered in escrow or pursuant to any agreement that it should not be effective until any condition precedent or subsequent has been complied with. This Agreement shall be operative and binding notwithstanding that it is not executed by any proposed signatory.

# 11.3 Receipt and Waiver

The Obligor acknowledges receipt of a copy of this Agreement. The Obligor waives any notice of acceptance of this Agreement by Servus. The Obligor also waives the right to receive a copy of any financing statement or financing change statement that may be registered in connection with this Agreement or any verification statement issued with respect to a registration, if waiver is not otherwise prohibited by law. The Obligor agrees that Servus may from time to time provide information regarding this Agreement and the Obligations to persons that Servus believes in good faith are entitled to the information under applicable law.

# 12. GOVERNING LAW

# 12.1 Governing Law

This Agreement and any dispute arising from or in relation to this Agreement shall be governed by, and interpreted and enforced in accordance with, the law of the province of Alberta and the laws of Canada applicable in that province, excluding the conflict of law rules of that province.

# 12.2 Obligor's Exclusive Dispute Resolution Jurisdiction

The Obligor agrees that the courts of the Province of Alberta have exclusive jurisdiction over any dispute arising from or in relation to this Agreement and the Obligor irrevocably and unconditionally attorns to the exclusive jurisdiction of that province. The Obligor agrees that the courts of that province are the most appropriate and convenient forum to settle disputes and agrees not to argue to the contrary.

# 12.3 Secured Parties Entitled to Concurrent Jurisdiction

Despite Section 12.2, Servus is permitted to take proceedings in relation to any dispute arising from or in relation to this Agreement in any court of another province or another state with jurisdiction and to the extent allowed by law may take concurrent proceedings in any number of jurisdictions.

# 13. SUCCESSORS AND ASSIGNS

# 13.1 Successors and Assigns

The Obligor may not assign or transfer all or any part of its liabilities under this Agreement. This Agreement shall enure to the benefit of Servus and its successors and assigns and be binding on the Obligor and its successors and any permitted assigns.

IN WITNESS OF WHICH the Obligor has duly executed this Agreement.

**CRELOGIX ACCEPTANCE** CORPORATION , Per: ι 2 Name: KARL SIGERIST

Title: DIRECTOR

Per:

Name:

Title:

IWe have the authority to bind the corporation

[Signature page of guarantee of Crelogix Acceptance Corporation]

20077064.1

# GUARANTEE

# TO: SERVUS CREDIT UNION LTD.

DATE: As of September 14, 2016

# RECITALS:

- A. The undersigned obligor (the "**Obligor**") is required to deliver this guarantee and indemnity (the "**Agreement**") under the terms of the Note.
- B. The Obligor will derive substantial direct and indirect benefits and advantages from the financial accommodations to the Company under the Transaction Documents, and it will be to the Obligor's direct interest and economic benefit to deliver this Agreement in order to allow the Company to obtain those financial accommodations.
- C. The Obligor acknowledges the value of that benefit.

**NOW THEREFORE**, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Obligor agrees as follows:

# 1. INTERPRETATION

# 1.1 Capitalized Terms

In this Agreement, except where the context otherwise requires, capitalized terms that are used and not otherwise defined have the meanings defined in the Servus (as defined below), and:

- (a) "Company" means Crelogix Credit Group Inc.
- (b) **"Event of Default**" means the occurrence of (i) an Event of Default as defined in the Note, (ii) a "**default**," "**event of default**" or similar circumstance identified in any Transaction Document other than the Note that entitles the Company's counterparty to enforce its rights under that Transaction Document, (iii) the failure of the Company to pay any of the Obligations when due.
- (c) "Note" means the unsecured convertible note agreement dated as of the date hereof in the principal amount of Two Million Canadian Dollars (CAD \$2,000,000) made by the Company as borrower in favour of Servus as lender and note holder, as amended, supplemented, restated and replaced from time to time.
- (d) "Obligations" means all debts, liabilities and obligations of the Company to Servus under or in connection with the Note and other Transaction Documents, whether present or future, direct or indirect, absolute or contingent, matured or not, at any time owing or remaining unpaid by the Company to Servus in any currency under or in connection with the Transaction Documents, whether arising from dealings between Servus and the Company or from other dealings or proceedings by which Servus may be or become in any manner whatever creditors of the Company under or in connection with the Transaction Documents, and wherever incurred, and whether incurred by the Company alone or with another or others and whether as principal or surety (including obligations

under or in connection with any guarantee or indemnity given by the Company under or in connection with the Transaction Documents), and all interest, fees, commissions and legal and other costs, charges and expenses owing or remaining unpaid by the Company to Servus in any currency under or in connection with the Transaction Documents.

(e) "Transaction Documents" has the meaning ascribed thereto in the Note.

# 1.1 No Contra Proferentum

This Agreement has been negotiated by the Obligor and Servus with the benefit of legal representation, and any rule of construction to the effect that ambiguities are to be resolved against the drafting party shall not apply to the construction or interpretation of this Agreement.

# 1.2 Conflict With Note

If there is any conflict or inconsistency between the terms of the Note and the terms of this Agreement, the provisions of the Note shall govern to the extent necessary to remove the conflict or inconsistency.

# 1.3 Other Interpretation Rules

In this Agreement:

- (a) The division into Sections and the insertion of headings are for convenience of reference only and do not affect the construction or interpretation of this Agreement.
- (b) Unless otherwise specified or the context otherwise requires, (i) "including" or "includes" means "including (or includes) but is not limited to" and shall not be construed to limit any general statement preceding it to the specific or similar items or matters immediately following it, (ii) a reference to any legislation, statutory instrument or regulation or a section thereof is a reference to the legislation, statutory instrument, regulation or section as amended, restated and re-enacted from time to time, and (iii) words in the singular include the plural and vice-versa and words in one gender include all genders.
- (c) Unless otherwise specified or the context otherwise requires, any reference in this Agreement to payment of the Obligations includes performance of the Obligations.

# 2. GUARANTEE AND INDEMNITY

# 2.1 Guarantee

The Obligor unconditionally guarantees payment to Servus of the Obligations.

# 2.2 Indemnity

The Obligor also unconditionally agrees that, if the Company does not unconditionally and irrevocably pay any Obligations when due and those Obligations are not recoverable from the Obligor for any reason under Section 2.1, the Obligor shall indemnify Servus immediately on demand against any cost, loss, damage, expense or liability suffered by Servus as a result of the Company's failure to do so.

- 3 -

# 2.3 Separate Liabilities

The liabilities of the Obligor under Sections 2.1 and 2.2 are separate and distinct from each other, but the provisions of this Agreement shall apply to the liabilities under both of those Sections unless the context otherwise requires.

# 2.4 Limit on Liability

The liability of the Obligor under this Agreement is unlimited.

# 2.5 Irrevocable

This Agreement is irrevocable by the Obligor, and the Obligor expressly and unconditionally waives any right to terminate this Agreement.

# 3. CONTINUING AGREEMENT AND REINSTATEMENT

# 3.1 Continuing Agreement

This Agreement is a continuing guarantee and indemnity for a current or running account and will extend to the ultimate balance of the Obligations, regardless of any intermediate payment or discharge of the Obligations in whole or in part. Without limiting the foregoing, the Obligations may include advances and readvances under revolving credit facilities, which permit borrowing, repayment of all or part of the amount borrowed and re-borrowing of amounts previously paid.

# 3.2 Payments in Gross

Until this Agreement has been terminated in accordance with Section 3.4, all amounts of any kind received by Servus from any source in respect of the Obligations shall be regarded for all purposes as payments in gross without any right on the part of the Obligor to claim the benefit of those amounts in reduction of its liabilities under this Agreement.

# 3.3 Reinstatement

If at any time any payment of the Obligations is or must be rescinded or returned by Servus as a result of insolvency or reorganization of the Company or any other person, or for any other reason whatsoever, the Obligations will be deemed to have continued in existence and this Agreement shall continue to be effective, or be reinstated, as if the payment had not occurred. Servus may concede or compromise any claim that any payment ought to be rescinded or returned without diminishing the liability of the Obligor under this Section.

# 3.4 Termination

If the Obligations have been indefeasibly paid in full (whether by Conversion as defined in in and in accordance with the Note or in cash) and if all obligations of Servus to extend credit under the Transaction Documents have been cancelled, then Servus shall, at the request and expense of the Obligor, execute and deliver whatever documents are reasonably required to acknowledge the termination of this Agreement.

# 4. WAIVER OF DEFENCES AND OTHER MATTERS

# 4.1 In Addition to Other Rights; No Marshalling

This Agreement is in addition to and is not in any way prejudiced by or merged with any other guarantee, indemnity or security now or subsequently held by Servus in respect of any Obligations. Servus shall be under no obligation to marshal in favour of the Obligor any other guarantees or other securities or any money or other property that Servus may be entitled to receive or may have a claim upon.

# 4.2 Liabilities Unconditional

The liabilities of the Obligor under this Agreement are absolute and unconditional, and will not be affected by any act, omission, matter or thing that, but for this Section, would reduce, release or prejudice any of its liabilities under this Agreement, or that might constitute a legal or equitable defence to or a discharge, limitation or reduction of the Obligor's liabilities under this Agreement, including the following, whether or not known to it or Servus or consented to by it or Servus:

- (a) any discontinuance, reduction, increase, extension or other variance in the credit granted by Servus to the Company or any time, waiver or consent granted to, or any release of or compromise with, the Company or any other person;
- (b) any amendment, supplement or restatement (however fundamental) or replacement of any Transaction Document;
- any unenforceability, illegality or invalidity of any obligation of any person under or in connection any Transaction Document, including any bar to recovery under any statute of limitations;
- (d) the loss of capacity of the Company, any change in the name of the Company, or in the ownership, objects, capital structure or constitution of the Company, the sale of all or any part of the Company's business or the Company being amalgamated or merged with one or more other entities, but shall, notwithstanding any such event, continue to apply to all Obligations whether previously or subsequently incurred; and in the case of the Company being amalgamated or merged with one or more other entities, this Agreement shall also apply to the liabilities of the resulting or continuing entity, and the term "Company" shall include each resulting or continuing entity;
- (e) any credit being granted or continued by Servus purportedly to or for the Company after the loss of capacity, bankruptcy or insolvency of the Company;
- (f) any lack or limitation of power, incapacity or disability of the Company or of the directors or agents of the Company, or the Company not being a legal or suable entity, or any irregularity, defect or lack of formality in the obtaining of credit by the Company;
- (g) any bankruptcy, insolvency or similar proceedings, including any stay of or moratorium on proceedings;
- (h) any impossibility, impracticability, frustration of purpose, *force majeure*, illegality or act of governmental authority affecting any Transaction Document;

- (i) any taking or failure to take security, any loss of or loss of value of security for the Obligations, any invalidity, lack of perfection or unenforceability of any security, or any enforcement of, failure to enforce or irregularity or deficiency in the enforcement of any security; or
- (j) the existence of any claim, set-off or other right that the Obligor may have against the Company, Servus or any other person, whether in connection with the Transaction Documents or otherwise.

Each of the defences mentioned above is waived by the Obligor to the fullest extent permitted under applicable law.

# 4.3 Information Concerning Company

The Obligor acknowledges that it is currently familiar with the Transaction Documents and the terms thereof, the financial condition of the Company and any other circumstances affecting the risk incurred by the Obligor in connection with this Agreement. The Obligor shall be solely responsible for keeping itself informed concerning those matters in the future. The Obligor acknowledges that Servus has no obligation to provide any information concerning those matters now or in the future and that, if it does so at any time, it shall have no obligation to update the information or provide other information subsequently.

# 4.4 No Obligation to Enforce Other Rights

The Obligor waives any right it may have of requiring Servus (or any trustee or agent on its behalf) to proceed against or enforce any other rights or security or claim payment from any person before claiming from the Obligor under this Agreement and the Obligor waives all benefits of discussion and division. These waivers apply irrespective of any law or any provision of any Transaction Document to the contrary.

# 4.5 Saskatchewan

*The Limitation of Civil Rights Act* (Saskatchewan) shall not have any application to this Agreement, or to any agreement or instrument renewing, extending or collateral to this Agreement, or to the rights, powers or remedies of Servus under this Agreement.

# 5. USE OF AMOUNTS RECEIVED

# 5.1 Use of Amounts Received

Until this Agreement has been terminated in accordance with Section 32.4, Servus (or any trustee or agent on its behalf) may:

- (a) refrain from applying any money received or enforcing any other security or rights held by or on behalf of Servus in respect of the Obligations, or apply any money and enforce any other security or rights in any manner and order as they see fit;
- (b) change any application of money received in whole or in part from time to time; and
- (c) hold in a suspense account any money received from the Obligor or on account of the Obligor's liabilities under this Agreement.

# -6-

# 6. POSTPONEMENT OF OBLIGOR'S RIGHTS

# 6.1 Postponement of Subrogation

; 4

Until this Agreement has been terminated in accordance with Section 3.4, the Obligor shall not exercise any rights that it may have by reason of performance by it of its liabilities under this Agreement:

- (a) to be indemnified by the Company;
- (b) to claim contribution from any other guarantor of the Obligations; or
- (c) to take the benefit (in whole or in part and whether by way of subrogation or otherwise) of any rights of Servus under any Transaction Document.

# 6.2 Postponement of Set-Off

Until this Agreement has been terminated in accordance with Section 3.4, the Obligor shall not claim any set-off or counterclaim against the Company as a result of any liability of the Company to the Obligor, or claim or prove in the bankruptcy or insolvency of the Company in competition with Servus.

# 6.3 Postponement and Assignment

The Obligor postpones payment of all present and future debts, liabilities and obligations of the Company to the Obligor until this Agreement has been terminated in accordance with Section 3.4. The Obligor assigns to Servus all present and future debts, liabilities and obligations of the Company to the Obligor as security for payment of the Obligor's liabilities under this Agreement, and agrees that all money received by the Obligor in respect of those debts, liabilities and obligations shall be received in trust for Servus and forthwith upon receipt shall be paid over to Servus, all without in any way lessening or limiting the liabilities of the Obligor under this Agreement. The provisions of this Section 6.3 are independent of the other provisions of this Agreement and shall remain in full force and effect until this Agreement has been terminated in accordance with Section 3.4, notwithstanding that the other liabilities of the Obligor under this Agreement may have been discharged or terminated.

# 7. OBLIGATION TO MAKE PAYMENT

# 7.1 Payment Immediately After Demand

The Obligor's liability to make a payment under this Agreement shall arise immediately after demand for payment has been made in writing on the Obligor. In connection with any demand, Servus may treat all Obligations as due and payable and may demand immediate payment from the Obligor of the total amount of its liabilities under this Agreement, whether or not all Obligations are otherwise due and payable at the time of demand.

# 7.2 Right to Enforce

Demand under this Agreement may be made from time to time upon the occurrence of an Event of Default and during its continuance, and the liabilities of the Obligor under this Agreement may be enforced, irrespective of:

- (a) whether any demands, steps or proceedings are being or have been made or taken against the Company and/or any third party; or
- (b) whether or in what order any security to which Servus may be entitled in connection with any Transaction Document is enforced.

# 7.3 Certificate as to Amount

A certificate of Servus specifying the outstanding amount of the Obligations shall be conclusive evidence of that amount against the Obligor in the absence of any manifest error.

# 7.4 Interest

The Obligor's liabilities under this Agreement shall bear interest from the date of demand at the highest rate of interest per annum that is applicable to any part of the Obligations.

# 7.5 Rights Cumulative

No failure on the part of Servus to exercise, nor any delay in exercising, any right or remedy under any Transaction Document or this Agreement shall operate as a waiver, nor shall any single or partial exercise of any right or remedy prevent any further or other exercise or the exercise of any other right or remedy. Neither the taking of any judicial or extra judicial proceeding nor the exercise of rights under any security held from the Obligor shall extinguish the liability of the Obligor to pay and perform its liabilities under this Agreement, nor shall the acceptance of any payment or security create any novation. No covenant, representation or warranty of the Obligor in this Agreement shall merge in any judgment. The rights and remedies provided in this Agreement are cumulative and do not exclude any rights and remedies provided by law or otherwise.

# 7.6 Limitation Periods

To the extent that any limitation period applies to any claim for payment of the Obligations or remedy for enforcement of the Obligations, the Obligor agrees that:

- (a) any limitation period is expressly excluded and waived entirely if permitted by applicable law;
- (b) if a complete exclusion and waiver of any limitation period is not permitted by applicable law, any limitation period is extended to the maximum length permitted by applicable law;
- (c) any applicable limitation period shall not begin before an express demand for payment of the Obligations is made in writing by the Servus to the Obligor; and
- (d) any applicable limitation period shall begin afresh upon any payment or other acknowledgment of the Obligations by the Obligor.

# 8. PAYMENTS

# 8.1 Withholdings

Any payment made by the Obligor under this Agreement shall be made without any deduction or withholding for or on account of tax and without any setoff or counterclaim

of any kind. However, if the Obligor is required by law to deduct, withhold or pay any tax in respect of any payment under this Agreement, then (a) the Obligor shall pay additional sums under this Agreement as necessary so that, after making or allowing for all required deductions, withholdings and payments (including deductions, withholdings and payments applicable to additional sums payable under this Section), each Secured Party receives an amount equal to the sum it would have received had no deductions, withholdings or payments been required, (b) the Obligor shall make any deductions, withholdings or payments required by law to be made by it, and (c) the Obligor shall timely pay the full amount required to be deducted, withheld or paid to the relevant governmental authority in accordance with applicable law.

# 8.2 Currency and Place of Payment

Payment shall be made in the currency or currencies specified in the demand for payment to Servus at Servus's address specified in the Note or another address or account that Servus may specify by written notice to the Obligor from time to time.

# 8.3 Currency Indemnity

If a judgment or order is rendered by any court or tribunal for the payment of any amount owing to Servus under or in connection with this Agreement and the judgment or order is expressed in a currency (the "Judgment Currency") other than the currency payable under or in connection with this Agreement (the "Agreed Currency"), the Obligor shall indemnify and hold each Secured Party harmless against any deficiency in terms of the Agreed Currency in the amount received by that Secured Party arising or resulting from any variation as between (a) the rate at which the Agreed Currency is converted into the Judgment Currency for the purposes of the judgment or order, and (b) the rate at which the Secured Party is able to purchase the Agreed Currency in accordance with normal banking practice with the amount of the Judgment Currency actually received by the Secured Party on the date of receipt. The indemnity in this Section shall constitute a separate and independent liability from the other liabilities of the Obligor under this Agreement, shall apply irrespective of any indulgence granted by Servus, and shall be secured by-any security held by Servus from the Obligor.

# 8.4 Set-Off

Servus is authorized at any time and from time to time to set off and apply any and all deposits (general or special, time or demand, provisional or final, in whatever currency) at any time held and other obligations (in whatever currency) at any time owing by Servus or affiliates to or for the credit or the account of the Obligor against any and all of the liabilities of the Obligor now or in the future existing under this Agreement, irrespective of whether or not Servus has made any demand under this Agreement and although those liabilities of the Obligor may be contingent or unmatured. The rights of Servus under this Section 8.4 are in addition to other rights and remedies (including other rights of set-off, consolidation of accounts and bankers' lien) that Servus may have.

# 9. NOTICES

# 9.1 Notices in Writing

Any communication to be made under this Agreement shall be made in accordance with the procedures set out in the Note.

# 9.2 Address for Notice

The Obligor's address for notice in accordance with the Note is:

900 - 4445 Lougheed Highway Burnaby, BC V5C 0E4 Attention: Karl Sigerist

Fax: (604) 357-1095 Email: ksigerist@crelogix.com

# 10. ENTIRE AGREEMENT; SEVERABILITY

# 10.1 Entire Agreement

This Agreement embodies all the agreements between the Obligor and Servus relating to the guarantee, indemnity, assignment and postponement contemplated in this Agreement. No party shall be bound by any representation or promise made by any person relating to this Agreement that is not embodied in it. It is specifically agreed that Servus shall not be bound by any representation or promise made by the Company to the Obligor. Any waiver of, or consent to departure from, the requirements of any provision of this Agreement shall be effective only if it is in writing and signed by Servus, and only in the specific instance and for the specific purpose for which it has been given.

# 10.2 Severability

If, in any jurisdiction, any provision of this Agreement or its application to any circumstance is restricted, prohibited or unenforceable, that provision shall, as to that jurisdiction, be ineffective only to the extent of that restriction, prohibition or unenforceability without invalidating the remaining provisions of this Agreement, without affecting the validity or enforceability of that provision in any other jurisdiction and, if applicable, without affecting its application to other circumstances.

# 11. DELIVERY OF AGREEMENT

# 11.1 Delivery and Counterparts

To evidence the fact that it has executed this Agreement, the Obligor may send a signed copy of this Agreement or its signature to this Agreement by facsimile transmission or email in portable document format (PDF) and the signature sent in that way shall be deemed to be its original signature for all purposes. This Agreement may be executed in any number of counterparts and all counterparts taken together shall be deemed to constitute a single agreement.

# 11.2 No Conditions

Possession of this Agreement by Servus shall be conclusive evidence against the Obligor that the Agreement was not delivered in escrow or pursuant to any agreement that it should not be effective until any condition precedent or subsequent has been complied with. This Agreement shall be operative and binding notwithstanding that it is not executed by any proposed signatory.

# 11.3 Receipt and Waiver

The Obligor acknowledges receipt of a copy of this Agreement. The Obligor waives any notice of acceptance of this Agreement by Servus. The Obligor also waives the right to receive a copy of any financing statement or financing change statement that may be registered in connection with this Agreement or any verification statement issued with respect to a registration, if waiver is not otherwise prohibited by law. The Obligor agrees that Servus may from time to time provide information regarding this Agreement and the Obligations to persons that Servus believes in good faith are entitled to the information under applicable law.

# 12. GOVERNING LAW

# 12.1 Governing Law

This Agreement and any dispute arising from or in relation to this Agreement shall be governed by, and interpreted and enforced in accordance with, the law of the province of Alberta and the laws of Canada applicable in that province, excluding the conflict of law rules of that province.

# 12.2 Obligor's Exclusive Dispute Resolution Jurisdiction

The Obligor agrees that the courts of the Province of Alberta have exclusive jurisdiction over any dispute arising from or in relation to this Agreement and the Obligor irrevocably and unconditionally attorns to the exclusive jurisdiction of that province. The Obligor agrees that the courts of that province are the most appropriate and convenient forum to settle disputes and agrees not to argue to the contrary.

# 12.3 Secured Parties Entitled to Concurrent Jurisdiction

Despite Section 12.2, Servus is permitted to take proceedings in relation to any dispute arising from or in relation to this Agreement in any court of another province or another state with jurisdiction and to the extent allowed by law may take concurrent proceedings in any number of jurisdictions.

# 13. SUCCESSORS AND ASSIGNS

# 13.1 Successors and Assigns

The Obligor may not assign or transfer all or any part of its liabilities under this Agreement. This Agreement shall enure to the benefit of Servus and its successors and assigns and be binding on the Obligor and its successors and any permitted assigns.

**IN WITNESS OF WHICH** the Obligor has duly executed this Agreement.

CREL	OGIX PC	RTFOLI	OSERVI	CES
CORP	ORATIO	N		
Per:	-			<u> </u>
	N	1/ . 0 .	a	-

Name: KARL SIGERIST Title: DIRECTOR

Per:

Name:

Title:

I/We have the authority to bind the corporation

[Signature page of guarantee of Crelogix Portfolio Services Corporation]

20077150.1

۰**۰**,

This is **Exhibit "K"** referred to in the Affidavit of Darcy Peelar, sworn before me this 5<sup>th</sup> day of July, 2017

A Commissioner for Oaths in and for Alberta

Patrick T. Harnett Barrister and Solicitor

25337254.1

# GENERAL SECURITY AGREEMENT

**∇5C 6A7** 

	Community	_
	Credit Union Ltd. (the "Credit Union")	
	Parkland Square Branch	
	4901 48th Street	
	Red Deer	, Alberta
. •	T4N 6N4	, Branch
	Member No(s).:	

1. DEFINITIONS

Borrower(s):

All capitalized terms used in this agreement ("Agreement") including any schedules ("Schedules") annexed hereto shall, except where (a)defined herein, be interpreted pursuant to their respective meanings when used in the Personal Property Security Act of Alberta in force at the date of this Agreement ("PPSA").

- (b) In this Agreement:
  - Account Debtor" means a debtor of the Debtor on an Intangible, Chattel Paper or Account, or any obligor of the Debtor on an (1) Instrument:
  - "Agreed Rate" means the rate of interest payable under the document(s) evidencing the indebtedness and in the event such document(s) bear different rates of interest the "Agreed Rate" shall mean the highest of such interest rates; (ii)
  - "Consumer Goods" means those goods that are used or acquired by the Debtor for use primarily for his personal, family or (ii)) household purposes:
  - Debtor" means the "Borrower(s)"; (iv)

Travelers Acceptance Corporation Suite 500, 4180 Lougheed Highway

Burnaby, British Columbia

- "Encumbrances" means any Security Interest, mortgages, liens claims, charges and other encumbrances affecting the Collateral including Permitted Encumbrances but excluding the Security Interest created hereby; "Permitted Encumbrances" means any Encumbrances which are described in Schedule "C" and any others approved in writing by (v)
- (vi) the Credit Union prior to their creation or assumption; and
- (vii) "Receiver" includes a Receiver-Manager.

2. SECURITY INTEREST

- For value received, the Debtor hereby grants to the Credit Union, by way of mortgage, charge and assignment, a Security Interest in (a) For value received, the behavior hereby grants to the creat Union, by way or mortgage, charge and assignment, a Security Interest in the undertaking of the Debtor and in all Goods (including all parts, accessories, attachments, special tools, additions and Accessions thereto). Chaltel Paper, Documents of Title (whether negotiable or not), Instruments, Intangibles, Money, Securities and other property now or hereafter owned or acquired by or on behalf of the Debtor (including such as may be returned to or repossessed by the Debtor) and in all Proceeds thereof, (herein collectively called the "Collateral"), including, without limitation, all of the following property now or hereafter owned or acquired by or on behalf of the Debtor:
  - all property described in Schedule "A"; 0

  - (ii) all inventory;
     (iii) all Equipment, Including, without limitation, all machinery, tools, apparatus, plant, furniture, Fixtures and vehicles of whatsoever nature or kind;
  - all Accounts, including, without limitation, all book accounts and book debts and generally all accounts, debts, dues, claims, (iv) choses in action, judgments and demands of every nature and kind howsoever arising or secured including letters of credit and advices of credit, which are now due, owing or accruing due to or owned by or which may hereafter become due, owing or accruing due to or owned by the Debtor;
  - all deeds, documents, writings, papers and books of account and other books relating to or being records of Accounts, Chattel Paper, Instruments or Documents of Title or by which such are or may hereafter be secured, evidenced, acknowledged or made (v) payable;
  - all contractual rights and insurance claims and all goodwill, patents, trademarks, copyrights, and other intellectual property; and (vi)
  - (vii) all crops and livestock including all crops that are or which hereafter become crops on any real property described in Schedule

 (b) Any reference to "Collateral" shall, unless its context otherwise requires, be deemed a reference to "Collateral or any part thereof"
 (c) The Security Interest granted heraby shall not extend or apply to and the Collateral shall not include the last day of the terms of any lease or agreement therefore but upon the enforcement of the Security Interest the Debtor shall stand possessed of such last day in trust to assign the same to any person acquiring such term.

\* (d) The Security interest granted hereby shall not extend or apply to and the Collateral shall not include the Debtor's Consumer Goods except for any described in Schedule "A".

3. INDEBTEDNESS SECURED

The Security Interest granted hereby secures payment of any and all obligations, indebtedness and liabilities of the Debtor to the Credit Union whether present or future, direct or indirect, absolute or contingent, matured or not, extended or renewed, wheresoever and howsoever incurred and any ultimate unpaid balance thereof and whether the same is from time to time reduced and thereafter increased or entirely extinguished and thereafter incurred again and whether the Debtor is bound alone or with another or others and whether as principal or surety (hereinafter collectively called the "indebtedness").

## 4. DESTOR'S REPRESENTATIONS AND WARRANTIES

The Debtor represents and warrants and so long as this Agreement remains in effect continuously represents and warrants that:

the Collateral is genuine and owned by the Debtor free of all Encumbrances except Permitted Encumbrances; each Account, Chattel Paper and Instrument constituting Collateral is enforceable in accordance with its terms against the party (b) obligated to pay the same and the amount represented by the Debtor to the Credit Union from time to time as owing by each Account Debtor is the correct amount actually and unconditionally owing by such Account Debtor,

(c) there is no litigation, proceeding or dispute pending or to the knowledge of the Debtor threatened against or affecting the Debtor or the Collateral, the adverse determination of which might materially and adversely affect the Debtor's financial condition or impair the Debtor's ability to perform its obligations hereunder,

\* Delete and initial if inapolicable.

10112 ND (05/2001 ST)

-

REV. (09/90) B.A.S.

 (d) the name(s) of the Debtor is(are) accurately and fully set out above, and the Debtor is not known by any other name(s); and
 (e) there is no provision in any agreement to which the Debtor is a party, nor to the knowledge of the Debtor is there any statute, rule or regulation, or any judgment, decree or order of any court binding on the Debtor, which would be contravened by the execution and delivery of ils Aareement.

## 5. DEBTOR'S COVENANTS

The Debtor covenants and agrees:

(a) to defend the Collateral (except Collateral dealt with as permitted by clause 7 hereof) against the claims and demands of all other (b) to keep the Collateral free from all Encumbrances except Permitted Encumbrances;

(c) subject to clause 7 hereof, not to sell, exchange, transfer, assign, lease or otherwise dispose of Collateral or any Interest therein without the prior written consent of the Credit Union;
 (d) to notify the Credit Union promptly of:

- (1) any change in the information contained herein or in the Schedules relating to the Debtor, the Debtor's name, the Debtor's business or Collateral:
- the details of any significant acquisition of Collateral;
- (iii) the details of any claims or litigation affecting the Debtor or Collateral; (iv) any loss or damage to Collateral;
- (v) any default by any Account Debtor in payment or other performance of his obligations with respect to Collateral; and
   (vi) the return to or repossession by the Debtor of Collateral;

(e) to keep the Collateral in good order, condition and repair and not to use Collateral in violation of the provisions of this Agreement or any other agreement relating to the Collateral or any policy insuring the Collateral or any applicable statute, law, bylaw, rule, regulation or ordinance:

(f) to pay all taxes, rates, levies, assessments and other charges of every nature which may be lawfully levied, assessed or imposed against or in respect of the Debtor or Collateral as and when the same become due and payable; (g) to punctually make all payments and perform all obligations in any lease by the Debtor and under any agreement charging property of

the Debtor,

(h) to prevent Collateral from being or becoming a Fixture or an Accession to other property that is not Collateral;
 (i) to carry on and conduct the business of the Debtor in a proper and efficient manner so as to protect and preserve the Collateral and to keep, in accordance with generally accepted accounting principles, consistently applied, proper books of account for the Debtor's business as well as accurate and complete records concerning the Collateral;
 (ii) to diverse the Core in the Content of the test of the preserve the Collateral;

(i) to deliver to the Credit Union from time to time promptly upon request;

- any Documents of Title, Instruments, Securities and Chattel Paper constituting, representing or relating to the Collateral;
- (ii) all books of account and all records, ledgers, reports, correspondence, schedules, documents, statements, lists and other writings relating to the Collateral for the purpose of inspecting, auditing or copying the same;
   (iii) all financial statements prepared by or for the Debtor regarding the Debtor's business;
   (iv) all policies and certificates of inspectively. In the Debtor have business and effektive and the Certificates are relating to the Collateral; and

- such information concerning the Collateral, the Debtor and the Debtor's business and affairs as the Credit Union may reasonably (v) request;
- (k) not to remove any of the Collateral from the Province of Alberta without the prior written consent of the Credit Union;
   (i) In the event the value of the Collateral shall be materially reduced, to Immediately reduce the amount of the Indebtedness by an amount determined by the Credit Union;
   (m) if the Collateral includes crops and livestock, in addition to the Debtor's other obligations regarding Collateral:

- to do all acts which may be necessary to attend to, care for, raise and fatten the livestock and to grow, cultivate, spray, irrigate, (ī) cut, harvest, pick, clean, preserve and protect the crops, all according to the most approved methods of farming husbandry and to keep the farm(s) on which the Collateral is located free of noxious weeds and grasses, and maintain the present buildings and
- (ii) to pay, when due, all obligations incurred for labour or material or otherwise in the care or feeding or shearing of such livestock;
- and
- (iv) at the request of the Credit Union, to deliver to the Credit Union the Debtor's Canadian Wheat Board producer's permit book and to assign to the Credit Union all of the Debtor's rights thereunder, and

(n) to permit the Credit Union, by its officers or authorized agents, at any time, and from time to time, as often as the Credit Union in its sole discretion may determine, to enter the premises owned or occupied by the Debtor for the purpose of inspecting the Collateral and the operation of the Debtor's business.

## 6. INSURANCE

The Debtor shall insure and keep insured against loss or damage by fire or other insurable hazards the Collateral to the extent of its full insurable value, and shall maintain such other insurance as the Credit Union may reasonably require. The loss under the policies of insurance shall be made payable to the Credit Union as its interest may appear and the insurance shall be written by an insurance company approved by the Credit Union in terms satisfactory to the Credit Union and the Debtor shall provide the Credit Union with copies of the same. The Debtor shall premiums and other sums of money necessary for such insurance steps become vitlated. Upon the happening of any loss or damage the Debtor shall furnish at its expense all necessary proofs and shall do all necessary acts to enable the Credit Union to obtain payment of the insurance of the same. insurance monies.

## 7. DEALING WITH COLLATERAL

The Debtor shall not sell, exchange, transfer, assign, lease or otherwise dispose of that Collateral described in Schedule "A" except with the prior written consent of the Credit Union which consent may be arbitrarily withheld. Until but not after Default the Debtor may deal with Collateral, other than that Collateral described in Schedule "A", in the ordinary course of the Debtor's business in any manner not inconsistent with the provisions of this Agreement, provided that the Debtor may only sell, exchange, transfer, assign, lease or otherwise dispose of such Collateral for fair value on commercially reasonable terms and provided that all cash Proceeds therefrom are immediately deposited with the Credit Linion

#### COLLATERAL IN POSSESSION OF CREDIT UNION. RECEIVER OR SHERIFF 8.

a the second as in

If Collateral is at any time in the possession of the Credit Union, a Receiver or Sheriff, the Credit Union, Receiver or Sheriff in possession, as the case may be:

(a) shall not be required to take any steps to preserve any rights against other parties to any Chattel Paper, Security, or Instrument constituting Collateral;

(b) shall not be required to keep the Collateral identifiable; and

(c) may use the Collateral in any manner and to any extent the Credit Union in its sole discretion, deems advisable.

-----

## 9. SECURITIES

If the Collateral at any time includes Securities, the Debtor authorizes the Credit Union to transfer the same or any part thereof into its own name or that of its nomine(s) so that the Credit Union or its nomine(s) may appear of record as the sole owner thereof; provided that, until Default, the Credit Union shall promptly deliver to the Debtor all notices or other communications received by it or its nomine(s) as such registered owner and, upon demand and receipt of payment of any necessary expenses thereof, shall issue to the Debtor or its order a proxy to vote and take all action with respect to such Securities. After Default, the Debtor walves all rights to receive any notices or communications received by the Credit Union or its nominee(s) as such registered owner and agrees that no proxy issued by the Credit Union to the Debtor or its order as aforesaid shall thereafter be effective.

## 10. COLLECTION FROM ACCOUNT DEBTORS

Before or after Default, the Credit Union may notify any Account Debtor of this Security interest and may direct such Account Debtor to make all payments to the Credit Union. The Debtor acknowledges that any payments on or other Proceeds of the Collateral received by the Debtor from any Account Debtor, whether before or after notice of this Security interest is given to such Account Debtor and whether before or after Default, shall be received and held by the Debtor in trust for the Credit Union and shall be turned over to the Credit Union upon request. The Debtor agrees that it will not commingle any Proceeds of or payments on the Collateral with any of the Debtor's funds or property, but will hold them separate and apart.

## 11. OTHER TERMS

This Agreement includes the terms, if any, which are contained in Schedule "D".

## 12. APPLICATION OF MONIES

All Monies collected or received by the Credit Union pursuant to or in exercise of any right it possesses with respect to the Collateral shall be applied on account of the Indebtedness in such manner as the Credit Union may in its sole discretion determine or, at the option of the Credit Union, may be held unappropriated in a collateral account or released to the Debtor, all without prejudice to the liability of the Debtor or the rights of the Credit Union hereunder, and any surplus shall be accounted for as required by law.

## 13. DEFAULT

The happening of any of the following events shall constitute default ("Default") hereunder:

nonpayment when due, whether by acceleration, demand or otherwise, of any amount forming part of the indebtedness; (a)

failure of the Debtor to observe or perform any term contained in this Agreement or in any other agreement between the Debtor and the Credit Union;

(c) declaration of incompetency by a court of competent jurisdiction or death of an individual Debtor or an individual partner of a partnership Debtor:

(d) bankruptcy or insolvency of the Debtor; filing against the Debtor of a petition in bankruptcy; making of an assignment for the benefit of creditors by the Debtor; appointment of a Receiver or trustee for the Debtor of rany property of the Debtor or institution by or against the Debtor of any proposal, plan of arrangement or other type of insolvency proceeding under the Bankruptcy Act or otherwise;
 (e) institution by or against the Debtor of any formal or informal proceeding for the dissolution or liquidation of, settlement of claims against the Debtor.

against, or winding up of the affairs of the Debtor; (f) any of the Encumbrances becomes enforceable;

(g) the Debtor ceases or threatens to cease to carry on business or makes or agrees to make a bulk sale of assets or commits or threatens to commit an act of bankruptcy; (h) any execution, sequestration, extent or other process of any court becomes enforceable against the Debtor or If a distress or

 (h) Bity execution, sequestration, extent or other process or any part thereof;
 (i) any certificate, statement, representation, warranty or audit report heretofore or hereafter furnished by or on behalf of the Debtor pursuant to or in connection with this Agreement, or otherwise (including, without limitation, the representations and warrantles contained herein) or as an inducement to the Credit Union to extend any credit to or to enter into this or any other agreement with the Debtor, is faise in any material respect at the time as of which the facts therein set forth were stated or certified, or omits any substantial contingent or investigations and warrantles behaviors. uniquidated liability or claim against the Debor; or if upon the date of execution of this Agreement, there shall have been any material adverse change in any of the facts disclosed by any such certificate, representation, statement, warranty or audit report, which change shall not have been disclosed to the Credit Union at or prior to the time of such execution;

(i) any of the licenses, permits or approvals granted by any government or any governmental authority and essential to the business of the Debtor is withdrawn, cancelled or significantly altered;

 (k) at any time, there is a material adverse change in the financial condition of the Debtor; or
 (l) the Credit Union considers that it is insecure, or that the prospect of payment or performance by the Debtor of the Indebtedness is or is about to be impaired, or that the Collateral is or is about to be placed in jeopardy.

## 14. ACCELERATION

In the event of Default, the Credit Union, in its sole discretion, may without demand or notice of any kind, declare all or any of the Indebtedness which is not by its terms payable on demand, to be immediately due and payabla. This clause does not apply to or affect any of the indebtedness payable on demand.

## 15. REMEDIES

## On Default:

 (a) the Credit Union may enforce this Agreement by any method provided for in this Agreement, in the PPSA or as otherwise permitted by law or in equity, and without limitation, may dispose of Collateral by lease or deferred payment;
 (b) the Credit Union may selze or otherwise take possession of the Collateral or any part thereof and sell the same by public or private sale at such price and upon such terms as the Credit Union in its sole discretion may determine and the proceeds of such sale less all costs, charges and expenses of the Credit Union (including costs as between a solicitor and his own client on a full indemnity basis) shall be applied on the indebtedness and the surplus, if any, shall be disposed of according to law; (c) the Credit Union may take proceedings in any court of competent jurisdiction for the appointment of a Receiver;

(c) the Credit Union may care proceedings in any court of comparisin persons to the opposition of a court of

## (e) unless otherwise restricted by his appointment, any Receiver shall have the power,

to take possession of any Collateral and for that purpose to take any proceedings, in the name of the Debtor or otherwise; íi)

- to carry on or concur in carrying on the business of the Debtor; (ii)
- àb to sell or lease any Collateral;
- (iii) to sell or lease any Collateral;
  (iv) to make any arrangement or compromise which he may think expedient in the interest of the Credit Union;
  (v) to pay all itabilities and expenses connected with the Collateral, including the cost of Insurance and payment of taxes or other costs, charges or expenses Incurred in obtaining, maintaining possession of any preserving the Collateral, and the same shall be added to the Indebtedness and secured by the Collateral;
  (vi) to hold as additional security any increase or profits resulting from the Collateral;
  (vii) to exercise all rights and remedies that the Credit Union may have under this Agreement, the PPSA or otherwise at law or in condition.
- equity:

#### 10112 ND (06/2000 JB)

## Page 3 of 6

(viii) with the written consent of the Credit Union, to borrow money for the purpose of carrying on the business of the Debtor or for maintenance of the Collateral or any part thereof or for other purposes approved by the Credit Union, and any amount so borrowed together with interest thereon shall form a charge upon the Collateral in priority to the Security Interest created by this Agreement: and

(ix) to do any other act or thing as may be considered to be incidental or conducive to any of the matters and powers aforesaid;

the Debtor hereby appoints each Receiver appointed by the Credit Union to be its attorney to effect a sale or lease of any Collateral (f)

and any deed, lease, agreement or other document signed by a Receiver pursuant to this power of atomey shall have the same effect as if it had been executed by and under the seal of the Debtor;
 (g) a Receiver appointed by the Credit Union shall be deemed to be the agent of the Debtor, and the Debtor shall be solely responsible for his acts or defaults and for his remuneration and expenses, and the Credit Union shall not be in any way responsible for any misconduct or

his acts or defaults and for his remuneration and expenses, and the Credit Union shall not be in any way responsible for any misconduct or negligence on the part of any Receiver; (h) all monies received by the Receiver after providing for payment of all costs, charges and expenses of or incidental to the exercise of any of the powers of the Receiver shall be paid to the Credit Union and applied on account of the indebtedness; (i) the Credit Union may enter upon, use and occupy all premises owned or occupied by the Debtor wherein the Collateral may be situate; (j) before, during or after realizing on the Collateral, the Credit Union may recover and enforce judgment against the Debtor for the indebtedness and all costs, charges and expenses reasonably incurred by the Credit Union (including, without limitation, costs as between a solicitor and his own client on a full indemnity basis) in recovering or enforcing judgment against the Debtor; and

(k) the Credit Union may, but shall not be bound to, realize on the Collateral.

## 16. DEFICIENCY

If the Credit Union realizes on the Collateral and the realization is not sufficient to satisfy all the Indebtedness, the Debtor acknowledges and agrees that the Debtor shall continue to be liable for any indebtedness remaining outstanding and the Credit Union shall be entitled to pursue full payment thereof.

#### 17. COSTS AND EXPENSES

(a) Upon the Debtor's failure to perform any of its obligations under this Agreement then the Credit Union may, but shall not be obligated to perform the same and in the event of performance thereof by the Credit Union the Debtor shall pay to the Credit Union forthwith upon written demand therefor an amount equal to all costs, charges and expenses incurred by the Credit Union in performing the Debtor's obligations plus interest thereon at the Agreed Rate from the date such costs, charges and expenses are incurred by the Credit Union until paid by the Debtor's plus interest. Debtor.

(b) The Debtor shall pay all costs, charges and expenses reasonably incurred by the Credit Union or any Receiver appointed by it (including, but without restricting the generality of the foregoing, costs as between a solicitor and his own client on a full indemnity basis), in preparing, registering financing statements regarding or enforcing this Agreement, inspecting taking custody of, preserving, repairing, maintaining, processing, preparing for disposition and disposing of Collateral and in enforcing or collecting the indebtedness and all such costs, charges and expenses together with any monies owing as a result of any borrowing by any Receiver appointed by the Credit Union shall be a first charge on the proceeds of realization, collection or disposition of Collateral and shall be secured hereby. Such costs, charges and expenses shall bear interest at the Agreed Rate from the date the same were incurred to the date of payment by the Debtor.

(c) The Credit Union may pay or satisfy any Encumbrances or pay any sum necessary to clear title to any Collateral, and the Debtor agrees to repay the same on demand, plus interest thereon at the Agreed Rate.

All amounts paid by the Credit Union pursuant to this clause together with interest thereon at the Agreed Rate shall form part of the Indebtedness and be secured hereby.

## 18. SET OFF

Without limiting any other right the Credit Union may have, the Credit Union may, in its sole discretion at any lime and without notice, set off any and all amounts owed to the Debtor by the Credit Union in any capacity and, whether or not due, against any and all indebtedness including any contingent or non-matured indebtedness and indebtedness as principal or guarantor.

#### 19. FURTHER ASSURANCES

The Debtor agrees to execute and deliver to the Credit Union such further assurances, conveyances and supplemental deeds as may be necessary to properly carry out the intention of this Agreement, as determined by the Credit Union, or as may be required by the Credit Union from time to time.

#### 20. NOTICE

Any notice or demand required or permitted to be made or given by the Credit Union to the Debtor may be validly served by leaving the same with, or by mailing the same by prepaid registered mail to, the Debtor at his address as set out herein (or at such other address as the Debtor may in writing notify the Credit Union of as the Debtor's address for service under his Agreement) or by leaving such notice with any officer or director of the Debtor as shown on the records of the Credit Union, and in the case of mailing such notice or demand shall be deemed to have been received by the Debtor on the third business day following the date of malling.

#### 21. GENERAL

(a) The Credit Union may grant extensions of time and other indulgences, take and give up security, accept compositions, compound, compromise, settle, grant releases and discharges and otherwise deal with the Debtor, Account Debtors, surges and otherwise deal with the Debtor, Account Debtors, surges and others and with Collateral and other security as the Credit Union may see fit without prejudice to the liability of the Debtor or the Credit Union's right to hold and realize upon the Security Interest. The Credit Union may demand, collect and sue on Collateral in either the Debtor's or the Credit Union's name and may endorse the Debtor's name on any and all cheques, commercial paper, and any other instruments pertaining to or constituting Collateral.

may endorse the Debtor's name on any and all cheques, commercial paper, and any other instruments pertaining to or constituting Collateral.
The Credit Union shall not be liable for any failure to exercise its remedies, take possession of, collect, enforce, realize, sell, lease or otherwise dispose of Collateral or to initiate any proceedings for such purposes.
(b) No delay or omission by the Credit Union in exercising any right or remedy hereunder or with respect to any indebtedness shall operate as a waiver thereof or of any other right or remedy, and no single or partial exercise thereof shall preclude any other or further exercise thereof or the exercise of any other right or remedy, and no single or partial exercise thereof shall preclude any other or further exercise thereof or the exercise of any other right or remedy. The Credit Union may remedy any Default by the Debtor hereunder in any manner without waiving the Default remedied and without waiving any other prior or subsequent Default by the Debtor. All rights and remedies of the Credit Union granted or recognized herein are cumulative and may be exercised at any time and from time to time independently or in combination.
(c) The Debtor waives protest of any instrument constituting Collateral at any time held by the Credit Union in which the Debtor Is in any way liable and notice of any other by the tore the prior.

(d) This Agreement shall enure to the benefit of and be binding upon the parties hereto and their respective heirs, executors, administrators, successors and assigns. In any action brought by an assignee of this Agreement and the Security Interest or any part thereof to enforce any rights hereunder, the Debtor shall not assert against the assignee any claim or defence which the Debtor now has or hereafter may

have against the Credit Union.

(e) If more than one Debtor executes this Agreement the obligations of such Debtors hereunder shall be joint and several.

(f) No modification, variation or amendment of any term of this Agreement shall be bloding or effective unless made by written agreement, executed by the parties hereto and no waiver of any term hereof shall be binding or effective unless in writing.

(g) This Agreement Is in addition to and not in substitution for any other agreements, securities or Security interests now or hereafter held by the Credit Union and all such other agreements, securities and Security Interests shall remain in full force and effect.

standard and a standard to see the

(h) The headings used in this Agreement are for convenience only and are not to be considered part of this Agreement and do not in any way limit, explain or amplify the terms of this Agreement.
 (i) When the context so requires, the singular shall be read as if the plural were expressed and vice versa and the terms hereof shall be ad with all grammatical changes necessary dependant upon the person referred to being a male, female, partnership or corporation.

(i) In the event any terms of this Agreement, as amended from time to time, shall be deemed invalid, void or unenforceable, in whole or in part, by any court of competent jurisdiction, the remaining terms of this Agreement shall remain in full force and effect.
 (k) Nothing herein contained shall in any way obligate the Credit Union to grant, continue, renew or extend time for payment of the total to the shall be deemed invalid.

Indebtedness

## 22. ATTACHMENT

(a) Subject to subclause (b), the Security Interest created hereby is intended to attach the Collateral when this Agreement is executed by the Debtor and delivered to the Credit Union. (b) With respect to that Collateral acquired by the Debtor after the date this Agreement is executed and delivered to the Credit Union, the

Security Interest created hereby in such Collateral is intended to attach at the same time as the Debtor acquires rights in such Collateral.

## 23. DISCLOSING INFORMATION

This Agreement and any information pertaining thereto or to the indebtedness may be disclosed by the Credit Union as required by the PPSA.

## 24. GOVERNING LAW AND JURISDICTION

This Agreement shall be interpreted in accordance with the laws of the Province of Alberta, and the Debtor irrevocably agrees that any suit Alberta or in any court of competent jurisdiction, as the Credit Union may elect, and the Debtor agrees to attorn to the same.

## 25. COPY OF AGREEMENT

Debtor acknowledges receipt of a copy of this Agreement and waives any right it may have to receive a Financing Statement or Financing Change Statement relating to it.

IN WITNESS WHEREOF the	Debtor has hereunto executed this / <del>Alberta:</del>	Agreement this <u>31</u> day of	Mush 2008	ət,
------------------------	--	---------------------------------	-----------	-----

• IF DEBTOR IS AN INDIVIDUAL: Signature of Debtor(s)

\* IF DEBTOR IS A PARTNERSHIP:

(Partner)

By:

Print Exact Name of Partnership or Trade Name

. 1

Witness		
Witness	 	
Witness	 	
Wilness	 	

Witness	 	
Witness	 ÷	 
Witness	 	
Witness		

By:	(Partner)	
* IF DE	BTOR IS A CORPORATION:	
<u>Tr</u> Print E By:	avelers Acceptance Corporation xact Name of Corporation	
By:	(Officer of Corporation	
FULL ADDRESS OF DEBTOR		

de la respecta a companya a r

\* Complete Affidavit of Execution if Debtor is an individual or partnership.

. . . . .

# SCHEDULE "A" (DESCRIBED PROPERTY)

\* Obtain serial numbers for all motors vehicles, trailers, mobile homes, farm machinery, equipment and airplanes

All present and after acquired property and all proceeds thereof

10112 ND (06/2000 JB)

Page 5 of 6

# SCHEDULE "B" (DESCRIBED REAL PROPERTY) \* Obtain legal description

# SCHEDULE "C" (PERMITTED ENCUMBRANCES AFFECTING COLLATERAL)

# SCHEDULE "D" (OTHER TERMS AND CONDITIONS)

10112 (06/2000 JB)

Page 6 of 6

# This is **Exhibit "L"** referred to in the Affidavit of Darcy Peelar, sworn before me this 5<sup>th</sup> day of July, 2017

A Commissioner for Oaths in and for Alberta

Patrick T. Harnett Barrister and Solicitor

25337254.1

GENERAL	SECURITY
AGREEME	NT

GENERAL SECURITY	SERVUS		
AGREEMENT	Credit Union Ltd. (the "Credit Union")		
	#601-4901-48TH STREET		
	RED DEER, ALBERTA		
orrower(s): CRELOGIX CREDIT GROUP INC.	T4N 6N4	, Alberta	
SUITE 500, 4180 LOUGHEED HIGHWAY	PARKLAND SQUARE	.Branch	
BURNABY, BRITISH COLUMBIA	Member No(s <u>).;</u>		
V5C 6A7			

## 1. DEFINITIONS

Borrower(s):

(a) All capitalized terms used in this agreement ("Agreement") including any schedules ("Schedules") ennexed hereto shall, except where defined herein, be interpreted pursuant to their respective meanings when used in the Personal Property Security Act of Alberta in force at the date of this Agreement ("PFSA").

(b) In this Agreement:

- "Account Debtor" means a debtor of the Debtor on an Intangible, Chattel Paper or Account, or any obligor of the Debtor on an (1) Instrument:
- "Agreed Rate" means the rate of interest payable under the document(s) evidencing the indebtedness and in the event such document(s) bear different rates of interest the "Agreed Rate" shall mean the highest of such interest rates; "Consumer Gooda" means those goods that are used or acquired by the Debtor for use primarily for his personal, family or (ii)
- (iii) household purposes;
- "Debtor" means the "Borrower(s)"; (iv)
- "Encumbrances" means any Security Interest, mortgages, liens claims, charges and other encumbrances affecting the Colleteral Including Permitted Encumbrances but excluding the Security Interest created hereby;
- "Permitted Encumbrances may be unbrances which are described in Schedule "C" and any others approved in writing by the Credit Union prior to their creation or assumption; and (vi)
- (vii) "Receiver" includos e Receiver-Manager.

## 2. SECURITY INTEREST

- For value received, the Debtor hereby grants to the Credit Union, by way of montgage, charge and assignment, a Security interest in the undertaking of the Debtor and in all Goods (including all parts, accessories, attachments, special tools, additions and Accessions thereto), Chettel Paper, Documents of Title (whether nagotiable or not), instruments, intangibles, Money, Securities and other property now or hereafter owned or acquired by or on behalf of the Debtor (including such as may be returned to or repossessed by the Debtor) and in all Proceeds thereof, (herein collectively called the "Colleteral"), including, without limitation, all of the following property now or hereafter owned or acquired by or on behalf of the Debtor: (a)
  - all property described in Schedula "A"; (i)
  - all Inventory;
  - ài all Equipment, including, without limitation, all machinery, tools, apparatus, plant, furniture, Fixtures and vehicles of whatsoever nature or kind:
  - (iv) all Accounts, including, without limitation, all book accounts and book debts and generally all accounts, debts, dues, claims, choses in action, locating, many and demands of every nature and kind howsever arising or secured including letters of credit and advices of credit, which are now due, owing or accruing due to or owned by or which may hereafter become due, owing or accruing due to or owned by the Debtor,
  - all deeds, documents, writings, papers and bocks of account and other books relating to or being records of Accounts, Chattel Paper, instruments or Documents of Title or by which such are or may hereafter be secured, evidenced, acknowledged or made payable
  - all contractual rights and insurance claims and all goodwill, patents, trademarks, copyrights, and other intellectual property; and (vi)
  - (vii) all crops and livestock including all crops that are or which hereafter become crops on any real property described in Schedule

(b) Any reference to "Collateral" shall, unless its context otherwise requires, be deemed a reference to "Collateral or any part thereof" (c) The Security Interest granted hereby shall not extend or apply to and the Collateral shall not include the last day of the terms of any or agreement therefore but upon the enforcement of the Security Interest the Debtor shall stand possessed of such last day in trust to assign the same to any person acquiring such term.

\*(d) The Security Interest granted hereby shall not extend or apply to and the Collateral shall not include the Debtor's Consumer Goods except for any described in Schedule "A".

#### 3. INDERTEDNESS SECURED

The Security Interest granted hereby secures payment of any and ell obligations, indebtedness and liabilities of the Debtor to the Credit Union whether present or future, direct or indirect, absolute or contingent, matured or not, extended or renewed, wheresoaver and howsoaver incurred and any ultimate unpaid balance thereof and whether the same is from time to time reduced and thereafter incurred agen and whether the Debtor is bound alone or with another or others and whether as principal or surety (hereinafter collectively called the "indebtedness").

#### 4. DEBTOR'S REPRESENTATIONS AND WARRANTIES

The Debtor represents and warrants and so long as this Agreement remains in effect continuously represents and warrants that:

the Collateral is genuine and owned by the Debtor free of all Encumbrances except Permitted Encumbrances; each Account, Chattel Paper and instrument constituting Collateral is enforceable in accordance with its terms against the party íh obligated to pay the same and the amount represented by the Debtor to the Cradit Union from time to time as owing by each Account Debtor is the correct amount actually and unconditionally owing by such Account Debtor;

 (c) there is no litigation, proceeding or dispute pending or to the knowledge of the Debtor threatened against or affecting the Debtor or the Collateral, the adverse determination of which might materially and adversely affect the Debtor's financial condition or impair the Debtor's ability to perform its obligations hereunder;

\* Delete and Initial if inapplicable.

10112 ND (06/2001 ST)

REV. (09/90) B.A.6.

(d) the name(s) of the Debtor is(are) accurately and fully set out above, and the Debtor is not known by any other name(s); and (e) there is no provision in any agreement to which the Debtor is a party, nor to the knowledge of the Debtor is there any statute, rule or regulation, or any judgment, decree or order of any court binding on the Debtor, which would be contravened by the execution and delivery of this Agreement.

## δ. DEBTOR'S COVENANTS

The Debtor covenants and agrees:

 (b) to keep the Collateral (except Collateral dealt with as permitted by clause 7 h parties claiming the same or an interest therein;
 (b) to keep the Collateral free from all Encumbrances except Permitted Encumbrances;
 (c) subject to clause 7 hereof, not to sail, exchange transfer except restored in the same transfer except restored in the (a) to defend the Collateral (except Collateral dealt with as permitted by clause 7 hereof) against the claims and demande of all other

(c) subject to clause 7 hereof, not to sell, exchange, transfer, essign, lease or otherwise dispose of Collateral or any interest therein without the prior written consent of the Credit Union;

(d) to notify the Credit Union promptly of:

- any change in the information contained herein or in the Schedules relating to the Debtor, the Debtor's name, the Debtor's (i) business or Collataral:
- (ii) the details of any significant acquisition of Collateral; (iii) the details of any claims or litigation affecting the Debtor or Collateral;
- (iv) any loss or damage to Collateral;
- any default by any Account Debtor in payment or other performance of his obligations with respect to Collateral; and
- (vi) the return to or repossession by the Debtor of Collateral;

(e) to keep the Collatoral in good order, condition and repair and not to use Collateral in violation of the provisions of this Agreement or any other agreement relating to the Collateral or any policy insuring the Collateral or any applicable statute, law, bylaw, rule, regulation or ordinance

to pay all taxes, rates, levies, assessments and other charges of every nature which may be lawfully levied, assessed or imposed (f) against or in respect of the Debtor or Collateral as and when the same become due and payable; (g) to punctually make all payments and perform all obligations in any lease by the Debtor and under any agreement charging property of

the Debtor;

(h) to prevent Collateral from being or becoming a Fixture or an Accession to other property that is not Collateral;
 (i) to carry on and conduct the business of the Debtor in a proper and efficient manner so as to protect and preserve tha Collateral and to keep, in accordance with generally accepted accounting principles, consistently applied, proper books of account for the Debtor's business as well as accurate and completa records concerning tha Collateral;
 (j) to deliver to the Credit Union from time to time promptly upon request;

- any Documents of Title, Instruments, Securities and Chattel Paper constituting, representing or relating to the Collateral;
- all books of account end all records, ledgers, reports, correspondence, schedules, documents, statements, lists and other writings relating to the Collateral for the purpose of inspecting, auditing or copying the same;
- (iii) all financial statements prepared by or for the Debtor regarding the Debtor's business;
   (iv) all policies and certificates of insurance relating to the Collateral; and
- such information concerning the Collateral, the Debtor and the Debtor's business and alfairs as the Credit Union may reasonably (v) request;
- (k) not to remove any of the Collateral from the Province of Alberta without the prior written consent of the Credit Union;
   (i) in the event the value of the Collateral shall be materially reduced, to immediately reduce the amount of the indebtedness by an

amount determined by the Credit Union; (m) if the Collateral includes crops and livestock, in addition to the Debtor's other obligations regarding Collateral:

- to do all acts which may be necessary to attend to, care for, raise and fatten the livestock and to grow, cultivate, spray, irrigate, cut, harvest, pick, clean, preserve and protect the crops, all according to the most approved methods of farming husbandry and to keep the farm(s) on which the Collateral is located free of noxious weeds and grasses, and maintain tho present buildings and improvements on the said farm(s) in good condition and repeir;
- (ii) to provide suitable range, pasture and feed for all ilvestock and care for end protect them from disease, damage, injury, deeth, destruction by weether, wild enimals, theft or other cause; (iii) to pay, when due, all obligations incurred for labour or material or otherwise in the care or feeding or shearing of such livestock;
- and
- (iv) at the request of the Credit Union, to deliver to the Credit Union the Debtor's Canadian Wheat Board producer's permit book and to assign to the Credit Union all of the Debtor's rights thereunder, and

(n) to permit the Credit Union, by its officers or authorized egents, at any time, and from time to time, as often as the Credit Union in its sole discretion may determine, to enter the premises owned or occupied by the Debtor for the purpose of inspecting the Collateral and the operation of the Debtor's business.

## 6. INSURANCE

The Debtor shall insure and keep insured against loss or damage by fire or other insurable hazards the Collateral to the extent of its full insureble value, and shall maintain such other insurance es the Credit Union may reasonably require. The loss under the policies of insurance shall be made payeble to the Credit Union as its interest may appear and the insurance shall be written by an insurance company approved by the Credit Union in terms satisfactory to the Credit Union and the Debtor shall provide the Credit Union with copies of the seme. The Debtor shell pay all premiums and other sums of money necessary for such insurance as they become due and deliver to the Credit Union proof of said payment, and shall not allow anything to be done by which the policies may become villated. Upon the happening of any loss or demage the Debtor shall formish at its expense all necessary proofs and ahall do all necessary acts to enable the Credit Union to obtain payment of the insurance monies.

#### 7. DEALING WITH COLLATERAL

The Debtor shall not sell, exchange, transfer, assign, lease or otherwise dispose of that Collateral described in Schedule "A" except with the provisions of the Credit Union which consent may be arbitrarily withheid. Until but not after Default the Debtor may deal with Collateral, other than that Collateral described in Schedule "A", in the ordinary course of the Debtor's business in any manner not inconsistent with the provisions of this Agreement, provided that the Debtor may only sell, exchange, transfer, assign, lease or otherwise dispose of such Collateral for fair value on commercially reasonable terms and provided that all cash Proceeds therefrom are immediately deposited with the Credit Union

#### 8. COLLATERAL IN POSSESSION OF CREDIT UNION, RECEIVER OR SHERIFF

If Colleteral is at any time in the possession of the Credit Union, a Receiver or Sherliff, the Credit Union, Receiver or Sherliff in possession, as the case may be:

shall not be required to take any steps to preserve any rights against other perties to any Chattel Paper, Security, or Instrument constituting Collataral;

(b) shall not be required to keep the Collateral identifiable; and

(c) may use the Collateral in any manner and to any extent the Credit Union In its sole discretion, deems advisable.

## 9. SECURITIES

If the Collateral at any time includes Securities, the Debtor authorizos the Credit Union to transfer the same or any part thereof into its own name or that of its nominee(s) so that the Credit Union or its nominee(s) may appear of record as the sole owner thereof; provided that, until Default, the Credit Union shall promptly deliver to the Debtor all notices or other communications received by it or its nominee(s) as such registered owner and, upon demand and receipt of payment of any necessary expenses thereof, shall issue to the Debtor or its order a proxy to vote and take all action with respect to such Securities. After Default, the Debtor walves all rights to receive any notices or communications received by the Credit Union or its nominee(s) as such registered owner and agrees that no proxy issued by the Credit Union to the Debtor or its order as aforesaid shall thereafter be effective.

## 10. COLLECTION FROM ACCOUNT DEBTORS

Before or after Default, the Credit Union may notify any Account Debtor of this Security Interest and may direct such Account Debtor to make all payments to the Credit Union. The Debtor acknowledges that any payments on or other Proceeds of the Collateral received by the Debtor from any Account Debtor, whether before or after notice of this Security interest is given to such Account Debtor and whether before or after Default, shall be received and held by the Debtor in trust for the Credit Union and shall be turned over to the Credit Union upon request The Debtor agrees that it will not commingle any Proceeds of or payments on the Collateral with any of the Debtor's funds or property, but will hold them separate and apart.

## 11. OTHER TERMS

This Agreement includes the terms, if any, which are contained in Schedule "D".

## 12. APPLICATION OF MONIES

All Monies collected or received by the Credit Union pursuant to or in exercise of any right it possesses with respect to the Collateral shall be applied on account of the Indebtedness in such manner as the Credit Union may in its sole discretion determine or, at the option of the Credit Union, may be held unappropriated in a collateral account or released to the Debtor, all without prejudice to the liability of the Debtor or the rights of the Credit Union hereunder, and any surplus shall be accounted for as required by law.

## 13. DEFAULT

The happening of any of the following events shall constitute default ("Default") hereunder:

(a) nonpayment when due, whether by acceleration, demand or otherwise, of any amount forming part of the indebtedness; (b) failure of the Debtor to observe or perform any term contained in this Agreement or in any other agreement between the Debtor and

the Credit Union: (c) declaration of incompetency by a court of competent jurisdiction or death of an individual Debtor or an individual partner of a partnership Debtor;

(d) bankruptcy or insolvency of the Debtor; filing against the Debtor of a patition in bankruptcy; making of an assignment for the benefit of creditors by the Debtor; appointment of a Receiver or trustee for the Debtor of or any property of the Debtor or institution by or against the Debtor of any proposal, plan of arrangament or other type of insolvency proceeding under the Bankruptcy Act or otherwise;
 (e) Institution by or against the Debtor of any formal or informal proceeding for the dissolution or liquidation of, settlement of claims are a property of the Debtor of any formal or informal proceeding for the dissolution or liquidation of, settlement of claims

against, or winding up of the affairs of the Debtor; (f) any of the Encumbrances becomes enforceable;

(g) the Debtor caases or threatens to cease to carry on business or makes or agrees to make a bulk sale of assets or commits or threatens

(g) the Debtor cases or threatens to cease to carry on business or makes or agrees to make a pulk sale of essets of commits or threatens to commit an act of bankruptcy;
(h) any execution, sequestration, extent or other process of any court becomes enforceable against the Debtor or if a distress or analogous process is leviad upon the property of the Debtor or any part thereof;
(i) any certificate, statement, representation, warranty or audit report heretofore or hereafter furnished by or on behalf of the Debtor pursuant to or in connection with this Agreement, or otherwise (including, without limitation, the representations and warranties contained herein) or as an inducement to the Credit Union to extend any credit to or to enter into this or any other agreement with the Debtor; is faise in any material respect at the time as of which the facts therein set forth were etated or cartified, or ormits any substantial contingent or unliquidated liebility or claim against the Debtor; or if upon the date of execution of this Agreement, there shall have been any material adverse change in any of the facts disclosed by any such certificate, representation, statement, warranty or audit report, which change shall not have been disclosed to the Credit Union at or prior to the time of such execution;
(j) any of the licenses, permits or approvals granted by any government or any governmental authority and essential to the business of the Debtor is withdrawn, cancelled or significantly eitered;

(k) any or the incenses, permits or approvals granted by any government or any governmental authority and essential to the business of the Debtor is withdrawn, cancelled or significantly altered;
 (k) at any time, there is a material adverse change in the financial condition of the Debtor; or
 (i) the Credit Union considers that it is insecure, or that the prospect of payment or performance by the Debtor of the indebtedness is or is about to be impaired, or that the Collaterel is or is about to be placed in jeopardy.

## 14. ACCELERATION

In the event of Default, the Credit Union, in its sole discretion, may without demand or notice of any kind, declare all or any of the indebtedness which is not by its terms payable on demand, to be immediately due and payable. This clause does not apply to or affect any of the indeptedness payable on demand.

#### 15. REMEDIES

## On Default:

 (a) the Credit Union may enforce this Agreement by any method provided for in this Agreement, in the FPSA or as otherwise parmitted by law or in equity, and without limitation, may dispose of Collateral by lease or deferred payment;
 (b) the Credit Union may seize or otherwise take possession of the Collateral or any part theraof and sell the same by public or private sala at such price and upon such terms as the Credit Union in its sole discretion may determine and the proceeds of such sale less all costs, charges and expenses of the Credit Union (including costs as between a solicitor and his own client on a full indemnity basis) shell be applied on the backted terms and the proceeds of such sale less all costs. Indebtodness and the surplus, if any, shall be disposed of according to law; (c) the Credit Union may take proceedings in any court of competent jurisdiction for the appointment of a Receiver;

(d) the Credit Union may appoint by instrument any person or persons to be a Receiver of any Collateral, and may remove any person so appointed and appoint enother in his stead;

(e) unless otherwise restricted by his appointment, any Receiver shall have the power;

- to take possession of any Collateral and for that purpose to take any proceedings, in the name of the Debtor or otherwise; to carry on or concur in carrying on the business of the Debtor;
- to sell or lease any Collateral;
- /lv1
- to make any arrangement or compromise which he may think expedient in the interest of the Credit Union; to pay all labilities and expenses connected with the Collateral, including the cost of insurance and payment of taxes or other costs, charges or expenses incurred in obtaining, maintaining possession of any preserving the Collateral, and the same shall be added to the indebtedness and secured by the Collaterel; (vi) to hold as additional security any increase or profits resulting from the Collaterel; (vii) to exercise all rights and remedias that the Credit Union may have under this Agreement, the PPSA or otherwise at law or in
- equity:

- (vill) with the written consent of the Credit Union, to borrow money for the purpose of carrying on the business of the Debtor or for maintenance of the Collateral or any part thereof or for other purposes approved by the Credit Union, and any amount so borrowed together with interest thereon shall form a charge upon the Collateral in priority to the Security interest created by this Acreement: and
- to do any other act or thing as may be considered to be incidental or conducive to any of the matters and powers aforesaid

 (f) the Debtor hereby appoints each Receiver appointed by the Credit Union to be its attorney to effect a sale or lease of any Collateral and any deed, lease, agreement or other document signed by a Receiver pursuant to this power of attorney shall have the same effect as if it had been executed by end under the seal of the Debtor;
 (g) a Receiver appointed by the Credit Union shall be deemed to be the agent of the Debtor, and the Debtor shall be solely responsible for his acts or defaults and for his remumeration and expenses, and the Credit Union shall not be in any way responsible for any misconduct or public to the text of the Debtor. negligence on the part of any Receiver; (h) all monies received by the Receiver after providing for payment of all costs, charges and expenses of or incidental to the exercise of

(i) all mones received by the receiver and provident of provident of the costs, charges and expension of the Receiver shall be paid to the Credit Union and applied on account of the indebtedness;
 (i) the Credit Union may enter upon, use and occupy all premises owned or occupied by the Debtor wherein the Collateral may be situate;
 (j) before, during or after realizing on the Collateral, the Credit Union may recover and enforce judgment against the Debtor for the Indebtedness; and all costs, charges and expenses reasonably incurred by the Credit Union (including, without limitation, costs as between a solicitor and his own client on a full indemnity basis) in recovering or enforcing judgment against the Debtor; and
 (k) the Credit Union mey, but shall not be bound to, realize on the Collateral.

## 16. DEFICIENCY

If the Credit Union realizes on the Collateral and the realization is not sufficient to satisfy all the indebtadness, the Debtor acknowledges and egrees that the Debtor shell continue to be liable for any indebtedness remaining outstanding and the Credit Union shall be entitled to pursue full payment thereof.

#### 17. COSTS AND EXPENSES

(a) Upon the Debtor's failure to parform any of its obligations under this Agreement then the Credit Union may, but shall not be obligated to perform the same and in the event of performance thereof by the Credit Union the Debtor shall pay to the Credit Union forthwith upon written demand therefor an emount equal to all costs, charges and expenses incurred by the Credit Union in performing the Debtor's obligations plus interest thereon at the Agreed Rate from the date such costs, charges and expenses are incurred by the Credit Union until paid by the Debtor

(b) The Debtor shall pay all costs, charges and expenses reasonably incurred by the Credit Union or any Receiver appointed by it (including, but without restricting the generality of the foregoing, costs as between a solicitor and his own client on a full indemnity basis), in preparing, registering financing statements regarding or enforcing this Agreement, inspecting taking custody of, preserving, repairing, preparing, registering financing statements regarding or enforcing this Agreement, inspecting taking costody or, preserving, reparing, maintaining, processing, preparing for disposition and disposing of Collateral and in enforcing or collecting the indebtedness and all such costs, charges and expenses together with any monies owing as a result of any borrowing by any Receiver appointed by the Credit Union shall be a first charge on the proceeds of realization, collection or disposition of Collateral and shall be secured hereby. Such costs, charges and expenses shall bear interest at the Agreed Rate from the date the same were incurred to the date of payment by the Dobtor.

(c) The Credit Union may pay or satisfy any Encumbrances or pay any sum necessary to clear title to any Collateral, and the Debtor agrees to repay the same on demend, plus interest thereon at the Agreed Rate.

(d) All amounts paid by the Credit Union pursuant to this clause together with interest thereon at the Agreed Rate shall form part of the indebtedness and be secured hereby.

## 18. SET OFF

Without limiting any other right the Credit Union may have, the Credit Union may, in its sole discretion at any time and without notice, set off any and all amounts owed to the Debtor by the Credit Union in any capacity and, whether or not due, against any and all indobtedness including any contingent or non-matured indebtedness and indebtedness as principal or guarantor.

#### 19. EURTHER ASSURANCES

The Debtor agrees to execute and deliver to the Credit Union such further assurances, conveyances and supplemental deeds as may be necessary to properly carry out the intention of this Agreement, as determined by the Credit Union, or as may be required by the Credit Union from time to time.

#### 20. NOTICE

Any notice or demand required or permitted to be made or given by the Credit Union to the Debtor may be velidly served by leaving the same with, or by mailing the same by prepaid registered mail to, the Debtor at his address as set out herein (or at such other address as the Debtor may in writing notify the Credit Union of as the Debtor's address for service under his Agreement) or by leaving such notice with any officer or director of the Debtor as shown on the records of the Credit Union, and in the case of mailing such notice or demand shall be deemed to have been received by the Debtor on the third business day following the dato of mailing.

## 21. GENERAL

(a) The Credit Union may grant extensions of time and other indulgences, take and give up security, accept compositions, compound, compromise, settle, grant releases and discharges and otherwise deal with the Debtor, Account Debtors, suretles and others and with Collateral and other security as the Credit Union may see fit without prejudice to the liability of the Debtor or the Credit Union's right to hold and realize upon the Sacurity Interest. The Credit Union may demand, collect and sue on Collateral in either the Debtor's or the Credit Union's name and may endorse the Debtor's name on any and all cheques, commercial paper, and any other instruments pertaining to or constituting Collateral. The Credit Union is to exercise its remedies, take possession of, collect, enforce, realize, sell, lease or otherwise dispose of Collateral or to initiate any proceedings for such purposes.
(b) No delay or ornission by the Credit Union in exercising any right or premedy hereunder or with respect to any ladebtedness shall operate as a waiver thereof or of any other right or remedy, and no single or partial exercise thereof shall preclude any other remedy. The Credit Union may remedy any Default by the Debtor. All rights and remedies of the Credit Union granted and without waiving eny other prior or subsequent Default by the Debtor. All rights and remedies of the Credit Union, (c) The Debtor waives protest of any and the credit Union.

 (c) The Deptor Waives protect or any instrument constituting Collateral at any time held by the Credit Union in Which the Deptor is in any way liable and notice of any other action taken by the Credit Union.
 (d) This Agreement shall enure to the benefit of and be binding upon the parties hereto and their respective heirs, executors, administrators, successors and assigns. In any action brought by an assignee of this Agreement and the Security Interest or any part thereof to enforce any rights hereunder, the Debtor shall not essert against the assignee any claim or defence which the Debtor now has or hereofter may have against the Credit Union, (e) If more than one Debtor executes this Agreement the obligations of such Debtors hereunder shall be joint and several

(f) No modification, variation or amendment of any term for this Agreement shall be binding or effective unless made by written agreement, executed by the perties harde and no waiver of any term hereof shall be binding or effective unless in writing.
(g) This Agreement is in addition to and not in substitution for any other agreements, securities or Security Interests now or hereafter held

by the Credit Union and all such other agreemants, securities and Security Interests shall remain in full force and effect.

(h) The headings used in this Agreement are for convenience only and are not to be considered part of this Agreement and do not in any (i) When the context so requires, the singular shall be read as if the plurel were expressed and vice verse and the terms hereof shall be

(i) When the context so requires, the singular shall be read as it the pural were expressed and vice versa and the terms neces shall be read with all grammatical changes necessary dependant upon the person referred to being a male, female, partnership or corporation.
 (j) In the event any terms of this Agreement, as amended from time to time, shall be deemed invalid, void or unenforceable, in whole or in part, by any court of competent jurisdiction, the remaining terms of this Agreement shall remain in full force and effect.
 (k) Nothing herein contained shall in any way obligate the Credit Union to grant, continue, renew or extend time for payment of the

22. ATTACHMENT

Indebtedness.

(a) Subject to subclause (b), the Security interest created hereby is intended to attach the Collateral when this Agreement is executed by the Dabtor and delivered to the Credit Union. (b) With respect to that Collateral acquired by the Debtor after the date this Agreement is executed and delivered to the Credit Union, the

Security interast created hereby in such Collateral is intended to attach at the same time as the Debtor acquires rights in such Collateral.

## 23. DISCLOSING INFORMATION

This Agreement and any information pertaining thereto or to the indebtedness may be disclosed by the Credit Union as required by the PPSA

#### 24. GOVERNING LAW AND JURISDICTION

This Agreement shell be interpreted in accordance with the laws of the Province of Alberta, and the Debtor irrevocably agrees that any suit In suggestion sharped in account and the twee with the lows of the rownice of Automa, and the Descon introdeady agrees that any suit or proceeding with respect to any matters arising out of or in connection with this Agreement may be brought in the courts of the Province of Alberta or in any court of competent jurisdiction, as the Credit Union may efect, and the Oebtor agrees to attom to the same.

## 25. COPY OF AGREEMENT

Debtor acknowledges receipt of a copy of this Agreement and waives any right it may have to receive a Financing Statement or Financing Change Statement relating to it.

IN WITNESS WHEREOF the Debtor has here unto executed this Agreement this <u>30TH</u> day of <u>OCTOBER</u> VANCOUVER, B.C. Alberts 2009 at

\* IF DEBTOR IS AN INDIVIDUAL: Signature of Debtor(s)

Vitnesa	
Witness	
Vilness	
Miness	* IF DEBTOR IS A PARTNERSHIP;
Vilness	Print Exact Name of Partnership or Trade Name
Milness	By: (Partner)
Winess	- By: IF DEBTOR IS A CORPORATION: CRELOGIX CREDIT GROUP INC.
	Print Exact Name of Corporation
	By: (Officer of Corporation
	FULL ADDRESS OF DEBTOR
* Complete Affidavit of Execution if Debtor is an individual or	

# SCHEDULE "A" (DESCRIBED PROPERTY)

Oblain serial numbers for all motors vehicles, trailers, mobile homes, farm machinery, equipment and airplanes
 ALL PRESENT AND AFTER-ACQUIRED PROPERTY

nartnershin.

Page 5 of 6

Page 6 of 6

# SCHEDULE "D" (OTHER TERMS AND CONDITIONS)

SCHEDULE "C" (PERMITTED ENCUMBRANCES AFFECTING COLLATERAL)

SCHEDULE "B" (DESCRIBED REAL PROPERTY) \* Obtain legal description This is **Exhibit "M**" referred to in the Affidavit of Darcy Peelar, sworn before me this 5<sup>th</sup> day of July, 2017

A Commissioner for Oaths in and for Alberta

Patrick T. Harnett Barrister and Solicitor

25337254.1

# **GENERAL SECURITY** AGREEMENT

	SERVUS	
	Credit Union Ltd. (the "Credit Union")	-
	#601-4901-48TH STREET	
	RED DEER, ALBERTA	
ATION	T4N 6N4	, Alberta
	PARKLAND SQUARE	, Branch
	Member No(s).:	

CRELOGIX PORTFOLIO SERVICES CORPOR SUITE 500, 4180 LOUGHEED HIGHWAY

BURNABY, BRITISH COLUMBIA

V5C 6A7

Borrower(s):

## 1. DEFINITIONS

All capitalized terms used in this agreement ("Agreement") including any schedules ("Schedules") annoxed hereto shall, except where defined herein, be interpreted pursuant to their respective meanings when used in the Personal Property Security Act of Alberta in force at the date of this Agreement ("PPSA").

- (b) In this Agreement:
  - "Account Debtor" means a debtor of the Debtor on an intangible, Chattel Paper or Account, or any obligor of the Debtor on an (1) Instrument:
  - "Agreed Rate" means the rate of interest payeble under the document(s) evidencing the indebtedness and in the event such document(s) bear different rates of interest the "Agreed Rate" shall mean the highest of such interest rates;
     (iii) "Consumer Goods" means those goods that are used or acquired by the Debter for use primarily for his personal, family or
  - household purposes;

  - "Debtor" means the "Borrower(s)"; "Encumbrances" means any Security Interest, mortgages, liens claims, charges and other encumbrances affecting the Collateral (v) including Permitted Encumbrances but excluding the Security Interest created hereby; "Permitted Encumbrances" means any Encumbrances which are described in Schedule "C" and any others approved in writing by
  - (vi) the Credit Union prior to their creation or assumption; and (vii) "Receiver" includes a Receiver-Manager.

## 2. SECURITY INTEREST

- (a) For value received, the Debtor hereby grants to the Credit Union, by way of mortgage, charge and assignment, a Security Interest in the undertaking of the Debtor and in all Goods (including all parts, accessories, attachments, special tools, edditions and Accessions thereto), Chattel Paper, Documents of Title (whether negotiable or not), Instruments, Intangibles, Money, Securities and other property now or hereafter owned or acquired by or on behalf of the Debtor (including such as may be returned to or repossessed by the Debtor) and in all Proceeds thereof, (herein collectively called the "Collateral"), including, without limitation, all of the following property now or hereafter owned or acquired by or on behalf of the Debtor:
  - i) all property described in Schedule "A";

  - (ii) all Inventory;
     (iii) all Equipment, Including, without limitation, all machinery, tools, apparatus, plant, furniture, Fixtures and vehicles of whatsoever nature or kind;
  - (iv) all Accounts, including, without limitation, all book accounts and book debts and generally all accounts, debts, dues, claims, choses in action, judgments and demands of every nature and kind howsoever arising or secured including letters of credit and advices of credit, which are now due, owing or accruing due to or owned by or which may hereafter become due, owing or eccruing due to or owned by the Debtor;
  - all deeds, documents, writings, papers and books of account and other books relating to or being records of Accounts, Chattel (v) Paper, instruments or Documents of Title or by which such are or may hereafter be secured, evidenced, acl owledged or payable:

  - (vi) all contractual rights and insurance claims and all goodwill, petents, trademarks, copyrights, and other intellectual property; and (vii) all crops and livestock including all crops that are or which hereafter become crops on any real property described in Schedule

 (b) Any reference to "Collateral" shall, unless its context otherwise requires, be deemed a reference to "Collateral or any part thereof"
 (c) The Security Interest granted hereby shall not extend or apply to and the Collateral shall not include the last day of the terms of any lease or agreement therefore but upon the enforcement of the Security interest the Debtor shall stand possessed of such last day in trust to

(a) a promotion function of the environment of the Security interest the Debtor shall stand possessed of such last day in trust to assign the same to any person acquiring such term.
 (d) The Security Interest granted hereby shall not extend or apply to and the Collateral shall not include the Debtor's Consumer Goods except for any described in Schedule "A".

## 3. INDEBTEDNESS SECURED

The Security interest granted hereby secures payment of any and all obligations, indebtedness and liabilities of the Debtor to the Credit Union whethar present or future, direct or indirect, absolute or contingent, matured or not, extended or renewed, wheresoever and how soever incurred and any ultimate unpaid balance thereof and whether the same is from time to time reduced and thereafter increased or entirely extinguished and thereafter incurred again and whether the Debtor is bound alone or with another or others and whether as principal or surely (hereinafter collectively celled the "Indebtedness").

## 4. DEBTOR'S REPRESENTATIONS AND WARRANTIES

The Debtor represents and warrants and so long as this Agreement remains in effect continuously represents and warrants that;

(a) the Collateral is genuine and owned by the Debtor free of all Encumbrances except Permitted Encumbrances;
 (b) each Account, Chattel Paper and instrument constituting Collateral is enforceable in accordence with its terms against the party obligated to pay the same and the amount represented by the Debtor to the Credit Union from time to time as owing by each Account Debtor is the correct amount actually and unconditionally owing by such Account Debtor;

 (c) there is no litigation, proceeding or dispute pending or to the knowledge of the Debtor threatened against or affecting the Debtor or the Collateral, the adverse determination of which might materially and edversely affect the Debtor's financial condition or impair the Debtor's ability to perform its obligations hereunder;

10112 ND (06/2001 ST)

REV. (09/90) B.A.S.

Delete and initial if inapplicable.

(d) the name(s) of the Debtor is(are) accurately and fully set out above, and the Debtor is not known by any other name(s); and

 (a) there is no provision in any agreement to which the Debtor is a party, nor to the knowledge of the Debtor is there any statute, rule or regulation, or any judgment, decree or order of any court binding on the Debtor, which would be contravened by the execution and delivery of this Agreement

## 5. DEBTOR'S COVENANTS

The Debtor covenants and agrees:

(a) to defend the Collateral (except Collateral dealt with as permitted by clause 7 hereof) against the claims and demands of all other parties claiming the same or an interest therein:

(b) to keep the Collateral free from all Encumbrances except Permitted Encumbrances;

(c) subject to clause 7 hereof, not to sell, exchange, transfer, assign, lease or otherwise dispose of Collateral or any interest therein without the prior written consent of the Credit Union;

(d) to notify the Credit Union promptly of;

- any change in the information contained herein or in the Schedules relating to the Debtor, the Debtor's name, the Debtor's (1) business or Collateral:
- (ii) the details of any significant acquisition of Collateral;
   (iii) the details of any claims or litigation affecting the Debtor or Collateral;
- (iv) any loss or damage to Collateral;
   (v) any default by any Account Debtor in payment or other performance of his obligations with respect to Collateral; and
- (vi) the return to or rapossession by the Debtor of Collateral;

(e) to keep the Collateral in good order, condition and repair and not to use Collateral in violation of the provisions of this Agreement or any other agreement relating to the Collateral or any policy insuring the Collateral or any applicable statute, law, bylaw, rule, regulation or ordinance:

(f) to pay all taxes, rates, levies, assessments and other charges of every nature which may be lawfully levied, assessed or imposed against or in raspect of the Debtor or Collateral as and when the same become due and payable;

(g) to punctually make all payments and perform all obligations in any lease by the Debtor and under any agreement charging property of the Debtor

 (h) to prevant Collateral from being or becoming a Fixture or an Accession to other property that is not Collateral;
 (l) to carry on and conduct the business of the Debtor in a proper and efficient manner so as to protect and preserve the Collateral and to keep, in accordance with generative accounting principles, consistently applied, proper books of account for the Debtor's business as well as accurate and complete records concerning the Collateral;

(i) to deliver to the Credit Union from time to time promptly upon request;

- any Documents of Title, Instruments, Securities and Chattel Paper constituting, representing or relating to the Collateral;
   all books of account and all records, ledgers, reports, correspondence, schedules, documents, statements, lists and other writings
- relating to the Collateral for the purpose of inspecting, auditing or copying the same; all financial statements prepared by or for the Debtor regarding the Debtor's business;
- άv).
- all policies and certificates of insurance relating to the Collateral; and such information concerning the Collateral, the Debtor and the Debtor's business and affairs as the Credit Union may reasonably (v) request:
- (k) not to remove any of the Collateral from the Province of Alberta without the prior written consent of the Credit Union;
- (i) In the event the value of the Collateral shall be materially reduced, to immediately reduce the emount of the Indebtedness by an amount determined by the Credit Union;
   (m) If the Collateral includes crops and livestock, in addition to the Debtor's other obligations regarding Collateral;

- to do all acts which may be necessary to attend to, care for, raise and fatten the livestock and to grow, cultivate, spray, irrigate, cut, harvest, pick, clean, preserve and protect the crops, all according to tha most approved methods of farming husbandry and to keep the farm(s) on which the Collateral is located free of noxious weeds and grasses, and maintain the present buildings and improvements on the seld farm(s) in good condition and repair;
- to provide suitable range, pasture and feed for all livestock and care for and protect them from disease, damage, injury, dasth, destruction by weather, wild enimals, theft or other cause;
- (iii) to pay, when due, all obligations incurred for labour or material or otherwise in the care or feeding or shearing of such livestock; and
- (iv) at the request of the Credit Union, to deliver to the Credit Union the Debtor's Canadian Wheat Board producer's permit book and to assign to the Credit Union all of the Debtor's rights thereunder, and

(n) to permit the Credit Union, by its officers or authorized agents, at any time, and from time to time, as often as the Credit Union in its sole discretion may determine, to enter the premises owned or occupied by the Debtor for the purpose of inspecting the Colleteral and the operation of the Debtor's business.

## 6. INSURANCE

Debtor shall furnish at its expense all necessary proofs and shall do all necessary acts to enable the Credit Union to obtain payment of the insurance monies

## 7. DEALING WITH COLLATERAL

The Debtor shall not sell, exchange, transfer, assign, lease or otherwise dispose of that Collateral described in Schedule "A" except with the prior written consent of the Credit Union which consent may be arbitrarily withheld. Until but not after Default the Debtor may deal with Collateral, other than that Colleteral described in Schedule "A", in the ordinary course of the Debtor's business in any manner not inconsistent Collateral for fair value on commercially reasonable terms and provided that all cash Proceeds therefrom are immediately deposited with the Credit Union

## 8. COLLATERAL IN POSSESSION OF CREDIT UNION, RECEIVER OR SHERIFF

If Collateral is at any time in the possession of the Credit Union, a Receiver or Sheriff, the Credit Union, Receiver or Sheriff in possession, as the case may be:

(a) shall not be required to take any steps to preserve any rights against other parties to any Chattel Paper. Security, or instrument constituting Collateral;

(b) shall not be required to keep the Collateral identifiable; and

(c) may use the Collateral in any manner and to any extent the Credit Union in its sole discretion, deems advisable.

#### 9. SECURITIES

If the Collateral at any time includes Securities, the Debtor authorizes the Credit Union to transfer the same or any part thereof into its own If the Contacta at any time includes Securities, the before authorizes the Creat Union to transfer the same of any part thereof into its bwin name or that of its nominee(s) so that the Credit Union or its nominee(s) may appear of record as the sole owner thereof; provided that, until Default, the Credit Union shall promptly deliver to the Debtor all notices or other communications received by it or its nominee(s) as such registered owner and, upon demand and receipt of payment of any necessary expenses thereof, shall issue to the Debtor or its order a proxy to vote and take all action with respect to such Securities. After Default, the Debtor weives all rights to receive any notices or communications received by the Credit Union or its nominee(a) as such registered owner and agrees that no proxy issued by the Credit Union to the Debtor or its order as aforesaid shall thereafter be effective.

## 10. COLLECTION FROM ACCOUNT DEBTORS

Before or after Default, the Credit Union may notify any Account Debtor of this Security Interest and may direct such Account Debtor to make all payments to the Credit Union. The Debtor acknowledges that any payments on or other Proceeds of the Collateral received by the Debtor from any Account Debtor, whether before or after notice of this Security interest is given to such Account Debtor and whether before or after Default, shall be received and held by the Debtor in trust for the Credit Union and shall be turned over to the Credit Union upon request. The Debtor agrees that it will not commingle any Proceeds of or payments on the Collateral with any of the Debtor's funds or property, but will hold them seperate and apart.

## 11. OTHER TERMS

This Agreement includes the terms, if any, which are contained in Schedule "D".

## 12. APPLICATION OF MONIES

All Monies collected or received by the Credit Union pursuant to or in exercise of any right it possesses with respect to the Collateral shall be applied on account of the Indebtednoss in such manner as the Credit Union may in its sole discretion determine or, at the option of the Credit Union, may be held unappropriated in a collateral account or released to the Debtor, all without prejudice to the liability of the Debtor or the rights of the Credit Union hereunder, and any surplus shall be accounted for as required by law.

#### 13. DEFAULT

The happening of env of the following events shell constitute default ("Default") hereunder:

 (a) nonpayment when due, whether by acceleration, demend or otherwise, of any amount forming part of the indebtedness;
 (b) failure of the Debtor to observe or perform any term contained in this Agreement or in any other agreement between the Debtor and the Credit Union:

declaration of incompetency by a court of competent jurisdiction or death of an individual Debtor or an individual partner of (c) partnership Debtor:

(a) backruptcy or insolvency of the Debtor, filling against the Debtor of a petition in bankruptcy; making of an assignment for the benefit of creditors by the Debtor; appointment of a Receiver or trustee for the Debtor of or any property of the Debtor or institution by or against the Debtor of any proposal, plan of arrangement or other type of insolvency proceeding under the Bankruptcy Act or otherwise;

(a) institution by or against the Debtor of any formal or informal proceeding for the dissolution or liquidation of, settlement of cleims against, or winding up of the affeirs of the Debtor;

any of the Encumbrances becomes enforceable;

(g) the Debtor ceases or threatens to case to carry on business or makes or agrees to make a bulk sale of assets or commits or threatens to commit an act of bankruptcy; (h) any execution, sequestration, extent or other process of any court becomes enforceable against the Debtor or if a distress or

 (i) any exaction, squestration, extent or other process of any could become sensitive better of it a distress of analogous process is levied upon the property of the Debtor or any part thereof;
 (i) any certificate, statement, representation, warranty or audit report heretofore or hereafter furnished by or on behalf of the Debtor pursuant to or in connection with this Agreement, or otherwise (including, without limitation, the representations and warranties contained herein) or as an inducement to the Credit Union to extend any credit to or to enter into this or any other agreement with the Debtor, is false in herein) or as an inducement to the creat which the facts therein set forth were stated or certified, or omits any substantial contingent or any material respect at the time as of which the facts therein set forth were stated or certified, or omits any substantial contingent or uniquidated liability or claim against the Debtor; or if upon the date of execution of this Agreement, there shall have been any material adverse change in any of the facts disclosed by any such certificate, representation, statement, werranty or audit report, which change shall not have been disclosed to the Credit Union at or prior to the time of such execution;

any of the licenses, permits or approvals granted by any government or any governmental authority and essential to the business of the Debtor is withdrawn, cancelled or significantly eltered;

Debtor is withdrawin, carceneou or significantly encirce. (k) at any time, there is a material adverse change in the financial condition of the Debtor; or (i) the Credit Union considers that it is insecure, or that the prospect of payment or performance by the Debtor of the indebtedness is or is about to be impaired, or that the Collateral is or is about to be placed in jeopardy.

## 14. ACCELERATION

In the event of Default, the Credit Union, in its sole discretion, may without demand or notice of any kind, declare all or any of the Indebtedness which is not by its terms payable on demand, to be immediately due and payable. This clause does not apply to or effect any of the indebtedness payable on demand.

## 16. REMEDIES

## On Default:

(a) the Credit Union may enforce this Agreement by any method provided for in this Agreement, in the PPSA or as otherwise permitted by

 (a) the Credit Union may shore this Agreement by any network you do for in this Agreement, in the PSA of as otherwise permitted by law or in equity, and without limitation, may dispose of Collateral by lease or deferred paymant;
 (b) the Credit Union may selze or otherwise take possession of the Collateral or any part thereof and sell the same by public or private sale at such price and upon such terms as the Credit Union in its sole discretion may determine and the proceeds of such sale less all costs, charges and expenses of the Credit Union (including costs as between a solicitor and his own client on a full indemnity basis) shall be applied on the and expenses of the oreal onion (including costs as between a solicitor and his own client on a full indemnity basis) shall be applied on the Indebtedness and the surplus, if any, shall be disposed of according to law;
 (c) the Credit Union may take proceedings in any court of competent jurisdiction for the appointment of a Receiver;
 (d) the Credit Union may appoint by instrument any person or persons to be a Receiver of any Collateral, and may remove any person so appointed and appoint another in his stead;

(e) unless otherwise restricted by his appointment, any Receiver shall have the power;

- to take possession of any Collateral and for that purpose to take any proceedings, in the name of the Debtor or otherwise;
- (ii) to carry on or concur in carrying on the business of the Debtor, (iii) to sell or lease any Collateral;

- (iv) to make any arrangement or compromise which he may think expedient in the interest of the Credit Union;
   (v) to pay all liabilities and expenses connected with the Colleteral, including the cost of insurance and payment of taxes or other (v) (v) to pay an maintee and expenses incurred in obtaining, maintaining possession of any perserving the Collateral, and the same shall be added to the indebtedness and secured by the Collateral;
   (vi) to hold as additional security any increase or profits resulting from the Colleteral;
   (vii) to exercise all rights and remedies that the Credit Union may have under this Agreement, the PPSA or otherwise at law or in
- equity;

(viii) with the written consent of the Gredit Union, to borrow money for the purpose of carrying on the business of the Debtor or for maintenance of the Collateral or any part thereof or for other purposes approved by the Credit Union, and any amount so borrowed together with interest thereon shall form a charge upon the Collateral in priority to the Security Interest created by this hne treement:

(ix) to do any other act or thing as may be considered to be incidental or conducive to any of the matters and powers aforeasid;

(f) the Debtor hereby appoints each Receiver appointed by the Credit Union to be its attorney to effect a sale or lease of any Collateral and any deed, lease, agreement or other document signed by a Receiver pursuant to this power of attorney shall have the same effect as if it had been executed by and under the seel of the Debtor;
(g) a Receiver appointed by the Credit Union shall be deemed to be the agent of the Debtor, and the Debtor shall be solely responsible for his acts or defaults and for his remuneration and expenses, and the Credit Union shall not be in any way responsible for any misconduct or

(h) all monies received by the Receiver after providing for payment of all costs, charges and expenses of or incidental to the exercise of any of the powers of the Receiver shall be paid to the Credit Union and applied on account of the indebtedness;
 (i) the Credit Union may enter upon, use and occupy all premises owned or occupied by the Debtor wherein the Collateral may be situate;

(i) before, during or after realizing on the Collateral, the Credit Union may recover and enforce judgment against the Debtor for the Indebtedness and all costs, charges and expenses reasonably incurred by the Credit Union (including, without fimitetion, costs as between a solicitor and his own client on a full indemnity basis) in recovering or enforcing judgment against the Debtor; and

(k) the Credit Union may, but shall not be bound to, realize on the Collateral

#### 16. DEFICIENCY

If the Credit Union realizes on the Collateral and the realization is not sufficient to satisfy all the Indebtedness, the Debtor acknowledges and agrees that the Debtor shall continue to be liable for any indebtedness remaining outstanding and the Credit Union shall be entitled to pursue full payment thereof.

## 17. COSTS AND EXPENSES

(a) Upon the Debtor's failure to perform any of its obligations under this Agreement then the Credit Union may, but shall not be obligated to perform the same and in the event of performance thereof by the Credit Union the Debtor shall pay to the Credit Union forthwith upon written demand therefor an amount equal to all costs, charges and expanses incurred by the Credit Union in performing the Debtor's obligations. plus interest thereon at the Agreed Rate from the date such costs, charges and expenses are incurred by the Credit Union until paid by the Debtor.

The Debtor shall pay all costs, charges and expenses reasonably incurred by the Credit Union or any Receiver appointed by it (b) (including, registering financing statements regarding or enforcing this Agreement, inspecting taking custody of, preserving, repairing, maintaining, procassing, preparing for disposition and disposing of Collateral and in enforcing or collecting the indebtedness and all such costs, charges and expenses together with any monies owing as a result of any borrowing by any Receiver appointed by the Credit Union shall be a first charge on the proceeds of realization, collection or disposition of Collateral and shall be secured hereby. Such costs, charges and expenses

shall bear interest at the Agreed Rate from the date the same were incurred to the date of payment by the Debtor. (c) The Credit Union may pay or satisfy any Encumbrances or pay any sum necessary to clear title to any Collateral, and the Debtor agrees to repay the same on demand, plus interest thereon at the Agreed Rate. (d) All amounts paid by the Credit Union pursuant to this clause together with Interest theraon at the Agreed Rate shall form part of the

Indebtedness and be secured hereby.

## 18. SET OFF

Without limiting any other right the Credit Union may have, the Credit Union may, in its sole discretion at any time and without notice, set off any and all amounts owed to the Debtor by the Credit Union in any capacity and, whether or not due, against any and all indebtedness including any contingent or non-matured indebtedness and indebtedness as principal or guarantor.

#### 19. FURTHER ASSURANCES

The Debtor agrees to execute and deliver to the Credit Union such further assurances, conveyances and supplemental deeds as may be necessary to properly carry out the intention of this Agreement, as determined by the Credit Union, or as may be required by the Credit Union from time to time.

#### 20, NOTICE

Any notice or demand required or permitted to be made or given by the Credit Union to the Debtor may be validly served by leaving the same with, or by mailing the same by prepaid registered mail to, the Debtor at his address as set out herein (or at such other address as the Debtor may in writing notify the Credit Union of as the Debtor's address for service under his Agreement) or by leaving such notice with any officer or director of the Debtor as shown on the records of the Credit Union, and in the case of meiling such notice or demand shall be deemed to have been raceived by the Debtor on the third business day following the date of mailing.

## 21. GENERAL

(a) The Credit Union may grant extensions of time and other Indulgences, take and give up security, accept compositions, compound, (a) The Credit Union may grant extensions of time and other Indulgences, take and give up security, accept compositions, compound, compromise, settle, grant releases and discharges and otherwise deal with the Debtor, Account Debtors, sureties and others and with Collateral and other security as the Credit Union may see fit without prejudice to the liability of the Debtor of the Credit Union's right to hold and realize upon the Security interest. The Credit Union may demand, collect and sue on Collateral in either the Debtor's or the Credit Union's name and may endorse the Debtor's name on any and all cheques, commercial paper, and any other instruments partialning to or constituting Collateral. The Credit Union shall not be liable for any failure to exercise its remedies, take possession of, collect, enforce, realize, sell, lease or otherwise dispose of Collateral or to initiate any proceedings for such purposes.
 (b) No delay or omission by the Credit Union in exercising any right or remedy hereunder or with respect to any Indebtedness shall operate as a walver thereof or of any other right or remedy, and no single or partial exercise thereof shall preclude any other or further exercise thereof or the exercise of any other right or remedy. The Credit Union may remedy any Default by the Debtor. All rights and remedies of the Credit. Union granted or recognized herein are cumulative and may be exercised at any time and from time to time independentily or in combination.
 (c) The Debtor walves protest of any Instrument constituting Collateral at any time to the Debtor is in any

(c) The Debtor walves protect of any instrument constituting Collateral at any time held by the Credit Union in which the Debtor is in any

way liable and notice of any other action taken by the Credit Union.

(d) This Agreement shall enure action taken by the creat Union.
(d) This Agreement shall enure to the benefit of and be binding upon the parties hereto and their respective helds, executors, administrators, successors and assigns. In any action brought by an assignee of this Agreement and the Security Interest or any part thereof to enforce any rights hereunder, the Debtor shall not assert egainst the assignee any claim or defence which the Debtor now has or hereafter may have against the Credit Union.

(e) If more than one Debtor executes this Agreement the obligations of such Debtors hereunder shall be joint and several.
 (f) No modification, variation or amendment of any term of this Agreement shall be binding or effective unless made by written agreement, executed by the parties herete and no waiver of any term hereof shall be binding or effective unless in writing.

(g) This Agreement is in addition to end not in substitution for any other agreements, securities or Security Interests now or hereafter held Credit Union and all such other agreements, securities and Security Interests shall remain in full force and effect.

(h) The headings used in this Agreement are for convenience only and are not to be considered part of this Agreement and do not in any way limit, explain or amplify the terms of this Agreement. (i) When the context so requires, the singular shall be read as if the plural were expressed and vice versa and the terms hereof shall be

 (i) When the context so requires the angula shall be read as in the place were expressed and were easily and the context so recovery and the context so recov Indebtedness,

## 22. ATTACHMENT

(a) Subject to subclause (b), the Security interest created hereby is intended to attach the Collateral when this Agreement is executed by the Debtor and delivered to the Credit Union.
 (b) With respect to that Collateral acquired by the Debtor effer the date this Agreement is executed and delivered to the Credit Union, the Security interest created hereby in such Collateral is intended to attach at the same time as the Debtor acquires rights in such Collateral.

## 23. DISCLOSING INFORMATION

This Agreement and any information pertaining thereto or to the indebtedness may be disclosed by the Credit Union as required by the PPSA.

## 24. GOVERNING LAW AND JURISDICTION

This Agreement shell be interpreted in accordance with the laws of the Province of Alberta, and the Debtor irrevocably agrees that any suit or proceeding with respect to any matters arising out of or in connection with this Agreement may be brought in the courts of the Province of Alberta or in any court of competent jurisdiction, as the Credit Union may elect, and the Debtor agrees to attom to the same.

## 25. COPY OF AGREEMENT

Debtor acknowledges receipt of a copy of this Agreement and waives any right it may have to receive a Financing Statement or Financing Change Statement relating to it.

IN WITNESS WHEREOF the Debtor has hereunto executed this Ag VANCOUVER, B.C. <del>Vilberte</del> .	reement this <u>30TH</u> day of <u>OCTOBER</u> 2009 at
	* IF DEBTOR IS AN INDIVIDUAL: Signature of Debtor(s)
Mitness	
Minese Witnass	
Mthess	* IF DEBTOR IS A PARTNERSHIP:
VTI.1635	Print Exact Name of Partnership or Trade Name
Winess	By: (Pertner)
Miness	By: (Partner) * IF DEBTOR IS A CORPORATION:
	CRELOGIX PORTFOLIO SERVICES CORPORATION Print Exer Name Despective By:
	(Officer of Corporation By: (Officer of Corporation
	FUILL ADDRESS OF DEBTOR
* Complete Affidavit of Execution if Debtor is an individual or partnership.	

# SCHEDULE "A" (DESCRIBED PROPERTY)

\* Obtain serial numbers for all motors vehicles, trailers, mobile homes, farm machinery, equipment and airplanes ALL PRESENT AND AFTER-ACQUIRED PROPERTY

SCHEDULE "B" (DESCRIBED REAL PROPERTY) \* Obtain legal description

SCHEDULE "C" (PERMITTED ENCUMBRANCES AFFECTING COLLATERAL)

# SCHEDULE "D" (OTHER TERMS AND CONDITIONS)

10112 (06/2000 JB)

Page 6 of 6

This is **Exhibit "N"** referred to in the Affidavit of Darcy Peelar, sworn before me this 5<sup>th</sup> day of July, 2017

A Commissioner for Oaths in and for Alberta

Patrick T. Harnett Barrister and Solicitor

25337254.1
# Government of Alberta ■

### Personal Property Registry Search Results Report

Page 1 of 4

Search ID#: Z09280920

**Transmitting Party** MILLER THOMSON LLP Attention: Accounts Payable 2700, 10155 102 STREET EDMONTON, AB T5J 4G8

Party Code: 50000066 Phone #: 780 429 1751 Reference #: 138667.138 PTH

Search ID #: Z09280920

Date of Search: 2017-Jul-04

Time of Search: 13:58:41

### Business Debtor Search For:

CRELOGIX ACCEPTANCE CORPORATION

Exact Result(s) Only Found

### NOTE:

A complete Search may result in a Report of Exact and Inexact Matches. Be sure to read the reports carefully.



Government of Alberta <b>=</b>		Personal Property Registry Search Results Report Search ID#: Z09280920		Page 2 of 4
Busin	ess Debtor Search For:			
	OGIX ACCEPTANCE CORPOR ID #: Z09280920	RATION Date of Search: 2017-Jul-04	Time of Search:	13:58:41
Registration Number: 08032830685 Registration Date: 2008-Mar-28		Registration Type: SECUR Registration Status: Current Expiry Date: 2018-M		
	Exact Match on: Debtor	No: 3		
Amen	dments to Registration			
09120114543		Amendment	Amendment 2009-Dec-01	
11080	326435	Amendment	2011-Aug-03	
16091	311286	Amendment	2016-S	ep-13
<u>Debtc</u>	<u>or(s)</u>			
<u>Block</u>				<u>Status</u>
1	TRAVELERS ACCEPTANCE ( 500-4180 LOUGHEED HIGHW BURNABY, BC V5C 6A7			Current
<u>Block</u>				<u>Status</u>
2	TRAVELERS CONSUMER FIN 500-4180 LOUGHEED HIGHW BURNABY, BC V5C 6A7			Current
<u>Block</u>				<u>Status</u>
3	CRELOGIX ACCEPTANCE CO 4445 LOUGHEED HWY, SUIT BURNABY, BC V5C 0E4			Current by 11080326435
	<u>red Party / Parties</u>			01.1
<u>Block</u> 1	COMMUNITY CREDIT UNION			<u>Status</u> Deleted by
	601-4901 48 STREET			09120114543



### Personal Property Registry Search Results Report

Page 3 of 4

Search ID#: Z09280920

<u>Block</u>	
2	SERVUS CREDIT UNION LTD. 601 4901 48 STREET RED DEER, AB T4N 6M4

<u>Status</u>

Current by 09120114543

Phone #: 403 343 0144 Fax #: 403 342 4547

### **Collateral: General**

<u>Block</u>	Description	<u>Status</u>	
1	All present and after acquired personal property and all proceeds including crops, fixtures and licences.	Current	
2	PROCEEDS: ALL OF THE DEBTOR'S PRESENT AND AFTER ACQUIRED GOODS, MOTOR VEHICLES, ACCOUNTS, MONEY, CHATTEL PAPER, DOCUMENTS OF TITLE, INVESTMENT PROPERTY, INSTRUMENTS AND INTANGIBLES AS DEFINED IN THE PERSONAL PROPERTY SECURITY ACT, INSURANCE PROCEEDS AND ALL OTHER SUBSTITUTIONS, RENEWALS, ALTERATIONS OR PROCEEDS OF EVERY DESCRIPTION AND OF ANY KIND WHATSOEVER DERIVED DIRECTLY OR INDIRECTLY FROM ANY DEALINGS WITH THE GENERAL COLLATERAL OR SERIAL NUMBER COLLATERAL (IF ANY) DESCRIBED ABOVE, OR PROCEEDS THEREFROM.	Current By 16091311286	

Government of Alberta ■	Personal Property Registry Search Results Report Search ID#: Z09280920	Page 4 of 4	
Business Debtor Search For: CRELOGIX ACCEPTANCE CORPO Search ID #: Z09280920	DRATION Date of Search: 2017-Jul-04 Time of Search: 7		
Registration Number: 17042612131 Registration Date: 2017-Apr-26	Registration Type: SECURITY AGREEMENT Registration Status: Current Expiry Date: 2022-Apr-26 23:59:59		
Exact Match on: Debtor	No: 1		
Debtor(s) Block 1 CRELOGIX ACCEPTANCE ( #900-4445 LOUGHEED HIG BURNABY, BC V5C0E4	CORPORATION HWAY	<u>Status</u> Current	
Secured Party / PartiesBlock1DEALNET CAPITAL CORP. #300-325 MILNER AVE. TORONTO, ON M1B5N1		<u>Status</u> Current	
Collateral: GeneralBlockDescription1ALL OF THE DEBTOR'S PRI	ESENT AND AFTER-ACQUIRED PERSONAL PROPERTY. Result Complete	<u>Status</u> Current	

# Government of Alberta ■

### Personal Property Registry Search Results Report

Page 1 of 4

Search ID#: Z09280945

Transmitting Party MILLER THOMSON LLP Attention: Accounts Payable 2700, 10155 102 STREET EDMONTON, AB T5J 4G8

Party Code: 50000066 Phone #: 780 429 1751 Reference #: 138667.138 PTH

Search ID #: Z09280945

Date of Search: 2017-Jul-04

Time of Search: 14:02:07

### Business Debtor Search For:

CRELOGIX CREDIT GROUP INC.

Exact Result(s) Only Found

## NOTE:

A complete Search may result in a Report of Exact and Inexact Matches. Be sure to read the reports carefully.



	nment erta ∎	Personal Property Re Search Results Rep Search ID#: Z09280945	Page 2 of 4
CRELOG	Bebtor Search For: IX CREDIT GROUP INC. #: Z09280945	Date of Search: 2017-Jul-04	Time of Search: 14:02:07
	tion Number: 09111226536 stration Date: 2009-Nov-12	Registration Type: SECURITY A Registration Status: Current Expiry Date: 2025-Nov-12	
Exa	act Match on: Debtor	No: 1	
Amendn	nents to Registration		
14101616 16091311		Amendment And Renewal Amendment	2014-Oct-16 2016-Sep-13
Debtor(s	<u>5)</u>		
;	CRELOGIX CREDIT GROUP 500-4180 LOUGHEED HIGH\ BURNABY, BC V5C 6A7		<u>Status</u> Current
	<u> Party / Parties</u>		
1	SERVUS CREDIT UNION LT 601-4901 48TH STREET RED DEER, AB T4N 6N4	D.	<u>Status</u> Deleted by 14101616342
<u>Block</u>			<u>Status</u>
	SERVUS CREDIT UNION LT 151 KARL CLARK ROAD NW EDMONTON, AB T6N 1H5		Current by 14101616342
	Phone #: 780 641 8934	Fax #: 780 433 8629	
Collater	al: General		
<u>Block</u>	<u>Description</u>		<u>Status</u>



# Personal Property Registry Search Results Report

Page 3 of 4

Search ID#: Z09280945

1	ALL PRESENT AND AFTER-ACQUIRED PERSONAL PROPERTY OF THE DEBTOR.	Current
2	PROCEEDS: ALL OF THE DEBTOR'S PRESENT AND AFTER ACQUIRED GOODS, MOTOR VEHICLES, ACCOUNTS, MONEY, CHATTEL PAPER, DOCUMENTS OF TITLE, INVESTMENT PROPERTY, INSTRUMENTS AND INTANGIBLES AS DEFINED IN THE PERSONAL PROPERTY SECURITY ACT, INSURANCE PROCEEDS AND ALL OTHER SUBSTITUTIONS, RENEWALS, ALTERATIONS OR PROCEEDS OF EVERY DESCRIPTION AND OF ANY KIND WHATSOEVER DERIVED DIRECTLY OR INDIRECTLY FROM ANY DEALINGS WITH THE GENERAL COLLATERAL OR SERIAL NUMBER COLLATERAL (IF ANY) DESCRIBED ABOVE, OR PROCEEDS THEREFROM.	Current By 16091311408

Government of Alberta ■	Personal Property Registry Search Results Report Search ID#: Z09280945	Page 4 of 4
Business Debtor Search For: CRELOGIX CREDIT GROUP IN Search ID #: Z09280945	NC. Date of Search: 2017-Jul-04 Time of Search: 14	  4:02:07
Registration Number: 17042612 Registration Date: 2017-Apr-;		
Exact Match on: Debtor	No: 1	
Debtor(s) Block 1 CRELOGIX CREDIT GR #900-4445 LOUGHEED BURNABY, BC V5C0E4		<u>Status</u> Current
Secured Party / Parties Block 1 DEALNET CAPITAL CO #300-325 MILNER AVE. TORONTO, ON M1B5N		<u>Status</u> Current
Collateral: GeneralBlockDescription1ALL OF THE DEBTOR'S	PRESENT AND AFTER-ACQUIRED PERSONAL PROPERTY. Result Complete	<u>Status</u> Current

\_\_\_\_

# Government of Alberta ■

### Personal Property Registry Search Results Report

Page 1 of 4

Search ID#: Z09280929

Transmitting Party MILLER THOMSON LLP Attention: Accounts Payable 2700, 10155 102 STREET EDMONTON, AB T5J 4G8

Party Code: 50000066 Phone #: 780 429 1751 Reference #: 138667.138 PTH

Search ID #: Z09280929

Date of Search: 2017-Jul-04

Time of Search: 13:59:52

**Business Debtor Search For:** 

CRELOGIX PORTFOLIO SERVICES CORPORATION

Exact Result(s) Only Found

## NOTE:

A complete Search may result in a Report of Exact and Inexact Matches. Be sure to read the reports carefully.



	ernment Iberta ∎	Personal Property Regi Search Results Repo Search ID#: Z09280929	istry ort Page 2 of 4
Busine	ess Debtor Search For:		
	DGIX PORTFOLIO SERVICES ID #: Z09280929		ne of Search: 13:59:52
	stration Number: 09103020468 egistration Date: 2009-Oct-30	Registration Type: SECURITY AG Registration Status: Current Expiry Date: 2025-Oct-30 23	
E	Exact Match on: Debtor	No: 1	
Amen	dments to Registration		
14093041232 16091311386		Amendment And Renewal Amendment	2014-Sep-30 2016-Sep-13
<u>Debto</u> <u>Block</u>	<u>r(s)</u>		<u>Status</u>
1	CRELOGIX PORTFOLIO SER' 500-4180 LOUGHEED HIGHW BURNABY, BC V5C 6A7		Current
<u>Block</u> 2	CORPORATION DE RECOUV 500-4180 LOUGHEED HIGHW BURNABY, BC V5C 6A7		<u>Status</u> Current
	ed Party / Parties		
<u>Block</u> 1	SERVUS CREDIT UNION LTD 601-4901 48 STREET RED DEER, AB T4N 6M4		<u>Status</u> Deleted by 14093041232
<mark>Block</mark> 2	SERVUS CREDIT UNION LTD 151 KARL CLARK ROAD NW EDMONTON, AB T6N 1H5		<u>Status</u> Current by 14093041232
		Fax #: 780 433 8629	

## Government of Alberta

### Personal Property Registry Search Results Report

Page 3 of 4

Search ID#: Z09280929

### **Collateral: General**

#### Block Description **Status** 1 ALL PRESENT AND AFTER-ACQUIRED PERSONAL PROPERTY OF THE DEBTOR. Current 2 PROCEEDS: ALL OF THE DEBTOR'S PRESENT AND AFTER ACQUIRED GOODS, Current By 16091311386 MOTOR VEHICLES, ACCOUNTS, MONEY, CHATTEL PAPER, DOCUMENTS OF TITLE, INVESTMENT PROPERTY, INSTRUMENTS AND INTANGIBLES AS DEFINED IN THE PERSONAL PROPERTY SECURITY ACT, INSURANCE PROCEEDS AND ALL OTHER SUBSTITUTIONS, RENEWALS, ALTERATIONS OR PROCEEDS OF EVERY DESCRIPTION AND OF ANY KIND WHATSOEVER DERIVED DIRECTLY OR INDIRECTLY FROM ANY DEALINGS WITH THE GENERAL COLLATERAL OR SERIAL NUMBER COLLATERAL (IF

ANY) DESCRIBED ABOVE, OR PROCEEDS THEREFROM.

Government of Alberta ■	Personal Property Registry Search Results Report Search ID#: Z09280929	Page 4 of 4	
Business Debtor Search For:		-	
CRELOGIX PORTFOLIO SERVICE Search ID #: Z09280929	ES CORPORATION Date of Search: 2017-Jul-04 Time of Search: 1	- 3:59:52	
Registration Number: 17042612300	Registration Type: SECURITY AGREEMENT		
Registration Date: 2017-Apr-26	Registration Status: Current		
	Expiry Date: 2022-Apr-26 23:59:59		
Exact Match on: Debtor	No: 1		
Block		<u>Status</u>	
1 CRELOGIX PORTFOLIO SE #900-4445 LOUGHEED HIG BURNABY, BC V5C0E4		Current	
Secured Party / Parties		Chatura	
Block1DEALNET CAPITAL CORP.		<u>Status</u> Current	
1 DEALNET CAPITAL CORP. #300-325 MILNER AVE. TORONTO, ON M1B5N1		Guirent	
Collateral: General			
Block Description		<u>Status</u>	
1 ALL OF THE DEBTOR'S PR	ESENT AND AFTER-ACQUIRED PERSONAL PROPERTY.	Current	
	Result Complete		

\_\_\_\_\_

\_\_\_\_\_

This is **Exhibit "O"** referred to in the Affidavit of Darcy Peelar, sworn before me this 5<sup>th</sup> day of July, 2017

A Commissioner for Oaths in and for Alberta

Patrick T. Harnett Barrister and Solicitor

25337254.1

Active Session Products Account Setup Contact Us
Personal Property Registry Selection List
For: [ PH96238 ] [ ELDOR-WAL REGISTRATIONS (1987) LT ] Jul 04, 2017
01:01:10 PM
Folio: BC On Line Mailbox
Business Name: CRELOGIX ACCEPTANCE CORPORATION <b>Exact Matches: 4</b> Local Print Limit: 200
BSR101 - NO MORE INFORMATION TO DISPLAY
Debtor Name
CRELOGIX ACCEPTANCE CORPORATION
Display Selection

Page: 1

BC OnLine: PPRS SEARCH RESULT 2017/07/04 Lterm: XPSP0050 For: PH96238 ELDOR-WAL REGISTRATIONS (1987) LT 13:01:10 Index: BUSINESS DEBTOR Search Criteria: CRELOGIX ACCEPTANCE CORPORATION Reg. Date: MAR 28, 2008 Reg. Length: 10 YEARS Reg. Time: 14:05:05 Expiry Date: MAR 28, 2018 Base Reg. #: 268048E Control **#:** B8633833 Block# S0001 Secured Party: COMMUNITY CREDIT UNION LTD. 601-4901-48 STREET RED DEER ALTA T4N 6M4 D0001 Base Debtor: TRAVELERS ACCEPTANCE CORPORATION (Business) 500 4180 LOUGHEED HIGHWAY BURNABY B.C. V5C 6A7 D0002 Bus. Debtor: TRAVELERS CONSUMER FINANCE 500 4180 LOUGHEED HIGHWAY BURNABY B.C. V5C 6A7 General Collateral: ALL PRESENT AND AFTER ACQUIRED PERSONAL PROPERTY AND ALL PROCEEDS INCLUDING CROPS, FIXTURES AND LICENCES. Registering Party: EDWARDS KENNY & BRAY LLP 1900 1040 WEST GEORGIA STREET VANCOUVER BC V6E 4H3 Req. #: 589862F Reg. Date: JUN 02, 2010 Reg. Time: 10:49:01 Control #: B9993519 Base Reg. Type: PPSA SECURITY AGREEMENT Base Reg. #: 268048E Base Reg. Date: MAR 28, 2008 General Collateral: \*\* DELETED \*\* THE SECURED PARTY'S SECURITY INTEREST IN THE ASSETS OF THE DEBTOR HAVE BEEN DISCHARGED WITH RESPECT TO THOSE ASSETS WHICH HAVE BEEN SOLD BY THE DEBTOR TO BANK WEST PURSUANT TO THE TERMS AND CONDITIONS SET OUT IN THE FUNDING AGREEMENT DATED OCTOBER 16, 2009 BETWEEN THE DEBTOR AND BANK WEST. Registering Party: DAVIS LLP/DAVIS MANAGEMENT LTD. 2800 PARK PLACE 666 BURRARD ST VANCOUVER BC V6C 2Z7

Continued on Page 2

Search Criteria: CRELOGIX ACCEPTANCE CORPORATION Page: 2 \*\*\* Name/Address Changed on April 17, 2015 to: Registering Party: DLA PIPER (CANADA) LLP/DAVIS MANAGEMENT LTD. 2800 PARK PLACE 666 BURRARD ST VANCOUVER BC V6C 2Z7 ----- AMENDMENT / OTHER CHANGE -----Reg. Date: JUN 02, 2010 Reg. #: 589866F Reg. Time: 10:52:10 Control #: B9989164 Base Reg. Type: PPSA SECURITY AGREEMENT Base Reg. #: 268048E Base Reg. Date: MAR 28, 2008 Details Description: SUBORDINATION AGREEMENT IN FAVOUR OF BANK WEST SECURITY AGREEMENT REGISTERED DECEMBER 21, 2009 AS REGISTRATION NUMBER 332240F GRANTING PRIORITY OVER COLLATERAL DESCRIBED IN REGISTRATION NUMBER 268048E. Registering Party: DAVIS LLP/DAVIS MANAGEMENT LTD. 2800 PARK PLACE 666 BURRARD ST VANCOUVER BC V6C 2Z7 \*\*\* Name/Address Changed on April 17, 2015 to: Registering Party: DLA PIPER (CANADA) LLP/DAVIS MANAGEMENT LTD. 2800 PARK PLACE 666 BURRARD ST VANCOUVER BC V6C 2Z7 ----- A M E N D M E N T / O T H E R C H A N G E ------ A M E N D M E N T / O T H E R C H A N G E ------Reg. #: 286648G Reg. Date: AUG 05, 2011 Reg. Time: 14:50:53 Control #: D0708125 Base Reg. Type: PPSA SECURITY AGREEMENT Base Reg. #: 268048E Base Reg. Date: MAR 28, 2008 Details Description: NAME CHANGE EFFECTIVE MARCH 14, 2010 Block# \*\*\* ADDED \*\*\* =D0003 Bus. Debtor: CRELOGIX ACCEPTANCE CORPORATION 4445 LAUGHEED HWY, STE 900

BURNABY BC V5C 0E4

Registering Party: PARLEE MCLAWS LLP 1500, 10180 - 101 STREET EDMONTON AB T5J 4K1

Continued on Page 3

Search Criteria: CRELOGIX ACCEPTANCE CORPORATION Page: 3 ----- AMENDMENT / OTHER CHANGE------Reg. Date: SEP 07, 2016 Reg. #: 524189J Reg. Time: 17:32:42 Control #: D4023183 Base Reg. Type: PPSA SECURITY AGREEMENT Base Reg. #: 268048E Base Reg. Date: MAR 28, 2008 Details Description: ADDITION OF SECURED PARTY AND GENERAL COLLATERAL. Block# \*\*\* ADDED \*\*\* S0002 Secured Party: SERVUS CREDIT UNION LTD 151 KARL CLARK ROAD NW EDMONTON AB T6N 1H5 General Collateral: \*\*\* ADDED \*\*\* PROCEEDS: ALL OF THE DEBTOR'S PRESENT AND AFTER ACQUIRED GOODS, MOTOR VEHICLES, ACCOUNTS, MONEY, CHATTEL PAPER, DOCUMENTS OF TITLE, INVESTMENT PROPERTY, INSTRUMENTS AND INTANGIBLES AS DEFINED IN THE PERSONAL PROPERTY SECURITY ACT, INSURANCE PROCEEDS AND ALL OTHER SUBSTITUTIONS, RENEWALS, ALTERATIONS OR PROCEEDS OF EVERY DESCRIPTION AND OF ANY KIND WHATSOEVER DERIVED DIRECTLY OR INDIRECTLY FROM ANY DEALINGS WITH THE GENERAL COLLATERAL OR SERIAL NUMBER COLLATERAL (IF ANY) DESCRIBED ABOVE, OR PROCEEDS THEREFROM. Registering Party: MILLER THOMSON LLP 1000 840 HOWE STREET VANCOUVER BC V6Z 2M1 -----AMENDMENT / OTHER CHANGE ------Reg. #: 525261J Reg. Date: SEP 08, 2016 Reg. Time: 09:59:36 Control #: D4024113 Base Reg. Type: PPSA SECURITY AGREEMENT Base Reg. #: 268048E Base Reg. Date: MAR 28, 2008 Details Description: ADDITION OF DEBTOR. Block#

DICC

\*\*\* ADDED \*\*\* D0004 Bus. Debtor: CRELOGIX ACCEPTANCE CORPORATION 2600 - 595 BURRARD ST VANCOUVER BC V7X 1L3

> Registering Party: MILLER THOMSON LLP 1000 840 HOWE STREET VANCOUVER BC V6Z 2M1

> > Continued on Page 4

Search Criteria: CRELOGIX ACCEPTANCE CORPORATION Page: 4 Reg. Date: APR 26, 2017 Reg. Length: 5 YEARS Reg. Time: 07:25:04 Expiry Date: APR 26, 2022 Base Reg. #: 963857J Control #: D4472660 Block# S0001 Secured Party: DEALNET CAPITAL CORP. #300-325 MILNER AVE. TORONTO ON M1B5N1 =D0001 Base Debtor: CRELOGIX ACCEPTANCE CORPORATION (Business) #900-4445 LOUGHEED HIGHWAY BURNABY BC V5C0E4 General Collateral: ALL OF THE DEBTOR'S PRESENT AND AFTER-ACQUIRED PERSONAL PROPERTY. Reg. Date: JUN 30, 2017 Reg. Length: 10 YEARS Reg. Time: 15:19:41 Expiry Date: JUN 30, 2027 Base Reg. #: 113174K Control #: D4624775 Block# S0001 Secured Party: PROSPERA CREDIT UNION 500- 32071 SOUTH FRASER WAY ABBOTSFORD BC V2T 1W3 S0002 Secured Party: PROSPERA HOLDINGS LTD. 500- 32071 SOUTH FRASER WAY ABBOTSFORD BC V2T 1W3 =D0001 Base Debtor: CRELOGIX ACCEPTANCE CORPORATION (Business) 900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4 D0002 Bus. Debtor: TRAVELERS ACCEPTANCE CORPORATION

900 - 4445 LOUGHEED HIGHWAY BURNABY BC V5C 0E4

General Collateral:

ALL PRESENT AND FUTURE ACCOUNTS, MONIES, DEBTS AND OTHER INTANGIBLES, INSTRUMENTS, CHATTEL PAPER, DOCUMENTS OF TITLE. ALL DEPOSITS FROM TIME TO TIME IN THE NAME OF THE DEBTOR OR OWED TO THE DEBTOR BY THE SECURED PARTY AND ALL THE DEBTOR'S RIGHTS IN AND TO THOSE DEPOSITS. ALL PROCEEDS INCLUDING, WITHOUT LIMITATION, ALL ACCOUNTS INVESTMENT PROPERTY, INSTRUMENTS, DOCUMENTS OF TITLE, CHATTEL PAPER, INTANGIBLES AND MONEY (ALL AS DEFINED IN THE PERSONAL PROPERTY SECURITY ACT, ANY REGULATIONS THEREUNDER AND ANY AMENDMENTS THERETO.

Registering Party: MCMILLAN LLP BOX 11117,1500 1055 W. GEORGIA VANCOUVER BC V6E 4N7

Continued on Page 5

Search Criteria: CRELOGIX ACCEPTANCE CORPORATION

Page: 5

### 

Some, but not all, tax liens and other Crown claims are registered at the Personal Property Registry (PPR) and if registered, will be displayed on this search result. HOWEVER, it is possible that a particular chattel is subject to a Crown claim that is not registered at the PPR. Please consult the Miscellaneous Registrations Act, 1992 for more details. If you are concerned that a particular chattel may be subject to a Crown claim not registered at the PPR, please consult the agency administering the type of Crown claim.

	re Session Products Account Setup Contact Us
Personal Property Registry	Selection List
For: [ PH96238 ] [ ELDOR-WAL REGISTRATIONS (1987) LT	] Jul 04, 2017 01:03:44 PM
Folio:	Send to Mailbox Help ? BC OnLine Mailbox
Business Name: CRELOGIX CREDIT GROUP INC.	Local Print Limit: 200
BSR101 - NO MORE INFORMATION TO	DISPLAY
Debtor Name	
CRELOGIX CREDIT GROUP INC	
CRELOGIX CREDIT GROUP INC.	
CRELOGIX CREDIT GROUP INC.	
<ul> <li>CRELOGIX CREDIT GROUP INC.</li> </ul>	

Display Selection

Page: 1

BC OnLine: PPRS SEARCH RESULT 2017/07/04 Lterm: XPSP0050 For: PH96238 ELDOR-WAL REGISTRATIONS (1987) LT 13:03:44 Index: BUSINESS DEBTOR Search Criteria: CRELOGIX CREDIT GROUP INC. Reg. Date: MAY 16, 2016 Reg. Length: 5 YEARS Reg. Time: 09:19:31 Expiry Date: MAY 16, 2021 Base Reg. #: 289385J Control #: D3783403 Block# S0001 Secured Party: RCAP LEASING INC. 5575 NORTH SERVICE RD BURLINGTON ON L7L 6M1 =D0001 Base Debtor: CRELOGIX CREDIT GROUP INC. (Business) 4445 LOUGHEED HWY BURNABY BC V5C 6A7 D0002 Bus. Debtor: CRELOGIX CREDIT GROUP INC. 630 BOUL RENE-LEVESQUE MONTREAL QC H3B 1S6 General Collateral: ALL OFFICE FURNITURE EQUIPMENT FROM TIME TO TIME LEASED BY THE SECURED PARTY TO THE DEBTOR AS DESCRIBED ON LEASES, CONDITIONAL SALES AGREEMENTS AND ANY OTHER FINANCING AGREEMENTS ENTERED INTO BETWEEN THE SECURED PARTY AND THE DEBTOR FROM TIME TO TIME AND ANY PROCEEDS THEREOF, TOGETHER WITH ALL REPLACEMENT PARTS, ACCESSORIES AND ATTACHMENTS. Registering Party: (REGISTRY=RECOVERY) TM INC. 1551 THE QUEENSWAY TORONTO ON M8Z 1T8 \*\*\*\* P P S A A G R E E M E N T \*\*\*\*\*\*\*\*\*\*\*\*\*\*\* SECURITY Reg. Date: SEP 07, 2016 Reg. Length: 5 YEARS Reg. Time: 17:27:37 Expiry Date: SEP 07, 2021 Base Reg. #: 524187J Control #: D4023173 Block# S0001 Secured Party: SERVUS CREDIT UNION LTD 151 KARL CLARK RD NW EDMONTON AB T6N 1H5 Base Debtor: CRELOGIX CREDIT GROUP INC =D0001 (Business) 2600 - 595 BURRARD ST VANCOUVER BC V7X 1L3 General Collateral: ALL PRESENT AND AFTER ACQUIRED PERSONAL PROPERTY OF THE DEBTOR AND, WITHOUT LIMITATION, ALL FIXTURES, CROPS, AND LICENCES. PROCEEDS: ALL OF THE DEBTOR'S PRESENT AND AFTER ACQUIRED GOODS,

MOTOR VEHICLES, ACCOUNTS, MONEY, CHATTEL PAPER, DOCUMENTS OF TITLE, INVESTMENT PROPERTY, INSTRUMENTS AND INTANGIBLES AS DEFINED IN THE

Continued on Page 2

Search Criteria: CRELOGIX CREDIT GROUP INC.

PERSONAL PROPERTY SECURITY ACT, INSURANCE PROCEEDS AND ALL OTHER SUBSTITUTIONS, RENEWALS, ALTERATIONS OR PROCEEDS OF EVERY DESCRIPTION AND OF ANY KIND WHATSOEVER DERIVED DIRECTLY OR INDIRECTLY FROM ANY DEALINGS WITH THE GENERAL COLLATERAL OR SERIAL NUMBER COLLATERAL(IF ANY) DESCRIBED ABOVE, OR PROCEEDS THEREFROM.

Registering Party: MILLER THOMSON LLP 1000 840 HOWE STREET VANCOUVER BC V6Z 2M1

Rec	J.	Dat	te:	APR	26,	2017
Reg	J.	Tir	ne:	07:2	25:25	5
Base	Re	∋g.	#:	9638	359J	

Reg. Length: 5 YEARS Expiry Date: APR 26, 2022 Control #: D4472662

Block#

- S0001 Secured Party: DEALNET CAPITAL CORP. #300-325 MILNER AVE. TORONTO ON M1B5N1
- =D0001 Base Debtor: CRELOGIX CREDIT GROUP INC. (Business) #900-4445 LOUGHEED HIGHWAY BURNABY BC V5C0E4

General Collateral: ALL OF THE DEBTOR'S PRESENT AND AFTER-ACQUIRED PERSONAL PROPERTY.

Some, but not all, tax liens and other Crown claims are registered at the Personal Property Registry (PPR) and if registered, will be displayed on this search result. HOWEVER, it is possible that a particular chattel is subject to a Crown claim that is not registered at the PPR. Please consult the Miscellaneous Registrations Act, 1992 for more details. If you are concerned that a particular chattel may be subject to a Crown claim not registered at the PPR, please consult the agency administering the type of Crown claim.

Page: 2



Display Selection

Page: 1 BC OnLine: PPRS SEARCH RESULT 2017/07/04 For: PH96238 ELDOR-WAL REGISTRATIONS (1987) LT 13:02:06 Lterm: XPSP0050 Index: BUSINESS DEBTOR Search Criteria: CRELOGIX PORTFOLIO SERVICES CORPORATION \*\*\*\*\* P P S A Reg. Date: SEP 07, 2016 Reg. Length: 5 YEARS Reg. Time: 17:28:52 Expiry Date: SEP 07, 2021 Base Reg. #: 524188J Control #: D4023177 Block# S0001 Secured Party: SERVUS CREDIT UNION LTD 151 KARL CLARK RD NW EDMONTON AB T6N 1H5 =D0001 Base Debtor: CRELOGIX PORTFOLIO SERVICES (Business) CORPORATION 2600 - 595 BURRARD STREET VANCOUVER BC V7X 1L3 General Collateral: ALL PRESENT AND AFTER ACQUIRED PERSONAL PROPERTY OF THE DEBTOR AND, WITHOUT LIMITATION, ALL FIXTURES, CROPS, AND LICENCES. PROCEEDS: ALL OF THE DEBTOR'S PRESENT AND AFTER ACQUIRED GOODS, MOTOR VEHICLES, ACCOUNTS, MONEY, CHATTEL PAPER, DOCUMENTS OF TITLE, INVESTMENT PROPERTY, INSTRUMENTS AND INTANGIBLES AS DEFINED IN THE PERSONAL PROPERTY SECURITY ACT, INSURANCE PROCEEDS AND ALL OTHER SUBSTITUTIONS, RENEWALS, ALTERATIONS OR PROCEEDS OF EVERY DESCRIPTION AND OF ANY KIND WHATSOEVER DERIVED DIRECTLY OR INDIRECTLY FROM ANY DEALINGS WITH THE GENERAL COLLATERAL OR SERIAL NUMBER COLLATERAL (IF ANY) DESCRIBED ABOVE, OR PROCEEDS THEREFROM. Registering Party: MILLER THOMSON LLP 1000 840 HOWE STREET VANCOUVER BC V6Z 2M1 Reg. Date: APR 26, 2017 Reg. Length: 5 YEARS Reg. Time: 07:25:46 Expiry Date: APR 26, 2022 Base Reg. #: 963860J Control #: D4472663 Block# S0001 Secured Party: DEALNET CAPITAL CORP. #300-325 MILNER AVE. TORONTO ON M1B5N1 =D0001 Base Debtor: CRELOGIX PORTFOLIO SERVICES (Business) CORPORATION #900-4445 LOUGHEED HIGHWAY BURNABY BC V5C0E4 General Collateral: ALL OF THE DEBTOR'S PRESENT AND AFTER-ACQUIRED PERSONAL PROPERTY.

Continued on Page 2

Search Criteria: CRELOGIX PORTFOLIO SERVICES CORPORATION Page: 2

Some, but not all, tax liens and other Crown claims are registered at the Personal Property Registry (PPR) and if registered, will be displayed on this search result. HOWEVER, it is possible that a particular chattel is subject to a Crown claim that is not registered at the PPR. Please consult the Miscellaneous Registrations Act, 1992 for more details. If you are concerned that a particular chattel may be subject to a Crown claim not registered at the PPR, please consult the agency administering the type of Crown claim.